

Mobile-assisted Medication Adherence Support Intervention for Tuberculosis Patients: Effectiveness, Experiences and Adoption Intention

By

Zekariyas Sahile Nezenega

Thesis Submitted to Flinders University for the degree of Doctor of Philosophy

Doctor of Philosophy

College of Nursing and Health Sciences January 17, 2024

Table of Contents

LIST OF FIGURESXII
LIST OF TABLESXIII
ABBREVIATIONS AND ACRONYMS XVI
PUBLICATIONS FROM THE THESISXVII
CONFERENCE PRESENTATIONS XVIII
ABSTRACT PUBLICATION XIX
TRIAL REGISTRATIONXX
ABSTRACTXXI
DECLARATIONXXIII
ACKNOWLEDGEMENTSXXIV
CHAPTER 1: INTRODUCTION1
1.1. Introduction1
1.2. Background1
1.2.1. Overview of Tuberculosis1
1.2.2. TB Treatment: Why medication adherence is a critical issue
1.2.3. TB treatment administration options and recommendations
1.2.4. Evidence on the use of mobile health interventions5
1.3. The rationale for the study6
1.4. Research aims and objectives10
1.4.1. Aims
1.4.2. Objectives
1.5. Hypothesis

1.5.1. Primary hypothesis	11
1.5.2. Secondary hypothesis	11
1.6. Research questions	12
1.7. Overview of the chapters	13
CHAPTER 2: LITERATURE REVIEW	17
2.1. Introduction	17
2.2. Factors associated with non-adherence to TB medication	17
2.2.1. Patient-centred factors	
2.2.2. Social factors	18
2.2.3. Economic factors	19
2.2.4. Health system factors	19
2.2.5. Therapy factors	20
2.2.6. Lifestyle factors	20
2.2.7. Geographical access factors	20
2.3. Adherence Interventions	21
2.3.1. Patient education	21
2.3.3. Reminders and tracers	22
2.3.4. Incentives and enablers	22
2.3.5. Staff education	22
2.3.6. Patient-centred care	22
2.3.7. Digital health	23
2.4. Mobile health Intervention	23
2.4.1. Definition of terms	23
2.4.2. Effectiveness of mobile health intervention	24

2.4.3. Interactivity of mobile health intervention	25
2.4.4. SMS text messages frequency, time, and duration	25
2.4.5. Feasibility and acceptability of mobile health intervention	26
2.5. Adherence definition and measurement	35
2.5.1. Definition of Adherence	35
2.5.2. Adherence measurement	
2.5.3. Selection of adherence measurement	
CHAPTER 3: METHODOLOGY AND METHODS	40
3.1. Introduction	40
3.2. Research philosophy	40
3.2.1. Ontology	40
3.2.2. Epistemology	40
3.2.3. Researcher position	41
3.2.4. Research paradigm	42
3.3. Mixed methods	43
3.3.1. Definition of mixed methods	43
3.3.2. Mixed methods evaluation	43
3.4. Study setting and period	46
3.5. Study one: Development and validation of Ma-MAS intervention: An MRC framew	ork
approach	47
3.5.1. MRC framework for complex interventions	47
3.5.2. Ma-MAS Intervention development process	48
3.5.3. Study design	56
3.5.4. Participants	

3.5.5. Sample size and technique	57
3.5.6. Data collection instrument and process	58
3.5.7. Data analysis	60
3.6. Study two: Feasibility and acceptability of Ma-MAS intervention	61
3.6.1. Study design	61
3.6.2. Participants	61
3.6.3. Sample size and technique	61
3.6.4. Data collection instrument and process	62
3.6.5. Data analysis	62
3.7. Study three: Effectiveness of Ma-MAS intervention	63
3.7.1. Study design	63
3.7.2. Participants	63
3.7.3. Sample size and techniques	64
3.7.4. Recruitment	64
3.7.5. Randomisation and allocation	65
3.7.6. Intervention group	65
3.7.7. Control group	66
3.7.8. Intervention implementation	66
3.7.9. Outcome measurement	67
3.7.10. Blinding	70
3.7.12. Data collection process	70
3.7.13. Data Analysis	71
3.8. Study four: Perceptions, experiences, and adoption intention	72
3.8.1. Unified theory acceptance and use of technology	72

3.8.2. Study design	74
3.8.3. Participants	75
3.8.4. Sample size and technique	75
3.8.5. Data collection instrument and process	75
3.8.6. Data analysis	76
3.8.7. Trustworthiness of qualitative research	78
3.9. Ethical process	80
CHAPTER 4: RESULTS OF STUDY ONE:	
DEVELOPMENT AND VALIDATION OF MA-MAS INTERVENTION: AN MRC FI	RAMEWORK
APPROCH	82
4.1. Introduction	
4.2. SMS text messages development	82
4.3. Demographic characteristics	84
4.4. SMS text messages validation	85
CHAPTER 5: RESULTS OF STUDY TWO:	92
FEASIBILITY AND ACCEPTABILITY OF MA-MAS INTERVENTION	92
5.1. Introduction	92
5.2. Demographic characteristics	92
5.3. Clinical characteristics	94
5.4. Feasibility of Ma-MAS intervention	94
5.5. Acceptability of Ma-MAS intervention	95
CHAPTER 6: RESULTS OF STUDY THREE:	97
EFFECTIVENESS OF MA-MAS INTERVENTION	97
6.1. Introduction	

6.2. Background	97
6.3. Demographic characteristics	99
6.4. Socio-economic characteristics	100
6.5. Substance use chracteristics	
6.6. Health service and geographic access characteristics	
6.7. Clinical characteristics	
6.8. Effectiveness of Ma-Mas intervention	
6.8.1. Effectiveness of Ma-Mas intervention on primary ouctome	
6.8.2. Effectiveness of Ma-Mas intervention in multivariable analysis	
6.8.3. Effectiveness of Ma-Mas intervention on secondary outcomes	
CHAPTER 7: RESULTS OF STUDY FOUR:	
PERCEPTIONS, EXPERIENCES AND ADOPTION INTENTION	
7.1. Introduction	109
7.1. Introduction	
7.2. Background	
7.2. Background	109 109 111
 7.2. Background 7.2.1. Demographic characteristics 7.2.2. Theme, sub-theme, and categories 	
 7.2. Background 7.2.1. Demographic characteristics 7.2.2. Theme, sub-theme, and categories 7.3. Theme 1: Acceptance: Attitude and adoption intention 	
 7.2. Background 7.2.1. Demographic characteristics 7.2.2. Theme, sub-theme, and categories 7.3. Theme 1: Acceptance: Attitude and adoption intention 7.3.1. Attitude 	
 7.2. Background 7.2.1. Demographic characteristics	
 7.2. Background 7.2.1. Demographic characteristics	
 7.2. Background 7.2.1. Demographic characteristics	

7.4.5. Sense of professional care	122
7.4.6. General health consultation	
7.5. Theme 3: Convenience: Experience of adherence intervention design	
7.5.1. Evaluation of the contents of the messages	125
7.5.2. Suggested content of messages	
7.5.3. Comprehension	
7.5.4. Time of delivery	
7.5.5. Frequency of SMS text messages and phone calls	
7.5.6. Adherence Intervention period	141
7.5.7. Privacy and confidentiality	
7.6. Factors associated with adoption intention	
7.7. Demographic characteristics of the participants	
7.8. Measurement model	149
7.8.1. Construct validity and reliability of Unified theory of acceptance and use	of technology
	149
7.8.2. Confirmatory factor analysis	151
7.8.3. Construct correlation coefficient and discriminant validity	151
7.9. Structural model of Unified theory of acceptance and use of technology	152
7.9.1. Original UTAUT model analysis	152
7.9.2. Modified UTAUT model analysis	152
CHAPTER 8: DISCUSSION OF THE FINDINGS	154
8.1. Introduction	154
8.2. Key findings	
8.3. Factors affecting non-adherence to TB medication	156

8.4. Effectiveness of Ma-MAS intervention157
8.5. Ma-MAS intervention design161
8.6. Feasibility and acceptability of Ma-MAS intervention
8.7. Factors affecting adoption intention of Ma-MAS intervention
8.8. Facilitators and challenges of Ma-MAS intervention
8.9. Strengths and limitations of the study169
CHAPTER 9: CONCLUSIONS AND RECOMMENDATIONS
9.1. Introduction
9.2. Conclusions
9.3. Recommendations175
9.3.1. Implications for policy makers and program managers
9.3.2. Implications for health education and communication
9.3.3. Implications for health science education178
9.3.4. Implications for future research179
BIBLIOGRAPHY
APPENDIX ONE: FACTORS INFLUENCING PATIENT ADHERENCE TO TB MEDICATION IN
ETHIOPIA
APPENDIX TWO: QUALITATIVE ASSESSMENT FOR MA-MAS INTERVENTION
DEVELOPMENT
Demographic characteristics
Theme, sub-theme and categories
Theme 1: Adherence to TB medication and influencing factors
Adherence to TB Medication
Factors affecting adherence209

Theme 2: History of health system practice	224
Adherence Intervention and mentoring mechanism	224
The DOT standard care during COVID-19 emerged	235
Theme 3: Acceptability of mobile-assisted adherence intervention	238
Acceptability and benefits of mobile-assisted adherence intervention	238
Acceptability and preference: SMS text versus phone calls	244
Theme 4: Convenience: mobile-assisted adherence intervention design	249
Content of the message	249
Language	250
Comprehension	252
Interactivity	253
Frequency	255
Time of delivery	258
Duration of the Intervention	259
Privacy and confidentiality	
Theme 5: Barriers and facilitators of mobile-assisted adherence intervention	
Facilitators of mobile-assisted adherence intervention	
Challenges of mobile-assisted adherence intervention	268
Suggested solutions	275
Availing telephone at TB clinic	275
Briefing the mobile-assisted adherence intervention	275
Using a family member's or caregiver's mobile phone	277
Making a voice message or phone calls to an illiterate	278
Confirming and updating the patient's phone number	279

Giving the project a unique name280
APPENDIX THREE: PROTOTYPE SMS TEXT MESSAGES FOR MA-MAS INTERVENTION
USING BEHAVIOUR CHANGE TECHNIQUES TAXONOMY AND IMB MODEL
APPENDIX FOUR: VALIDATED SMS TEXT MESSAGES USING BEHAVIOUR CHANGE
TECHNIQUES AND IMB MODEL
APPENDIX FIVE: PHONE CALLS CHECKLIST GUIDE
APPENDIX SIX: DEMPGRAPHIC CHARACTEROSTICS OF ADULT TB PATIENTS WITHOUT
PERSONAL MOBILE PHONES
APPENDIX SEVEN: CHARACTERSTICS OF LOSS TO FOLLOW-UP PARTICIPANTS AND
SENSITIVITY ANALYSIS
Demographic characteristics of loss to follow-up participants
Sensitivity analysis
Effectiveness of Ma-Mas intervention on primary outcome
Assumptions
Effectiveness of Ma-Mas intervention on secondary outcomes
Assumptions
APPENDIX EIGHT: PUBLICATIONS FROM THE THESIS
Article 1: Factors influencing patient adherence to Tuberculosis treatment in Ethiopia: a literature
review
Article 2: Protocol of a parallel group Randomised Control Trial (RCT) for Mobile-assisted
Medication Adherence Support (Ma-MAS) intervention among Tuberculosis patients
Article 3: Mobile-assisted medication adherence support intervention among Tuberculosis
patients: a parallel group randomised control trial (Conference Abstract)
APPENDIX NINE: QUESTIONNAIRE
In-depth Interview Guide for Ma-MAS Intervention Development

Feasibility and Acceptability Questionnaire	337
Ma-MAS Intervention Trial Questionnaire	340
Adoption Intention of Ma-MAS Intervention Questionnaire	344
In-depth Interview Guide for Post-Ma-MAS Intervention Experiences	347
APPENDIX TEN: INFORMED CONSENT FORM	348
APPENDIX ELEVEN: ETHICS APPROVAL	374
APPENDIX TWELVE: CONFERENCE PRESENTATIONS CERTIFICATES	379

LIST OF FIGURES

Figure 1: Schematic presentation of the thesis chapters
Figure 2: Schematic presentation of the mixed methods evaluation design
Figure 3: Map of Addis Ababa, Ethiopia (Source: Shapefile from Ethiopian mapping Agency) 46
Figure 4: Key element in the development and evaluation process of complex Intervention adapted
from Craig P. et al. 2008 (76)
Figure 5: Adapted MRC framework development Phase. Source: Bleijenberg et al. 2018 (161) 49
Figure 6: Information, Motivation and Behavioural skills model to TB medication adherence adapted
from Fisher et al. 2006 (160)
Figure 7: Schematic presentation of study objective, design, and participants
Figure 8: Mobile-assisted Medication Adherence Support (Ma-MAS) intervention trial design diagram
Figure 9: Timeline for Ma-MAS intervention trial69
Figure 10: Unified theory of acceptance and use of technology adopted from Venkatesh et al.
2003. (170)
Figure 11: The modified Unified theory of acceptance and use of technology adapted from Venkatesh
et al. (2003) and modified by Dwivedi et al. 2019 ⁽¹⁷³⁾
Figure 12: Schematic presentation of qualitative analysis adapted from Raskind et al. 2019 ⁽¹⁸³⁾ 77
Figure 13: SMS text messages development and validation process
Figure 14: Ma-MAS trial flow diagram (adapted from CONSORT 2010) ⁽¹⁹⁶⁾
Figure 15: The predictive probability of adherence to TB medication after four weeks and eight weeks
of follow-up in Ma-MAS trial, Addis Ababa, Ethiopia107

LIST OF TABLES

Table 1: Hypothesis on the construct relationships within UTAUT framework 11
Table 2: Randomised clinical trials of mobile health interventions for TB extended from Lester et al.
2019 ⁽²⁷⁾ and Nglazi et al. 2013 ⁽⁸¹⁾
Table 3: The characteristics of SMS text message interventions in randomised control trials extended
from Lester et al. 2019 ⁽²⁷⁾ and Nglazi et al. 2013 ⁽⁸¹⁾
Table 4: Acceptability and feasibility of mobile health interventions for TB treatment support 33
Table 5: Mobile-assisted medication adherence support intervention development phases: An
adapted MRC framework approach ⁽¹⁶¹⁾ 50
Table 6: SMS text messages relevance rating description
Table 7: Definition for UTAUT constructs used in the proposed theoretical model. source: Dwivedi et
al. 2019 ⁽¹⁷³⁾
Table 8: Development and validation process of SMS text messages
Table 9: Demographic characteristics of health professionals participated in SMS text messages
validation, Addis Ababa, Ethiopia85
Table 10: Validated SMS text messages for promoting TB medication adherence
Table 11: Socio-demographic characteristics of adult TB patients, Addis Ababa, Ethiopia93
Table 12: Clinical characteristics of adult TB patients, Addis Ababa, Ethiopia
Table 13: Feasibility of mobile-assisted medication adherence support among adult TB patients,
Addis Ababa, Ethiopia95
Table 14: Acceptability of mobile-assisted medication adherence support (Ma-MAS) among adult TB
patients, Addis Ababa, Ethiopia96
Table 15: Demographic characteristics of adult TB patients enrolled in Ma-MAS trial, Addis Ababa,
Ethiopia

Table 16: Social and economic characteristics of adult TB patients enrolled in Ma-MAS trial, Addis
Ababa, Ethiopia
Table 17: Substance use characteristics of adult TB patients enrolled in Ma-MAS trial, Addis Ababa,
Ethiopia102
Table 18: Health service and geographic access characteristics of adult TB patients enrolled in Ma-
MAS trial, Addis Ababa, Ethiopia 103
Table 19: Clinical characteristics of adult TB patients enrolled in Ma-MAS trial, Addis Ababa, Ethiopia
Table 20: Ma-MAS intervention effect on TB medication adherence measured by IsoScreen urine
test, Addis Ababa, Ethiopia 105
Table 21: Multivariable analysis of covariates in Ma-MAS intervention effectiveness after eight weeks
of follow-up, Addis Ababa, Ethiopia106
Table 22: Ma-MAS intervention effect on self reported TB medication adherence, Addis Ababa,
Ethiopia108
Table 23: Demographic characteristics of adult TB patients, Addis Ababa, Ethiopia 110
Table 23: Demographic characteristics of adult TB patients, Addis Ababa, Ethiopia110Table 24: Theme, sub-theme, and categories112
Table 24: Theme, sub-theme, and categories 112
Table 24: Theme, sub-theme, and categories 112 Table 25: Demographic characteristics of adult TB patients, Addis Ababa, Ethiopia 149
Table 24: Theme, sub-theme, and categories112Table 25: Demographic characteristics of adult TB patients, Addis Ababa, Ethiopia149Table 26: Construct validity and reliability of unified theory of acceptance and use of technology150
Table 24: Theme, sub-theme, and categories 112 Table 25: Demographic characteristics of adult TB patients, Addis Ababa, Ethiopia 149 Table 26: Construct validity and reliability of unified theory of acceptance and use of technology150 150 Table 27: Descriptive statistics, correlation coefficient, and discriminant validity of UTAUT 151
Table 24: Theme, sub-theme, and categories 112 Table 25: Demographic characteristics of adult TB patients, Addis Ababa, Ethiopia 149 Table 26: Construct validity and reliability of unified theory of acceptance and use of technology150 150 Table 27: Descriptive statistics, correlation coefficient, and discriminant validity of UTAUT 151 Table 28: Construct path relationship in unified theory of acceptance and use of technology 153
Table 24: Theme, sub-theme, and categories
Table 24: Theme, sub-theme, and categories 112 Table 25: Demographic characteristics of adult TB patients, Addis Ababa, Ethiopia 149 Table 26: Construct validity and reliability of unified theory of acceptance and use of technology150 149 Table 27: Descriptive statistics, correlation coefficient, and discriminant validity of UTAUT 151 Table 28: Construct path relationship in unified theory of acceptance and use of technology 153 Table 29: Summary of reviewed articles showing factors influencing non-adherence or default or loss to follow up from TB treatment 196

Table 33: Prototype SMS text for Ma-MAS Intervention using Behaviour Change Techniques (BCT)
taxonomy (Michie et al. 2013) ⁽¹⁵⁸⁾ and IMB model ⁽¹⁶⁰⁾
Table 34: Validated SMS text for Ma-MAS Intervention using Behaviour Change Techniques (BCT)
taxonomy (Michie et al. 2013) ⁽¹⁵⁸⁾ and IMB model (Fisher et al. 2006) ⁽¹⁶⁰⁾
Table 35: Socio-demographic characteristics of adult TB patients without personal mobile phones,
Addis Ababa, Ethiopia
Table 36: Demographic characteristics of lost to follow-up adult TB patients enrolled in Ma-MAS trial,
Addis Ababa, Ethiopia
Table 37: Sensetivity analysis of Ma-MAS effect on TB medication adherence meausred IsoScreen
urine test in the worst scenario
Table 38: Sensetivity analysis of Ma-MAS intervention effect on TB medication adherence meausred
IsoScreen urine test in best scenario
Table 39: Sensetivity analysis of Ma-MAS intervention effect on self report TB medication adherence
in worst senario
Table 40: Sensetivity analysis of Ma-MAS intervention effect on self report TB medication adherence
in best scenario

ABBREVIATIONS AND ACRONYMS

ACTG	AIDS Clinical Trial Group
ART	Antiretroviral Therapy
BCT	Behavioural Change Techniques
CI	Confidence Interval
DOT	Directly observed Treatment
DR TB	Drug-resistant TB
EDHS	Ethiopia Demography and Health Survey
HIV	Human Immunodeficiency Virus
IMB	Information-Motivation-Behavioural skills
INH	Isoniazid
IRB	Institutional Review Board
Ma-MAS	Mobile-assisted Medication adherence support
MDR TB	Multidrug-resistant TB
MEMS	Medication Event Monitoring System
MOH	Ministry of Health
MRC	Medical Research Council
RCT	Randomised Control Trial
RR	Relative Risk
SAT	Self-Administered Treatment
SMS	Short Message Service
ТВ	Tuberculosis
UTAUT	Unified Theory of Acceptance and Use of Technology
VAS	Visual Analogues scale
VOT	Video Observed Treatment
WHO	World Health Organization

PUBLICATIONS FROM THE THESIS

- Nezenega ZS, Perimal-Lewis L, Maeder AJ. Factors influencing patient adherence to Tuberculosis treatment in Ethiopia: a literature review. International Journal of Environmental Research and Public Health. 2020 Aug;17(15):5626.¹
- Sahile Z, Perimal-Lewis L, Arbon P, Maeder AJ. Protocol of a parallel group Randomised Control Trial (RCT) for Mobile-assisted Medication Adherence Support (Ma-MAS) intervention among Tuberculosis patients. Plos one. 2021 Dec 31;16(12):e0261758.¹

¹ **NB**: The first author's name listed above as Nezenega ZS and Sahile Z is the same. The author's full name is Zekariyas Sahile Nezenega

CONFERENCE PRESENTATIONS

- Development and validation of SMS text messages for mobile-assisted medication adherence support intervention for Tuberculosis patients
 Conference: the 16th Tuberculosis Research Advisory Committee (TRAC) Annual Conference and 2022 World TB Day at Hawassa University in Ethiopia
- Mobile-assisted Medication Adherence Support intervention among Tuberculosis Patients: A Parallel Group Randomised Control Trial

Conference: the 14th International Conference on Health Policy Statistics in Scottsdale,

Arizona, USA, from January 9-11, 2023

ABSTRACT PUBLICATION

Sahile Z, Maeder AJ, Perimal-Lewis L, Arbon P. Mobile-assisted medication adherence support intervention among Tuberculosis patients: a parallel group randomised control trial. Population Medicine. 2023;5(Supplement): A585. <u>https://doi.org/10.18332/popmed/163638</u>

TRIAL REGISTRATION

The trial has been registered in the Pan-Africa Clinical Trials Registry under the trial identifier PACTR202002831201865. <u>https://pactr.samrc.ac.za/TrialDisplay.aspx?TrialID=9729</u>

ABSTRACT

Tuberculosis (TB) control programs face serious challenges in resource-limited settings, particularly in Ethiopia, where 20% of TB patients do not adhere to prescribed TB medication. Mobile phonebased adherence interventions can address challenges associated with non-adherence to TB medication because they allow direct communication between providers and patients. However, there is conflicting evidence regarding the efficacy of SMS text message interventions on TB medication adherence. Additionally, most adherence interventions based on mobile phone messaging are not tailored to the local context and needs of the target audience.

A pragmatic mixed-methods evaluation design was employed to systematically develop and evaluate the Mobile-assisted medication adherence support (Ma-MAS) intervention in Addis Ababa, Ethiopia. The Medical Research Council (MRC) framework including the information-motivation-behavioural skills (IMB) model and behavioural change techniques, was employed to develop and validate the Ma-MAS intervention. The feasibility and acceptability of the intended intervention were assessed by a cross-sectional survey. A parallel-group randomised control trial design was used to evaluate the effectiveness of Ma-MAS intervention among a sample of randomly assigned (n=186) participants. Participants in the Ma-MAS group received a daily SMS text message and a weekly phone call regarding their daily medication intake and reminders to attend clinic visits for eight weeks. Participants in the control group did not receive SMS texts or phone calls but received the same routine standard care as the Ma-MAS group. The primary outcome was the proportion of adherence measured by urine tests for isoniazid (INH) metabolites at the end of the fourth and eighth week of follow-up period. A post intervention interpretive study with qualitative descriptive approach and quantitative cross-sectional survey were also employed to inform the study findings on utility and acceptability.

In total, 40 SMS text messages were selected by experts to use for TB medication adherence based on content validity indexes above 0.80. According to a cross-sectional study, 88.5% of TB patients have their own mobile phone, and 93.5% of mobile phone owners were willing to receive SMS text message as reminders for medication intake and 92% were willing to receive SMS text

xxi

message as reminder for clinic appointments. Most participants were also willing to receive phone calls as reminders for medication intake (91%) and clinic appointments (93%).

In the experimental study, Ma-MAS significantly improved adherence to TB medications by 15.25% (95%CI: 5.38, 25.12; P-value=0.0065) after eight weeks of the intervention compared to the standard care alone in the control group. At the end of the eight weeks follow-up period, the predicted probability of adherence to TB medication in the Ma-MAS group was 86% (95%CI: 81, 93), and in the control group was 70% (95%CI: 61, 79). Ma-MAS also improved adherence to TB medication by 15.30% (95%CI: 6.68, 23.90; P-value=0.0022) after four weeks of the intervention compared to the control group. Ma-MAS intervention was shown to be useful to improve TB medication adherence as measured by a urine test for isoniazid (INH) metabolites. The Ma-MAS intervention had high acceptance and is a useful for reminder of TB medication, habit formation, awareness creation, motivation, enhancement of perceived professional care, and health consultation. This study therefore provides evidence for the effectiveness of mobile-assisted approaches to medication adherence through a rigorously planned and executed randomised control trial and interpretive study in a particular setting where best practice in medication adherence is critical to health outcomes but difficult to achieve by conventional approaches.

DECLARATION

I certify that this thesis: Mobile-assisted Medication Adherence Support Intervention for

Tuberculosis Patients: Effectiveness, Experiences and Adoption Intention

1. does not incorporate without acknowledgment any material previously submitted for a degree or diploma in any university

2. and the research within will not be submitted for any other future degree or diploma without the permission of Flinders University; and

3. to the best of my knowledge and belief, does not contain any material previously published or written by another person except where due reference is made in the text.

Signed

Zekariyas Sahile Nezenega

Date. January 17, 2024

ACKNOWLEDGEMENTS

First and foremost, I would like to express my gratitude to my all-powerful God. I am also deeply grateful to my family for their unwavering support and encouragement throughout my doctoral journey. I am especially thankful to the Flinders International Postgraduate Research Scholarship for providing full sponsorship of my doctoral study, and to the Flinders University Digital Health Centre for partially funding my research. The funding support of the NIH/Fogarty International Centre Global Infectious Disease has also been helpful in the success of this research.

I am deeply appreciative of the Ministry of Health of Ethiopia and the Addis Ababa Health Bureau, as well as their staff members working in health offices and health facilities in each sub-city, for their invaluable contributions that made this research possible.

My heartfelt thanks go to my supervisors, Professor Paul Arbon, Dr. Lua Perimal-Lewis, and Professor Anthony Maeder, for their invaluable guidance and feedback throughout the research process.

I am also grateful to Dr. Kidist Bobosha, Mrs. Hiwet Eyob, and other staff members at Armauer Hansen Research Institute (AHRI) in Addis Ababa, Ethiopia, for their assistance in shipping the adherence measurement kit from abroad and providing necessary equipment. Additionally, I would like to thank Shahid Ullah from Flinders University's College of Medicine and Public Health for his advice on sample size determination.

I would also like to extend my thanks to all my friends who have been instrumental in sharing information and resources. My deepest appreciation goes to the research team who worked tirelessly to complete the work, and to all the study participants who made this research possible. Finally, I am grateful to all the professionals who participated in the Delphi technique for SMS text message validation. Thank you all for your invaluable contributions to this research.

CHAPTER 1: INTRODUCTION

1.1. Introduction

This chapter provides the definition of Tuberculosis (TB), the general epidemiology of TB, TB treatment regimen, the crucial role of medication adherence, the risk of non-adherence to TB medication, and the prevalence of non-adherence to TB medication in Ethiopia. Additionally, it discussed the TB treatment administration options and recommendations, explores evidence on the use of mobile health for TB treatment support, and outlines the rationale for the study. The chapter also presented the research aim, objectives, hypotheses, and research questions.

1.2. Background

1.2.1. Overview of Tuberculosis

Tuberculosis (TB) is "an infectious bacterial disease caused by Mycobacterium tuberculosis, which is transmitted between humans through the respiratory route and most commonly affects the lungs, but can damage any tissue."⁽¹⁾

Living with TB is a challenging experience that affects the person physically, emotionally, and economically. The impact of TB extends beyond the physiological aspects, as TB patients may face social stigma, discrimination, and psychological distress.^(2, 3) The person affected by TB requires a long treatment regimen that necessitates strict adherence for a cure. The TB treatment may have drug side effects, which are additional challenges for the patients. The prolonged treatment regimen is associated with economic losses for the patients and their families due to the cost of food supplements, transportation, and associated medical expenses.⁽³⁾

TB is still the biggest infectious cause of mortality worldwide, despite the presence of vaccine and chemotherapy.⁽¹⁾ TB is a major global health problem, accounting for millions of people's disease each year. According to the 2022 World Health Organization (WHO) global TB report,

in 2021 a TB incidence of 119 per 100,000 population was detected in Ethiopia. The mortality rate of HIV negative TB was 16 per 100,000 population and the mortality rate of HIV plus TB was 1.7 per 100,000 population.⁽⁴⁾

1.2.2. TB Treatment: Why medication adherence is a critical issue

The World Health Organization recommends a six-month treatment regimen for drugsusceptible TB. This treatment regimen involves the administration of rifampicin, isoniazid, pyrazinamide, and ethambutol during the initial phase of two months, followed by a continuation phase consisting of rifampicin and isoniazid for the subsequent four months (2HRZE/4HR).⁽⁵⁾

To be cured of TB, strict adherence to medication for the whole treatment regimen is necessary. The level of adherence to TB medication is defined differently in various studies, but according to the World Health Organization's guidelines for the treatment of drug-susceptible TB and patient care report, patients who take more than 90% of their medications under the observation of another person was considered adherent.⁽⁵⁾ Achieving a high level of adherence to TB medication is challenging for many patients due to various complex reasons.⁽⁶⁾

Non - adherence with TB medication increases the risk of additional transmission, treatment failure, relapsed TB, multidrug resistance or extensively drug-resistant TB, and mortality. ⁽⁷⁻⁹⁾ Drug-resistant TB is treatable, but it needs massive use of therapy that is toxic to the patients, too expensive and less effective as compared to drug-sensitive TB.⁽¹⁰⁾ According to one study, non-adherent patients had a longer period until they tested negative to TB (254 versus 64 days), and had a higher risk of developing multi-drug resistance TB, would have an extended treatment regimen (560 versus 324 days) and lower treatment completion rate.⁽⁸⁾

In Ethiopia, TB prevention and control programs are challenged by widespread of nonadherence to TB treatment. One systematic review conducted in Ethiopia indicated that the pooled prevalence of non-adherence to TB medication was 21.3%.⁽¹¹⁾ In the same geographic

area, other systematic reviews indicated that overall prevalence of non-adherence to TB medication and lost to follow-up rates was 20% and 5%, respectively.⁽¹²⁾ The prevalence of non-adherence to TB medication in Addis Ababa, where the current research was undertaken, was 19.5%.⁽¹³⁾

1.2.3. TB treatment administration options and recommendations

The World Health Organization (WHO) recommends Direct Observation of Treatment (DOT) by a trained supervisor to address the problem non-adherence to TB medication over self-Administered Treatment (SAT). In DOT administration, a patient takes medication every day under the observation of a healthcare provider to ensure adherence.⁽¹⁴⁾ While in Self-Administered Treatment (SAT) administration, a patient takes medication without the observation of a healthcare provider.⁽¹⁵⁾ But evidence showed that DOT does not provide an optimal solution to poor adherence in TB treatment.⁽¹⁶⁾ For example, a one systematic review indicated that the difference between DOT and Self-Administered Treatment (SAT) on lost to follow-up, treatment failure, cure, treatment completion, and mortality is small and not statistically significant.⁽¹⁵⁾

In Ethiopia, implementing DOT is challenging to patients as it requires daily visits to the health facility, which has an undesirable impact on their work, social life and transportation costs.^(17, 18) Healthcare professionals highlight the difficulties of implementing a daily DOT at the health facility.⁽¹⁷⁾ Thus, DOT has not been fully implemented throughout the treatment regimen.^(17, 18) DOT and non-daily DOT has been implemented in Ethiopia, in which during the first two months of intensive phase of treatment patients take their daily medication with health carer provider observation, and in the continuation of phase treatment, from two to six months, patients take all their medication at home and obtain weekly medication refill at health facilities.⁽¹⁷⁻¹⁹⁾ However, the DOT's standard schedule for both intensive and continuation phase of treatment were interrupted by the COVID-19 pandemic.⁽²⁰⁾

The current WHO recommendations are realised using DOT implementation with different treatment administration options. These include Daily DOT, Video Observed Treatment (VOT), and non-daily DOT where not every dose is overseen (overseen weekly or a few times per week).⁽⁵⁾ The WHO has also revised the definition of DOT to include patients taking the medication under the observation of any person in real time.

It is desirable to measure and monitor the patient's treatment adherence, identify alternative strategies to DOT ⁽¹⁶⁾ and design interventions such as social support including material, food, or financial and transport help; psychological support; tracers through home-to-home visits or digital health interventions such as Short Message Service (SMS) texts and phone calls, to improve TB medication adherence.^(5, 21, 22)

The World Health Organization (WHO) has developed a conceptual framework for designing digital health interventions in response to the growing recognition of their benefits.⁽²³⁾ The framework focuses on four key areas: patient care, surveillance and monitoring, program management, and e-learning, all of which can be integrated into the patient pathway of healthcare seeking, diagnosis, treatment, and follow-up.⁽²³⁾ Short Message Service (SMS), Video Observed Treatment (VOT) and Event Monitoring Device for Medication support (EMM) have already been used in TB program in different geographic settings to help to complete treatment and monitor the daily dose of medication.⁽²³⁾ A synthesis of literature reviews showed that one randomised trial of medication monitors and two observational studies of VOT were not found to have a statistical effect on cure or treatment completion.⁽²⁴⁾ However, the existing evidence supports the use of mobile health intervention to promote medication adherence since these are at least relatively effective as standard care.⁽²⁴⁻²⁶⁾ VOT has emerged as an alternative method to DOT, but it is inaccessible to patients in some settings as it requires a smartphone with an internet connection.⁽²⁷⁾

1.2.4. Evidence on the use of mobile health interventions

Mobile phones increasingly enable access even for those living on US\$1 per day in high TB burden countries.⁽⁹⁾ In Ethiopia, access to mobile phones is expanding widely: according to the Ethiopia Demography and Health Survey (EDHS) 2019, 87% of urban households and 59% of rural households had a mobile phone access.⁽²⁸⁾ This is higher when compared to the EDHS 2011, which showed 65% of urban households and 13% rural households have mobile phone.⁽²⁹⁾ Mobile health interventions are low cost, easy to use, widely available and use the existing infrastructure including network, internet, and mobile channels.⁽³⁰⁾

The increased availability of mobile phones creates opportunities to address the challenges of implementing DOT, as mobile health can facilitate the communication between healthcare providers and TB patients via SMS and voice phone calls, in addition to DOT or non-daily DOT.^(22, 27) Digital health interventions through mobile health can create new opportunities for healthcare especially where there is limited infrastructure, expertise, and human resource in the healthcare system.⁽³¹⁻³⁴⁾ Digital adherence technologies including phone-based and smartphone-based technologies are expected to facilitate a more patient-centred approach, although evidence is still limited.⁽³⁵⁾

Mobile based SMS text messaging has been used for reminders, awareness creation and behaviour change purposes.^(30, 34, 36, 37) Previous SMS text message experiments have proven to be beneficial in helping people quit smoking,⁽³⁸⁻⁴³⁾ in the continuation of oral contraception,⁽⁴⁴⁾ breastfeeding practice,⁽⁴⁵⁾ Antenatal care (ANC) visit,^(46, 47) Postnatal care (PNC) visit, ⁽⁴⁷⁻⁴⁹⁾ skilled delivery attendance,^(47, 49-52) lower perinatal mortality and stillbirth,⁽⁵³⁾ self-management of long-term illness,⁽⁵⁴⁾ cardiovascular risk factor awareness,⁽⁵⁵⁾ diabetes management,^(56, 57) systolic blood pleasure management,⁽⁵⁸⁾ malaria case management,⁽⁵⁹⁾ and medication reminders for chronic disease.⁽⁶⁰⁻⁶²⁾

Most previous mobile-based SMS text reminder studies in Ethiopia have focussed on maternal and child health services improvement.^(47-49, 63-65) An SMS-based intervention was found effective in improving Antenatal Care (ANC) in one randomised trial ⁽⁴⁷⁾, postnatal care (PNC) in another a randomised control trial and one non-randomised control trial ^(48, 49) and childbirth attended by a health worker in one randomised and one non-randomised trial.^(47, 49)

Making mobile health interventions user-friendly, with good user acceptance and perceived usefulness are important underpinning of success.⁽⁶⁶⁾ Previous studies showed that SMS messaging was found acceptable in TB ⁽⁶⁷⁻⁶⁹⁾ and antiretroviral therapy (ART)-medication adherence support,⁽⁷⁰⁻⁷⁴⁾ self-management of diabetes,⁽⁵⁶⁾ maternal health promotion,^(52, 63) and had high satisfaction in supporting weight loss, physical activity and dietary behaviour change.⁽⁷⁵⁾ Moreover, an SMS-based TB adherence intervention was found to be technically feasible in terms of access to a mobile phone and familiarity with the use of SMS text message, and has a high rate of acceptance.⁽⁶⁹⁾

1.3. The rationale for the study

The design of an adherence intervention needs to address the underlying factors that cause medication non-adherence.⁽³⁵⁾ Digital health interventions may enhance TB medication adherence and have health benefits to the community by lowering the incidence of disease relapse, acquired drug resistance, and infection transmission. Digital health interventions can be an alternative to observable medication adherence (DOT), and may be more feasible with the expansion and the low cost of technology and because TB treatment is for limited duration as compared to treatment for other chronic diseases.⁽³⁵⁾

The Medical Research Council (MRC) framework for complex intervention recommend using appropriate theory and evidence when developing an intervention.^(76, 77) However, most adherence interventions were developed without a theoretical base which may be one of the reasons they are sometimes not effective.⁽⁷⁸⁾ Literature indicates that ineffectiveness of mobile health SMS-based interventions might be due to the intervention not being developed systematically ⁽⁷⁹⁾ and not being customized to the local context and population needs.⁽³⁰⁾ A mobile-assisted medication adherence support intervention needs to be developed systematically and evaluated for effectiveness through rigorous research design, to provide quality evidence.

The scientific base for designing SMS text and phone call-based adherence intervention is founded in the premise that adherence information, motivation and behavioural skills can make a direct positive effect on adherence,⁽⁸⁰⁾ and can be used for awareness creation and behavioural change.⁽³⁰⁾ There is some evidence that adherence interventions like educating and counselling the patient, medication monitors, and SMS text reminders and tracers can improve treatment adherence and success.^(21, 22, 81, 82)

In Ethiopia, for example, a randomised control trial found that psychological counselling and educational intervention can significantly improve a patient's treatment adherence. This intervention combined anxiety and depression counselling along with patient education based on health belief model.⁽⁸³⁾ However, implementing psychological counselling and educational intervention at health facilities might require high level of expertise and human resources compared to delivering intervention through mobile device. Another trial in the northwest Ethiopia was investigated the effect of a graphic based daily reminder and weekly medication refill reminders on patient's medication adherence reported that the web-based graphic reminder was found to be effective on treatment adherence and patient provider interaction.^(84, 85) The graphic-based reminder intervention can be effective in addressing forgetfulness related to medication intake and medication refill reminders, but it might not address other behavioural factors that cause non-adherence to medication such as knowledge, motivation and other barriers.

The current intervention, Ma-MAS, is different from the previous two randomised interventions conducted in Ethiopia in three ways: first, the intervention itself was systematically developed based on evidence from the local context through conducting a

literature review to identify factors affecting TB medication adherence in Ethiopia; second, the intervention was delivered through mobile-based SMS text messages and phone calls, which is feasible to scale up on a large scale; third, the Ma-MAS intervention is designed to serve beyond simple reminders because it developed educational messages based on the factors affecting TB medication adherence. As for this, the combination of SMS text and phone calls intervention using a behavioural change model and techniques has not been previously designed and tested in Ethiopia. Additionally, this study assessed the primary outcome of the intervention trial using the most reliable measurement of adherence (IsoScreen kit) ⁽⁸⁶⁻⁸⁹⁾ Therefore, in this study, a mobile-assisted medication adherence support (Ma-MAS) intervention was systematically designed using the Medical Research Council (MRC) conceptual framework with appropriate behavioural change model and techniques. It was assessed its effectiveness on medication adherence and its acceptance by patients.

For mobile health to achieve its full potential, it must be integrated into the current health care system. In Ethiopia, the existence of national eHealth strategy is an opportunity to conduct a mobile-based adherence intervention trial and to integrate this into the health system. The Ethiopia Ministry of Health (MOH) identified eHealth as one of the top priorities in a five-year health transformation plan, and mobile health is one of the priority areas of the country's eHealth strategy.^(90, 91)

Generally, this study may have three positive effects. First, rational ^(R1) the intervention may supplement the effort of the Ministry of Health by designing mobile-assisted medication adherence support intervention systematically with the local evidence, needs and practices that may promote medication adherence, Second, ^(R2) the mobile-based SMS text messages and phone calls intervention have been demonstrated in this research to improve the treatment adherence, therefore it may also improve treatment outcome, reduce disease relapse, drug resistance, and infection transmission to others. Third, ^(R3) the study provides a research question for future studies and can be used as educational material. Thus, this project is believed to make a major potential contribution to the development plan for the health

sector, country and at large globally. The evidence generated from this study is valuable for policymakers, program managers, healthcare providers, education sectors and non-governmental organizations (NGOs) working in Ethiopia and beyond.

1.4. Research aims and objectives

1.4.1. Aims

This study aimed to systematically develop and evaluate a mobile-assisted medication adherence support intervention using SMS and phone calls among TB patients in Addis Ababa, Ethiopia.

1.4.2. Objectives

Objective-1: To systematically develop and validate mobile-assisted medication adherence support intervention for TB patients in Addis Ababa, Ethiopia

Objective-2: To assess the feasibility and acceptability of mobile-assisted medication adherence support intervention among TB patients in Addis Ababa, Ethiopia

Objective-3: To measure the effect of mobile-assisted medication adherence support intervention on treatment adherence among TB patients in Addis Ababa, Ethiopia

Objective-4: To assess perceptions, experiences and intention to adopt mobile-assisted medication adherence support intervention among TB patients in Addis Ababa, Ethiopia

1.5. Hypothesis

1.5.1. Primary hypothesis

Mobile-assisted medication adherence support intervention using mobile SMS text reminders and phone calls improves TB medication adherence in addition to routine standard non-daily DOT care.

1.5.2. Secondary hypothesis

The Unified Theory of Acceptance and Use of Technology (UTAUT) explains and predicts how people adopt and use new technologies. Therefore, the secondary hypothesis of this study examined how factors predict mobile-assisted medication adherence support intervention acceptance (see Table 1). In this study, the term 'Adoption intention' is used synonymously with 'behavioural intention', which is an indicator of acceptance of intervention.

Original UTAUT	Modified UTAUT
H1: Performance expectancy has a positive	H1: Performance expectancy has a positive
effect on behavioural intention	effect on attitude
H2: Effort expectancy has a positive effect on	H2: Performance expectancy has a positive
behavioural intention	effect on behavioural intention
H3: Social influence has a positive effect on	H3: Effort expectancy has a positive effect on
behavioural intention	attitude
	H4: Effort expectancy has a positive effect on
	behavioural intention
	H5: Social influence has a positive effect on
	attitude
	H6: Social influence has a positive effect on
	behavioural intention
	H7: Facilitating condition has a positive effect on
	attitude
	H8: Facilitating condition has a positive effect on
	behavioural intention

Table 1: Hypothesis on the construct relationships within UTAUT framework
1.6. Research questions

- 1. What factors affect non-adherence to TB medication in Ethiopia?
- 2. How can a mobile-assisted medication adherence support intervention be designed to address factors of non-adherence among TB patients in Addis Ababa, Ethiopia?
- 3. Is a mobile-assisted medication adherence support intervention feasible and acceptable in Addis Ababa, Ethiopia?
- 4. Is a mobile-assisted medication adherence support intervention effective in improving medication adherence among TB patients in Addis Ababa, Ethiopia?
- 5. What factors explain the effectiveness of a mobile-assisted medication adherence support intervention among TB patients in Addis Ababa, Ethiopia?
- 6. How do patients perceive and experience a mobile-assisted medication adherence support intervention in Addis Ababa, Ethiopia?
- 7. What factors affect the adoption intention of a mobile-assisted medication adherence support intervention among TB patients in Addis Ababa, Ethiopia?

1.7. Overview of the chapters

Chapter one provides an informative introduction to the topic, presenting an overview of TB, TB treatment regimen, the problem of non-adherence to TB medication, the TB treatment administration options, and explores evidence on the use of mobile health interventions for TB treatment support. It explains the rationale behind conducting mobile-assisted medication adherence support intervention study, emphasizing the significance of this research. It also provides an overview of the research aim, objectives, hypothesis, and research questions.²

Chapter two provides a comprehensive literature review on adherence to TB medication. It covers factors influencing adherence and explores the use of mobile health (mHealth) interventions. The chapter provides valuable insights into the challenges of adherence to TB medication, interventions, and the potential of mHealth to improve TB medication adherence. The chapter covers the feasibility, acceptability, and effectiveness of mobile-based adherence interventions. The chapter also covers the definition of adherence, various adherence measurements, and the adherence measurements used in the current study.

Chapter three discusses research methodology, a fundamental component of the research. It encompasses essential elements such as the research philosophy, the researcher position, the research paradigm, and the utilization of a mixed methods design. This chapter further presents a comprehensive overview of the study design, including the methods employed for data collection and analysis in both empirical and interpretive research approaches. This chapter provides a comprehensive explanation of the methodology and methods for conducting the research and for generating and analyzing data systematically and rigorously.

Chapter four presents the results of the development and validation of Ma-MAS intervention using Medical Research Council (MRC) framework. This chapter gives general overview of the intervention development and validation process. Each phase of development and

² The introduction chapter contains material originally published article of protocol of a parallel group Randomized Control Trial (RCT) for Mobile-assisted Medication Adherence Support (Ma-MAS) intervention among Tuberculosis patients. Plos one. 2021 Dec 31;16(12):e0261758.

validation process is discussed in detail in the method section of the thesis. The purpose of this chapter is to present the key result of Ma-MAS intervention development and validation to establish a solid foundation for understanding its implementation and effectiveness.

Chapter five presents the results of the feasibility and acceptability of the mobile-assisted medication adherence support (Ma-MAS) intervention. The chapter focuses on two key elements: feasibility and acceptability. Feasibility was evaluated by examining participants' access to mobile phones and their level of digital literacy in utilizing SMS text messages and phone calls. The acceptability of Ma-MAS was measured by assessing participants willingness to receive SMS text messages and phone calls as a means of support for their TB medication adherence and clinic appointments. The findings of this evaluation provide insight into the practicality and acceptability of Ma-MAS intervention to enhance medication adherence and clinic attendance for TB patients.

Chapter six presents the results of randomised control trial that aimed to assess the effectiveness of the mobile-assisted adherence support (Ma-MAS) intervention in improving TB medication adherence. The study involved two groups: the Ma-MAS group and the control group. Participants in the Ma-MAS group received daily SMS text messages and weekly phone calls for eight weeks, along with standard care during the continuation phase of treatment. On the other hand, the control group received standard care without Ma-MAS intervention. This chapter will examine the results of this trial, providing valuable insight into the effects of the Ma-MAS intervention on TB medication adherence compared to standard care alone.

Chapter seven presents an overview of patients' perceptions, experiences, and adoption intention regarding the Mobile-assisted Medication Adherence Support (Ma-MAS) intervention. The qualitative analysis of this chapter focuses on patients' perceptions and experiences, while the unified theory of acceptance and use of technology (UTAUT) was used

to identify key factors influencing adoption intentions. This chapter provides valuable insights into patients' experiences and the determinants of Ma-MAS intervention adoption.

Chapter eight provides a discussion and interpretation of the research findings on the effectiveness of the mobile-assisted medication adherence support (Ma-MAS) intervention for TB medication adherence. The chapter discusses factors influencing adherence to TB medication, the intervention development process, acceptability and feasibility assessment. The randomised control trial findings and patients' experiences are discussed in this chapter.

Chapter nine presents the conclusions and recommendations based on the research findings. This chapter presents conclusions about the effectiveness of the mobile-assisted medication adherence support (Ma-MAS) intervention for improving TB medication adherence. This chapter also provides recommendations to improve Ma-MAS intervention implementation, design, and patient engagement. The chapter emphasizes the potential use of mobile technology in supporting TB medication adherence and suggests its integration into healthcare systems for improving health outcomes. The schematic presentation of thesis chapters is presented in Figure 1.



Figure 1: Schematic presentation of the thesis chapters

CHAPTER 2: LITERATURE REVIEW

2.1. Introduction

The literature review section explores various aspects related to adherence to TB medication and interventions aimed at promoting adherence. It begins by discussing the literature review of factors affecting adherence to TB medication in Ethiopia. Next, the review provides an overview of adherence interventions, including the use of mobile health (mHealth) interventions to promote medication adherence. This chapter also discussed the effectiveness, feasibility, and acceptability of mobile health interventions for TB treatment support, drawing on relevant studies in the field. At the end of the chapter, the definition of adherence and the various measures used to assess adherence are discussed.

Overall, the literature review provides a comprehensive understanding of the key issues related to adherence to TB medication, including the challenges faced by patients, the different interventions that have been used to address these challenges, and the potential of mHealth interventions to improve medication adherence.

2.2. Factors associated with non-adherence to TB medication

A literature review was conducted to identify factors influencing non-adherence to TB medication in Ethiopia. Because we develop adherence interventions based on the evidence available locally, the literature review was focused only on Ethiopia. This section contains material originally published article of factors Influencing Patient Adherence to Tuberculosis Treatment in Ethiopia: A Literature Review.⁽⁶⁾ The copy of the published article is available in the Appendix Eight. The synthesised findings from both quantitative and qualitative studies are presented below, grouped in seven dimensions of adherence influencing factors based on the Ogundele et al. (2015) classification of adherence influencing factors ⁽⁹²⁾. The individual studies included in the literature review are found in the Appendix One.

2.2.1. Patient-centred factors

Forgetfulness ^(19, 93-100) and inadequate knowledge about TB and its treatment regimen ^(94, 98, 101-104) were the two major patient-centred factors. Three studies conducted in Oromia ^(105, 106) and Southern region of Ethiopia ⁽¹⁰⁰⁾ regions showed that the patient's educational status was associated with non-adherence to TB medication: the more the patient is educated the less likely non-adherence to TB medication. Psychological distress was another factor: two studies conducted in Addis Ababa reported that this indirectly positively influences the non-adherence to TB medication. ^(13, 107) Another qualitative study conducted in Addis Ababa also stated that poor mental health status of a patient would make them reluctant to regularly attend follow up and clinic appointments.⁽⁹⁸⁾

2.2.2. Social factors

Several studies reported that patients not getting social support from families and neighbours in remembering to take their medication, food and financial assistance was the major social factor that influences non-adherence to TB medication.^(18, 98, 99, 103, 105, 108-110) Additionally, one study in Addis Ababa conducted among latent TB-HIV co-infected patients reported that the patient's friends' decision to take the medication would make them less likely to be non-adherent to isoniazid preventive therapy (IPT).⁽¹¹¹⁾ Another study among the same subjects and setting found that patients who were comfortable to take IPT in front of other people were less likely to be non-adherent. ⁽¹¹²⁾ Busy with work ⁽⁹⁴⁾ and away from home for work or other social-related activities were also found to influence non-adherence to TB medication.⁽⁹⁴⁻⁹⁶⁾ Perceived and experienced stigma and discrimination also led the patient to non-adherence. ^(98, 113-115) These particular factors were highly noted in studies conducted among TB-HIV co-infected patients.^(98, 109, 114, 115) As one study indicated, because of fear of stigma and discrimination, the patients were not disclosing their HIV status to their family, which in turn influenced their non-adherence to TB medication.⁽⁹⁸⁾

Beliefs about the disease and treatment, such as perceived wellness or cured, perceived risk, and perceived barriers over the benefits, were influencing factors for non-adherence to TB medication.^(13, 93, 105, 107) One study conducted in Addis Ababa reported that a patient's belief in curability and severity of TB in the presence of HIV infection would make them less likely to be non-adherent.⁽¹⁰⁹⁾ Another study in Addis Ababa found that the perceived risk of discontinuing TB medication was the reason for adherence while perceived wellness was the reason for patients have the intention to discontinue TB treatment.⁽⁹³⁾ One study conducted in Southern region of Ethiopia also reported that belief in traditional healing influenced non-adherence to TB medication.⁽¹¹³⁾

2.2.3. Economic factors

The patient's economic constraints (which impact the financial burden) were the main economic factor that influences non-adherence to TB medication.^(13, 97, 108, 109) Economic constraints limit the patient's ability to have adequate food which influences non-adherence.^(109, 113) The cost of medication other than TB medication is also a factor for non-adherence in one study conducted in Southern region of Ethiopia. Another study conducted in Southern region of Ethiopia reported that the patient being not employed was associated with non-adherence.⁽¹¹⁶⁾

2.2.4. Health system factors

Poor healthcare provider-patient relationship with communication gaps was a major factor that influences non-adherence to TB medication.^(94, 109, 110, 113, 116) For example, one study conducted in Addis Ababa among loss to follow up patients reported that the healthcare providers were seen as disrespectful of their patients and less committed to their profession.⁽¹¹⁰⁾ The quality of healthcare service and a patient's satisfaction with healthcare service affect non-adherence to TB medication.^(96, 108, 116) When patients perceived that they received less professional care, less time spent with the healthcare providers, and waiting a long time to receive the healthcare service, they were more likely to be non-adherent.^(96, 110, 110, 110, 110, 110)

^{116, 117)} Health information/education is also crucial for adherence: a few studies showed that the patients who did not receive health information/education from health facilities were more likely to be non-adherent.^(95, 98, 101) Additionally, one study done in Addis Ababa found that lack of cues to action were reported as a factor for non-adherence.⁽¹⁰⁷⁾ Lack of supervision and healthcare provider incapable to manage the patient's illness was also reported as influencing factors for interruption and default from TB treatment.⁽¹¹⁵⁾

2.2.5. Therapy factors

Many studies reported that drug side-effects were the major therapy-related reason for nonadherence to TB medication.^(95, 96, 102, 103, 105) Pill burden was also reported as a factor for nonadherence to TB medication among active TB and TB-HIV co-infected patients.^{(109, 113).} The presence of more than one co-morbidity including TB-HIV co-infection was also reported as a factor for non-adherence to TB medication.^(19, 94, 96, 105, 114) One study conducted in Addis Ababa also found that being on Antiretroviral Therapy (ART) was a factor for non-adherence to TB medication.⁽¹³⁾ Symptom presence after initiation of anti-TB treatment and slow progression of the health status were also found as non-adherence factors.^(18, 19, 99, 105) Being in the continuation phase of the treatment (after the initial 2-month clinic-based treatment period) was a factor for non-adherence and default.^(19, 94, 103, 111, 118) This might be due to the patient's perceived wellness or cured and because of a daily DOT was not implemented after the first two months of treatment in Ethiopia.

2.2.6. Lifestyle factors

Alcohol consumption was reported as a factor that influences non-adherence to TB medication in three studies.^(13, 94, 96, 106) Cigarette smoking ⁽⁹⁶⁾ and Khat chewing were also found as factors associated with non-adherence.^(96, 106)

2.2.7. Geographical access factors

Healthcare inaccessibility from residence location was a major geographical access factor for non-adherence and default from TB treatment.^(97, 99, 101, 105, 108, 113, 117) Due to inaccessibility,

patients were unable to keep regular clinic appointments and follow up treatment in two studies, conducted in Addis Ababa and Amhara regions.^(96, 112) Distance of the health facility was related to transportation cost, which was also a factor for non-adherence.^(18, 100, 101, 104)

2.3. Adherence Interventions

Adherence interventions that enhance medication adherence and treatment outcomes can be categorized as follows: patient education, psychological interventions, reminders and tracers, incentives and enablers, staff education, patient-centred care, and digital health.^(5, 21, 119) Adherence interventions and their effectiveness on medication adherence and treatment outcomes are summarised below based on a systematic review and meta-analysis conducted by Alipanah et al. 2018.⁽²¹⁾

2.3.1. Patient education

Patient education and counselling are a recommended adherence intervention to improve medication adherence and treatment outcomes. The counselling here is educational, not psychological.⁽⁵⁾ A systematic review showed that patient education and counselling interventions, involving both oral and written educational materials along with counselling, demonstrated positive associations with higher rates of treatment completion, cure, and adherence. However, these interventions did not influence on mortality, treatment success, failure, or loss to follow-up.⁽²¹⁾

2.3.2. Psychological supports

Psychological supports such as counselling for alcohol cessation and establishing a group and clubs as social support network are used to improve adherence and treatment outcomes.⁽⁵⁾ A systematic review showed that the utilization of support groups was associated with higher rates of treatment completion and lower rates of treatment failure and loss to follow-up.⁽²¹⁾

2.3.3. Reminders and tracers

Reminders and tracers are also be used to improve mediation adherence and treatment outcomes.⁽⁵⁾ A systematic review indicated that the use of reminders and tracers were associated with higher rates of treatment success, cure, medication adherence, and sputum conversion at two months. Additionally, there were lower rates of the development of drug resistance and loss to follow-up with the implementation of reminders and tracers.⁽²¹⁾

2.3.4. Incentives and enablers

Incentives and enablers such as food or financial support are a recommended patient support interventions to improve medication adherence and treatment outcomes.⁽⁵⁾ A systematic review found that incentives and enablers were associated with favourable outcomes, including lower rates of mortality, treatment failure, and loss to follow-up, as well as higher rates of treatment success, completion, cure, and sputum conversion at two months.⁽²¹⁾

2.3.5. Staff education

Staff education is also a recommended adherence intervention to enhance medication adherence and treatment outcomes. Staff education such as adherence education for staff, peer training for lay health workers, reminders to initiate adherence discussions, and the provision of educational tools and aids for decision-making.⁽⁵⁾ A systematic review showed that staff education interventions were associated with positive outcomes, including higher rates of treatment success and lower rates of loss to follow-up in the studies.⁽²¹⁾

2.3.6. Patient-centred care

Patient-centred care along with DOT was also used as adherence intervention in several studies to enhance medication adherence and treatment outcomes. ⁽²¹⁾ The interventions compared patient-centred directly observed treatment (DOT) with self-administered treatment (SAT). Patient-centred DOT was demonstrated benefits compared to DOT alone, showing lower rates of loss to follow-up and higher rates of treatment success and cure. When compared with SAT, patient-centred DOT was associated with higher rates of treatment

success, completion, cure, and sputum conversion at two months. However, no studies were found that directly compared patient-centred DOT to patient-centred SAT.⁽²¹⁾

2.3.7. Digital health

Digital health is one of the recommended adherence interventions to improve medication adherence and treatment outcomes.⁽²³⁾ Digital technologies such as daily reminder SMS texts or phone calls, video observed treatment (VOT), wireless pillbox/SMS reminder systems, and electronic medication monitor boxes are used for promoting adherence. A systematic review showed indicated that digital health interventions, such as SMS reminders, wireless pillbox/SMS reminder systems, and electronic medication monitor boxes and electronic medication monitor boxes, were associated with positive outcomes in terms of treatment completion rates, cure rates, and adherence, while VOT did not show a significant impact compared to DOT in treatment completion and mortality rates.⁽²¹⁾

The current research implemented a mobile health intervention by designing a combination of SMS text messages and phone calls. The mobile health intervention designed to have multiple benefits such as facilitate awareness creation through SMS text messaging and phone calls, remind the patient to take their medication and clinic appointment, facilitate the communication between healthcare professionals with patients and enable to reach the service to the patients where they live.

2.4. Mobile health Intervention

2.4.1. Definition of terms

The World Health Organization (2015) defined the following terms as:

mHealth (mobile Health): "the provision of health service and information via mobile technologies such as mobile phones, tablet computers and personal digital assistance (PDAs)."⁽¹²⁰⁾

eHealth (electronic health): "use of information and communication technology (ICT) for health and health-related activities." ⁽¹²⁰⁾

Digital health: "a collective term for eHealth and mHealth technology."(120)

2.4.2. Effectiveness of mobile health intervention

There is contradictory evidence about the effect of SMS text message interventions on treatment adherence and outcomes. Some studies showed that SMS text messages did not provide an improvement of TB medication adherence ^(69, 81, 121-123) and treatment outcomes,^(69, 121, 122) while other studies found SMS text messages were effective at improving clinic attendance^(22, 81) and treatment outcomes.^(124, 125)

In four randomised trials conducted in Pakistan,⁽¹²¹⁾ Argentina, ⁽⁶⁹⁾ China ⁽¹²³⁾ and Cameroon ⁽¹²²⁾ found that the use of SMS text did not show statistically significant improvement in medication adherence. However, one randomised trial in China found that one-way daily SMS text led to a statistically significant lower rate of interrupted treatment and missed does,⁽¹²⁵⁾ as well as another trial in China which showed a daily two-way SMS text led to a statistically lower patient lost to follow-up.⁽¹²³⁾ Medication event monitoring a combination with daily two-way SMS text also showed a statistically significant improvement in medication adherence.⁽¹²³⁾

In three randomised trials conducted in Pakistan,⁽¹²¹⁾ Argentina, ⁽⁶⁹⁾ and Cameron ⁽¹²²⁾ were found that a daily SMS text did not significantly improve treatment outcomes. Conversely, one clustered randomised trial in China and one non-randomised trial in South Africa found that SMS text led respectively to a statistically higher rate of smear conversion and cure rate,⁽¹²⁵⁾ and a higher treatment completion rate and a lower interrupted treatment and missed dose.⁽¹²⁴⁾ Furtherly, one randomised trial conducted in different geographic settings and one South African non-randomised study found that SMS text along with self-administered therapy did not show a statistically lower effect on completion of treatment ⁽²⁵⁾ and treatment outcomes ⁽²⁶⁾ compared with DOT. A phone call reminder was also found effective in improving the TB treatment success rate in one randomised trial study done in Thailand among MDR-TB and

DR-TB patients.⁽¹²⁶⁾ The individual studies summary findings are presented in the Table 2. A systematic review found that mobile-based SMS text messaging had a modest effect on TB treatment success rate, but the quality of the evidence was low.⁽¹²⁷⁾

2.4.3. Interactivity of mobile health intervention

Some evidence has shown that an interactive SMS texts may have a better impact than using simple one-way SMS reminders in improving medication adherence ^(122, 128-130) as it has potential benefits of improving the relationship between patient and healthcare provider.^(27, 122) However, there is evidence that two-way SMS reminders are also unable to improve treatment outcomes and adherence to medication ⁽¹²¹⁾ (¹²³⁾ (¹²³⁾ (⁶⁹⁾ and one-way SMS reminders were found to improve treatment adherence ⁽¹²⁵⁾ and treatment outcomes.⁽¹²⁵⁾ (¹²⁴⁾ The characteristics of SMS interventions are summarised in the Table 3.

It is suggested that a phone call to the patient along with SMS for medication reminders may enhance the healthcare provider and patient relationship, and treatment adherence^(121, 122) and thereby improve TB treatment success.⁽¹²⁶⁾ The WHO suggests investigating creatively how SMS text messages can influence adherence by combining it with other digital health interventions or blended digital technology such as video, and voice calls.⁽²³⁾ Therefore, it is recommended to use a phone call to improve a patient's treatment adherence, clinic attendance and treatment outcomes, in addition to SMS text messages, and this can make the SMS text messages intervention more interactive.^(121, 126)

2.4.4. SMS text messages frequency, time, and duration

Most of the previous interventional studies implemented daily SMS text messages to prompt TB medication adherence and improve treatment outcomes. Most of the studies also implemented the SMS text message intervention throughout the entire six-month period of TB treatment (see Table 3). Two randomised trials explicitly mentioned that the SMS text messages were sent in the morning time. ^(69, 125) Two randomised trials reported that the SMS text text messages were sent based on the patient's preferred time.^(121, 123) In one retrospective

analysis study, the SMS text message was sent when the patient did not open the monitoring device (SIMpills) to take the medication.⁽¹²⁴⁾ However, one mixed-methods study ⁽²⁶⁾, and two randomised trials did not report the specific times of the SMS text messages.^(25, 122, 131)

2.4.5. Feasibility and acceptability of mobile health intervention

On the other hand, SMS based adherence support intervention studies conducted at different geographical settings found that SMS text is acceptable and feasible to support TB medication adherence. The reasons for acceptance of SMS reminders were they feel cared for and supported,⁽⁶⁹⁾ and help them to avoid missed medication and remind them to refill their medication.^(67, 68, 132) In Tanzania, a real-time medication monitoring study also showed that patients were satisfied with SMS text reminders.⁽²⁷⁾ SMS intervention was also found as feasible in terms of mobile access and technical familiarity with SMS text messaging. ^(69, 133) Generally, SMS text messaging is inexpensive, uses existing infrastructure, is easy to use, has high acceptance, satisfaction and is valued by patients and healthcare professionals.⁽³⁰⁾ The acceptability and feasibility of SMS text intervention studies are summarised in the Table 4.

Trial ID	Location	Sample size	Study design	Population	Intervention	Control	Outcome	Findings
Mohammed 2016 ⁽¹²¹⁾	Pakistan	2207	Individual RCT, open label	Newly diagnosed	Two-way SMS reminders	DOT	Primary outcome: Treatment success-cured (i.e., sputum smear or culture negative in the last month) or treatment completion, death, transferred. Secondary outcome: self- reported adherence	Treatment success ARM1=83% and Control=83% Default; ARM1=10% and control=9% Died; ARM1=2% and control= 2% Treatment failure; ARM1=2% and control=3% Additionally, there was no significant difference on self- report medication difference
Farooqi RJ. Et al. 2017 ⁽¹³⁴⁾	Pakistan	148	Individual RCT	Newly diagnosed patients	Daily one way SMS reminder	DOT	Primary outcome: default (lost from treatment for 2 consecutive months)	Daily SMS text messages had no significant difference on default of TB patients, but lower number of default cases were reported in the intervention group compared to control group.
Liu 2015 (123)	China	4292	Cluster RCT	Newly diagnosed	Arm1-Two- way SMS reminders Arm2- medication monitor Arm3-SMS reminders plus	Self- administered treatment, family supervision, or DOT	Primary outcome: Poor adherence-at least 20% dose missed. Secondary outcome: Poor adherence-at least 47% dose missed (7/15 doses)	Arm1= 27.3%, Arm2=17.00%, Arm3=13.90 Control=29.90 SMS reminders not only have a statistically significant difference in adherence, but medication monitors and the combination of SMS

Table 2: Randomised clinical trials of mobile health interventions for TB extended from Lester et al. 2019 (27) and Nglazi et al. 2013(81)

					medication monitor		Drimonu cu tacara	reminders and medication monitors statistically significantly improved medication adherence. However, SMS reminders had a positive effect on lost to follow up. Audio reminded medication monitor box and the combination of medication monitor and text messages had significant improvement on TB medication adherence.
Bediang G. et al. ^(122, 131)	Cameroon	279	Individual RCT Single-blind of healthcare providers	Active TB	One-way SMS reminders	DOT	Primary outcome: Negative Microscopy at 5 months cured (microscopy) at 6 months. Secondary outcome: Adherence to medication and appointment	Arm1=81% and Control=74.6% SMS reminders did not provide a statistically significant difference in treatment success and adherence to medication and clinic appointments.
Belknap 2017 ⁽²⁵⁾	US, Spain, Hong Kong, and South Africa	964	Individual RCT	Latent TB	Arm1-Weekly SMS Reminders plus self- administered treatment	DOT (monthly)	Primary outcome: Treatment completion -11 or more doses within 16 weeks Secondary outcome: adverse events	Arm1=76.4%, Arm2=74.0% Control=87.2% Self-administered treatment and combination of self- administered treatment and SMS reminders not statistically lower compared to DOT.

					Arm2- self- administered treatment		Drimony outcomes	Arm1 06 20/ cod
Fang 2017 ⁽¹²⁵⁾	China	350	Cluster RCT	Active TB	One-way SMS reminders	DOT	Primary outcome: Treatment completion Secondary outcome: Missed dose, interrupted treatment, a reexamination of sputum after 2-, 5- and 6-months of treatment	Arm1=96.3% and Control=86.4% A significantly higher treatment completion rate in the SMS intervention group compared to DOT. Both the Interrupted treatment rate and missed dose rate significantly lower in SMS intervention compared with DOT
Hüsler 2005 ⁽²⁶⁾	South Africa		Mixed methods design	Active TB, clinic staff, TB experts	One-way SMS reminders for self- administered patient	DOT	Primary outcome: treatment outcome	Cure rate: Arm1=62.35%, control=66.44% Completion rate: Arm1=10.59%, control=3.0% Treatment success rate: Arm1=72.94%, control=69.4% Daily SMS reminders have similar effect treatment outcome effect with clinic- based DOT
Broomhead 2012 ⁽¹²⁴⁾	South Africa	120	Retrospective analysis	Active TB patients	One-way SMS SIMpill	DOT	Primary outcome: smear conversion rate and TB cure rate	Smear conversion rate Arm1=62.5% and control=38.4% Cure rate Arm1=75.0% and control=32.3% Smear conversion rate and cure rate were significantly

Iribarren, 2013 ⁽⁶⁹⁾	Argentina	37	Mixed method design including RCT	Active TB	Two-way SMS intervention	SAT (monthly)	Primary outcome: Feasibility, acceptability and self-report treatment adherence Secondary outcome: Microscopy test result from positive to negative, treatment outcomes	higher among SMSpill based medical adherence support. Arm1=77% Self-reported adherence Arm2=53% calendar-based self-report A similar finding obtained for sputum smear or culture between intervention and control group.
Kunawararak, 2011 ⁽¹²⁶⁾	Thailand	38 MDR- TB 61 non MDR- TB	Randomised Clinical trial	MDR- TB and non MDR-TB	Daily Phone call reminder	DOT	Primary outcome: Treatment outcomes	In sputum conversion rate at one-month Among MDR-TB Arm1=90% and control group =20% (statistically significant). Among non MDR_TB Arm1 =37% and control group= 52% (not statistically significant). In treatment success Arm1 was a statistically significant higher in both MDR-TB and non MDR-TB (100%) compared with control group (73.7% and 96.7%)

Table 3: The characteristics of SMS text message interventions in randomised control trials extended from Lester et al. 2019⁽²⁷⁾ and Nglazi et al. 2013⁽⁸¹⁾

Trial ID	SMS Intervention	Intervention duration (Months)	Message Frequency	Motivational message	Message content	Message time	Follow-up reminder
Mohammed 2016 ⁽¹²¹⁾	Two-way SMS reminders	6	Daily	Yes	"Your health is in your hands. Take your medication and remember to respond by SMS or a missed call."	Reminders sent based on preferred time of patients	A second reminder was sent if the patient did not react within two hours. After two more hours of no answer, a third and final reminder for the day was delivered. Participants who had not responded in seven days were contacted by members of the research team.
Liu 2015 ⁽¹²³⁾	Two-way SMS reminders	6	Daily	No	"Please take the medication on time"	Reminder sent based on preferred time patients	Patients are expected to replay up to three SMS messages, and reminders stop once the patient answers.
Bediang G. et al. ^{(122,} ¹³¹⁾	One-way SMS reminders	6	Daily	Yes (every two weeks)	"Good morning it is important to take your drug against TB every day" "Good morning, taking drug daily increases your chance of healing"	Not Reported	
Belknap 2017 ⁽²⁵⁾	One -way SMS Reminders	4	Weekly	No	Not Reported	Not Reported	
Hüsler 2005 ⁽²⁶⁾	One-way SMS reminders	6	Daily	No	Not Reported	Not Reported	
Broomhead 2012 ⁽¹²⁴⁾	One-way SMS reminder	6	Daily	NO	Not Reported	Morning	When patient don't take their medication

						When patient	
						did not take	
						their medication	
Fang	One-way SMS						Contained TB-related knowledge
2017 ⁽¹²⁵⁾	reminders	6	Daily	No	Not Reported	Morning	for establishing a healthy
2017	Terminuers						lifestyle.
Iribarren,				Yes			Confirmation of message receipt
2013 ⁽⁶⁹⁾	Two-way SMS	2	Daily	(twice	Not Reported	Morning	and a reminder sent to the
2013(00)				weekly)			patient if they did not send

Source	Location	Sample size	Study design	Population	Intervention	Outcome measure	Major findings
Iribarren, 2013 ⁽⁶⁹⁾	Argentina	37	Mixed- methods design	Active TB	Two-way SMS intervention	Feasibility, acceptability for treatment adherence	SMS intervention was found feasible as it has high access to mobile phones, patient familiarity with texting, and low refusal. SMS was also found as acceptable as patients had a feeling of care, supported and responsible.
Albino 2014 (67)	Peru	4 FGD	Qualitative	Active TB	One-way SMS	Perception and acceptability for treatment adherence	Patients have positive perception and accept mobile based SMS for improving medication adherence through transmit motivational text and simple reminders.
Nhavoto 2017 ⁽⁶⁸⁾	Mozambique	140 patients 40 Health professional	Qualitative	Active TB & Healthcare professional	Two-way SMS	Usefulness, perceived benefits, ease of use, Perception satisfaction, risks of the SMS	SMS was found useful and reliable because majorly reduce failure to collect medication and avoiding a missing appointment. Patients were confident on SMS and two-way SMS (questions and appropriate answer) likable features. Unintentional disclosure of health status was the risk in case patient use shared mobile phone.
Hoffman 2010 ⁽¹³³⁾	Kenya	11 patients 3 Health professionals	Mixed- methods design	Active TB & Healthcare professionals	Mobile Video capture and transmission (Mobile DOT)	Technical feasibility and patient preference and receptivity	Mobile DOT was found to be technically feasible, and both patients and health professionals preferred communicating with one another and were open to remote MDOT and health messaging.
Sumari-de 2016 ⁽¹³⁵⁾	Tanzania	10 Patients	Mixed- methods design	Active TB	One-way SMS reminder at	Process measure,	Patients reported that they were satisfied with SMS reminder, but few reported that SMS reminder may be seen by others.

 Table 4: Acceptability and feasibility of mobile health interventions for TB treatment support

					real-time medication monitoring	adherence, perception	However, patients were disturbed by incorrect SMS reminders.
Mohammed 2012 ⁽¹³²⁾	Pakistan	30 patients	Qualitative	Active TB	Two-way SMS	Acceptability and engagement	Interactive two-way SMS reminders were acceptable by patients with TB support medication adherence. However, the average response rate for SMS was 57% (62% during the first message of the week and 49% during the last ten days of messaging)

2.5. Adherence definition and measurement

2.5.1. Definition of Adherence

Measuring adherence is a complex task for the TB control program because of the various meanings given for adherence, as well as various methods of measurement. The term 'adherence' is extensively used in the field of psychology and medicine. The very general definition of adherence by Howren (2013) is "the extent to which an individual's behaviour coincides with health-related instructions or recommendations given by a healthcare provider in the context of a specific disease or disorder".⁽¹³⁶⁾ The World Health Organization (WHO) defines adherence as "the extent to which a person's behaviour – taking medication, following a diet, and/or executing lifestyle changes, corresponds with agreed recommendations from a healthcare provider".⁽¹³⁷⁾ Osterberg and Blaschke (2005) define adherence to a treatment program as "the extent to which patients take medication as prescribed by their health care providers".⁽¹¹⁹⁾

A review of literature by Valencia (2017)⁽¹³⁸⁾ summarised the definition of adherence related to TB treatment at different levels of adherence. Most commonly used definitions of adherence to TB medication are:

- Lost to follow-up (former called default)- "treatment interrupted for two consecutive months or more" which measured through direct observation of the patient during treatment regimen.
- **Recent non-adherent** "missing one dose in the last four days" measured by the number of doses taken through pill count or report.
- Non-adherent in the last month- "missing 10% of prescribed doses" measured by the number of doses taken through pill count or report.
- Adherent- "all test positive" measured by fluid detection.

- Adherent- "attended all visits" measured by the number of visits attended through follow-ups or medication refill visits.
- Adherent- "at least taking 70-90% of the prescribed doses" measured through percentage/number of doses taken by pill count or report.

2.5.2. Adherence measurement

Adherence has been measured in a variety of ways in addition to the definition, and there is no gold standard measurement for adherence.⁽¹³⁷⁾ The methods of measuring adherence broadly classified into direct and indirect methods.⁽¹¹⁹⁾ Valencia (2017) summarised these different adherence methods of measurement as below.⁽¹³⁸⁾

2.5.2.1. Direct methods

- Directly Observed Treatment (DOT)- Institutional-based DOT (IBDOT) which made at the health facility level is the most commonly used form of DOT. IBDOT is expensive and logistical barriers. Community-based DOT (CBDOT) in which observation performed by family or community members or community health workers. Video Observed Treatment (VOT) has also been implemented in the TB control program which records and transmits medication ingestion video to health care provider so that enable to watch the patient remotely. VOT is not feasible in low resource settings because it requires smartphone and internet accessibility.⁽¹³⁸⁾
- Detection of drug and metabolites' concentration in urine Urine sample colour change occurs with the presence of rifampicin (RIF). However, it is less effective because of reduction in RIF present in urine a few hours after drug ingestion. The Arkansas and IsoScreen methods detect isoniazid (INH) metabolites in urine. Arkansas and IsoScreen methods are fast, low-cost and easy to interpret, but limited to recent drug use.⁽¹³⁸⁾

 Ingestible sensor-based system – This is an ingestible drug sensor, the signal of which is detected by on-body wearable sensor upon ingestion. It can verify the use of pills, but it is expensive especially in high TB burden settings.⁽¹³⁸⁾

2.5.2.2. Indirect methods

- Patient self-report- This includes questionnaires, scales, interviews and patient diaries. The most commonly used questionnaire types are; Morisky Medication Adherence Scale (MMAS),⁽¹³⁹⁾ Visual Analogue Scale (VAS),⁽¹⁴⁰⁾ Medication Adherence Rating Scale (MARS),⁽¹⁴¹⁾ and Brief Medication Questionnaire (BMQ).⁽¹⁴²⁾ The AIDS Clinical Trial Group adherence questionnaire (ACTG)⁽¹⁴³⁾ is also used for TB medication adherence assessment. ⁽¹⁴⁴⁾ These approaches can indicate long-term adherence status, but they are unreliable, which is one of the most significant drawbacks of self-reported adherence assessments. Self-report adherence measures are also open to different types of biases such as recall bias, social desirability bias, assuming the patient taking the medication as reported, as well as interviewer skill and construction of questions affecting the adherence measurement.⁽¹³⁸⁾
- Pill count and refill visit If there are any remaining pills in the blister pack of pill bottles, patients bring them to their controller, who compares the total number of pills given with the number of pills left in the blister pack. These methods are less expensive, easy to measure and can show long term adherence. However, it has limitations such as patients may dump their medication deliberately or forget bringing blisters during visits.⁽¹¹⁹⁾ Recording the patient's appointment attendance for refill medication is therefore not accurate as a measure of ingestion of medication.⁽¹¹⁹⁾
- Electronic pill bottles- Medication Event Monitoring System (MEMS) are one of the most reliable markers of adherence and can predict long-term adherence. Other adherence methods have been validated using MEMS as a reference. However, MEMS is expensive especially to use in low resource settings and it does not verify the actual medication intake. ^{(119) (108)}

2.5.3. Selection of adherence measurement

When testing an intervention with randomised trials, correctly measuring outcomes in the same way for both the experiment and control groups is crucial. For example, in an Argentinian individual randomised study, adherence was measured in flawed way; it was measured differently in the intervention group (measured by replying SMS confirmation) and the control group (measured by calendar diaries and self-report), and a large proportion of patient's information was missed by the calendar diaries. ⁽⁶⁹⁾ It is therefore important to carefully select the measurement tools based on their pros and cons.

Biochemical analysis of anti-TB drug metabolites in urine is a modern approach, and one method is using the Arkansas method of analysis of isoniazid (INH) metabolites in urine. This is a useful and reliable laboratory-based assay, which has been used in many studies to determine drug treatment adherence.⁽¹⁴⁵⁾ However, the required laboratory equipment and expertise make difficulties for this test. An alternative is to use a point of care test that incorporates the chemicals for the Arkansas method into a simple disposable plastic testing device. ⁽¹⁴⁶⁾ The test called IsoScreen which has been used in Tanzania, ^(144, 147) Brazil ⁽⁸⁶⁾ and Canada.⁽⁸⁷⁾ IsoScreen is a rapid and safe point-of-care test that provides accurate detection of adherence for the patient.⁽¹⁴⁶⁾ The existing evidence shows that Isoscreen has high sensitivity (83%-100%) and specificity (over 95%) in the detection of adherence to TB medication.⁽⁸⁶⁻⁸⁹⁾ For example, one study indicated that the sensitivity and specificity are very high especially at 12 and 24 hours for purple or blue color change (100% and 99%) and at 48 hours for green color change (78% and 100%). Isoniazid metabolites detection by IsoScreen was not influenced by demographic characteristics, smoking status, alcohol consumption and additional medication uses.⁽⁸⁹⁾

The accuracy of adherence measurement is very crucial to detect the true effect of an intervention on medication adherence. The most commonly used method to measurement adherence is considered to be the indirect method. ⁽¹⁴⁸⁾ It is necessary to have a valid and reliable adherence measure with a low cost and acceptable to use in a limited-resource

setting. Indirect measures can be used over a longer period of adherence, but it assumed that the patients are actually taking their medication as they reported.^(138, 148) Direct method adherence measures that detect drugs and metabolites concentration in urine are quick, easy, and inexpensive as compared with MEMS. Evidence shows that the IsoScreen NIH metabolites test has been effective in detecting non-adherence ^(88, 138) and it can also be used to validate the indirect adherence measures in poor resource settings.⁽⁸⁸⁾

In the current study, we choose to use the Isoscreen NIH test over urine color change by the presence of rifampicin (RIF) because evidence showed that the RIF effect is very shortlived (peaking 2-6 hour) and is only seen in <50% of patients.⁽⁸⁹⁾ For example, one study found that the sensitivity of RIF was 43.5% compared with 93.2% in INH urine test.⁽⁷⁵⁾ The IsoScreen test is easy to use and convenient especially in busy outpatient clinics.

As IsoScreen test only indicates a recent medication adherence, other validated indirect methods were also used for showing long term effects of mobile-assisted adherence intervention. The previous studies showed the four scale Morisky medication adherence was reported as less sensitive among TB patients.^(88, 144) The revised eight scales Morisky medication adherence (MMAS-8) has better sensitivity and specificity compared to the four scales but is not licensed to individuals.⁽¹³⁹⁾ The AIDS Clinical Trial Group Adherence Questionnaire (ACTG) was validated among TB patients in Tanzania and is recommended for use in adherence interventions, along with other indirect methods of adherence.⁽¹⁴⁴⁾ The Visual Analogue Scale (VAS) is a recommended adherence scale for clinical practice, demonstrating a significant correlation with other methods of adherence measurements.⁽¹⁴⁹⁾ Thus, in the randomised trial, adherence was measured using the combination of direct method of the IsoScreen test and indirect methods: Visual Analogue Scale (VAS),⁽¹⁴⁰⁾ and AIDS Clinical Trial Group adherence questionnaire (ACTG).⁽¹⁴³⁾

CHAPTER 3: METHODOLOGY AND METHODS

3.1. Introduction

This chapter is focused on the research methodology and methods, which is a critical aspect of the research project. The methodology includes various elements, such as the research philosophy, the researcher's position, the research paradigm, and the mixed methods design. Additionally, this chapter provides details of the study design, method of data collection and analysis for each study.

3.2. Research philosophy

3.2.1. Ontology

Ontology deals with the nature of reality. It is the theory of existence or reality.^(150, 151) The two major ontological positions are realism and anti-realism. Realism states that reality exists independent of our perceptions or theories. Whereas anti-realism rejects the belief that reality is independent of us or our consciousness.⁽¹⁵⁰⁾ In this study, the realist ontological position was applied in part because the effect of a mobile-assisted medication adherence support (Ma-MAS) intervention on patient's adherence was empirically examined, and the anti-realism ontological position was also applied because the patients' perceptions and experiences of the Ma-MAS intervention were studied with an interpretive research approach.

3.2.2. Epistemology

The concept of epistemology pertains to the meaning and the process of acquiring knowledge. Epistemology is concerned with the nature and justification of knowledge.⁽¹⁵⁰⁾ It asks what knowledge is? What is the justification for knowledge and how are the known and the knower related?⁽¹⁵⁰⁾ The most common epistemological positions raised in the literature are objectivism and constructivism. The Objectivist approaches view the meaning of realist existence as independent of consciousness (objective meaning). Objectivists understand the phenomenon using a positivist research approach with deductive reasoning and empirical

methods.⁽¹⁵⁰⁾ Whereas the constructivist approaches view meaning as socially and experientially constructed by an inductive reasoning and interpretive approach.⁽¹⁵⁰⁾ The constructivist supports the existence of multiple meanings of a phenomenon.

3.2.3. Researcher position

In this research, the phenomenon under investigation is adherence to the prescribed TB medication. An intervention was designed to improve the patients' adherence to medication. The phenomenon of 'adherence' or 'non-adherence' can exist independent of our consciousness or assessment. Adherence is defined as the act of taking the prescribed medication at a consistent time and regularly. In a positivist approach, adherence is measured and analysed statistically to determine whether it has been occurred. For example, adherence can be defined as "a test positive" measured by urine detection, and in self-reported measurement, adherence can be defined as "when a patient is taking 90% and above of the prescribed dose of a medication". The interpretivist would ask why adherence or nonadherence has occurred and what beliefs led to people adhering or not adhering. The interpretivist would also ask what the patients perceived and experienced during the adherence intervention. Thus, both the positivist and the interpretivist approaches can be used depending on the focus of the research questions. In this study, I used truth of knowledge as "what works at the time" and truth of knowledge is relative, not absolute, as indicated in the pragmatism research paradigm.⁽¹⁵²⁾ My primary interest is to solve the problem of adherence to TB medication through designing an intervention using a behavioural change model and techniques.

The epistemic questions here are: what is adherence or non-adherence? Is non-adherence to TB medication a problem? What is the justification for that conclusion? What are the causes of non-adherence? Why are people adhering or not adhering? How can resolve the problem of non-adherence to TB medication? How does the intervention improve patient's adherence to TB medication (mechanism of intervention)? What other factors explain the effectiveness of the adherence intervention? What do patients perceive and experience about the

adherence intervention? Some of these questions are answered using objectivist epistemology, others by constructivism, and others by a mixture of both. Therefore, the research is open to different world views, assumptions, and methods of investigation of the problem and focused on identifying a practical solution.

3.2.4. Research paradigm

The current research was conducted using the pragmatic research paradigm. It is important to understand two other research paradigms before explaining pragmatism. The positivist paradigm determines cause-effect relationships. In the positivist research paradigm, hypothesis and research questions are tested through deductive reasoning and statistical methods.⁽¹⁵²⁾ The epistemic position of the positivist research paradigm is objectivist. Constructivists hold the assumption that the participant constructs the meanings of the phenomenon being studied. Participants develop a subjective meaning for their experience towards certain object or things. Indeed, there may be multiple meanings given to the phenomenon being studied. Constructivism understands the meaning of the phenomenon through inductive reasoning and an interpretive approach. Thus, constructivism is often combined with interpretivism.⁽¹⁵²⁾

Pragmatism is centred on the problem, the action, and the consequences of the action.⁽¹⁵²⁾ In pragmatism, truth is "what works at the time", it is not a dichotomy between a reality independent of the mind or a reality within the mind.⁽¹⁵²⁾ Pragmatism accepts both singular and multiple realities that are open to inquiry to solve a problem while avoiding the controversies created by positivism and constructivism.⁽¹⁵³⁾ The pragmatist ontological position on reality is close to "existential reality" and refers to an experiential world with elements that are somewhat objective, somewhat subjective, or a mix of both.⁽¹⁵³⁾ The pragmatic approach allows for different worldviews, assumptions, and methods that are best suited to solve the problem.^(152, 153) In pragmatism, a mixed-methods research approach is dominant, with both quantitative and qualitative research data being used to identify solutions for the problem at

hand and the associated research questions.^(152, 153) Thus, pragmatism supports the use of mixed research methods and different methods of analysis to generate socially useful knowledge to identify practical solutions.^(153, 154) Pragmatism is an outcome-oriented approach that focuses on finding shared meaning and identifying practical solutions to the problems being investigated using both empirical and interpretive research approaches.⁽¹⁵⁴⁾ In a pragmatist approach, theories can be contextual and generalisable, and support the transferability of results to other situations.⁽¹⁵⁴⁾ In addition, theory can be developed before and after data collection, which is known as abductive reasoning;⁽¹⁵⁴⁾ meaning that the approach uses both inductive and deductive reasoning.

3.3. Mixed methods

3.3.1. Definition of mixed methods

Creswell John and Creswell David's (2018) text on research design, defines qualitative, quantitative, and mixed methods research as follows. Qualitative research is "an approach for exploring and understanding the meaning individuals or groups ascribe to a social or human problem,"⁽¹⁵²⁾ quantitative research is "an approach for testing objective theories by examining the relationship among variables,"⁽¹⁵²⁾ and mixed-methods research is "an approach of inquiry collecting both quantitative and qualitative data, integrating the two forms of data, and using distinct designs that may involve philosophical assumptions and theoretical frameworks".⁽¹⁵²⁾

3.3.2. Mixed methods evaluation

In this study, a mixed-methods evaluation design was used to develop and evaluate a mobileassisted medication adherence support intervention.⁽¹⁵²⁾ Mixed-methods design uses both quantitative and qualitative research data to answer the research questions. The mixedmethods design was employed through the Medical Research Council's (MRC) framework for the development and evaluation of complex interventions.

The Ma-MAS intervention was developed using an exploratory mixed methods design with a literature review and interpretive study using in-depth interviews (qualitative). A Delphi technique was used to validate the proposed Ma-MAS intervention using expert opinion (qualitative) and a survey (quantitative). The feasibility and acceptability of the Ma-MAS intervention were evaluated through a cross-sectional survey (quantitative) before the Ma-MAS intervention trial. An experimental design with a parallel group randomised control trial employed to evaluate the effectiveness of Ma-MAS intervention. A convergence mixed-methods design, composed of in-depth interviews (qualitative) and cross-sectional survey (quantitative), was used to evaluate the patients' perceptions, experiences, and adoption intention of the Ma-MAS intervention. The data collection, analysis, and interpretation were undertaken separately, but the results were discussed together (see Figure 2). The reasons for using a mixed-methods design with both quantitative and qualitative research approaches are to: describe the change process, explain how the intervention worked, assess whether the intervention was implemented as planned, evaluate the acceptance of the intervention, explain the benefits of the Ma-MAS intervention, receive patient feedback on the intervention design, and examine whether the intervention has a long-term desire.

The methods section presented below in study one and three contains material originally published article of protocol of a parallel group Randomised Control Trial (RCT) for Mobile-assisted Medication Adherence Support intervention.⁽¹⁵⁵⁾ The copy of published article is found in the Appendix Eight.



Figure 2: Schematic presentation of the mixed methods evaluation design

3.4. Study setting and period

The research was carried out in public health institutions in Addis Ababa, the capital city of Ethiopia. Addis Ababa has 10 administrative sub-city regions (see Figure 3). Based on population projections, the total population of Addis Ababa in 2017 was 3,433,999 with 1.624,999 males and 1,809,000 females.⁽¹⁵⁶⁾ In Ethiopia, TB treatment care is provided through direct observed treatment (daily and non-daily DOT). The DOT program has two phases, the intensive and the continuation phase of treatment. In the intensive phase, the patient takes the medication in front of the healthcare worker every day at the health facility. In the continuation phase, the patient is given the responsibility of taking the medication at home, and comes to a health facility at fixed regular intervals (weekly refill visits), which is known as non-daily DOT.⁽¹⁵⁷⁾ However, the daily DOT and non-daily DOT program schedules were disrupted as a result of the emergence of and response to the COVID-19 pandemic.⁽²⁰⁾ Thus, throughout this research study period, the regular program of daily DOT and non-daily DOT was not implemented; instead, self-administered treatment administration was implemented with irregular medication refill visit schedules. More information is presented in qualitative research findings in the Appendix Two. The research was conducted from August 27, 2020, to October 29, 2021.



Figure 3: Map of Addis Ababa, Ethiopia (Source: Shapefile from Ethiopian mapping Agency)

3.5. Study one: Development and validation of Ma-MAS intervention: An MRC framework approach

3.5.1. MRC framework for complex interventions

Behavioural interventions involving multiple interacting components are complex, and complex interventions are most commonly used in health research.^(76, 158) In 2000, the Medical Research Council (MRC) developed a framework to help the researcher to develop and evaluate complex intervention ⁽¹⁵⁹⁾ that was updated in 2008.⁽⁷⁶⁾ Complex interventions are commonly described as "interventions that contain several interacting components". However, complex interventions have also other features.⁽⁷⁶⁾ These are: the interconnected components in the intervention and control groups, the target behaviours difficulty, the target groups number in the intervention, the outcomes number and variability, and the flexibility or tailoring of the intervention.

Complex interventions may be effective when it is designed to the home-grown circumstance rather than entirely standardised, and develop systematically using evidencebased and proper behavioural theory and testing in a phased approach before evaluation. ⁽⁷⁶⁾ The framework has four elements in the development and evaluation process of complex intervention. These are (1) Development; (2) Feasibility and acceptability; (3) Evaluation; and (4) Implementation (see Figure 4). The intervention development process using this framework described in this section.


Figure 4: Key element in the development and evaluation process of complex Intervention adapted from Craig P. et al. 2008 ⁽⁷⁶⁾

3.5.2. Ma-MAS Intervention development process

To generate quality evidence from the trial of the mobile-assisted medication adherence support (Ma-MAS), the intervention is developed systematically with the best available evidence and using the behavioural model of information-motivation-behavioural skills (IMB)⁽¹⁶⁰⁾ and behavioural change techniques taxonomy.⁽¹⁵⁸⁾ Thus, the Ma-MAS intervention is developed and evaluated based on the Medical Research Council (MRC) framework for complex intervention in health care.⁽⁷⁶⁾ The development phase of the framework was adapted from Bleijenberg et al. (2018) to be a more specific to the development of an intervention.⁽¹⁶¹⁾ The adapted development phase of the MRC framework guideline has seven different phases in a nonlinear iterative process:

- 1. Phase 1: problem definition (identification and analysis)
- 2. Phase 2: systematically identifying the evidence
- 3. Phase 3: identifying and developing a theory
- 4. Phase 4: determining the needs
- 5. Phase 5: examining the practice

- 6. Phase 6: modelling process and outcome and
- 7. Phase 7: intervention design (see Figure 5).

The development phase of the MRC framework along with the means of evidence collection is summarised in Table 5.



Figure 5: Adapted MRC framework development Phase. Source: Bleijenberg et al. 2018 (161)

Table 5: Mobile-assisted medication adherence support intervention development phases: An adapted MRC framework approach⁽¹⁶¹⁾

Development phases	Methods		
Phase 1: Problem definition (identification and	Review of literature		
analysis)	 Qualitative descriptive research 		
	Review of literature		
Phase 2: Systematically identifying the evidence	 Qualitative descriptive research 		
	Quantitative survey research		
Phase 3: Identifying or developing theory	Review of literature		
Phase 4: Determine the needs	Review of literature		
Filase 4. Determine the needs	Qualitative descriptive research		
Phase 5: Examine the current practice and	Qualitative descriptive research		
context			
	 Modelling a prototype of intervention 		
Phase 6: Modelling process and outcome	components		
	 Identifying interrelation with the outcome 		
	Experts' opinion through Delphi technique		
Phase 7: Intervention design	Experts' opinion		
Thase 7. Intervention design	Refine the intervention		

3.5.2.1. Phase 1: Problem identification and definition

The problem of non-adherence to TB medication has been identified and defined through a review of the literature conducted in Ethiopia where the study was conducted. The literature review is published under the title of "Factors Influencing Patient Adherence to Tuberculosis Treatment in Ethiopia; A Literature Review" (see Appendix Eight).⁽⁶⁾ Along with problem identification and definition, it is recommended to describe the context for an intervention, such as socioeconomic background, the health service system, and the characteristics of the patient and their influence on adherence and the adherence intervention which would benefit the success of the intervention.⁽⁷⁹⁾ A further qualitative descriptive research was conducted to extend on the literature review evidence. The qualitative research findings are presented in the Appendix Two. The prototype SMS text messages are developed based on factors influencing adherence to TB medication. The prototype SMS text messages are found in the Appendix Three.

3.5.2.2. Phase 2: Identifying the evidence base

Evidence has been collected through a review of literature, to identify the effectiveness of mobile health SMS interventions including acceptability, feasibility, and characteristics of the intervention. A further acceptability and feasibility study was conducted using qualitative descriptive and quantitative survey research before the intended trial was implemented. This phase aimed to identify past interventions and their effectiveness, acceptability and feasibility of the target groups of the interventions, as well as the features of the intervention. In the literature review evidence, there was contradictory evidence regarding the effectiveness of SMS text message interventions on TB medication adherence and treatment outcomes. Nevertheless, the SMS text messages and phone calls intervention for TB medication adherence promotion had high acceptance and was considered feasible to implement. This information is presented in the literature review, in chapter five of study two results and in the Appendix Two.

3.5.2.3. Phase 3: Identifying or developing theory

The behavioural change model that can best explain the proposed intervention's mechanism of action has been identified through the review of literature. This phase of development aims to identify the theory or model that can best explain the intervention mechanism of action. Information, motivation, and behavioural skills model was selected⁽¹⁶⁰⁾, and the relationship between the model constructs with adherence behaviour was logically connected based on factors influencing TB medication non-adherence (see Figure 6).

Information, Motivation and Behavioural Skills (IMB) model

The information, Motivation and Behavioural skills (IMB) model was applied to develop the mobile-assisted medication adherence support intervention (see Figure 6). The IMB model is developed based on the analysis and integration of social and health psychology theories that consider the psychological determinants of adherence, and focuses broadly on information, motivation and behavioural skills factors linked to adherence. ⁽¹⁶⁰⁾ The IMB model explains the

relationships between these constructs and adherence. The model's assumption is that adherence related information, motivation and behavioural skill are the major causes for non-adherence. According to this model if a patient is informed, motivated to act and has the required behavioural skills to act, he/she is highly likely to adhere to TB medication and thereby will have a good treatment outcome.⁽¹⁶⁰⁾ This model was used to guide the development of SMS text messages, and phone calls designed to support the SMS text messages in the Ma-MAS intervention.

Information: Information is an antecedent construct for adherence behaviour. Information about TB and its treatment can influence adherence behaviour directly and indirectly mediated through individual motivation and behavioural skills.⁽¹⁶⁰⁾ Information about TB and its treatment regimens, when and how to take their medication, what to do a dose is missed, the link between medication adherence and treatment outcomes, the link between substance use and treatment outcomes, the side effects of the drug, the misconception about TB and its medications and other barriers are the major information required for medication adherence.⁽⁶⁾

Motivation: The IMB model argues that an individual's motivation to adhere is influenced by both personal and social motivations. Personal motivation to adhere to medication includes attitude towards the treatment regimen and beliefs about the consequence of adherence and/or non-adherence to TB medication and the evaluation of these outcomes.⁽¹⁶⁰⁾ Thus, favourable attitudes and beliefs towards the treatment such as perceived curability, severity, risk, and benefit are assumed to be associated with adherence to TB medication.⁽⁶⁾ Individual perceptions of social support for adherence from significant others such as families, friends, and healthcare workers, as well as individuals' motivation to comply with the wishes of significant others are also associated with adherence to TB medication. The Ma-MAS intervention itself can also be considered as a social support to patients and positively influence adherence to TB medication.

Behavioural skills: The behavioural skills are an antecedent construct for adherence that also determines the success of adequately informed and motivated individuals to adhere to their medication. The activities here include setting goals and perceived abilities and self-management for integrating the treatment regimen into daily life, managing the side effects of the drug, acquiring social and health care support, self-reinforcement, and problem-solving skills. ⁽¹⁶⁰⁾ To develop the desired behavioural skills, the information provided through SMS text messages was supported by phone calls.

Moderating factors: Factors other than information, motivation, and behavioural skills are also expected to influence the model and adherence to TB medication. Although patients are properly educated, motivated to act and have the required behavioural skills, they may still not adhere to their medication because of other moderating factors such as economic constraints, inadequate food, co-morbidities, travelling distance, time and cost, social support, stigma and discrimination, psychological distress and others. Thus, moderating factors have a direct and indirect effect, through the influence of the constructs, on adherence to TB medication. However, individuals with a moderate level of moderating factors will theoretically respond to an adherence support intervention.⁽¹⁶⁰⁾ In the Ma-MAS intervention trial, the randomisation of participants into the intervention and control groups is expected to uniformly distribute moderating factors into both groups, ensuring that they do not affect the intervention's effectiveness.



Figure 6: Information, Motivation and Behavioural skills model to TB medication adherence adapted from Fisher et al. 2006 ⁽¹⁶⁰⁾

3.5.2.4. Phase 4: Determine the needs

The needs, preferences, perceptions, and capacities of the patient and healthcare providers with the identified problems and proposed solutions were determined through the review of literature and qualitative descriptive research using in-depth interviews. This approach generates information about which characteristics of the intervention are likeable and adaptable, and how the SMS texts and phone calls can be tailored regarding language, content, frequency, length, and time of the day. The frequency of SMS text messages and

phone calls was determined to be daily and weekly respectively based on the need assessment. It was also decided that all SMS text messages, and phone calls would be made in the national language of the country. The content of the SMS text messages suggested by health care providers and patients was included. It was decided that the daily SMS text message and weekly phone call are to be delivered in the morning from 7:30am to 8:00am. The intervention length was agreed to be eight weeks as a first trial to test the effectiveness of the intervention, though most healthcare professionals believed the SMS text messages and phone calls should be used throughout the entire six months of treatment. The details findings of the qualitative research are presented in the Appendix Two.

3.5.2.5. Phase 5: Examine current practice and context

Barriers and facilitators of the mobile assisted adherence support intervention were identified by examining the current practice and context of care from the patient's and healthcare provider's perspectives. This step assures that the functionality of the proposed mobile health intervention best fits with existing practice and the local context. Qualitative descriptive research, using in-depth interviews, was employed to examine the current practice and context. The barriers, facilitators and suggested solutions were considered in the design of the Ma-MAS intervention and some of these are especially useful for future scale-up. The barriers and facilitators of the mobile-assisted adherence support intervention and suggested solutions are presented in the qualitative research findings in the Appendix Two.

3.5.2.6. Phase 6: Modelling process and outcome

Modelling the active component of the mobile-assisted adherence intervention and its relationship with outcomes was refined with multidisciplinary team involvement. The initial set of prototype SMS text messages were drafted applying the taxonomy of behaviour change techniques⁽¹⁵⁸⁾ and linked to the constructs of IMB model ⁽¹⁶⁰⁾ (see Appendix Three). Factors associated with non-adherence to TB medication were logically connected to the behaviour change technique and IMB model. A Delphi technique was employed to validate the content

of the SMS text messages by senior experts and health professionals who work at ground level.

3.5.2.7. Phase 7: Intervention design

A full mobile-assisted medication adherence support intervention using SMS text messages and phone calls was developed. A total of 40 SMS text messages were selected for the Ma-MAS intervention for eight weeks of intervention (see Appendix Four). The experts gave their comments on the timing, frequency of information, duration of the intervention, and arrangement order of the SMS text messages. Based on the experts' opinions the final mobileassisted medication adherence support intervention design was prepared and tested.

3.5.3. Study design

To systematically develop and validate the mobile-assisted adherence support intervention, an exploratory mixed-methods approach was employed, together with a review of relevant literature, and qualitative descriptive research. The SMS text messages developed for Ma-MAS intervention were validated using a four round Delphi technique.

3.5.4. Participants

To systematically develop and validate Ma-MAS intervention different groups of participants were utilised. These participants were: (1) Adult TB patients aged 18 years and above involved in qualitative in-depth interviews and (2) Primary healthcare providers at service delivery points and experts at health offices who have experience related to TB prevention and control involved in qualitative in-depth interview (see Figure 7).

For validation of the SMS text messages, experts comprised of professional backgrounds in behavioural science, healthcare systems, pulmonary disease, and digital health were involved in the SMS text messages content validation in the round one to three Delphi technique. In round four, health professionals working as TB experts from health facilities and health offices in Addis Ababa, Ethiopia were invited to rate the relevance of the proposed SMS text messages.



Figure 7: Schematic presentation of study objective, design, and participants

3.5.5. Sample size and technique

The number of qualitative in-depth interviews with patients, healthcare providers and experts were determined based on idea saturation. That means interviews were continued until no further new information was obtained.^(162, 163) However, based on the literature review recommendation, the minimum number of interviews was not below 15.⁽¹⁶⁴⁾ Purposive sampling method was employed to select the patients for the in-depth interview.

Eight senior experts involved in the validation process of Delphi technique from rounds one to three. An additional 47 health professionals who work as TB focal person in 36 health facilities and TB prevention and control experts in 11 health offices in Addis Ababa, Ethiopia, were invited for further validation of the SMS text messages in Delphi round four. Both expert groups were selected purposively based on their professional background, work experience, and geographical location.

3.5.6. Data collection instrument and process

An in-depth interview guide was developed through review of relevant literature and the focus of the research questions, to systematically develop the mobile-assisted adherence support intervention. An open-ended interview guide and audio-recorder were used for qualitative data collection. The qualitative interview audio was recorded using digital recorder. The qualitative data was collected by the principal investigator who has experience with interpretive research and the culture and languages of the location. In-depth interviews with patients and health professionals were conducted at a convenient place and time for discussion and privacy.

To validate the SMS text messages a four round Delphi technique was applied.⁽¹⁶⁵⁾ This are:

Delphi round one: In the first round, the prototype SMS text messages were given to experts who were asked for their opinion about the consistency of the prototype SMS text messages content with the IBM model and behavioural change techniques (BCT), and asked to develop additional SMS text messages according to the IBM model and BCT.

Delphi round two: In the second round, the experts received the modified prototype SMS text messages and were asked to rate the relevance of the SMS text messages for TB medication adherence using a four-point Likert scale. The experts rated the relevance of SMS text messages ranging from 1 to 4, with 1 for "not relevant" to 4 for "highly relevant". The rating criteria were defined and given to experts as seen in the Table 6 below.

Delphi round three: In the third round, the experts received summary statistical reports such as mean, standard deviation (SD), and item-level content validity index (I-CVI) based on the second-round rating. Experts were asked to revise their rating if they would like to change. The experts were also informed about the cut-off point for the I-CVI (0.80) to retain the SMS text message that applied in the third-round rating.

Delphi round four: In the fourth round, the selected SMS text messages in round three were translated to the national language of Amharic and checked the consistency by another professional translator. The translated SMS text messages were given to health professionals to rate the relevance of SMS text messages using the four-point Likert scale. I-CVI above 0.80 was used to retain the SMS text messages for the Ma-MAS intervention.

Scale	Relevance	Description
1	Not relevant	A text message is not at all important to promote adherence to TB medication behaviour, and failure to communicate this message to the patient will not have any consequence on non-adherence TB medication behaviour.
2	Somewhat relevant	A text message that can be used on a daily or non-daily basis and has limited importance to promote adherence to TB medications, and failure to communicate this message to the patient will have minor consequences on non-adherence TB medication behaviour.
3	Quite relevant	A text message that can be used on a non-daily basis and is important to promote adherence to TB medication, and failure to communicate this message to the patient will have a major consequence on TB medication non-adherence.
4	Highly relevant	A text message can be used on a daily basis and is highly important to promote adherence to TB medication, and failure to communicate this message to the patient will have a major consequence on non-adherence TB medication behaviour.

Table 6: SMS text messages relevance rating description

3.5.7. Data analysis

The qualitative data analysis involves verbatim transcription of in-depth interviews, coding, categorizing themes and sub-themes, and thematic analysis was employed using NVivo software version 12. The detail of the qualitative data analysis process is explained in section 3.8.6. The trustworthiness of qualitative studies is explained in section 3.8.7 in this thesis.

The experts' ratings of SMS text messages data was entered into Microsoft excel and exported to SPSS version 27 for analysis. To retain SMS text messages for the next round of Delphi, the Item level content validity index (I-CVI) of 0.80 was used as the cut-off point.^{(166).} Consensus on the retained SMS text messages was reached in Delphi rounds three and four. Scale-level content validity index (S-CVI) was calculated in two ways: (I) S-CVI/Ave is the "average of the I-CVIs for all items on the scale" and (II) S-CVI/UA is the "percentage of items on a scale that achieve a relevant rating of 3 or 4 by all experts." This is also known as 'universal agreement'^{(166).} To retain the SMS text messages for TB medication adherence intervention, we used 90% for the Scale-level content validity index (S-CVI) for S-CVI/Ave.⁽¹⁶⁶⁾

3.6. Study two: Feasibility and acceptability of Ma-MAS intervention

3.6.1. Study design

A quantitative cross-sectional survey was used to assess access and use of mobiles for SMS texts and phone calls, and the acceptability of a mobile-assisted medication adherence support intervention.

3.6.2. Participants

To assess the feasibility and acceptability of the mobile-assisted medication adherence support intervention, the participants were adult TB patients receiving anti-tuberculosis medication in 36 primary public health facilities chosen from 10 sub-cities of Addis Ababa, Ethiopia. The eligibility for this study was a diagnosis with TB and commenced on anti-TB treatment at the primary public health facilities, and at least 18 years old. Participants were excluded from this study if their treatment was prescribed for more than six months, or they had been diagnosed with MDR-TB.

3.6.3. Sample size and technique

To assess the feasibility and acceptability of the mobile-assisted medication adherence support intervention, the sample size was determined using Epi Info statistical software and a single population proportion calculation. The proportion of acceptance of mobile-assisted medication adherence intervention was 50%, 95% confidence level (alpha 0.05) and 5% margin of error were used for sample size calculation. The initial sample size calculated was 384, and with a 10% non-response rate, the final sample size was 422. A total of 36 health facilities were randomly selected from 10 sub-cities of Addis Ababa, Ethiopia. The number of participants in each health facility was determined based on the probability proportional to size (PPS) of the patients in the selected 36 health facilities in Addis Ababa, Ethiopia. Eligible participants were sequentially recruited and invited to be a part of the survey when they came to the clinic for medication refill visits (convenience sampling).

3.6.4. Data collection instrument and process

The feasibility and acceptability of the mobile SMS and phone call intervention assessment tool was developed based on the study objective and a review of similar studies.⁽¹⁶⁷⁾⁽⁶³⁾ The instruments were pretested, and appropriate correction was made during the training of data collectors. The data was collected electronically using Open Data Kit (ODK).

3.6.5. Data analysis

The electronically collected data was cleaned in excel and exported to SPSS version 27 for statistical analysis. To assess the feasibility and acceptability of the Ma-MAS intervention, descriptive statistics with frequency and percentages was employed.

3.7. Study three: Effectiveness of Ma-MAS intervention

3.7.1. Study design

A parallel-group individual randomised controlled trial (RCT) with two groups was employed to investigate the effect of the Ma-MAS intervention on TB medication adherence. Participants were individually randomised into one of the two groups (intervention or control group) in a 1:1 allocation ratio.

3.7.2. Participants

To measure the effect of the mobile-assisted medication adherence support intervention, participants were adult TB patients diagnosed with TB and commenced on anti-tuberculosis treatment at public health facilities of Addis Ababa, Ethiopia. To be included in the randomised control trial (RCT) participants were required to be enrolled in a primary public health facility for anti-TB treatment and have completed their first two months of the intensive phase of treatment, aged at least 18 years, have their own mobile phone, and able to read and understand an SMS text that is written in the national official language (Amharic). Participants who did not have a mobile phone also were included if they could share partner's mobile phone in the household with the collaborative agreement.

Participants were excluded from the randomised trial if they met the following conditions:

- Patients whose anti-tuberculosis treatment had been prescribed for more than six months. This is because of the need to ensure homogeneity in treatment duration, which could affect adherence.
- Patients who enrolled or agreed to enrol in any other interventional study at the same time as this study was conducted.

3.7.3. Sample size and techniques

To measure the effect of the mobile-assisted medication adherence support intervention a sample size was calculated using STATA code *power two proportions* considering the primary outcome of TB medication non-adherence. The rate of TB medication non-adherence in Ethiopia varies considerably in the literature ranging from 10% to 26% depending on the patient's treatment phase and different settings.^(13, 19, 83, 94, 116, 117) We found the rate of non-adherence was 25.6% (po=0.256%) during the continuation phase of treatment from a previous study conducted at Addis Ababa, Ethiopia in the control group.⁽⁸³⁾ Previous mobile SMS text interventions in Africa used 15% to 20% differences.^(84, 122, 131) Thus, we use a 15% absolute reduction of TB non-adherence due to a mobile SMS text intervention found in a similar study ⁽⁸⁴⁾ that would yield a non-adherence rate of 10.6% (pa=0.106) in the intervention group. Considering simple randomised controlled trial parallel-group design, 95% confidence level (alpha=0.05), 80% power (beta=0.80) and a one-sided p-value of 0.025 the initial sample size is n=81 participants per arm. Assuming an attrition rate of 15%, the final sample size would become the total n=186 with arm-1 n=93, and arm-2 n=93.

A total of 36 primary public health facilities were included in the trial by random selection with the assumption being that an average of 6 participants would be found in each health facility.

3.7.4. Recruitment

Patients who completed the intensive phase of anti-TB treatment were recruited by the research support team of healthcare professionals from each health facility, based on eligibility and exclusion criteria. The recruitment was continued until the required sample size was achieved. All the recruitment process details were recorded electronically in Open Data Kit (ODK) platform and retained securely by the research team as required approvals.

3.7.5. Randomisation and allocation

Participants who met the selection criteria were centrally randomised into one of the two groups using simple randomisation in a computer-generated algorithm, with an equal allocation ratio (1:1), by member of the research support team, who was not involved in measuring the outcome. The number of participants allocated to each health facility was determined based on the number of patients enrolled in the intensive phase of treatment. The investigators, the data collectors, the intervention administer, and the patients did not know where the next eligible patients would be assigned (Ma-MAS or control group) before randomisation.

3.7.6. Intervention group

Ma-MAS intervention (Arm-1) group participants received a daily SMS text message and weekly phone calls for medication intake and medication refill visit reminders (see Figure 8). All participants received the intervention at a similar time for eight weeks during the continuation phase of the TB treatment. The phone calls information was guided by questions such as 'Good morning, do you take your daily medication?' 'Did you have a question regarding your medication?' 'Good morning, this is to remind you of your daily medication intake' 'Do you have a question regarding your medication?' The phone calls discussions were open and not limited to these questions (see Appendix Five).



Figure 8: Mobile-assisted Medication Adherence Support (Ma-MAS) intervention trial design diagram

3.7.7. Control group

The control group (Arm-2) participants received the same routine standard treatment care as the intervention group but did not receive the SMS text and phone call intervention. Patients in the control group were followed for the same eight-week period as the intervention group during the continuation phase of the TB treatment (see Figure 8).

3.7.8. Intervention implementation

All patients assigned to any group received the same standard treatment care based on the Ethiopia national TB treatment guideline which is rifampicin, isoniazid, pyrazinamide and ethambutol for two months followed by rifampicin and isoniazid for four months.⁽¹⁶⁸⁾ Patients screening, enrolment, consenting and randomisation were undertaken by the research support team, who have health professional backgrounds. An open SMS automation software platform was used to store, monitor, and send SMS texts. SMS text traffic was managed and monitored in one centralised computer database server by trained IT personnel. The maximum size of an SMS text was not more than 160 characteristics. A weekly phone call was made to the intervention group participants from the health facility by a delegated person from the research team. This person was trained in TB clinical management. A checklist of questions was recorded each time the weekly phone calls were made to the participants. Both the SMS

text and Phone calls intervention were made using the national official language of Amharic. For participants who did not answer the phone calls, three repetitive attempts were made at five-minute intervals. All SMS texts and phone calls were delivered at a similar time between 7:30am to 8:00am at the average medication time. The message delivery reports were continuously monitored to ensure the participants received the SMS texts as planned. Participants were also asked whether they were receiving the SMS texts every week through phone calls and when they came for their medication refill visit. Participants were briefed about the duration, frequency, and timing of the SMS texts and phone calls and advised to turn on their mobile phone in the morning. Participants were informed about the intervention and instructed not to tell and/or share the SMS texts or phone calls with other people. For participants who lost or changed their phone numbers, the new phone number was recorded immediately after they informed a support team member. The trial was audited on participant enrolment, consent, allocation, and during the intervention period.

3.7.9. Outcome measurement

The measured outcome of the trial is the medication adherence of the patient, measured by both direct and indirect methods. The primary outcome was measured using an IsoScreen test (GFC Diagnostics Ltd, Bicester, England) for the presence of isoniazid (INH) drug metabolites in urine in accordance with the vendor-supplied user manual. The IsoScreen test result is negative when there is no colour change (i.e., the strip remains yellow) after 5 minutes, and the test result is positive when the colour changes to dark purple, or blue/ green in that time, depending on the concentration of the medication (i.e., if the medication was taken in the last 24 hours or 48 hours respectively). The proportion of adherence measured by the IsoScreen test at the eighth week of the intervention (primary end point) was used as the primary interest of analysis.^(88, 89) The proportion of adherence measured by the IsoScreen test at the fourth week of the intervention was used as the secondary interest of analysis.

For the secondary outcome of the trial, indirect methods of self-report medication adherence were used along with direct methods. The AIDS Clinical Trial Group adherence questionnaire (ACTG)⁽¹⁴³⁾ and Visual Analogue Scale (VAS) were used.⁽¹⁴⁰⁾ The adapted version of ACTG has three sections: section B (social support), section C (possible reasons for non-adherence) and section D (adherence behaviour). Participant scores less than "somewhat satisfied" to any questions in section B, or other than "never" to any questions in sections C and D were considered as non-adherence.⁽¹⁴⁴⁾ VAS is a single adherence scale that requires a patient to estimate their adherence level from 0% to 100%: 0% means the patient has taken no medication, 50% means the patient has taken half of their prescribed medications, and 100% means the patient has taken all of their prescribed medications, for the last 30 days. Participant scores less than 90% for VAS were considered as non-adherence.^(83, 138) The ACTG and VAS questions were asked at the baseline assessment and at the end of eight weeks of intervention (see Figure 9). The difference in proportion of adherence measured by self-report from the baseline to the eighth week of the intervention was used as the secondary interest of analysis.

	Study	Period			
	Enrolment	Allocation Post-allocation			close-out
Timeline	- t1	to	t1	t2	t3
	Week 1	Week 12	Week 16	Week 20	week 21
Enrolment					
Eligibility screening	*	★			
Informed consent	★	₩			
Allocation	≍	₩			
Intervention					
Daily SMS text					
Weekly Phone calls					
Assessment					
Baseline					
Socio-demographic	≍	₩			
Clinical characterstics	★	★			
Substance Use-related information	★	*			
Health Service-related information	≍	*			
AIDS Clinical Trial Group adherence questionnaire (ACTG)	★	★			
Adherence measured by Visual Analogue Scales (VAS)	₩	★			
Outcome					
Adherence measured by IsoScreen to	est			*	*
AIDS Clinical Trial Group adherence questionnaire (ACTG)					₩
Adherence measured by Visual Analogue Scales (VAS)					₩

Figure 9: Timeline for Ma-MAS intervention trial

3.7.10. Blinding

Single blind was applied to avoid bias in outcome measurement by differentiating the person who measures the outcome from the person who randomises and provides the intervention. Thus, in this study, the person who measured the outcome was blind to which group participants were assigned. Participants were not blinded to which group they were assigned, due to the nature of the trial intervention.

3.7.12. Data collection process

The research team was healthcare professionals who have a relevant Diploma and/or above higher degree. The research team were recruited for both data collection and supervision based on their previous experience. The data collectors and supervisors were recruited through the Addis Ababa Health Bureau, and they were responsible for managing any time allocation and remuneration impacts on their staff. The research team (intervention providers, data collectors and supervisors) was trained by the principal investigator for two days on the research procedures of the intervention, data collection, and the quality and security of the data as per the research protocol.

To ensure the quality of the data, the data collection instruments were pre-tested and appropriate corrections were made based on the pre-test result to ensure the accuracy and precision of the measurement.

Data collection was made after the patient received the routine DOT services in a separate room from the TB clinic room to ensure the confidentiality and privacy of the patient's information. The data was collected electronically using a smart mobile app by designing the questionnaire on Open Data Kit (ODK). To assess adherence status using the IsoScreen test a research support team member with a laboratory technology background collected samples of the urine and tested for the presence of Isoniazid metabolites in urine. The Ministry of Health laboratory facilities were used for this adherence assessment.

3.7.13. Data Analysis

The data gathered electronically was cleaned in Excel before being exported to STATA software version 16 for analysis. For continuous variables, descriptive statistics such as mean, standard deviation, median, and range were utilized, whereas for categorical variables, frequency and percentage were used. For binary variables, a chi-square test was employed, and for continuous variables, a student t-test was used to compare the background characteristics between the intervention and control groups participants.

All randomised individuals were subjected to an intention to treat (ITT) analysis based on their group. The intervention effect was analysed using a log binomial regression model with a 95% confidence interval. For primary outcome, multivariable log binomial regression model analysis was employed to adjust confounding factors in the statistical effect. A P-value of less than 0.05 was used as a cut-off point to show a statistically significant association between Ma-MAS intervention and medication adherence with a 95% confidence interval. The predicted probability of TB medication adherence was calculated by the intervention and control group and covariates after multivariable log binomial regression model analysis. For examining the effect of Ma-MAS intervention on self-reported adherence (secondary outcome) difference in difference statistical technique analysis model was used to calculate the difference of proportion of self-reported adherence from the baseline to the end of the intervention, between intervention and control group participants. A sensitivity analysis was performed by including missing and lost to follow-up patients in the analysis as the best and worst scenario cases of the outcome (see Appendix Seven).

3.8. Study four: Perceptions, experiences, and adoption intention

3.8.1. Unified theory acceptance and use of technology

Research has focused on the acceptance of an information system and information technology. There are numerous theoretical models to assess the acceptance and use of information systems/ technology.⁽¹⁶⁹⁾ After reviewing and synthesizing eight theoretical models, Venkatesh et al. (2003) proposed the unified theory acceptance and use of technology (UTAUT). The original constructs of UTAUT are effort expectancy, performance expectancy, social influence, and facilitating conditions (see Figure 10).⁽¹⁷⁰⁾ The Unified theory of acceptance and use of technology was applied to assess doctors' acceptance and usage of electronic medical records and found to be a useful model that predicted 44% variance in behavioural intention.⁽¹⁷¹⁾ Another similar study found that the explanatory power to predict behavioural intention and usage of electronic medical records was 51.1% and 28.2%, respectively.⁽¹⁷²⁾



Figure 10: Unified theory of acceptance and use of technology adopted from Venkatesh et al. 2003. ⁽¹⁷⁰⁾

Recently the unified theory of acceptance and use of technology was updated by adding attitude construct into the original UTAUT (see Figure 11).⁽¹⁷³⁾ Attitude is included in UTAUT because it is regarded as significantly associated with the behavioural intention and use of information systems or information technology. The explanatory power improved when the attitude construct was added into the model: from 38% to 45% variance in behavioural intention, and from 21% to 27% in usage of behaviour. It was also observed that attitude has a direct effect on the information technology usage.⁽¹⁷³⁾ Thus, in this study the patient's intention to adopt mobile-assisted medication adherence intervention was assessed through using both the original and modified version of unified theory of acceptance and use of technology (see Table 7).

The current research utilized both the original and modified Unified Theory of Acceptance and Use of Technology (UTAUT) to assess which factors influence the adoption of the Ma-MAS intervention (see Figure 10 and 11).



Figure 11: The modified Unified theory of acceptance and use of technology adapted from Venkatesh et al. (2003) and modified by Dwivedi et al. 2019 ⁽¹⁷³⁾

Table 7: Definition for UTAUT constructs used in the proposed theoretical model. source: Dwivedi et al. 2019 $^{(173)}$

Construct	Definition
Performance expectancy (PE)	Performance expectancy is defined as "the degree to which an individual believes that using the system will help him or her to attain gains in job performance" (Venkatesh et al. 2003). ⁽¹⁷⁰⁾
Effort Expectancy (EE)	Effort expectancy is defined as "the degree of ease associated with the use of the system" (Venkatesh et al. 2003). ⁽¹⁴⁷⁾
Social influence (SI)	Social influence is defined as "the degree to which an individual perceives that important others believe he or she should use the new system" (Venkatesh et al. 2003). ⁽¹⁴⁷⁾
Facilitating conditions (FC)	Facilitating conditions are defined as "the degree to which an individual believes that an organizational and technical infrastructure exists to support use of the system" (Venkatesh et al. 2003). ⁽¹⁴⁷⁾
Attitude (AT)	Attitude defined as "an individual's positive or negative feelings about performing the target behaviour" (Davis et al. 1989; Taylor and Todd 1995; Fishbein and Ajzen 1975). ⁽¹⁷⁴⁻¹⁷⁸⁾
Behavioural Intention (BI)	Behavioural intention is defined as "a measure of the strength of one's intention to perform a specific behaviour" (Fishbein and Ajzen 1975). ⁽¹⁷⁸⁾

3.8.2. Study design

A convergence mixed-methods design was employed to assess patients' perceptions, experiences, and adoption intention of Ma-MAS intervention. Patients' perceptions, experiences, and adoption intention were assessed interpretively using qualitative descriptive approach. And for the purpose of assessing the factors associated with the intention to adopt the Ma-MAS intervention, a quantitative cross-sectional survey design was used. The data collection, analysis, and interpretation were undertaken separately, and the results were merged into the discussion.

3.8.3. Participants

To assess patients' perceptions, experiences, and intention to adopt mobile-assisted medication adherence support intervention, adult TB patients who received the Ma-MAS intervention for eight weeks were included for both interpretive and survey study. The interpretive and survey studies did not include participants who enrolled in the mobile-assisted adherence intervention but were lost to follow up or discontinued.

3.8.4. Sample size and technique

To assess factors associated with adoption intention (acceptance) of the Ma-MAS intervention, all patients who received daily SMS messages and weekly phone calls for eight weeks were invited to be part of the cross-sectional survey. For the post-intervention qualitative in-depth interviews, participants were randomly selected from each sub-city until idea saturation was reached.

3.8.5. Data collection instrument and process

An in-depth interview guide developed based on the research question, to assess the patients' perceptions, experiences, and intention to adopt mobile-assisted medication adherence support intervention. The in-depth interview guide and audio-recorder were used to collect the qualitative data. The principal investigator was conducted the in-depth interview at the convenient place and time for the patients.

To assess the adoption intention and associated factors, a data collection tool was adapted from similar literature based on the unified theory of acceptance and use of technology.^{(170, 179-¹⁸¹⁾ It was originally developed in English and then translated into Amharic language by a professional linguist. This questionnaire consists of two sections: background characteristics (age, gender and education level) and items for the unified theory of acceptance and use of technology. In the proposed UTAUT model, 25 items were used to assess six constructs with two endogenous variables, attitude (AT) and behavioural intention (BI), and four exogenous variables namely performance expectancy (PE), effort expectancy (EE), social influence (SI),} and facilitating condition (FC). The construct items were measured using a five-point Likert scale, ranging from 1 for "strongly disagree" to 5 for "strongly agree". Trained data collectors pretested the data collection instruments, and the data was subsequently gathered electronically using Open Data Kit (ODK).

3.8.6. Data analysis

Thematic analysis approach was employed for qualitative data.⁽¹⁸²⁾ First, all the in-depth interviews were transcribed verbatim by a professional translator and the consistency of transcription checked with audio record. The principal investigator read all transcriptions to familiarize themselves with the qualitative data before commencing coding. Nvivo software version 12 was used for coding and analysis of the qualitative data. Coding started with first level coding of broader domains. Under the broader domain an initial deductive coding was made based on the research question and the literature review. The second level domain open coding was done under the broader domain. New emerging concepts from the broader domain were also coded (inductive coding). All codes were reviewed, and similar codes were categorized to form themes and sub-themes. The characteristics of the theme, sub-theme and codes were reviewed. Themes, sub-themes, and codes were compared across all cases to confirm the qualitative data was correctly matched with each theme and sub-theme.⁽¹⁸³⁾ An interpretation of the data was made under each theme and sub-theme by the principal investigator and given to three other experts to review if the coding, categorization, analysis, and interpretation were appropriate (see Figure 12). In the report editing in the transcript is indicated by three full stops (...), reported speech indicated by inverted commas ("") and strong feeling indicated by (!).⁽¹⁸²⁾



Figure 12: Schematic presentation of qualitative analysis adapted from Raskind et al. 2019⁽¹⁸³⁾

Confirmatory factor analysis and structural equation modelling (SEM) were employed to assess the correlation between behavioural intention with constructs of attitude, performance expectancy, effort expectancy, social influence and facilitating conditions. A structural equation modelling analysis was employed using STATA software version 16. In order to check the construct validity and reliability of the UTAUT measurements model, convergent validity, discriminant validity, and internal reliability were evaluated using a variety of indices including factor loading weight, average variance extracted (AVE), square root of AVE, composite reliability (CR), and Cronbach alpha (α). As a standard criterion, AVE value over 0.5 was considered acceptable for convergent validity, and CR value always greater than over AVE value (AVE>0.5 and CR>AVE).^(184, 185) A factor loading value of 0.5 and above was considered as practically significant to remain the item in the construct and is evidence of convergent validity.⁽¹⁸⁶⁾ Pearson correlational analysis was completed to examine the relationship between the constructs of unified theory of acceptance and use of technology and correlation coefficient and a value between 0.10-0.29 was considered week, 0.30-0.49

medium, and 0.5-1 a strong correlation.^(185, 187) For acceptable standard of discriminant validity (DV), the square root of AVE of the construct must be greater the value of the correlation coefficient of the same construct with other constructs of UTAUT.⁽¹⁸⁸⁾ For acceptable construct reliability, the composite reliability (CR) and Cronbach alpha (CA) value of 0.70 and above was used as cut off point.⁽¹⁸⁶⁾

The items under each construct were summed and analysed using structural equation modelling path analysis to test the statistically significant relationship between the constructs of UTAUT. The root mean square error of approximation (RMSEA) and the standardized root mean square residual (SRMR) were used to assess model fitness. RMSEA <0.06 and SRMR<0.08 were used as cut-off values for acceptable goodness of model fit for the standard modeling criteria.^(189, 190) Additionally, comparative fit index (CFI) and Tucker–Lewis index (TLI) greater or equal to 0.95 were used as cut off point to accept the model.⁽¹⁸⁶⁾

3.8.7. Trustworthiness of qualitative research

The trustworthiness refers to the level of confidence in data, interpretation, and methods used to assure the quality of the qualitative research.^(191, 192) The trustworthiness of the qualitative research was examined according to Lincoln and Guba's (1985)⁽¹⁹³⁾ framework of quality criteria: credibility, dependability, confirmability and transferability. Authenticity, later added as additional criteria by Guba and Lincoln (1994) was also used.⁽¹⁹⁴⁾ These framework of quality criteria are considered a gold standard for quality assessment of interpretive research.⁽¹⁹²⁾

3.8.7.1. Credibility

Credibility refers to the confidence that the findings are truthful and accurate for the participants and the context of the research.^(192, 193) An adequate amount of time was allocated for each in-depth interview of participants to ensure the credibility of the qualitative data. Participants were interviewed in Amharic with open-ended guided questions which enabled them to clearly explain their perceptions and experiences. The principal investigator who undertook the interview had experience of qualitative research and the culture and language

of the area. The transcription, codes, analysis, and interpretation of the qualitative data were reviewed with three other experts. Data codes, analyses, and interpretations of qualitative data include both positive and negative findings, which are indicators of credibility. Member checking was not done, and participants were not able to receive verbatim transcriptions due to a shortage of time.

3.8.7.2. Dependability

Dependability refers to the stability of the data over time and over the phenomenon of the study.⁽¹⁹¹⁻¹⁹³⁾ The principal investigator verified the consistency of the recorded audio with the verbatim transcription in order to ensure the dependability of the qualitative data. For each indepth interview case, the themes, sub-themes, and categories were cross-checked to ensure they were correctly matched. Additionally, three experts verified the accuracy of coding, analysis, and interpretation of the qualitative data.

3.8.7.3. Confirmability

Confirmability refers to the neutrality of the researcher or the degree to which findings are consistent and could be repeated by two or more independent people.⁽¹⁹¹⁻¹⁹³⁾ Participants' information should be represented in qualitative data, and the data interpretation should be free of researcher bias. In order to ensure the accuracy of qualitative data, an audit trial of the process which includes the verbatim transcription, coding, analysis and interpretation examined by three independent experts, and a review was completed with these experts.

3.8.7.4. Transferability

Transferability refers to the degree to which findings can be transferred to other persons in other settings.⁽¹⁹¹⁻¹⁹³⁾ Participants' characteristics and context can determine whether or not qualitative findings are transferable to other settings and people.^(191, 195) In this research, details about the study participants, the study context including the health system and TB treatment care, and the study location are described. It is therefore possible to transfer the findings of the qualitative inquiry to other similar settings and more readily adapt the work to new settings.

3.8.7.5. Authenticity

Authenticity is the extent to which researchers depict the range of different participants' realities completely and fairly.^(191, 192, 194) It is possible to ensure the authenticity of the study by selecting appropriate participants and samples, and using tick descriptions.⁽¹⁹¹⁾ This study therefore draws an adequate sample of participants as recommended in the literature (idea saturation), which can represent a range of different experiences. Additionally, a tick description was made to record and consider the range of participants' experiences.

3.9. Ethical process

Research ethics approval was obtained from the Southern Adelaide Clinical Health Research Ethics committee (SAC HREC) of Flinders University. The proposal was also approved by the Addis Ababa Health Bureau Ethics Committee on August 21, 2020, with reference number A/A/H/930/227, and an official permission letter was provided from Addis Ababa Health Bureau to respective sub-city health facilities (see Appendix Eleven). A change in the trial protocol was submitted to SAC HREC and Addis Ababa Health Bureau ethics committee and was approved on April 7, 2021, before implementing the change. The trial design was modified from three groups to two groups, and the frequency of phone calls was changed from daily to weekly after conducting the needs assessment. This update has been reflected in the trial registration at https://pactr.samrc.ac.za/TrialDisplay.aspx?TrialID=9729 and in the published trial protocol (see Appendix Eight).⁽¹⁵⁵⁾

This study had no forms of inducement, or coercion and the study did not bring any risks or incur compensation in cash or kind. Each participant completed a written informed consent after being informed about the research objectives, procedures, anticipated risks, and benefits of the research before the start of data collection. All respondent's information was kept confidential. Participants had the right to withdraw from the study at any time should they feel uncomfortable, concerned or not able to continue.

To safeguard the security of the patient's information, all data was kept in a password protected database. At the completion of the study, the participant's phone numbers, and any other identification of the participants was deleted. Participants were asked clearly at the start of the intervention to report immediately any changes in their phone number and a prompt process was made to unsubscribe and replace the mobile phone numbers. Further detailed information is attached to the participant information sheet and consent form (see Appendix Ten). All data collection was done in accordance with the national Ministry of Health of Ethiopia recommendations for COVID-19 pandemic safety and precautions.

CHAPTER 4: RESULTS OF STUDY ONE: DEVELOPMENT AND VALIDATION OF MA-MAS INTERVENTION: AN MRC FRAMEWORK APPROCH

4.1. Introduction

This chapter presents the results of the systematic development and validation of the mobileassisted medication adherence support (Ma-MAS) intervention for TB patients. The Ma-MAS intervention is developed using the Medical Research Council (MRC) framework with seven non-linear iterative phases that incorporate literature reviews, interpretive and empirical studies, and Delphi techniques. The details of each phase of intervention development are presented in the method section of this thesis. The Ma-MAS intervention consists of two parts: phone call and SMS text message. The phone call intervention was designed to support the SMS text messages in Ma-MAS intervention. The phone call discussions were based on a guided checklist of questions, which is found in the Appendix Five. This chapter will focus on the final results of the SMS text messages development and validation.

4.2. SMS text messages development

Initially, 57 prototype SMS text messages were drafted using the IMB model and behavioural change techniques. The set of prototype SMS text messages was prepared based on a review of factors that influence TB medication adherence in the local context, a review of relevant facts and guidelines, and the input of experts and patients using behavioural change techniques and the IMB model (see Table 8 and Figure 13).

	Purpose	Results
Drafting SMS text	To draft prototype SMS text	57 prototype SMS text messages were
messages	messages	drafted
Content validation	To validate SMS text messages content	
Delphi round one	To evaluate the content of prototype SMS text messages and prepare additional text messages if necessary	Additional 13 SMS texts were added by senior experts and several SMS text contents was modified as per experts' opinions
Delphi round two	To rate the relevance of the SMS text messages	70 SMS texts were rated by 8 senior experts, and low rated SMS text messages were improved but no SMS message was dropped in this round
Delphi round three	To reach on consensus on qualified SMS text messages rating	70 SMS texts rated by 8 senior experts the second time and 40 SMS texts messages were qualified to next round
Delphi round four To evaluate the SMS text messages at the ground level		40 SMS texts messages were evaluated by 44 health professionals and all SMS texts messages were qualified

Table 8: Development and validation process of SMS text messages


Figure 13: SMS text messages development and validation process

4.3. Demographic characteristics

Eight senior experts (1 female and 7 males) with expertise in behavioural science, health care systems, digital health, information technology, public health and pulmonary diseases participated in the SMS text message validation.

A total of 44 health professionals working at the health facility and health office completed the relevance rating for SMS text messages, with a response rate of 93.6% (44/47). The majority of respondents (61.4%) were females. The health professionals' age was between 18 and 35 years. Three quarters of the health professionals worked as TB focal person in the service delivery point of a health facility (See Table 9).

Variables	Frequency	Percentage
Gender	n=44	n=44
Male	17	38.6%
Female	27	61.4%
Age (year)	n=44	n=44
18-35	29	66%
36-55	14	31,8%
Above 55 years	1	2.2%
Job position	n=44	n=44
TB focal person	33	75%
TB experts	11	25%
Institution	n=44	n=44
Health facility	33	75%
Health Office	11	25%
Education level	n=44	n=44
Diploma	8	18.2%
Bachelor's Degree	30	68.2%
Master's degree	6	13.6%

Table 9: Demographic characteristics of health professionals participated in SMS text messages validation, Addis Ababa, Ethiopia

4.4. SMS text messages validation

In the first round of Delphi, 7 of 8 senior experts completed the tasks of qualitative evaluation of the prototype SMS text messages which was 88% of response rate. The experts prepared 13 additional text messages, adding to the first draft text messages. The experts also gave their opinions to improve several of the prototype the SMS text messages in accordance with the behavioural change techniques definition and IBM model. Thus, the SMS text messages were modified according to the experts' suggestions before the second-round quantitative rating.

Eight senior experts rated the relevance of 70 SMS text messages in the second Delphi round, and 38 (54.3%) SMS text messages were rated as 3 'quite relevant' or 4 'highly relevant' by more than 88% of the experts (I-CVI=0.88). A scale-level content validity index (S-CVI/Ave) of 81.6% was obtained for an overall score and all experts (n=8) agreed that 19 text messages contained relevant information based on the scale-level content validity index of expert agreement (S-CVI/UA) which was 27% of the total SMS text messages. Thirty-two (45.7%) SMS text messages were below the minimum acceptable item-level content validity index (I-

CVI<0.80), but no SMS text message was dropped in the second round Delphi. The low-rated I-CVI SMS text messages were modified before the third round (See Appendix Three).

In the third Delphi, 8 senior experts reviewed the summary statistics report of the second round. The individual expert's rating was not included in the summary report (anonymous). The experts rated the SMS text messages for the second time with the same rating criteria. In this round, 88% of experts (I-CVI=0.88) rated 40 SMS texts (57%) as 3 'quite relevant' or 4 'highly relevant'. Based on the acceptable standard content validity index (I-CVI≥0.80), decision was made to retain 40 SMS text messages for the next round. The scale level content validity index (S-CV/Ave) for these 40 SMS texts was 94.6%, which was higher than the minimum suggested score of 90%. A scale level content validation index of universal agreement (S-CV/UA) of 21 (52.5%) was obtained for the retained SMS text messages, indicating that all experts (100%) agreed that the 21 SMS texts were relevant for TB medication adherence promotion, as they were rated 3 'quite relevant' or 4 'highly relevant' (Table 10). Experts were informed of the final retained SMS text messages and final quality improvements were made.

In the fourth round Delphi, 44 of 47 invited health professionals rated the relevance of 40 SMS text messages. The item-level content validity for 40 SMS text messages was over 0.84, meaning that the item-level content validity fulfills the acceptable standard of 80%. Based on these 40 SMS texts, the scale level content validity index was 94.3%. According to the S-CV/UA index of universal agreement (S-CV/UA), 17.5% (7/40) of the retained SMS text messages were rated as relevant for TB medication adherence promotion by all experts (n=44). These SMS texts were rated 3 'quite relevant' or 4 'highly relevant' by all experts (see Table 10).

86

S.NO	BCT SMS text messages English Version		Experts rating at the 3 rd round (n=8)		Health professionals rating at the 4 th round (n=44)			
			No. of agreement	СЛ	UA	No. of agreement	CVI	UA
1	Goal setting (Behaviour)	A person infected with TB must take their medications every day at a regular time for at least six months to be cured. Please plan to take all your medications with no interruptions.	7	0.88	0	44	1	1
2	Problem-solving	Why do some patients forget to refill their medication? You can come to refill any time before you run out of medications. Keep your calendar card in a place where you can see it.	7	0.88	0	43	0.98	0
3	Problem-solving	Please ask your family member to pick up your medications for you if you are not able to come to the clinic to refill your medications.	8	1	1	42	0.95	0
4	Goal setting (outcomes)	If you want to be fully cured from TB, you must take all your medications for at least six months. Please do not forget to come to refill your medication supply.	8	1	1	44	1	1
5	Discrepancy between current behaviour and goal	Consider some form of self-tracking such as a medication calendar to mark the number of consecutive days that you are adhering. How many days in a row have you taken your medication as prescribed?	7	0.88	0	37	0.84	0
6	Behavioural contract	Every day when you wake up in the morning, remember to take your medication before your breakfast. Consider asking a family member to remind you.	7	0.88	0	41	0.93	0
7	Commitment	For how many days have you correctly taken your medication as prescribed? To be cured from TB, you must take all your medications for at least six months.	7	0.88	0	42	0.95	0
8	Feedback and monitoring	Thank you for picking up your medication as per your scheduled date. Keep it up, and take your daily medication as directed.	8	1	1	42	0.95	0

9	Self-monitoring of behaviour	Plan ahead to visit the health facility to refill your medication supply. Use a calendar card to check off the days when you have taken your medicine.	8	1	1	37	0.84	0
10	Feedback on outcome(s) of behaviour	Do you know that those who adhere will have a better chance to be cured? If you missed any dose of TB medications, please tell the healthcare provider.	8	1	1	41	0.93	0
11	Practical	Please tell the healthcare provider if you are traveling somewhere away from home. They can give you extra medications or link you with another healthcare worker, so you will not run out of medications	8	1	1	44	1	1
12	Practical	Please tell us if you are feeling ill while you are on your medications. Ask the healthcare provider about the common side effects.	7	0.88	0	44	1	1
13	Emotional	Your health status is our priority. Please Keep in touch with your healthcare provider and family support team for help when needed.	7	0.88	0	39	0.88	0
14	Instruction on how to perform a behaviour	It is common to experience some side effects while you are taking TB medication. Please tell to your healthcare provider if you experience mild side effects.	8	1	1	41	0.93	0
15	Information about Antecedents	People tend to forget to take their medications when they are away from home. Please remember to have an adequate amount of medications with you when you are away from home.	8	1	1	43	0.98	0
16	Information about Antecedents	TB can be cured in almost all cases by taking the medications as prescribed without interruption.	8	1	1	42	0.95	0
17	Information about Antecedents	Some patients will stop taking TB medications when they experience side effects. Please tell the health care provider if you have any side effects.	8	1	1	42	0.95	0

18	Reattribution	Please remember that TB bacteria die very slowly. Simply because you are feeling well does not mean you have been cured. You must maintain regular treatment for at least 6 months to be cured.	8	1	1	44	1	1
19	Information about Health consequences	Be aware that if you are unable to adhere to TB medication regularly as directed, this can lead to failed treatment, relapse, and multidrug resistance.	8	1	1	42	0.95	0
20	Salience of consequences	A person with TB who takes all the medications as directed will be cured and will not be a risk of TB transmission to his/her family and other communities.	8	1	1	42	0.95	0
21	Information about social and environmental consequences	Untreated TB can be transmitted to close family members, friends and other people. Please take your medications until the end of the treatment period as per prescription.	8	1	1	43	0.98	0
22	Monitoring of Emotional consequences	Did you take your medications at the same time every day? If you did, congratulations! If you didn't adhere, try again, but tell your healthcare provider the doses that you missed.	8	1	1	40	0.91	0
23	Anticipated regret	Do you know that if TB treatment fails, you will become sick again, and you may transmit TB to loved family members and friends? Treatment fails if a patient is not adhering to medications.	8	1	1	40	0.91	0
24	Self-assessment of affective consequences	Imagine what you would feel if you had a multi-drug resistant TB because of non-adherence to medications. It would need long-term treatment and may not ever be cured.	7	0.88	0	41	0.93	0
25	Demonstration of the behaviour	Consider using a strategy like setting an alarm in your mobile phone, or tell a family member to remember your daily TB medication and refill visit schedule.	8	1	0	42	0.95	0
26	Prompts/Cues	You must decide the first task in the morning is always to take your daily medication. Keep taking your daily medications at the same time every day as directed.	8	1	1	41	0.93	0

							1	1
27	Cue signalling reward	Good morning! Before you eat your breakfast, please take your daily medication.	8	1	1	42	0.95	0
28	Remove access to the reward	Cigarette smoking during TB treatment can lower your treatment success. Avoid people or places that make you want to smoke as much as possible.	7	0.88	0	44	1	1
29	Associative learning	Some patients take their medication in the morning immediately after they wash their faces. Keep taking your daily medication.	7	0.88	0	37	0.84	0
30	Behavioural practice/rehearsal	Please remember to take your medication regularly every day at the same time. Please keep your medications in a safe place where you can see it.	7	0.88	0	42	0.95	0
31	Habit formation	Please take your medications every day at the same time in the morning. This helps you to remember to take your medications regularly.	7	0.88	0	40	0.91	0
32	Generalization of target behaviour	Please keep taking your daily medications regularly until the end of your treatment. TB is a curable disease if you are fully adhering to treatment.	7	0.88	0	39	0.89	0
33	Social reward	Thank you for picking up your medications. Please remember to take your daily medications at the same time every day.	7	0.88	0	43	0.98	0
34	Future punishment	Do you know that non-adherence to medications may cause multi-drug resistant TB that may need complex treatment for a long time, has more side effects, and may not work or allow you to be cured	8	1	1	42	0.95	0
35	Avoidance/reducing exposure to cues for the behaviour	Please remember to keep your medications with you anytime when you are away from home.	8	1	1	43	0.98	0
36	Avoidance/reducing exposure to cues for the behaviour	Please remember to avoid alcohol and smoking that will affect your TB treatment success.	8	1	1	44	1	1
37	Body changes	You will gain body weight and become healthy if you appropriately adhere to your TB medications and lead a healthy lifestyle.	7	0.88	0	42	0.95	0

You have the capacity to fully adhere to all your TB						I
medications every day to be cured. Yes, you can do it!	7	0.88	0	40	0.91	0
Imagine that you can remember to take your daily TB medications to the end of treatment. Your imagination will become true!	7	0.88	0	39	0.89	0
=8) = 0.946 525						
onals rating (n=44) =0.943						
2	 medications every day to be cured. Yes, you can do it! Imagine that you can remember to take your daily TB medications to the end of treatment. Your imagination will become true! 8) = 0.946 525 	medications every day to be cured. Yes, you can do it!7Imagine that you can remember to take your daily TB medications to the end of treatment. Your imagination will become true!78) = 0.946525525nals rating (n=44) =0.943	medications every day to be cured. Yes, you can do it!70.88Imagine that you can remember to take your daily TB medications to the end of treatment. Your imagination will become true!70.888) = 0.946525525525nals rating (n=44) =0.9435.9435.943	medications every day to be cured. Yes, you can do it!70.880Imagine that you can remember to take your daily TB medications to the end of treatment. Your imagination will become true!70.8808) = 0.946525 nals rating (n=44) =0.943	medications every day to be cured. Yes, you can do it!70.88040Imagine that you can remember to take your daily TB medications to the end of treatment. Your imagination will become true!70.880398) = 0.946525 nals rating (n=44) =0.943	medications every day to be cured. Yes, you can do it!70.880400.91Imagine that you can remember to take your daily TB medications to the end of treatment. Your imagination will become true!70.880390.898) = 0.946525 nals rating (n=44) =0.943

S-CVI/UA by Health professionals rating (n=44) =0.175 BCT: Behaviour change technique⁽¹⁵⁸⁾, CVI: content validity index , UA: Universal agreement

CHAPTER 5: RESULTS OF STUDY TWO: FEASIBILITY AND ACCEPTABILITY OF Ma-MAS INTERVENTION

5.1. Introduction

In this chapter, the results of a cross-sectional survey evaluating the feasibility and acceptability of mobile-assisted medication adherence support (Ma-MAS) intervention for TB patients are presented. To determine the feasibility of Ma-MAS intervention, participants' access to mobile phones and their digital literacy for using SMS text messages and phone calls was assessed. Additionally, the acceptability of Ma-MAS was evaluated by asking participants whether they would be willing to receive SMS text message and phone call to support their TB medication adherence and routine attendance at clinic appointments.

5.2. Demographic characteristics

In total, 416 participants participated in the study, with 98.6% responding. More than half (54.3%) were males. The median age of the participants was 30 years old. Among the participants, 199 (47.8%) were never married, while 187 (45.0%) were married. Most participants were educated, only 15% of participants did not have a formal education. Occupationally, participants were daily laborers (20.4%), government employees (13.2%), unemployed (13%), merchants (12.5%), and housewives (11.8%). The median number of family members in the household was 3, and 19% of participants lived alone. At the time of the study, 176 (42.3%) participants had a monthly income below the poverty level, which was 2373 Ethiopian Birr, or less than 57 US dollars. The majority of participants (53.8%) walked to the health institution for treatment and medication refills (see Table 11).

Variables	Frequency (percent)
Gender	n=416
Male	226 (54.3%)
Female	190 (45.7%)
Marital status	n=416
Never married	199 (47.8%)
Married	187 (45.0%)
Divorced	19 (4.6%)
Widowed	11 (2.6%)
Educational level	n=416
No formal education	62 (14.9%)
Primary education (1-8 grade)	123 (29.6%)
Secondary education (9-10 grade)	86 (20.7%)
Technical/Vocational education (11-12 grade)	48 (11.5%)
Higher education (above 12 grade)	97 (23.3%)
Occupational status	n=416
Daily laborer	85 (20.4%)
Governmental employee	55 (13.2%)
Unemployment	54 (13.0%)
Merchant	52 (12.5%)
Housewife	49 (11.8%)
Student	41 (9.9%)
Driver	20 (4.8%)
Private employee	18 (4.3%)
Pensioner	12 (2.9%)
Guard	10 (2.4%)
Other*	20 (4.8%)
Family size	n=416
Living alone	79 (19.0%)
2-3 people	155 (37.3%)
4-5 people	120 (28.8%)
6 and above people	62 (14.9%)
Monthly income in ETB (USD)	n=416
Less than 2373 (57\$)	176 (42.3%)
2373 (57\$) and above	240 (57.7%)
Transportation used to reach the health facility	n=416
Walk	224 (53.8%)
Taxi	148 (35.6%)
Bajaj	32 (7.7%)
Bus/Public transport	9 (2.2)
Other**	3 (0.7%)

Table 11: Socio-demographic characteristics of adult TB patients, Addis Ababa, Ethiopia

*Farmer, House Servant, Cleaner, Mechanics, Non-governmental employee, Barberry, Volunteer worker

**Personal care, Bicycle/Motor bicycle ETB: Ethiopia Birr, USD: United States Dollar, ETB to USD exchange was made based on the data collection time conversion rate.

5.3. Clinical characteristics

A total of 254 (61%) participants had pulmonary TB, with 176 (42.3%) participants having smear-positive pulmonary TB and 78 (18.8%) participants having smear-negative pulmonary TB. Overall, 371 (89.2%) of the participants were newly diagnosed with TB, and 10% had relapsed TB. Among the total participants, 78 (18.8%) had TB-HIV co-infection, 11 (2.6%) had TB-diabetes, and 12 (2.9%) had TB-hypertension (see Table 12).

Table 12: Clinical characteristics of adult TB patients, Addis Ababa, Ethiopia

Variables	Frequency (percent)
Type of TB and AFB sputum smear positivity	n=416
Smear positive pulmonary TB	176 (42.3%)
Smear negative pulmonary TB	78 (18.8%)
Extra pulmonary TB	162 (38.9%)
History of TB	n=416
New case	371 (89.2%)
Relapsed	42 (10.1%)
Treatment failure	3 (0.7%)
TB-HIV co-infected	78 (18.8%)
Non-communicable disease	n=416
Diabetes	11 (2.6%)
Hypertension	12 (2.9%)
Other*	22 (5.3%)

*Allergic, Asthma, Gastritis, mental illness, Anemia, Liver disease, and kidney disease, AFB: Acid-Fast Bacillus tests

5.4. Feasibility of Ma-MAS intervention

In total, 368 (88.5%) of the participants had their own mobile phones. Two hundred (54.3%) of the participants with their own mobile phones have smartphones that can install and run applications. Of the total participants, 368 (76.4%) had a trusted family member or caregiver in the household with access to a mobile phone. Of the participants who did not have their own mobile phones, 30 (62.5%) had access to a mobile phone through a family member or caregiver in the household. Only 18 (4.3%) of the participants did not have their own mobile phone and did not have access to a mobile phone through a family member or caregiver in the household. The demographic characteristics of participants who did not have their own mobile phone is presented in the Appendix Six.

Participants were confident in their ability to receive and open (81.3%), read (82.7%), and send (80.0%) SMS text messages using their mobile phones. Three hundred and seventy-three (89.7%), 341(82.0%), and 329(79.1%) were likewise confident in using a mobile phone to make a phone call, please call me and set a reminder, respectively (see Table 13).

Variables	Response	Frequency (percent)
Having own mobile phone	Yes	368 (88.5%)
(n=416)	No	48 (11.5%)
Type of mobile phone	Smart phone (can install and use applications)	200 (54.3%)
(n=368)	Basic phone (can only use for voice and text message)	168 (45.7%)
A trusted family or caregiver have	Yes	318 (76.4%)
mobile phone (n=416)	No	98 (23.6%)
	to receive and open an SMS text message	388 (81.3%)
Confident to use a mobile phone	to read an SMS text message	344 (82.7%)
(n=416)	to send an SMS text message	333 (80.0%)
	to make a phone call	373 (89.7%)
	to set a reminder	329 (79.1%)
	to send a "please call me"	341 (82.0%)

Table 13: Feasibility of mobile-assisted medication adherence support among adult TB patients, Addis Ababa, Ethiopia

5.5. Acceptability of Ma-MAS intervention

Among those with their own mobile phones, 344 (93.5%) participants had a willingness to receive SMS text messages for medication reminders, and 339 (92.1%) participants had also willingness to receive SMS text messages for clinic appointments. Also, the majority of participants with their own mobile phones were willing to receive a phone call reminding them to take their medication (90.8%) and attend clinic appointments (93.25%).

Two hundred eighty-five (89.6%) and 287 (90.3%) participants were willing to receive SMS text messages for medication intake and clinic appointments reminders, respectively, through family member's or caregiver's mobile phone. Through a family member's or caregiver's mobile phone, participants were willing to receive phone calls for medication intake (89.3%) and clinic appointments (89.9%) reminders (see Table 14).

Of the participants who did not have their own mobile phones but had access to a mobile phone through a family member or caregiver in the household (n=30), 26 (86.7%) and 28 (93.3%) were willing to receive SMS text messages for medication intake and clinic appointment reminders, respectively, through a family member's or caregiver's mobile phone. Similarly, of the participants who did not have their own mobile phones but had access to a mobile phone through a family member or caregiver in the household (n=30), 27 (93.0%) of participants were willing to receive phone calls for medication intake and clinic appointment reminders.

Variables	Response	Frequency (percent)
Willing to receive SMS reminder (n=368)	for medication intake	344 (93.5%)
	for clinic appointments	339 (92.1%)
Willing to receive a phone call reminder	for medication intake	334 (90.8%)
(n=368)	for clinic appointments	343 (93.2%)
Willing to receive SMS reminder on shared trusted family member's mobile phone	for medication intake	285 (89.6%)
(n=318)	for clinic appointments	287 (90.3%)
Willing to receive a phone call reminder on	for medication intake	284 (89.3%)
shared trusted family member's mobile phone (n=318)	for clinic appointments	286 (89.9%)

Table 14: Acceptability of mobile-assisted medication adherence support (Ma-MAS) among adult TB patients, Addis Ababa, Ethiopia

CHAPTER 6: RESULTS OF STUDY THREE: EFFECTIVENESS OF Ma-MAS INTERVENTION

6.1. Introduction

In this chapter, the results of a randomised controlled trial evaluating the effectiveness of a mobile-assisted adherence support (Ma-MAS) intervention for TB medication adherence are presented. The study included two groups of participants: the Ma-MAS group and the control group. Participants in the Ma-MAS group received daily SMS text messages and weekly phone calls for eight weeks during the continuation phase of TB treatment, in addition to standard care. The control group received only the standard care for TB treatment, without the Ma-MAS intervention.

6.2. Background

A total of 206 adult TB patients were screened using the eligibility criteria to obtain 186 eligible participants. Twenty participants were dropped from the study; 18 did not meet the eligibility requirements, and two patients declined to participate. The 186 eligible individuals were randomly assigned to one of the two groups: Ma-MAS group or control group, with 93 participants in each group. Two participants from the Ma-MAS and control groups were lost to follow-up before the fourth week because their treatment care was transferred to other healthcare facilities, and in the same follow-up period one additional participant from the Ma-MAS group was discontinued because of death. Two participants from the Ma-MAS group and one from the control group were lost from the follow-up because they were unable to travel to the clinic for IsoScreen urine test at the end of eighth week of follow-up. The resulting loss to follow-up rate was 5.4% and 3.2% in the Ma-MAS and control groups, respectively. After eight weeks of follow-up, a total of 88 participants in the Ma-MAS group and 90 participants in the control group completed the study (see Figure 14).



Figure 14: Ma-MAS trial flow diagram (adapted from CONSORT 2010)⁽¹⁹⁶⁾

A total of 56 SMS text messages and 8 phone call reminders were sent to the participant during 8 weeks of Ma-MAs intervention. Of the total 5,208 SMS texts sent and 744 phone calls made for participants of intervention group during 8 weeks, 99% of the SMS text and 85.4% of the phone calls were successfully delivered to the recipients.

6.3. Demographic characteristics

Fifty-five (59.14%) of the Ma-MAS and 51 (54.84%) of the control group were males. The median age of patients in both Ma-MAS and control group was 28. Forty-four (47.3%) of the Ma-MAS group and 54 (58.0%) of the control group patients had never married. Thirty-nine (41.9%) of the Ma-MAS and 21 (22.5%) of the control group patients had attended elementary school (1–8), whereas 9 (9.6%) of the Ma-MAS and 17(18.2%) of the control group patients had attended secondary school (11-12 grade) (see Table 15).

Variables	Ma-MAS group	Control group	P value
Variables	n=93	n=93	
Gender			
Male	55 (59.14%)	51 (54.84%)	0.554
Female	38 (40.86%)	42 (45.16%)	0.554
Age in years			
Mean (Sd)	32.11 (11.24)	31.26 (11.07%)	0.6045
Median (IQR)	28 (14)	28 (15)	
Marital status			
Never Married	44 (47.31%)	54 (58.06%)	
Married	40 (43.01%)	35 (37.63%)	0.165
Divorced/separated	8 (8.60%)	3 (3.23%)	0.165
Widowed*	1 (1.08%)	1 (1.08%)	
Educational status			
No formal Education	3 (3.23%)	7 (7.53%)	
1-8 grade	39 (41.94%)	21 (22.58%)	
9-10 grade	17 (18.28%)	19 (20.43%)	0.043**
11-12 grade	9 (9.68%)	17 (18.28%)	
Above 12 grades	25 (26.88%)	29 (31.18%)	

Table 15: Demographic characteristics of adult TB patients enrolled in Ma-MAS trial, Addis Ababa, Ethiopia

*Widowed merged with divorced/separated for chi square test. ** P value statistically significant (P value<0.05), sd: standard deviation IQR: Interquartile range

6.4. Socio-economic characteristics

Nineteen (20.4%) patients in both Ma-MAS and control groups worked in a government organization, whereas 14 (15.0%) of the Ma-MAS and 17 (18.2%) of the control group patients were unemployed. Thirty-five (37.63%) patients in both Ma-MAS and control groups were identified as living below the poverty level (\$57 or 2738 ETB). The majority of patients, 78 (83.87%) of the Ma-MAS and 76(81.72%) of the control group, reported they had access to adequate food. Seventeen (18.2%) of the Ma-MAS and 9 (9.6%) of the control group patients had received food support from any sources. More than three-fourths of patients in both Ma-MAS and control groups had received a reminder from families or caregivers for medication intake and clinic appointments. A few patients, 10(10.7%) of the Ma-MAS and 12(12.90%) of the control group, had experienced stigma and discrimination related to their disease diagnosis (see Table 16).

Table 16: Social and economic characteristics of adult TB patients enrolled in Ma-MAS trial, Addis Ababa, Ethiopia

Variables	Ma-MAS group (n=93)	Control group (n=93)	P value		
Occupation					
Governmental	19 (20.43%)	19 (20.43%)			
Daily laborer	19 (20.43%)	15 (16.13%)			
Housewife	14 (15.05%)	10 (10.75%)			
Merchant	11 (11.83%)	8 (8.60%)	0.826		
Student	6 (6.45%)	8 (8.60%)			
Unemployment	14 (15.05%)	17 (18.28%)			
Other*	6 (6.45%)	10 (10.75%)			
Monthly family Income ETB (USD)					
Less than 2738 (57\$)	35 (37.63%)	35 (37.63%)	1.0		
2738 (57\$) and above	58 (62.37%)	58 (62.37%)	1.0		
Mean (Sd)	3900 (3266.49)	4090.32 (3341.14)	0.6949		
Median (IQR)	3000 (3000)	3000 (3000)			
Access to adequate food	· · ·				
Yes	78 (83.87%)	76 (81.72%)	0.000		
No	15 (16.13%)	17 (18.27%)	0.698		
Food support (any source)					
Yes	17 (18.27%)	9 (9.68%)	0.001		
No	76 (81.72%)	84 (90.32%)	0.091		
Financial support (any source)					
Yes	14 (15.05%)	11 (11.83%)	0 540		
No	79 (84.95%)	82 (88.17%)	0.519		
Family number per household					
Living alone	15 (16.13%)	27 (29.03%)			
2-3	42 (45.16%)	30 (32.26%)	0.400		
4-5	30 (32.26%0	28 (30.11%)	0.123		
6 and above	6 (6.45%)	8 (8.60%)			
Family or caregivers remind to take					
the medication					
Yes	75 (80.65%)	71 (76.34%)	0.475		
No	18 (19.35%)	22 (23.66%)	0.475		
Family or caregivers remind to					
clinic appointments					
Yes	77 (82.80%)	73 (78.49%)	0.458		
No	16 (16.20%)	20 (21.51%)	0.400		
Stigma and discrimination					
Yes	10 (10.75%)	12 (12.90%)	0.650		
No	83 (89.25%)	81 (87.10%)	0.650		

*Guard, self-employment, Barbery, House renter, and Farmer

ETB: Ethiopia Birr, USD: United States Dollar, ETB to USD exchange was made based on the data collection time conversion rate, sd: standard deviation, IQR: Interquartile range

6.5. Substance use chracteristics

Twenty-one (22.5 %) of the Ma-MAS and 26 (27.9%) of the control group patients had ever consumed alcohol. Few patients, 7.53% of the Ma-MAS and 6.45% of the control group, had ever smoked. When compared to the control group, the proportion of ever khat chewing in the Ma-MAS group (10.7%) was lower (27.9%), which was statistically significant. There was also a statistically significant difference between Ma-MAS and control groups in the proportion currently chewing khat (see Table 17).

Table 17: Substance use characteristics of adult TB patients enrolled in Ma-MAS trial, Addis Ababa, Ethiopia

Variables	Ma-MAS group n=93	Control group n=93	P value	
Ever drank alcohol				
Yes	21 (22.58%)	26 (27.96%)	0.200	
No	72 (77.42%)	67 (72.04%)	0.399	
Current drink alcohol				
Yes	10 (10.75%)	14 (15.05%)	0.382	
No	83 (89.25%)	79 (84.955)	0.362	
Ever smoke cigarette				
Yes	7 (7.53%)	6 (6.45%)	0.774	
No	86 (92.47%)	87 (93.55%)	0.774	
Current smoke cigarettes				
Yes	1 (1.1%)	4 (4.3%)	NA	
No	92 (98.9%)	89 (95.7%)		
Ever chewing khat				
Yes	10 (10.75%)	26 (27.96%)	0.003*	
No	83 (89.25%)	67 (72.04%)	0.003	
Current chewing khat				
Yes	5 (5.4%)	14 (15.0%)	0.029*	
No	88 (94.6%)	79 (85.0%)	0.029	
Ever use of shisha, ganja, or other				
substances				
Yes**	2 (2.15%)	0	NA	
No	91 (97.85%)	93 (100%)		

*P value statistically significant (P value <0.05)

** There was no current user of shisha, ganja, or other substances

6.6. Health service and geographic access characteristics

Most patients believed the waiting time to get treatment care was acceptable; 93.5% in the Ma-MAS group and 94.4% in the control group. Sixty-seven (72.0%) in the Ma-MAS and 76(81.7%) in the control group believed their relationship with the healthcare provider was very good. In both the Ma-MAS and control groups, more than three-quarters (76.3%) of patients believed the quality of treatment they received was very good. Nearly half of the patients, 46.4% in the Ma-MAS and 43.4% in the control group, walked to reach the health facility. The median transportation cost to reach the health facility was 4 and 5 Ethiopia Birr (ETB) for the Ma-MAS and control groups, respectively (see Table 18).

Variables	Ma-MAS group n=93	Control group n=93	P value
The Perceived waiting time			
Acceptable	87 (93.55%)	88 (94.62%)	0.757
Unacceptable	6 (6.45%)	5 (5.38%)	0.757
The Perceived relationship with			
healthcare provider*			
Good	26 (27.96%)	17 (18.28%)	0.118
Very good	67 (72.04%)	76 (81.72%)	0.110
The Perceived quality of care*			
Good	22 (23.66%)	22 (23.66%)	1
Very good	71 (76.34%)	71 (76.34%)	
Means of transportation			
Walk	46 (49.46%)	43 (46.24%)	0.904
Car	37 (39.78%)	39 (41.94%)	
Bajaj	10 (10.75%)	11 (11.83%)	
Travel cost (ETB)			
Mean (Sd)	6.04 (8.75)	8.96 (11.97)	0.058
Median (IQR)	4 (6)	5 (7)	
Travel time to reach health facility			
(minutes)			
Mean (Sd)	21.21 (12.57)	21.53 (13.74%)	0.867
Median (IQR)	20 (20)	20 (20)	

Table 18: Health service and geogra	ohic access characteristics of adult TB patients enrolled
in Ma-MAS trial, Addis Ababa, Ethiop	ia

*poor and very poor options were zero reponse, ETB: Ethiopian Birr

6.7. Clinical characteristics

Smear positive pulmonary TB was found in 44 (47.3%) patients in the Ma-MAS group and in

39 (41.9%) patients in the control group. Most patients were newly diagnosed; 91.4 % in the

Ma-MAS group and 89.2% in the control group. A total of 11.8% of the Ma-MAS and 15.1%

of control group patients had co-infections with TB and HIV (see Table 19).

Table 19: Clinical characteristics of adult TB patients enrolled in Ma-MAS trial, Addis Ababa, Ethiopia

Variables	Ma-MAS group n=93	Control group n=93	P value
TB diagnosis category			
Smear positive pulmonary TB	44 (47.31%)	39 (41.94%	
Smear negative pulmonary TB	15 (16.13%)	21 (22.58%)	0.518
Extra pulmonary TB	34 (36.56%)	33 (35.48%)	
History of TB			
New	85 (91.40)	83 (89.25%)	
Relapsed	7 (7.53%)	9 (9.68%)	0.620
Treatment failure*	1 (1.08%)	1 (1.08%)	
TB-HIV co-infection			
Yes	11 (11.83%)	14 (15.17%)	0.519
No	82 (88.17%)	79 (84.95%)	0.519
Any other non-communicable disease			
Yes**	5 (5.38%)	6 (6.45%)	0.750
No	88 (94.62%)	87 (93.55%)	0.756

*Treatment failure was merged with relapsed case for chi square test, ** Diabetes and Hypertension

6.8. Effectiveness of Ma-Mas intervention

6.8.1. Effectiveness of Ma-Mas intervention on primary ouctome

The mobile-assisted medication adherence support intervention significantly improved TB medication adherence by 15.25% (95%CI: 5.38, 25.12; P-value=0.0065) after eight weeks compared to the standard care provided to the control group. Ma-MAS also improved TB medication adherence by 15.30% (95%CI: 6.68, 23.90; P-value=0.0022) after four weeks of intervention compared to the control group. Although the proportion of adherence decreased by 5.86% in the Ma-MAS group and by 5.81% in the control group from week four to eight, the Ma-MAS intervention remained effective in improving TB medication adherence (see Table 20).

Table 20: Ma-MAS intervention effect on TB medication adherence measured by IsoScreen urine test, Addis Ababa, Ethiopia

Primary outcomes	Adherence measured by IsoScreen test	95% CI	P value, 1-tailed
After 8 weeks of follow-up			
Ma-MAS group (n=88)	86.36%	[79.19, 93.53]	
Control group (n=90)	71.11%	[61.74, 80.47]	
Absolute Difference (AD)	15.25%	[5.38, 25.12]	0.0065
After 4 weekss of follow-up			
Ma-MAS group (n=90)	92.22%	[86.68, 97.75]	
Control group (n=91)	76.92%	[68.26, 85.57]	
Absolute Difference (AD)	15.30%	[6.68, 23.90]	0.0022

Absolute Difference (AD) is calculated Ma-MAS group- control group with 95% confidence interval (CI)

6.8.2. Effectiveness of Ma-Mas intervention in multivariable analysis

In a multivariable analysis, mobile-assisted medication adherence support had a statistically significant association with higher TB medication adherence compared to the standard care after controlling for TB/HIV co-infection, reminders from family or caregivers for clinic appointments, monthly household income, and financial support. When compared to patients who only received the standard care, those who received mobile-assisted medication adherence support had 1.21 times higher adherence to TB medication (95% CI: 1.07, 1.36). After adjustment, none of the co-variates demonstrated a statistically significant association with TB medication adherence (see Table 21).

Variables	Adherence	Risk Ratio [95%Cl]	Adjusted Risk Ratio [95%CI]	P value	
Randomisation group					
Ma-MAS group	86.36%	1.21 [1.05, 1.37] *	1.21 [1.07, 1.36] *	0.0035*	
Control group	71.11%	1	1		
TB/HIV co-infection					
Yes	60.86%	0.74 [0.53, 1.04]	0.77 [0.54, 1.08]	0.140	
No	81.29%	1	1		
Family or caregivers remind					
to clinic appointments					
Yes	80.82%	1.17 [0.91, 1.50]	1.17 [0.93, 1.46]	0.178	
No	68.75%	1	1		
Monthly family income (ETB)					
<2738	69.69%	1	1		
2738 and above	83.92%	1.20 [1.01, 1.43] **	1.06 [0.92, 1.23]	0.380	
Financial support					
Yes	63.63	0.778 [0.56, 1.09]	0.82 [0.62, 1.09]	0.184	
No	80.76%	1	1		

Table 21: Multivariable analysis of covariates in Ma-MAS intervention effectiveness after eight weeks of follow-up, Addis Ababa, Ethiopia

*One sided tailed used since it is a superiority trial; **P value=0.004, ETB: Ethiopian Birr

After eight weeks of follow-up, the predicted probability of adherence to TB medication in the Ma-MAS group participants was 86% (95%CI: 81, 93), while the predicted probability of adherence to TB medication in the control group participants was 70% (95%CI: 61, 79), after controlling for TB/HIV, family monthly income, financial support, and reminders from family or care givers for clinic appointments (see Figure 15).³

After four weeks of follow-up, the predictive probability of adherence to TB medication in the Ma-MAS group participants was 93% (95%CI: 89, 98), while the predicted probability of adherence to TB medication in the control group participants was 77% (95% CI: 70, 84) after controlling for TB/HIV, TB diagnosis category, access to adequate food, mode of transportation to health facility, and relationship with healthcare provider (see Figure 15).³

³ This section contain material originally published abstract conference of Mobile-assisted medication adherence support intervention among tuberculosis patients: a parallel group randomized control trial. Population Medicine. 2023;5(Supplement):A585.



Figure 15: The predictive probability of adherence to TB medication after four weeks and eight weeks of follow-up in Ma-MAS trial, Addis Ababa, Ethiopia

6.8.3. Effectiveness of Ma-Mas intervention on secondary outcomes

The mobile assisted medication adherence support intervention had no statistically significant positive effect on self-reported TB medication adherence as measured by ACTG and VAS (P value >0.05) (see Table 22).

Table 22: Ma-MAS intervention effect on self reported TB medication adherence, Addis Ababa, Ethiopia

Secondary outcomes	Adherence measured by ACTG [95%CI]	P value, 2- tailed	Adherence measured by VAS [95%CI]	P value, 2- tailed
Before intervention				
Ma-MAS group	70.97%		91.40%	
(n=93)	[61.74, 80.19]		[85.69, 97.09]	
Control group	79.57%		92.47%	
(n=93)	[71.37, 87.76]		[87.11, 97.83]	
Absolute Difference	-8.6%	0.174	-1.0%	0 707
(AD)	[- 20.9, 3.7]		[-8.8, 6.7]	0.787
After intervention		P value 1-		P value
		tailed		1-tailed
Ma-MAS group	83.72%		96.51%	
(n=86)	[75.91, 91.52]		[92.63, 100]	
Control group	82.95%		92.05%	
(n=88)	[75.09, 90.81]		[86.39, 97.69]	
Absolute Difference	0.8%	0.446	4.5%	0 1029
(AD)	[-8.4, 9.9]		[-1.2, 10.2]	0.1028
	-9.4%	0.137	-5.5%	0 4 5 4 5
DIFF	[-23.3, 4.5]		[-14.3, 3.3]	0.1515

Absolute Difference (AD) calculated Ma-MAS group- control group with 95% confidence interval (CI). DIFF-Difference in difference, AIDS Clinical Trial Group adherence questionnaire, VAS: Visual Analogue Scale

CHAPTER 7: RESULTS OF STUDY FOUR: PERCEPTIONS, EXPERIENCES AND ADOPTION INTENTION

7.1. Introduction

This chapter presents the interpretive and cross-sectional survey studies assessing the patients' perceptions, experiences, and adoption intention of mobile-assisted medication adherence support (Ma-MAS) intervention, which involves daily SMS messages and weekly phone calls for eight weeks. The first section of the chapter presents the interpretive study results of patients' perceptions, experiences, and adoption intention of the Ma-MAS intervention. The second section presents the cross-sectional survey results of factors influencing patients' adoption intention of Ma-MAS intervention using the Unified Theory of Acceptance and Use of Technology (UTAUT). The model identified four key factors that affect patients' adoption intention of a technology-based intervention: performance expectancy, effort expectancy, social influence, and facilitating conditions. These factors were analysed to its influence in the context of Ma-MAS intervention.

7.2. Background

7.2.1. Demographic characteristics

Twenty TB patients (15 males and 5 females) who received the Ma-MAS intervention were interviewed to assess their perceptions and experiences regarding the adherence intervention. The participants' ages were uniformly distributed, ranging from 21 to 60 years. There were nine participants who had attended education above grade 12, five who had attended grade 9 to 12, and six who had attended grade 1 to 8. Most participants either worked in private organizations (n=9), were employed by government organizations (n=5) or were unemployed (n=3) (see Table 24).

		Partic	ipants																		
Background	n	А	В	С	D	Е	F	G	Н	1	J	K	L	М	Ν	0	Р	Q	R	S	Т
Gender																					
Male	15	Μ		М		Μ	М	М	Μ			Μ	М	М	Μ	М		Μ	М	Μ	Μ
Female	5		F		F					F	F						F				
Age group																					
20-24 years	3									21				22							22
25-29 years	5		29			28		27				27						27			
30-34 years	4				30		30		32							30					
35-39 years	4	38											35				38		35		
40+ years	4			49							60				53					60	
Educational																					
status																					
Grade 1-8	6	5				8			8		8			8					8		
Grade 9-12	5			12	10			10				10			10						
Grade 12+	9		В				В			В			В			M*	M*	D		D	В
Employment																					
status																					
Self-employed (SE)	1				SE																
Private organization employed (POE)	9	POE				POE		POE	POE			POE	POE		POE			POE		POE	
Government employed (GE)	5		GE				GE				GE*					GE	GE				
Unemployed (UE)	3			UE										UE					UE		
Student (S)	2									S											S

Table 23: Demographic characteristics of adult TB patients, Addis Ababa, Ethiopia

GE* is Pensioner, Education: D is Diploma, B is bachelor's degree, M is master's degree

Private organization employed (POE) are includes merchants, daily labour employees and other workers in private organization.

7.2.2. Theme, sub-theme, and categories

Based on the thematic analysis of the qualitative data, three overarching themes were identified: acceptance, benefits, and convenience. The theme of acceptance refers to the patients' willingness to adopt and use a particular Ma-MAS intervention. The theme of benefits includes the perceived benefits of Ma-MAS intervention. The theme of convenience relates to the ease and practicability of using the Ma-MAS intervention (see Table 24).

Table 24:	Theme,	sub-theme,	and	categories

Theme	Sub-theme	Categories and sub-categories
Acceptance: Attitude	Attitude	
and adoption intention	Adoption intention	
Benefits: The	Awareness of the disease and its treatment	 A. Awareness of TB medication and the consequence of discontinuing B. Awareness of precautions while taking TB medications C. Awareness creation through asking questions
perceived benefits of adherence	A reminder of TB medication	
intervention	Habit formation	
	Motivation and adherence	
	Sense of professional care	
	General health consultation	
	Content of the messages	 I. Evaluation of the content of the messages II.Suggested contents A. Cause and prevention of TB B. TB diagnosis categories and related symptoms C. TB treatment and its side-effects D. Food and nutrition E. Substance use F. Question and answer G. Courage to talk and psychological advice
Convenience:	Comprehension	
Experience of adherence intervention design	Time of delivery	A. Evaluation of the time of deliveryB. Suggested time of delivery
	Frequency of SMS text messages and phone calls	 A. Evaluation of the frequency of SMS text messages and phone calls B. Suggested frequency
	Adherence intervention period	
	Privacy and confidentiality	A. Confidence in the intervention B. Fear of stigma and risks

7.3. Theme 1: Acceptance: Attitude and adoption intention

7.3.1. Attitude

Most participants had a positive attitude towards the SMS text messages and phone calls intervention implemented to improve their TB medication adherence. In agreement with this point a 60-year-old male participant stated that

"Yes! I am very interested in the support. I have good appreciation. It is such great work to reach all the clients and contribute to their health recovery by providing advice. It is a great job and I appreciate your efforts."

Another 22-year-old male TB patient said that the adherence support intervention was not only useful for him, but was also useful for supporting other patients,

"I have been reading all the texts and obtained the lessons. Not only for ourselves, but the messages also gave us lessons that we can support other victims. It is my first time to use such a service. And it has been very useful to me. I have got it is critically important to me."

However, some participants preferred SMS text messages more than phone calls while others preferred the phone calls. Participants who liked the SMS text messages reported that the phone calls did not go beyond asking a few questions, nor did they encourage the patients to talk more about their issues. The phone call would end abruptly when the patient had no more questions. For example, the 35-year-old male participant stated that

"The text messages are so good especially for reminding me of my medication. That is very good because I receive them in the morning. It helps me not to be reckless. The phone call is just, "are you taking your medication?" It has nothing else, as to me. Except for this, the text message is very good... It contributes a lot to reminding them of their medication." Another 38-year-old male participant added that *"I take out and read the text right away it reaches me. I got everything useful, and I followed it happily."* On the other hand, one 21-year-old female participant reported that phone communication was more beneficial than SMS text messages, because the text messages provided no added information on what she already knew, *"the phone calls have been better. I have been asking the health worker about any issue I face, and they have been responding to me adequately. It was good."*

7.3.2. Adoption intention

Most of the participants had a willingness to adopt the mobile-assisted medication adherence support intervention not only for their TB treatment, but also for other healthcare services. Most of the participants believed that the intervention had a positive effect on their TB medication adherence. The 35-year male participant reported that the adherence intervention was beneficial and should be continued,

"It was a very beneficial follow up. I would like to say that it shall be kept up. I am also grateful to you. I appreciate you, really! It is very vital to follow up the conditions of patients like that of people who support them at home. I am very happy about your support. Keep it up."

The 29-year female participant also suggested that this type of intervention was

"Not only about TB treatment, but it is also good if such an opportunity of having close talk between the patients and physicians is practiced. It opens path to use to consult our physicians regarding our health progress and issues. Let it be applied in other ill cases as well. It is a good start."

Some participants were also willing to continue the Ma-MAS intervention because they found it useful for reminders and for assistance in avoiding undesirable behaviours related to TB treatment. The 53-year-old male participant said that "The text messages were more useful for reminding purpose for they were received on daily basis. The phone call got me encouraged additionally. I am very grateful! ... I have benefited from your intervention. The service will be better if continued. It should be kept up because it will help in preventing undesirable conditions and behaviour in taking the medication."

7.4. Theme 2: Benefits: Perceived benefits of adherence intervention

7.4.1. Awareness of the disease and its treatment

7.4.1.1. Awareness of the TB medication and the consequence of discontinuing

The SMS text messages were considered useful for promoting awareness of the importance of taking TB medication as directed by the physician, and the consequences of discontinuing the medication. *A* female participant aged 60 years stated that

"I have obtained good awareness from your messages; for example, the harm that is embraced if one quits the TB medication, and the improvements that I will get if I take the medication accordingly."

Another 53-year-old male participant reported that the Ma-MAS intervention enabled him to continue the TB medication as directed,

"You are writing us educative lessons. For example, it makes people aware that quitting the medication unduly will cause illness or body adaptation to the medication. And it has helped me to continue taking the medication so far and will continue to the end.... It will even create awareness for more people if enhanced ... It enabled me not to quit taking the medication for I got explanation of the effects if I quit it. They supported me not to quit taking the medication, not to be reckless, and to apply the prescriptions ..." Another male participant who was 35 years old also said that

"They give good awareness of how to take them, if side effects, stuff that we should not intake and others. Indeed, there was a certain repetition after some time. It has been good beyond that."

The advice from the phone calls was also found to be useful to patients for helping them to continue taking their TB medication as directed by the physician. The 22-year-old male participant reported that, at first, his intention was to discontinue his TB medication, but he later changed his mind after discussing the issue with his healthcare provider, particularly on the potential consequences of discontinuing the medication.

"It has benefited me a lot. Every message teaches you a lesson. I was faced with some boredom and intent to quit the TB medication in the beginning days. On the first few days, for example, the health worker advised me on the consequence of the illness. He notified me that the illness may spread to family members, friends, and other community members if I quit the medication. I decided to continue it not to be a source of health problems for other people. I was reckless on the first days. I began to pay attention when the healthcare professional continued to call and give me information."

Another 28-year-old male participant also reported that the phone communication had helped him to take the TB medication properly, stating that

> "I am taking the medication properly. The healthcare provider gives me useful information on how to follow my medication. And I can forward any point I have to them."

7.4.1.2. Awareness of precautions while taking TB medication

One of the benefits of the SMS text messages was the promotion of awareness about the precautions needed while taking TB medication. A 27-year-old male participant said that

116

"It is greatly beneficial to me by noting me about foods and drinks that should not be taken and things that are forbidden to do while taking the medication. The messages incorporated in the text are so vital. The messages have enabled me to stay away from things that are forbidden when one is taking the medication."

The 60-year male participant also reported that the messages raised awareness about the negative effects and consequences of substance use while taking TB medication,

"They were very important, because some people are reckless, and some other people may engage in drug addiction. The information beware the people how much drug addiction will affect them and the likelihood of getting heal will decline steeply if taking the medication is quit. It was very important."

This participant also explained how the messages made him aware of the effects of substance use while taking TB medication, stating that

> "Yes! For example, it deals with the gradual harm of substance abuse while taking the TB medication may result resistance, and problems resulted from poor adherence. Generally, they were educative."

7.4.1.3. Awareness creation through asking questions

The regular phone calls created an opportunity to ask questions and helped the patients to better understand TB treatment-related issues. The 27-year-old male reported that the phone calls were good for gaining greater awareness, stating that "You can ask the health worker whatever your concern is. There was a question that I asked and got an explanation. I believe that is good ...". Another 38-year female participant also mentioned that she was able to ask questions and understand more about the TB medication, "The phone call has supported me a lot. I have been asking if taking the medication is associated with one or another condition. It has supported me better." The 32-year-old male participant added that, "It aimed at

educating. Similarly, the checking out via the weekly phone call was good. I also have got clarity by asking different things."

7.4.2. A reminder of TB medication

The SMS text messages served to remind patients to take their TB medication if they forgot to do so. The 27-year-old male participant reported that the daily SMS text messages helped him to confirm if he had taken his TB medication,

> "Regarding reminding me to take my medication, it has been good for me. I should take my medication at 07:00am. Sometimes, I forget to do that on time. The text reaches me at about 07:30am. Thus, it reminds me if I have forgotten till 07:30am. That has been very vital to me. For the text message reminds me every day, there is no possibility for me to forget to take my medication currently ..."

Another 38-year-old female participant reported that the SMS text messages were useful for reminding her about her TB medication, despite it not being a real-time reminder,

"It is beneficial. Of course, I take the medication early because I leave for work earlier. I accept that the text message is good although it was not alerting enough to me. It is not as reminding as an alarm that is set for waking from sleep. Still, it has supported me to cross-check that I have taken my medication. It has been supporting me to remember and get sure whether I have taken the medication or not. I get a click and take a moment to remember if I have taken my medication, because I take the medication before the text comes."

The SMS text messages were found to be useful for avoiding forgetfulness related to being busy with work and other social activities. The 53-year-old male participant reported that the demands of his job could result in him forgetting to take his TB medication if he had not received a daily SMS text message reminder, *"Really, it has supported me a lot. Because of*

the nature of my work, it was a good reminder to me. I am glad about it!" The 22-year-old male participant also said that "there are days on which you may forget to take the medication because of several factors. For example, there was a day that I took it at about 10:00am after seeing the message." Another 35-year-old male participant reported that he once forgot to take his TB medication, but he remembered because of the SMS text message,

> "The text messages are so good, especially for reminding me of my medication. That is very good because I receive them in the morning. It helps me not to be reckless. It supports the people who are outside the health field of study. It contributes a lot to reminding them of their medication. There were such days. I forgot on some days and reminded by the text message. I faced that overwhelmed with work."

The phone calls also proved to be useful for helping the patients remember when they forgot to take their TB medication. There were instances where patients forgot to take their TB medication and remembered as a result of the phone calls. The 22-year-old male participant stated that

"I took the medication an hour late after a phone call was given to me. The benefit has been significant to me. Unfortunately, I forgot to take it on the day. I did not see the text message either because I was sleeping. It was the call ring that woke me. I meant that the benefit is significant."

Another 53-year-old male participant added that *"It had similar support with the text message. There were times that it woke me up in the morning. I could not hear the text message if I was sleeping."*

7.4.3. Habit formation

The daily SMS text messages were also found to be useful for developing the habit of taking their TB medication at a regular consistent time. A 29-year-old female participant reported that

119
she was not taking the TB medication consistently at the same time before the start of the daily SMS text messages, stating that,

"Previously, sometimes, I had been taking the medication on undue time. After I began to use the messages, I have become able to keep the timing. It supported me in that aspect. I used to stay as late as 09:00am. I have improved it to 08:00-08:30am with the support of the reminder of the messages. I did not stay as late as 09:00 after I began to receive the SMS text messages reminder ..."

Another 22-year-old male participant stated that,

"Very much! It makes you alerted about taking the medication for you receive it on daily basis. The reminder teaches you the habit of waking up early, taking medication 30 minutes before having breakfast. Even, a message is received at 04:00pm in the afternoon will contribute to remember that he/she should take the medication in the next morning. It gives a responsibility. That has been very important to me."

7.4.4. Motivation and adherence

The daily SMS text messages were found to be useful to motivate patients and enhance their commitment to properly adhere to their TB medication. The 53-year-old male participant reported that he read the daily SMS text message and complied with the advice, *"Currently, I have been taking the medication according to the advice that you have been giving me, by adhering to the timing you advise me."* The 32-year-old male participant also said that *"I have been taking the medication properly. I have not made a quit. I have been taking it on time … I am taking it properly. There has been no gap with my medication."*

Participants also reported that getting a phone call and being asked about their issues by a healthcare professional gave them hope and motivation. The 49-year-old male participant

reported that the phone calls were helpful for taking the medication properly, even if the TB medication was strong and had side-effects,

"It (phone calls) was very good. It was constructive. The medication was also good, although it is difficult for the patients who are physically weaker. It is painful because the pill is very powerful. Nevertheless, it is impossible to heal without the pills. On my part, I have been taking the medication with great attention and commitment. I will complete that accordingly."

Another 22-year-old male participant added that, "Talking through the phone is very important for some people even under normal circumstances, let alone for patients. It gives them hope and motivation for them."

Daily SMS text message was also useful for preventing negligence and to ensure adherence to the TB medication. The 60-year-old female participant said that *"I took the medication duly in the morning. It (SMS) has supported me that I should not be reckless about my medication."* Another 60-year-old male participant also mentioned that the daily SMS text message was very helpful for alerting him to the TB medication, *"Very much! The text messages give us advice to be in conditions conducive to taking our medication on time. I follow it attentively."* Another 30-year-old female participant added that,

> "They are good. I follow such things attentively. I take the medication carefully indeed. The information is useful for those who do not have good adherence. They (SMS text messages) are good reminders to them."

The phone discussions were also useful and encouraging for adhering to the TB medication. The 28-year-old male participant reported that he was taking the TB medication according to the phone advice and stated that *"I am taking the medication properly. The healthcare professional gave me useful information on how to follow my medication. And I can forward any point I have to them."* Another 27-year-old male participant added that, *"Its benefit was to* promote adherence. They give you different advice, I mean the health workers. That is good because it encourages you to take the medication properly."

7.4.5. Sense of professional care

The SMS text message and phone call intervention to improve TB medication adherence also proved useful for enhancing patients' feelings towards professional care. The 60-year female participant was happy with the professional healthcare, stating that

> "I understand how much you are concerned about our health, although it is the responsibility of the person to take care of I. Even, I have not ever expected that there is such kind of support. It is very good."

Another 22-year-old male participant also expressed his feelings about the professional care, "It is my first time to use such a service! And it has been very useful to me." Another 35-yearold male participant added that "... In addition to the text message, the healthcare professional was following me up by giving me a phone call. Giving attention to the patients that much itself is very good. I am happy about it!"

Participants were happy with the health advice on the phone calls because they were able to ask questions and get explanations from healthcare professionals. The 21-year-old female participant stated that, *"I have been asking the health workers about any issue I face, and they have been adequately responding to me. It was good."* Another 29-year-old female participant had experienced what she considered to be good professional care, because she had the opportunity to ask questions, *"It was very good in giving us the opportunity to ask if we face any concern in our medication. The physicians encourage us to call them any time and ask if we have any concerns. It was so good."*

Participants also felt that the professional care was good because the healthcare professional made the phone calls and asked them about their health status. The 27-year-old male participant stated that

"The phone calls also have been very good communication. They have been asking about my progress and if I have been getting better or not from time to time. They have been asking me about my progress when I took the medication. I consider that it has been so good."

Another 28-year-old male participant expressed his feelings about the care, *"They asked me if I was taking the medication properly and if there was any new problem with my health. I told them accordingly."* The 22-year-old male participant added that

"The phone call that you gave us is important in case there is certain concern. They were calling and asking me if I had an illness or any problem. I think that the communications were useful in this regard as well."

How the healthcare professional communicated through the phone calls was important for the patients' perceptions of their professional care. The 32-year-old male participant reported that he was happy with the communication of the healthcare professional and how they answered his questions, *"They were good. They asked me and had a good conversation with me. They also gave me good explanations for my questions. It was good."* A 30-year-old female participant also expressed her feelings about the professional care of the healthcare professional, *"They have been giving me a call. They have been asking me if I would have experience of different illness."*

The healthcare professionals' efforts to improve medication adherence enhanced the positive perceptions of the professional care, and the patients' commitment to adhering to their TB medication. The 60-year-old male participant shared his feelings of appreciation for the professional care team,

"Here one thing that I want to assure you is that I appreciate the struggles of your health team in supporting people. I am happy as an Ethiopian and about your health principles when I see the efforts exerted to support others."

A 27-year-old male participant also said that *"the health workers are doing well in giving us important advice and reminding us to take the medication properly. They are good".*

7.4.6. General health consultation

Having a health consultation through a phone call was one of the benefits of the Ma-MAS intervention. Patients could have a consultation on their health issues through a phone call without needing to attend the health facility. The 27-year-old male participant explained how the phone calls helped him in relation to his health consultation,

"I explain to them when they call and ask me about my health progress. Last time, for illustration, I moved to a certain area that has cold weather. I had some difficulty, perhaps because of the cold. As a result, I asked them on the phone about it. I told them about the bad sensation I experienced. And they advised me to visit my physician by guessing that it may be because of the cold weather. Accordingly, I visited my physicians and they explained to me that such sensations develop because of the cold weather. Finally, my health condition got back to good status after I returned to my original place."

The 21-year-old female participant also reported that she received a health consultation without needing to go to the health facility,

"I could get explanations on phone by asking the workers without going to a health facility. It was good to ask if I have any concerns. That has been helpful ... I had a cough in between and asked the health care professional at the health facility if it was among the side effects. She told me to wait sometime till I take a certain amount of the medication instead of giving me answers. I asked your healthcare professional through phone, and he advised me to have a Medical examination and I did so. Say to me thank you to him."

Another 22-year-old male participant also shared his experience, *"I consulted him through phone when I got swelling on my body. He told me to visit a health facility … To talk about the case itself is something."* Another 32-year-old male participant added that *"… I also have got clarity asking different things. For example, they supported me when I could not find vitamins because it was not available at the health center."*

7.5. Theme 3: Convenience: Experience of adherence intervention design

7.5.1. Evaluation of the contents of the messages

Most participants believed that the content of the SMS text messages was appropriate and useful for promoting TB medication adherence. The 38-year-old female participant reported that the information included in the SMS text messages was good for promoting TB medication adherence,

"It gives good notification to hold the medication along us if we move to another area, not to forget it. The messages also include the risks associated with failing to follow the prescription medication accordingly. It is not something that we can undertake recklessly. It also results that our body may adapting to the medication and that it may not be easily treated and cured. It is forbidden to drink alcohol. The messages tell us about all these issues."

Another 38-year-old male participant stated that

"The messages to take the medication properly and not to drink alcohol are very good. There was also the message of not to forget to grab the medication if we move away from home for some days, be aware not to forget to do this. That was very useful." Participants also reported that the overall content of the SMS texts was useful for awareness and promoting TB medication adherence but recommended that repeated SMS text messages should be avoided. The 22-year-old male participant stated that

> "I have not seen a defect in the messages. I read and got awareness. There has been some repetition of course. If possible, it would be good if that was reduced. I am satisfied with the messages beyond that. The text messages give good lessons every day. It gives us information about what to do when we move to another place, to take medication from other health facilities for example. I have been reading all the texts and obtained lessons."

Another 35-year-old male participant added that

"They are very good. They (SMS text messages) give you good awareness of how to take them, if side effects, stuff that we should not intake and others during the medication, the conditions that make people forget taking their medication, for example. Indeed, there was a certain repetition after some time. It has been good beyond that. I properly read them daily."

Similarly, most participants reported that the information from the phone calls was also useful because open questions were asked of the patients, which made it a two-way form of communication. The 27-year-old male participant said that

"This one (phone call) is open. You can ask the health worker whatever your concern is. There was a question that I asked and got an explanation. I believe that is good. It depends on the service seekers. It is good to me because I have got explanation when I asked about my concern."

Another 30-year-old female participant stated that the phone communication included open questions that asked the patient about any challenges they had encountered, *"They are questions if we are taking the medication properly and if there is any challenge. And they are*

good. They also advise us to visit our physicians if there is any concern. That was good." Another 30-year-old male participant added that,

> "Yes! I am happy about the question they ask, "is there any challenge you faced?". I have not faced any challenges. Yet, the question is very important for others. I believe that asking that question and inviting the patients to have explanation on their concern is good opportunity."

7.5.2. Suggested content of messages

7.5.2.1. Cause and prevention of TB

It was suggested that information about the causes and prevention of TB should be included in the SMS text messages. The 38-year-old female participant stated that

> "It is good if awareness is created for us regarding the causes behind the health problem. Nobody has explained that to us. Additionally, awareness is good for us to have regarding how it can be prevented."

Another 60-year-old female participant also suggested the inclusion of information about the causes and prevention of TB, and the precautions to be taken once a person developed the disease, *"It is good if the contents include information about the causes of the illness, how to protect oneself from it, and the cautions that we should take after contracting the illness."* The 30-year male participant also suggested that information about the risk factors for TB and prevention methods should be included,

"Yes, with respect to vulnerability, it is better if the information of how people become bound to the illness, so that they can refrain from the exposure. In addition to educating the people who have contracted the illness about the disease conditions, and there shall be some contents about how people can protect themselves from contracting the illness."

Some participants also suggested that information about the causes and prevention of TB should be included in the phone discussions. According to the 60-year-old male participant, the healthcare provider did not clearly explain to him how the disease was caused or how to prevent it,

"It is good if the contents include information about the causes of the illness, how to protect oneself from it, and the cautions that we should take after contracting the illness. They (healthcare professionals) said that it is contracted when our body immunity is poor. Yet, I am eager to know how much percent that is precisely."

7.5.2.2. TB diagnosis categories and their related symptoms

There was also a suggestion to include the symptoms associated with the different TB diagnosis categories. The 29-year-old female participant said the contents of the SMS messages were good, incorporating a variety of information in each day, but she added that it would be good if the diagnosis categories of TB and information about the related symptoms were included in the SMS text messages,

"The contents were good. Yet, it is important to include in the messages that TB illness has different types. For example, my sister who is taking TB medication along with me got only a swelling on her neck. She had no cough, no fever, only a swelling on her neck. We did not guess it was TB. We attributed it to Goiter for she had that case and the swelling developed on her neck. She got different examinations at different facilities. It was lastly she was diagnosed with TB. Thus, it is good to indicate that TB can have varying symptoms and to have a checkup if one or another symptom is experienced. A very few people know of different cases of TB illness. Very few! For example, we did not have such knowledge except after the case of my sister. Thus, I think that it is very

good if such information is included in the text messages so that better awareness will be created."

Some participants also suggested that information about the different TB diagnosis categories should be included in the phone communications. For example, the 38-year-old female participant said that *"The question about my health problem is swimming my mind. Mine is not a lung case. I would be happy if you give me information about my specific case."*

7.5.2.3. TB treatment and its side effects

The SMS text messages should also include more information about the side-effects of TB medication, according to some participants. The 21-year-old female participant reported that information on the side-effects was not well explained by the healthcare professional who worked at her health facility,

"It is good if awareness is created regarding the side-effects of taking the medication and the causes of the illness. I have asked the healthcare professional at the health facility multiple times about the side-effects of the medication. They did not like to explain to me clearly. They do not like to explain that and the cautions that we should take. I do not know why. It may be because it disturbs us if it is hazardous. I think that it would be better if that is included."

Similarly, a 30-year-old male participant shared his experience with drug side effects, noting that he was not aware of them beforehand. He recommended including a message about these side effects,

"It would be good if the symptoms and signs that the medication causes were included in the messages. For example, I experienced signs like sight delusion, loss of appetite, and others. Had that information been included in the SMS text messages, I would not have been confused if those signs were common sideeffects or particular to me. I think it is good if concerns around the medication are included."

According to a 60-year-old male participant, it would be useful to ask the patient about their side-effects and their level of adherence via the SMS text messages,

"Furthermore, it would be good if you add content about the side-effects of the medication upon our health. You better send us text messages about this. Not all people have similar experiences. It is better if you can add questions about the side-effects a patient has faced and the level of his/her adherence. This is a point that may be improved."

In addition, some participants suggested that messages regarding the use of drugs on TB, the consequences of not taking TB medication at the same time, and facts about the disease, should be included. The 53-year-old male participant stated that *"I would be happy if you give deeper information about the problem of taking the medication at any time, maybe in the morning or midday or afternoon."* The 30-year-old male participant said that the SMS text messages provided good highlights of the information provided by the health facility, but additional facts about the disease, such as the chances of curability and other facts related to the disease were needed,

"Furthermore, it is good if some facts about the illness are incorporated in the texts. Research findings about the curing possibility of the sick, for example! It is good if patients are given information of promising realities about their health conditions."

7.5.2.4. Food and Nutrition

There were some participants who stated that they would like to have nutritional information included in the SMS texts. A 60–year old female participant suggested,

"It is good if the content includes information about how to protect oneself from it and the cautions that we should take after contracting the illness. I have the idea that it is good if the information about the food items that are vital for the patients are included in the information. People will apply that according to their ability."

Another 35-year-old male participant suggested that

"It is good to be clearly identified and delivered to the patients if there are food items that are forbidden to eat. For instance, there are food items that are defined traditionally not to take while taking certain medication. On my part, I have asked the healthcare professional on the phone calls. They told me nothing."

7.5.2.5. Substance use

Some participants suggested the inclusion of more information about the adverse effects of smoking on TB medication, as well as the effects of smoking on general health. Some of this information was included in the SMS text messages, but participants suggested that more information about the negative effects of smoking should be added. The 22-year-old male participant said that *"It is good if you give more attention and address about the harmful effects of smoking cigarettes. I would be glad if you do so."* Another 60-year-old male participant added that *"the gradual harm of having substance abuse while taking TB medication can cause drug resistance and health problems as result of medication non-adherence"*.

7.5.2.6. Question and answer

Participants suggested that the phone intervention could be designed as a 'question' and answer' format to ensure the patients understand the SMS text messages and comply with the advice in the messages. The 27-year-old male participant said that

"It is better if questions of how much the patient is learning the benefits of the medication and if his/her adherence is good. Additionally, a patient should be asked if he/she is staying away from forbidden things when they take the medication."

Another 53-year-old male participant explained the importance of asking a questions in the SMS text messages and phone calls added that

"One thing that I want to say is it will be better if you can assess if a client has really understood and practice your recommendations... For example, you may ask him/her to explain the impacts of quitting the medication unduly after creating awareness for him/her. The patients will be able to have good awareness and even can teach others afterwards... The question-answer will enable the patients not to fear to discuss and get rid of doubts about the medication. ... Thus, I advise you to have certain question-answer pieces in both the text and phone call communications."

7.5.2.7. Courage to talk and psychological advice

Some participants reported that the healthcare professionals who made the phone calls should be more courageous and supportive in talking to the patients to avoid the phone conversations ending too abruptly. A 30-year-old male participant reported that the healthcare professional who gave calls had a list of questions for the patients, but if the patient had no questions to ask, the phone call would end. The participant stated that

> "Very short duration was allocated for the phone call. They would say goodbye and hang up if we say we do not have a question to ask. Anyway, it was useful. It would be good if more time were allocated, and the healthcare professional gave time to listen to the beneficiaries. To tell you the truth, I would not raise a question though I had something in my mind to be asked. I did not ask because of some reason. So far, it might end in less than a minute if the beneficiary

receiving the phone had no question, right? It was good thing to consider 5 minutes as an average, but it will be better if it is improved to 7 minutes. It will be good if the questions of what the patients experience in association with the medication, how they are taking their medication, and other related questions are incorporated. To get their experiences, by asking them what they face for example."

The 35-year-old male participant added that

"Moreover, it will be better if the dialing health workers provide some pieces of advice. Giving us the opportunity to ask them if we have a question is a good thing. Additionally, it will be better if some pieces of advice are provided along. It is better if the dialers talk friendly and encourage the patients to discuss actively. It may be because of me if I consider my particular case indeed. Yet, the health workers who make the phone call shall motivate and push the patients to talk actively. That is about the phone call. And I do not have recommendation about the text message."

7.5.3. Comprehension

Most participants reported that the SMS text messages were easy to read, and their meaning was easy to understand. The 27-year-old male participant said that "Yes, *it has been very clear. The language is very easy to read and understand*". Another 30-year-old male participant added that "They were clear and brief! They were good. It had nothing difficult. Nothing was puzzling; they set them clearly. Nothing was ambiguous." The 60-year-old male participant also said that "For sure, I have not seen any difficult concepts. They were advice to have some proper behaviour. I have not seen anything difficult to understand. It was a simple language."

In addition, participants reported that the SMS text messages were easy to understand because they were prepared in Amharic (the national official language of Ethiopia). The 29-

year-old female participant stated that "I have followed them accordingly. They were prepared in Amharic language. There has been no difficulty. They were clear enough to transmit the intended information. They were good." Another 38-year-old male participant added that

> "I read and used it properly. All the points are good. Nothing challenged me. I have used the messages accordingly. Yes, doing that in the Amharic language is a very good decision. I could not have read or understood had it been prepared in English language."

However, one male participant, aged 22 years, said that the SMS text messages were understandable, but some texts had spelling errors that sometimes made the message difficult to comprehend,

"Yes, they were understandable for I am a student. Yet, some errors exist in the texts, spelling errors for example. It is sometimes difficult to get the meaning. It needs to cut and merge to have meaningful sentence. It is okay for me indeed. I can understand them correctly. However, it may be difficult for those who are not educated or lower graders. On my part, I have managed to understand the messages of the texts. I am telling as a recommendation for you. I did not face a challenge on my part."

Similarly, other participants reported that the phone communication was clear and understandable. The 27-year-old male participant stated that *"There has been no terminology that I could not understand either. It has been going well."* Another 22-year-old male participant added that *"No! The dialogue is so brief indeed. They ask me shortly and I also respond to them shortly."*

7.5.4. Time of delivery

7.5.4.1. Evaluation of the time of delivery

Most participants reported that the delivery times of the SMS text messages were convenient for their TB medication reminders. The SMS text messages were scheduled for 7:30am, and most were delivered between 7:30am and 8:00am in the morning. The 29-year-old female participant said that *"I do not have complaint about the timing of the text messages. It was proper time to remind me to take my medication. It was in the morning and before you begin a daily life journey."* Another 35-year-old male participant added that *"It reminded me when I was about rising from bed. Commonly, it is the time that people get prepared to kick-off to job or else. I think it was a good time."* Another 60-year-old female participant also said that *"It is very suitable time. The clients are reminded right after they wake up. It is good timing.* Yet it sometimes arrives late. Generally, it is good. Keep it up." The 22-year-old male participant also reported that the SMS text messages and phone calls were made to him in the early morning before class time,

"The messages and phone calls were given early in the morning, before work or class time. To me for example, he gives me a phone call before 08:00am. You go nowhere before 08:00am, you are at home. It is similar for text messages as well. I could read right away at that moment."

However, some participants reported that sometimes, the SMS texts and phone calls had reached them late, after 8:00am.

7.5.4.2. Suggested time of delivery

Some participants suggested scheduling the delivery of the SMS text messages according to the patients' actual time of taking their medication. The 28-year-old male participant said that

"Generally, it is good for me. The thing I have mentioned for improvement is to make the text message delivery exactly fitting to the time that the patients take their medication, delivering the message on 07:00am if patients tend to take their medication at that hour."

Another 38-year-old male participant added that

"This depends on the time that the person wants to take the medication. Some may prefer to take it at 08:00am; some others may need to take it at 07:30am. On my part, I take it around 06:00am most often. I take at that time for I need to leave for job earlier. I took it earlier."

The 22-year-old male participant also said that

"It is good if you send the message in accordance with the time that patients take their medication. I will take it at 06:00am. For example, it is meaningful if the reminding message reaches a patient at 06:30am if he/she takes the medication at 07:00am. It may not be meaningful to receive the message at 07:30am for such a patient."

Other participants suggested sending the SMS text messages before 7:30am in the morning because people wake up early to get ready for work. The 38-year-old female participant reported that the SMS text messages were delivered while she was getting ready to go to work, so she had not seen the text messages immediately,

"As I have told you, it could have benefited me more if it had it been sent to me earlier. I took the medication earlier. And I received the messages somewhat late. It is around 07:30am. However, most often, I see it later. It may be because of the hustle that we have in the morning ... I see late, but I use the text for cross-checking that I have taken the medication. I believe that it is better if received earlier in the morning so that it will alert us in advance." Similarly, some participants recommended scheduling the timing of the phone calls in the same way as the SMS text messages at 07:00am before the person went to work because people may not pick up their phone when they are busy with their job or routine activities. The 27-year-old male participant said that *"I think that it is better if the phone call takes place at 07:00am instead of 07:30am. I am receiving the call after 07:30am now."* Another 29-year-old female participant suggested that

"It is better if given early in the morning for it will support as a reminder as well, if a person has forgotten but has not seen the text message for example. For they are overwhelmed by daily life hustles, some people may not read text messages received. As I have told you, it will be more useful for reminding if the phone call is given early in the morning like that of the text message (07:30am)."

The 60-year-old male participant added that

"I may not pick up my phone when you call me 07:30-08:00am. It is a time to drive to the office for work. I may not respond because I am driving at that time. However, I do not believe that all beneficiaries have similar conditions."

7.5.5. Frequency of SMS text messages and phone calls

7.5.5.1. Evaluation of the frequency of SMS text messages and phone calls

Most participants reported that the daily frequency of the SMS text messages was appropriate for reminding the patients about their medication because the medication is taken in the morning on a daily basis. The 27-year-old male participant stated that

> "Sending text messages everyday has been good because the medication is taken daily. You may forget it sometimes in some conditions, going to work early for example. And texting us every day is very good to be reminded and

prevent such a gap. The text supports us like that of alarm reminder. One will remember when he/she receives the text."

Another 38-year-old female participant added that

"You may forget to take the medication if the frequency is going to be less. It may be forgettable if the current frequency is reduced. And the current frequency is enough to remind us. I think that the logic of sending us the message every day is that the medication is taken one time in a day."

Another 22-year-old male participant also added that *"Sending text messages every day is very good. It enables the person to be alerted about his/her health if he/she is a patient. To give a reminder every day is a very determinant factor."*

On the other hand, only a few participants said that weekly phone calls to the patients for adherence support would be adequate. A 30-year-old male participant expressed interest in receiving weekly phone calls to prevent monotony,

> "I think it is good if the phone call is given 1 or 2 days in a week. Even, I would not remember the day of call. I mean that it has not been dull for it was only one day in a week. It is better if the current frequency is kept on. It may be dull if increased."

Another 27-year-old male participant also stated that *"I think the weekly-based phone call is also good enough to explain one's status and progress. One can explain the feeling that he/she experiences inside. I consider that it has been very good."* Some participants also reported that because the phone calls were made with the support of the daily SMS text message, the weekly phone call was adequate. The 35-year-old male participant said that

"As to me, it is very adequate including the phone call. Calling the patients every week is also good. It is adequate. Giving attention to the patients that

much itself is very good. I am happy about it. It was in addition to the text messages that were sent every day."

Another 60-year-old male participant said that *"I think it is better to have it weekly instead of daily. Along with the text messages, it is enough."*

7.5.5.2. Suggested frequency

There was a consensus among most of the participants to increase the frequency of the phone calls, but some suggested making the phone calls twice, while others suggested three times per week. The participants' reasons were that they could get additional time if they had any health concerns they wished to discuss with the healthcare professional, the weekly phone call may not be adequate for those who were unable to read a text message, the weekly phone calls may not be adequate for patients who do not give due attention to the TB medication, and some mobile phones cannot read Amharic text messages.

Most participants suggested making the phone calls at least twice per week. The 27-year-old male participant said that

"It will be better if the phone call is improved to two days a week. It gives you more opportunity to express all your feelings, including if you face some illness. Notably, the current one-day call is not bad that much, of course. Yet, I think that it will be better if it is given on two days a week. That would be better because a patient may face a challenge and seek explanation from the health workers more frequently."

Another 38-year-old male participant suggested that a weekly phone call may not be adequate because some people who were not very mindful of their TB medication may forget to take it,

"Let me tell you one thing here. It is adequate for me. The current frequency is enough for me. However, it is better if you add one more day and make the phone call two days a week. Some people may forget the medication with the current frequency. Not all people are similar. Some people may be less attentive to a message that is memorized only one day in a week."

Another 22-year-old male participant added that

"Regarding the phone call, it is enough to me. Indeed, it may be good for others to make the phone calls 2 days a week. You may decide that by hearing the voice others ... It will be better to have more consultation. New symptoms may happen from time to time which may cause fear. It is easier to consult the healthcare professional on the phone instead of going to health facilities. I say it is good. You may add if you think something more is needed."

Other participants suggested making the phone calls three days per week. The 29-year-old female participant said that

"The frequency has been enough for me. However, it may be needed to increase the frequency of phone calls for those clients who may not be able to read and those whose phones may not read the Amharic text letter. As a result, they cannot read and obtain the messages of your texts."

Another 53-year-old male participant said that

"Yes, importantly! For example, there may be beneficiaries who cannot read. Thus, it will be better if the phone call is increased to 3 days in a week, the 1 day shall be improved. Such beneficiaries will be encouraged with the calls they cannot read. There are people who prefer phone calls over text messages. I also think listening is preferrable than reading. On my part, I am okay with the text and phone calls. I benefited from both. However, there are some people who would prefer the phone call. They do not elicit equal messages from text and phone calls. Thus, it is better if we give more phone calls to such patients."

7.5.6. Adherence Intervention period

Most participants agreed that a two-month adherence intervention period was inadequate and needed to be increased to improve medication adherence. However, participants had various suggestions regarding the length of the adherence intervention. Most participants believed that the Ma-MAS intervention would be more effective and desirable if it was implemented for the entire six month of TB treatment period. Participants believed that a 2-month adherence intervention support may not be enough for sustainable medication adherence. The 38-year female participant had doubts about the effectiveness of the intervention because it had been implemented for only 2 months,

"I think that the effectiveness of the aim of the effort is questionable. It is good to investigate it from the perspective of its purpose. A person may take the medication rightly for the 2 months. What about in the remaining months? To me, my husband follows up and supports me eagerly and continuously. I would miss some medication if he were not around. I think it is better if the support is provided all the time."

The participants recommended the intervention to take place across the entire treatment period, and in particular, to start it at the beginning of the treatment. According to the 22-yearold male participant, the adherence intervention would work best if it was started at the beginning of the TB treatment because it would provide early health advice that would help him to continue taking the TB medication for the entire treatment period. The participant reported that he had the intention to stop taking the TB medication at the beginning of the treatment, but changed his mind after talking to the healthcare professional via the phone calls,

> "It is better if you can do it starting from the time the patients begin to take their medication. Even, it is in the beginning stage that it is more important. On my part, for example, I was to quit the medication at the beginning. I came to

continue it after consulting a healthcare professional. It is good thing to have consultation at the beginning time when it is new to follow the medication properly. It is a relief to consult them. And you make a well-informed decision after getting more information. It is better if you begin the service from the beginning."

Another 22-year-old male participant suggested that

"I think, it is better if provided for the whole months/duration. It is most important at the beginning of the medication particularly because the probability of forgetting to take the medication is higher at the beginning. It is challenging in the first 2 months particularly. It will be fine afterwards for it is habituated later. It is better if you work on the first 2 months with great attention."

Some participants also reported that starting the adherence intervention and then quitting was not advisable because some people may act recklessly, have addictive behaviours, may not have family members that would help them to remember their medication, or may not receive adequate advice about the treatment. The 60-year-old male recommended that the intervention should take place throughout the entire treatment period,

> "Some people may be reckless. Some people may not have family to remind them. Some may not have adequate advice. Some may have drug addiction behaviour. Your advice is very important for such people. It informs and guides them ... It is good to continue the support. It has no negative effect. It is not good to begin something and quit halfway. It is not bad if clients are provided with advice from the beginning to the end."

Another 49-year-old male participant added that

"It is good if increased. It is better if you stand by the patients and care for them. Providing the reminder service only for 2 months is not enough. Particularly,

patients who have drug addiction have poor adherence. It is difficult for them to have good adherences for all 6 months on their own. You should follow them up and take them forward ... It is just a human engagement. It is good if you continue to give them inspiration."

7.5.7. Privacy and confidentiality

7.5.7.1. Confidence in the intervention

Most participants reported that privacy and confidentiality were not a problem during the adherence intervention support program. Participants were even confident enough to show the text messages to their family and relatives. According to the 35-year-old male participant, good communication with the healthcare professional and having clear information about the disease and the treatment helped him to feel comfortable,

"It depends on your personal attitude. I did not have a challenge in that regard. I took cautions to prevent transmission when I experienced cough and other symptoms ... I even showed the text messages to others for it helps them to get awareness. Outside the family! (The text shown to outside the family). There is no problem with that. It is me that is the victim. I believe that they would be alerted to protect themselves from infection by getting the awareness from the texts I received. Thus, I did not feel uncomfortable."

Another 53-year-old male participant added that

"No! There was no such problem on my part. I have not faced that so far. Things went on due time. I have even been ready to share information with a person interested to have, because keeping hidden the disease after it has already happened does not have significance. And I did not expect that such a thing would be a risk." Another 35-year-old male participant reported that the messages and calls had no sensitive information that violated his privacy and confidentiality,

"No, there was no challenge on my side. There is transparency in my family. And I shared the information with them at the beginning when the first time I visited the health center and knew my diagnosis. I have given them the information that reminding messages are sent to me. I did not have that challenge for my family knows it. Moreover, the messages are not that risky."

The 60-year-old male participant reported that people infected with HIV (Human Immunodeficiency Virus) might not disclose their status to avoid stigma, but TB does not present such a problem,

"How! I did not have privacy doubt for it is a health issue. I cannot have such privacy at all because it is a common illness that anybody can contract. It is not an illness to be hidden. HIV/AIDs victims may hesitate to expose the illness fearing stigma. In contrast, this illness can be contracted by anybody in different circumstances. I have never thought that there is a problem in exposing one's illness. I have not ever faced a challenge amid exposing it. I have not. I will never face it either."

Another 28-year-old male participant added that

"No, there was no time that I was concerned if somebody may read the text message. No! Nothing is above life. Taking the treatment cannot be a secret. It cannot be beyond saving our life. No! Differently, I think that it is beneficial even for others to get awareness if they see the text I receive."

7.5.7.2. Fear of stigma and risks

Although most participants had confidence in the Ma-MAS intervention, some were afraid of stigma and did not want to disclose their health status, so they tried to hide the SMS texts from

others, delete the text messages after they had read them, and/or did not pick up some of the phone calls made in public places. The 22-year-old male participant had an unpleasant feeling when a person next to him looked at the SMS text messages on his mobile while he was travelling on public transport. The participant said that he did not want people to know his health status so that he could avoid unnecessary stigma,

"To speak the truth, there is a sort of discomfort. One day, I was on a transportation when a text message was dropped. I took out my phone to look for the message. It was a TB message. I opened it. A person sitting next to me saw it. I felt discomfort when he saw it. I do not feel good when people learn that I am TB patient for they may ostracize me for the illness spreads through breath."

This participant also mentioned that he felt bad because the passenger reacted after reading the SMS text message and seeing that he was wearing a face mask,

> "I felt bad when I saw his reaction. I noticed when marginalization happens practically. As a caution, I did wear a face mask. He distanced himself from me. Indeed, I do not blame people because the disease spreads by breath. I am taking cautions to prevent the spread and discomfort of others ... I am a student and I view it critically. I can resist the possible offense. I take caution to prevent its spread to others. I am okay beyond that ... It may be offensive to see when others feel discomfort because of me."

The participant, however, had no such experience during the phone communications because the phone calls were made while he was at home,

> "Most often, I receive the call when I am at home. My family knows it and they encourage me to communicate on the phone properly. They take care of me in relation to the phone I receive from the center."

Similarly, another 30-year-old female participant reported that the daily SMS text message reminded her that she was a patient, and so she read the SMS text messages immediately after receiving them and then deleted them to avoid being seen by other people because she did not disclose her health status to anyone else so as not to cause inconvenience,

"Every time I received the text message, my status, "I am a patient" comes to my mind. I say to myself, "they remind me that I am a patient". There was no such challenge regarding the phone call. With the text message however, I have been deleting the messages right away I read them because it is only me that knows my illness. I am not afraid for myself regarding their view of me, but I do not want to create some inconvenience for them. There is nothing harm or favor they do to me but not to cause inconvenience. I have been careful thus. You make interactions with people for man is a social animal. Your companion may request you to call with your phone. And it is the behaviour of our people that they get into their friends' phone and see into their stuff in their phone. That is why I have been deleting the messages."

The 30-year-old male participant reported that the SMS text messages had no privacy and/or confidentiality issues, but he did not pick up some of the phone calls made while he was on public transport or at the office, which were not considered a convenient place to discuss privacy concerns in public.

"I would see the text message when I meant to check time from my phone because I go to work early in the morning. There was no challenge regarding the text messages. Regarding the phone calls, I doubt I might have missed some calls because I might be on transportation when the phone rings, because I have not saved the phone number and I might not know for the number may be new. There was no such challenge. I am married and I live with my wife. She may have seen the message before I saw it and told me to be

reminded. It was only she that knew about the case. Nobody else knows it. Regarding the phone call, I would not pick some of them. I have been relying on text messages. If I am at the office for example, I may pick it. However, I just listen to the dialer. I do not ask questions for example."

According to the 38-year-old female participant, the SMS text messages created an emotional feeling of illness and discomfort, but these feelings diminished after a short time. The participant also said the risk of discontinuing the TB medication information had made her think about the risk of contracting multi-drug resistant TB, which made her feel quite depressed. Yet, the participant believed that the adherence intervention was beneficial because it alerted her to giving due attention to the TB medication,

"There is a feeling of shock that you face when you receive the message. However, you will manage it by cooling down. It gives some inconvenience when you remember that you are living with the illness. It is overwhelming. It is also frustrating when it is said that dropping the medication may put you to death. The messages thus make you think about the illness now and then. It makes you overwhelmed by the condition. For example, I worry that it may develop to the next stage of TB (multi-drug resistant TB). It gives you certain depression. There was a time that I sat down and cried contemplating that I am in too bad health condition that needs everyday messaging. It may have a good effect of course. You can manage depression and give due attention to the condition. You get back from the uncomfortable mood and pursuing courage."

7.6. Factors associated with adoption intention

In this section, the results of factors associated with the adoption intention of a mobile-assisted medication adherence support (Ma-MAS) intervention are presented. The Unified Theory of Acceptance and Use of Technology model was applied to assess the acceptance of the intervention and its associated factors. Throughout this thesis, the term 'adoption intention' 'behavioural intention' and 'acceptance' are used interchangeably.

7.7. Demographic characteristics of the participants

The adoption intention to use mobile-assisted adherence intervention and the associated factors were assessed by surveying 83 TB patients who received a daily SMS message and weekly phone calls. Among the respondents of the survey, 59% were males and 41% were females. The median age of the participants was 28 years. In each grade, the majority of respondents had completed primary education grade1-8 (43.37%) and followed by above grade 12, which was corresponding to 30.12% (see Table 25). All participants included in this survey had an experience for the mobile–assisted adherence support intervention for two months, and all participants were volunteered to receive a daily SMS text message and weekly phone calls related to TB medication adherence. Additionally, in the regression analysis, the demographic characteristics of the participants (age and gender) had no statistical significance in relation with the constructs of the unified theory of acceptance and use of technology.

Table OF, Demographic characteristics of	adult TD patients Addie Ababa Ethiopia
Table 25: Demographic characteristics of a	aduli TB Dallenis, Addis Adada, Finiodia
rabio zer zernegraphie enalaetenetiee er	

Variables	Frequency	Percentage	
Sex			
Male	49	59.04%	
Female	34	40.96%	
Age in years			
Mean (Sd)	31.98 (±11.10)		
Median (IQR)	28 (13)		
Educational status			
No formal Education	1	1.20%	
1-8 grade	36	43.37%	
9-10 grade	13	15.66%	
11-12 grade	8	9.64%	
Above 12 grades	25	30.12%	

Sd: standard deviation, IQR: interquartile range

7.8. Measurement model

7.8.1. Construct validity and reliability of Unified theory of acceptance and use of technology

For the construct of unified theory of acceptance and use of technology, the measurement items showed convergence validity and reliability within acceptable levels. The AVE value of all constructs in the model was greater than 0.5, and the CR value was higher than the AVE value. There was a greater level of reliability in both the composite and Cronbach alpha which was greater than in the acceptable standard of 0.7 (see Table 26).

Table 26: Construct validity and reliability of unified theory of acceptance and use of technology

Constructs and Items	Factor loading	R ²	AVE	CR	CA
Performance expectancy (PE)					
PE1: Using mobile SMS and calls would help me to	0.832	0.583			
remember my daily medication intake	-				
-	2: Using mobile SMS and calls would help me to 0.912 0.722				
remember my refill medication visits at the scheduled date	0.709	0.906	0.904		
	3: Mobile SMS and calls would be very useful for my 0.879 0.899				
follow-up treatment			_		
PE4: Mobile SMS and calls would provide very useful	0.734	0.573			
services and information to me					
Effort expectancy (EE)					
EE1: Learning to operate mobile SMS and calls would be	0.793	0.628			
easy for me			-		
EE2: My interaction with health professionals through	0.848	0.719	0.728	0.888	0.887
mobile SMS and calls would be clear and understandable	0.0.0	0	0	0.000	0.001
EE3: It will be easy for me to become skillful at sending,	0.915	0.836			
receiving, reading, and interpreting mobile SMS text	0.010	0.000			
Social Influence (SI)					
SI1: People who are important to me think that I should use	0.797	0.523			
mobile SMS and calls	0.797	0.525			
SI2: People who influence my behaviour think that I should	0.920	0.817		0.050	0.050
use mobile SMS and calls	0.920	0.017	0.504		
SI3: The health professionals of this health facility have	0.621 0.363		0.591	0.850	0.850
been helpful in explaining the use of mobile SMS and calls					
SI4: The health facility has indicated that it would support	0.704	0.477			
the use of mobile SMS and calls for my treatment follow-up	0.704	0.477			
Facilitating Conditions (FC)					
FC1: I have the resources necessary to use mobile SMS					
and calls	0.591	0.349		0.795	0.779
FC2: I have the knowledge necessary to use mobile SMS					
and calls	0.851	0.724			
FC3: I believe I can get reliable help from health					
professionals when I have difficulties using mobile SMS and	0.794	0.631			
calls		0.001			
Attitude (AT)					
AT1: Using mobile SMS and calls is a good idea for					
medication reminders	0.789	0.622			
AT2: I believe that mobile SMS and calls can be utilized to	0.844 0.713		0.606	0.821	0.817
improve my medication intake adherence and refill visits.					
AT3: Mobile SMS and calls are easy to work with	0.695	0.483	-		
-	0.095	0.403			
Behavioural Intention (BI)					
BI1: I predict I will use mobile SMS and calls in the next	0.866	0.749			
months for my treatment follow-up			-		
I prefer to use mobile SMS and calls to remember 0.876 0			0.740	0.895	0.881
medications intake and refill visits	0.839	0.704			
BI3: I intend to use the mobile SMS and calls for my					
treatment follow-up when it is available					

AVE : Average Variance Extracted, CR is Composite Reliability, CA is Cronbach Alpha

7.8.2. Confirmatory factor analysis

The structural equation modeling analysis found that the construct validity of the sampling measurement items met the acceptable standard. The items relationship with respective construct were statistically significant; performance expectancy (4 items), effort expectancy (3 items), social influence (4 items), facilitating conditions (3 items), attitude (3 items) and behavioural intention (3 items). The model fitness of the construct validity met the acceptable standard criteria; the RMSEA and SRMR were less than or equal to 0.06 and 0.08, respectively and the CFI and TLI were greater than 0.95.

7.8.3. Construct correlation coefficient and discriminant validity

The correlation matrix between the constructs of unified theory of acceptance and use of technology indicated that there was a statistically significant positive correlation among the constructs of performance expectancy, effort expectancy, social influence, facilitation conditions, attitude and behavioural intention (P value < 0.01) (see Table 27). The measurement model of UTAUT fulfils the discriminant validity standard criteria, the square root of AVE is greater than the correlation coefficient value of the construct with another construct.

Construct	No. Items	Mean (SD)	PE	EE	SI	FC	AT	BI
PE	4	18.15 (2.04)	1					
EE	3	13.59 (1.63)	0.74	1				
SI	4	17.81 (1.78)	0.58	0.65	1			
FC	3	13.32 (1.49)	0.67	0.75	0.63	1		
AT	3	13.49 (1.31)	0.69	0.65	0.69	0.71	1	
BI	3	13.18 (1.80)	0.62	0.62	0.69	0.58	0.72	1
Average Variance Extracted (AVE)		0.709	0.728	0.591	0.590	0.606	0.740	
The square root of AVE		0.84	0.85	0.77	0.76	0.78	0.86	

Table 27: Descriptive statistics, correlation coefficient, and discriminant validity of UTAUT

7.9. Structural model of Unified theory of acceptance and use of technology

7.9.1. Original UTAUT model analysis

In the original unified theory of acceptance and use of technology model, performance expectancy and social influence had a statistically significant positive direct effect on behavioural intention (β =0.22 and β =0.47, respectively, P value <0.05). Therefore, hypothesis 1 and 3 in the originally proposed unified theory of acceptance and use of technology are supported. However, there was no statistically significant relationship between effort expectancy and behavioural intention prediction of mobile-assisted medication adherence support intervention adoption. The original model of UTAUT explained 56% of variance in behavioural intention of mobile-assisted medication adherence support intervention adoption.

7.9.2. Modified UTAUT model analysis

In the modified unified theory of acceptance and use of technology, social influence and attitude had a direct influence on behavioural intention of mobile-assisted adherence intervention (β =0.30 and β =0.54, respectively, P value <0.01). However, performance expectancy and facilitating conditions had only indirect positive effects on behavioural intention mediating through attitude. Performance expectancy, social influence, and facilitating condition had a direct positive effect on attitudes towards mobile-assisted medication adherence support interventions (β =0.21, β =0.25, β =0.29, respectively, p value<0.01). However, the effort expectancy had no statistically significant positive direct effect on attitude and behavioural intention. Therefore, under the modified unified theory of acceptance and use of technology model, hypotheses 2, 5, 6, and 8 are supported (P value<0.01). The modified unified theory of acceptance and use of technology model explained 62% of variance in behavioural intention and 66% of variance in attitude of mobile-assisted medication adherence support intervention. (see Table 28)

Construct relationship	Original UTAUT model	Modified UTAUT model
РЕ → ВІ	0.22**	0.070
EE> BI	0.15	0.15
SI> BI	0.47***	0.30***
FC → BI		-0.089
АТ→ВІ		0.54***
PE → AT		0.21***
EE → AT		-0.045
SI AT		0.25***
FCAT		0.29***
CFI	1	1
TLI	1	1
RMSEA	0	0
R ² BI	0.56	0.62
R ² AT		0.66

Table 28: Construct path relationship in unified theory of acceptance and use of technology

*P value <0.10, ** P value <0.05, *** P value <0.01,

CHAPTER 8: DISCUSSION OF THE FINDINGS

8.1. Introduction

This research aims to investigate the effectiveness of a mobile-assisted medication adherence support (Ma-MAS) intervention on TB medication adherence. The intervention is developed using the MRC framework for complex intervention with seven non-linear steps and consists of two components: (1) SMS text messages and (2) phone calls. The Ma-MAS intervention is designed based on the Information, Motivation, and Behavioural skills (IMB) model and the behavioural change technique taxonomy. To ensure the quality of the intervention, senior experts and health professionals who worked at the ground level were involved in the validation of the SMS text messages and reviewing the checklist questions for phone calls. Before the effectiveness trial began, the feasibility and acceptability of the Ma-MAS intervention were also assessed.

The effectiveness of the Ma-MAS intervention was evaluated using a randomised control trial, while patients' perceptions, experiences, and adoption intention were further evaluated using empirical and interpretive research approaches. The findings of this study will make a valuable contribution to the promotion of TB medication adherence using mobile technology.

8.2. Key findings

The following major findings were discovered:

- The Mobile-assisted medication adherence support (Ma-MAS) intervention significantly improved adherence to TB medication by 15.25% after eight weeks of the intervention compared to the standard care alone in the control group. Ma-MAS also significantly improved adherence to TB medication by 15.30% after four weeks of the intervention compared to the control group.
- Although the Ma-MAS intervention using the information-motivation-behavioural skills model and behavioural change techniques was shown to be useful to improve TB medication adherence as measured by a urine test for isoniazid (INH) metabolites, Ma-

MAS did not have a statistically significant positive effect on self-reported TB medication adherence. However, in the post-intervention interpretive study most patients reported that Ma-MAS intervention had a positive effect on their TB medication adherence.

- 3. The Ma-MAS intervention was useful for awareness creation related to the consequence of discontinuing TB medication and the precautions while taking TB medication. Some patients changed their intention of quitting TB medication after discussing the potential consequence of discontinuing TB medication with their healthcare provider through phone calls. The daily SMS text message helped to avoid forgetfulness due to being busy with work or other social activities. The Ma-MAS intervention was also found to be useful to motivate patients to properly adhere to their TB medication and to develop the habit of taking TB medication at a consistent time.
- 4. The Ma-MAS intervention can also enhance professional care by improving the communication between health professionals and patients that motivates the patients to comply with health advice and take their medication. The Ma-MAS intervention was also found to be useful for health consultation through phone calls without needing to visit the health facility.
- 5. The Ma-MAS intervention was found to be feasible in terms of access to mobile phones and patients' ability to operate their mobile phone. Most patients were willing to receive SMS text messages and phone calls for TB medication and clinic appointment reminder.
- Most patients who received SMS text messages and phone calls had a positive attitude and willingness to adopt the intervention, not only for TB treatment care but also other healthcare services in the future.
- 7. Privacy and confidentiality issues were not a major concern in the implementation of the Ma-MAS intervention. Nevertheless, some patients were afraid of stigma and did not want to disclose their health status. This was particularly the case among TB-HIV co-infected patients.
8.3. Factors affecting non-adherence to TB medication

TB medication adherence is a complex and dynamic phenomenon because multiple factors influence it.^(6, 197) Non-adherence to TB medication and lost follow-up from TB treatment remain a public health problem in Ethiopia.^(11, 12) This research has identified multiple factors that have been found to affect non-adherence to TB medications in Ethiopia through literature review and qualitative interviews. In different geographical settings, these factors might have a different character or level of influence, but they overlap largely in developing countries.⁽¹⁹⁷⁾

In this study it was found that some of the influencing factors of non-adherence to TB medication reflect the findings of previous systematic reviews in Ethiopia.^(11, 12) These influencing factors for non-adherence to TB medication were forgetfulness, lack of awareness about the disease and its treatment, psychological distress, perceived barrier, long waiting time, drug side effects, pills burden, co-morbidity, being in the continuation phase of treatment and travel distance and associated transportation cost.

However, the current study identified several additional factors contributing to nonadherence to TB medication that had not been identified in the previous two systematic reviews. These factors were patient-centred factors such as perceived wellness, perceived risk and denial of the disease, negligence, and fatigue during the long treatment regime. The social factors were lack of social support, busyness with work, being away from home, perceived and experienced stigma and discrimination. The economic factors were inability to afford sufficient food and being homeless, work type such as daily labour and long-distance drivers. The health service factors were the DOT standard not being correctly implemented, health professional's low commitment to follow through patients' treatment, lack of counselling at the time of diagnosis of TB, poor patient-healthcare provider relationships such as communication gaps and disrespecting patients, quality of the healthcare service, patient (dis)-satisfaction, lack of incentives for healthcare providers to work over the weekend. The treatment related, and lifestyle factors were the persistence of symptoms after initiation of treatment, and the use of substances including the use of alcohol, smoking, and khat chewing.

The Mobile-assisted adherence support (Ma-MAS) intervention is designed considering these underlying factors of non-adherence to TB medication in the local setting. The Ma-MAS intervention includes key information about the disease, its treatment, misconceptions, drug side-effects, the link between non-adherence and treatment outcomes, and the link between substance use and treatment outcomes. The Ma-MAS intervention also includes motivational messages related to attitude and beliefs about the outcome of adherence or non-adherence for example the negative and positive effect of non-adherence and adherence on treatment outcomes. Behavioural skill information was also an element of the Ma-MAS intervention; designed to develop the self-efficacy to self-manage and to remember their TB medication, to manage the side-effects, and other problem-solving skills, such as what the patients can do when they away from home or unable to come to a health facility to collect their medications and other non-adherence challenges. Therefore, the mobile-assisted medication adherence support intervention was expected to bring positive improvement in adherence to TB medication because it was designed with locally available evidence to overcome factors of non-adherence to the prescribed TB medication. This intervention is particularly important in low resource settings like Ethiopia because there is no governmental or non-governmental organization sustainably implementing an adherence intervention to promote TB medication adherence. The Ma-MAS intervention may also be useful when the DOT standard program is interrupted or not correctly implemented such as occurred during the COVID-19 pandemic.

8.4. Effectiveness of Ma-MAS intervention

The experimental trial showed that Ma-MAS significantly improved adherence to TB medication after eight weeks of intervention. The Ma-MAS group participants had 15.25% higher TB medication adherence compared to the control group participants who received the standard care alone. After eight weeks of follow-up, the predicted probability of adherence to TB medication in the Ma-MAS group participants was higher (86%) compared to the control group participants (70%). Participants who received the mobile-assisted medication adherence support intervention had 1.2 times higher adherence to TB medication than those

who only received standard care alone. This improvement is a crucial finding, emphasizing the potential of Ma-MAS in enhancing adherence to TB medication. The positive effects of Ma-MAS were not limited to the eight-week, the experimental trial showed that Ma-MAS had similar effect after four weeks of intervention. The Ma-MAS group participants had 15.30% higher adherence to TB medication compared to participants who only received the standard care alone in the control group. After four weeks of follow-up, the predictive probability of adherence was higher in the Ma-MAS group participants (93%) than the control group participants (77%). The Ma-MAS intervention effect remained the same after sensitivity analysis which includes the missing and loss to follow up outcomes in the analysis (see Appendix Seven).

The experimental trial showed a notable decrease in adherence proportion for both the Ma-MAS and control groups participants, with reductions of 5.86% and 5.81%, respectively, from week four to eight. Despite these declines, the Ma-MAS intervention maintained its effectiveness in improving TB medication adherence. The decrease in the proportion of adherence observed in both the Ma-MAS and control groups from week four to eight might be due to the duration of treatment. This needs further study on the factors contributing to the observed decreases and the development of potential strategies to sustain the adherence over time.

Although the Ma-MAS intervention significantly improved TB medication adherence when measured by urine tests for isoniazid (INH) metabolites after four and eight weeks of intervention, Ma-MAS did not show statistically significant improvement on self-reported TB medication adherence; this needs further investigation. However, this might be because self-report adherence can be affected by various types of bias such as recall bias, social desirability bias, the way questions are asked, and the way the questionnaires are constructed, and thus might not be a reliable measurement for adherence to TB medication.⁽¹³⁸⁾

There is a conflict in the evidence regarding the effectiveness of SMS texts on self-reported TB medication adherence. Some SMS text message-based interventions did not show significant improvement on self-reported TB medication adherence. For example, in randomised control trials in Pakistan,⁽¹²¹⁾ Argentina⁽⁶⁹⁾, and Cameroon,⁽¹²²⁾ a daily SMS text message did not show significant improvement in self-reported TB medication adherence.

However, in India a quasi-experimental design found that daily SMS text message and weekly phone calls for 90-days significantly improved self-reported TB medication adherence,⁽¹⁹⁶⁾ and in China a randomised control trial, the combination of an audio reminded medication monitor and SMS text messages, showed a significant improvement in TB medication adherence.⁽¹²³⁾ Another randomised control trial in China found that a daily SMS text message caused a significant improvement, reducing missed doses of TB medication.⁽¹²⁵⁾ In Saudi Arabia a randomised control trial, a daily SMS text message during the first two months of treatment, significantly improved self-reported TB medication adherence.⁽¹⁹⁹⁾ In Indonesia, a quasi-experimental study indicated that SMS text messages can replace DOT supervision by a healthcare provider and be effective and efficient for TB medication adherence.^{(200).} Another Indonesian a quasi-experimental study found that a daily SMS text message significantly improved adherence to TB medication compared to the control group who only received the standard care.⁽²⁰¹⁾ In Ethiopia a randomised control trial, using text messages with a graphic based reminder, was found effective for self-reported TB medication adherence.⁽⁸⁵⁾

The Ma-MAS intervention was found useful to raise awareness for patients about the consequence of discontinuing TB medication and the necessary precautions while taking TB medication. As a result, some patients changed their intention to quit their TB medication after they discussed the possible consequences with a healthcare provider through phone calls. The phone calls allow the patient to ask questions related to TB disease and its treatment. Awareness about the seriousness of TB and discontinuing its treatment is important for adherence to TB medication.⁽⁸³⁾ In a study in Peru, SMS text messages were found useful for

raising awareness of the disease and its treatment.⁽⁶⁷⁾ A Sudan study also found that SMS text messages and phone calls were effective in improving knowledge about TB treatment.⁽²⁰²⁾ A study in India also found that phone calls were useful for patient communication about side-effects, symptoms of the disease and nutrition.⁽²⁰³⁾

The current study demonstrates that daily SMS text messages can remind patients to take their TB medication. The daily SMS text message helps to avoid forgetfulness due to being busy with work or other social activities. The SMS text messages reminded patients when they did not remember to take their medication. The same findings were reported in studies conducted in Uganda and Pakistan which reported patients believe that SMS text messages improve TB medication adherence because they are a good reminder about their medications.^(204, 205) The current study also found that the daily SMS text message was useful to embed the habit of taking TB medication at a consistent time.

The Ma-MAS intervention using SMS text messages and phone calls was also useful to motivate patients to properly adhere to their TB medication. The daily SMS text message was found to be useful to avoid negligence and ensure adherence to TB medication. The same finding was reported from Uganda, Mozambique and in Pakistan studies that reported adherence monitoring through SMS text messages enhance the patients' motivation and commitment to adherence to TB medication, which helps to develop the habit of taking TB medication at a consistent time.^(68, 132, 205)

The Ma-MAS trial also discovered that the SMS text message and phone call intervention also enhanced the patients' sense of receiving professional care. Patients were able to ask questions and obtain answers from a health professional through phone calls which enhance the patients' feeling of receiving professional care and satisfaction with care. Patients feeling cared for and having a good relationship with their healthcare provider should enhance the patient's commitment to adherence. This is consistent with a previous trial in Northwest Ethiopia, involving text messages with a graphic based reminder, that significantly improved

healthcare provider-patient relationship.⁽⁸⁵⁾ The same finding was also reported in Mozambique, Uganda and Pakistan studies which found SMS text messages enhance the patient-healthcare provider relationship; patients feeling cared for and supported by health professionals.^(68, 204, 205) A Canadian study also reported that SMS text messages for TB medication reminder were found to be useful and promoted a feeling of being cared for and having a good relationship with a healthcare provider.⁽²⁰⁶⁾ A Lesotho study also found that a mobile-based SMS text messages intervention enhanced patient-healthcare provider communication for HIV/TB treatment support.⁽²⁰⁷⁾

The Ma-MAS intervention (phone calls) was found to be useful for health consultations beyond adherence improvement. Patients were satisfied because they were able to consult health professionals regarding any health issues without going to a health facility. The central and northwest Ethiopia studies also showed that HIV Patients would like to receive health advice through mobile phone SMS text messages or phone calls. ^(71, 208)

8.5. Ma-MAS intervention design

The way the adherence intervention is designed and how the outcome (adherence to medication) is measured is important in randomised control trials. Ma-MAS showed promising results that could help to improve adherence to TB medication, but a large-scale test and more evidence are needed. Most previous adherence interventions were not designed based on a theoretical foundation, using local evidence or a behavioural change models and techniques, which might be the reason they are sometimes ineffective. However, notably, a trial Argentina that developed SMS text messages based on the IMB model, ⁽²⁰⁹⁾ and sent reminders twice a week for 8 weeks, did not show a significant improvement in TB medication adherence.⁽⁶⁹⁾ In the current study, the Ma-MAS intervention using validated SMS text messages sent daily for 8 weeks and weekly phone calls to reinforce the SMS text messages was found to be effective. The Ma-MAS intervention is designed to be broadly useful, beyond use as a simple reminder of TB medication adherence, building relationships, motivation, and self-efficacy.

Considering the pros and cons of SMS text messages or phone calls, the Ma-MAS intervention was designed to use both SMS text messages and phone calls for promoting TB medication adherence. Health professionals believed that adding a phone call intervention to the SMS text messages can make the adherence intervention more interactive and engaging compared to only using two-way SMS text messages, and it is also a good means of communication to get feedback from patients about their experience, side-effects and health status (see Appendix Two).

Sending SMS text messages or phone calls for TB medication reminders does not cause concern for most patients because they believe that their health professionals' intervention is for their benefit. However, most health professionals did not recommend making a phone call daily, due to the possibility of disturbing patients and the difficulty of reaching a large number of patients every day. Thus, in the Ma-MAS intervention a daily SMS text message and a weekly phone call were designed and implemented in the trial. However, after the trial was implemented, most patients wanted to increase the number of phone calls to at least twice a week because patients would have an opportunity to discuss any health issues encountered during the preceding week. For less educated people or those who struggle to adhere to their TB medication, a weekly phone call might not be adequate. But most participants believe that sending a daily SMS text message was appropriate. Most of the previous trials also used a daily SMS text message (see Table 3).

In this study, patients emphasized their preference for the time of sending SMS text messages and making phone calls to be aligned with their routine TB medication schedule. In pre-intervention assessment, both patients and health professionals believe that morning between 7:30am to 8:00am is appropriate time for sending SMS text messages and making phone calls for TB medication reminders. And, after the trial of Ma-MAS intervention, most patients stated the time of SMS text message delivery was convenient, even if it did not precisely align with their medication intake. Yet, some patients would like to receive SMS text messages and phone calls in the early morning, particularly at 7:00am and earlier. This

difference in preferences indicates the importance of scheduling SMS text messages and phone calls at the time of patient preference.

Most health professionals believe that an adherence intervention using SMS text messages and phone calls should be implemented for the whole TB treatment period. The reason for this was that the daily DOT (Directly Observed Treatment) was generally not implemented in the initial phase of TB treatment after the emergence of the COVID-19 pandemic, and patients became negligent because they feel well (relatively) and at times forget to take their medications or are lost from their treatment regimen during the continuation phase of treatment (see Appendix Two). The Ma-MAS intervention was implemented for eight weeks as an initial trial. After completion of the trial intervention, most patients stated they would like the Ma-MAS intervention implemented for the whole period of treatment. Several claimed this was because some patients may be reckless or have addictive behaviours, or they may not have family members support to remember their medication, and patients might not receive adequate on going advice about their treatment.

Participants in the Ma-MAS intervention believe that the content of the SMS text messages is appropriate and useful to promote TB medication adherence. However, some participants also recommended to avoid repetition of some SMS text messages. Most participants also accepted the information communicated through phone calls because it was designed in the form of open questions that support two-way communication. In the content of SMS text messages and phone calls, participants recommend adding content such as the cause and prevention of TB, TB diagnosis categories and their related symptoms, TB treatment and its side effects, food and nutrition requirements, substance use and mental health advice. Some participants also suggested designing the SMS text messages and phone calls in question-and-answer format. Designing a digital health intervention by considering the feedback of the targeted beneficiaries is highly recommended and should have a beneficial effect on TB medication adherence.⁽²¹⁰⁾ Thus, the suggested feedback of the patients' needs to be added in the future scale-up of a Ma-MAS intervention.

The potential privacy and confidentiality issues related to the SMS text messages and phone calls were not considered a major problem by patients. However, some patients were concerned about the stigma of TB treatment and did not want to disclose their health status to other people. As a result, they hid the SMS text messages and some patients did not pick up phone calls when they were in a public place. Privacy and confidentiality issues were especially of concern among TB-HIV co-infected patients.

A similar finding was reported in a Mozambique study which showed that TB and HIV patients had concerns about the privacy and confidentiality issues related with disclosure of their HIV status in SMS text messages.⁽⁶⁸⁾ In a Peru study the need for careful handling of patients information to avoid stigma related with the disease was also identified as an issue.⁽⁶⁷⁾ The qualitative findings of studies in Uganda and Pakistan also reflected patients concern about disclosure of their health status if other people see the SMS text messages.^(132, 205) The concern about privacy and confidentiality was high among TB-HIV co-infected patients.⁽²⁰⁵⁾ In central and northwest Ethiopia studies, privacy and confidentiality issues were a concern of some patients and related to SMS text messages for promoting antiretroviral therapy (ART) medication.^(71, 208)

Therefore, in an Ma-MAS intervention, privacy and confidentiality should be carefully managed, patients should have clear information about the intervention and provide informed-consent before the implementation of the intervention. Further protection mechanisms are also required to protect the privacy and confidentiality of the patient's information.

8.6. Feasibility and acceptability of Ma-MAS intervention

The Ma-MAS intervention was found to be technically feasible to implement because most patients have their own mobile phone or have access to a trusted family member's or caregiver's mobile phone; the proportion of patients with their own mobile phone was 88% and 62.5% of patients without personal mobile phones have access to a family member's or caregiver's mobile phone. On the other hand, a relatively small percentage (4.3%) of patients

did not have personal mobile phones and lacked access to a mobile phone through their family members or caregivers. This was consistent with the demography of Ethiopia and a health survey of 2019 which reports 68% of households in the country have their own mobile phone: 87% in urban and 59% in rural.⁽²⁸⁾ The current study also found that most patients were confident to operate a mobile phone to receive, read and send SMS text messages. Most patients were also confident to make a phone call (89.7%), please call me back (miss call) (82%) and set a reminder (79%). This indicates the growth in use of mobile phones and high digital literacy are an opportunity to use a mobile-based adherence intervention using SMS text messages and phone calls in Ethiopia.

The mobile adherence intervention was also well accepted by both patients and healthcare providers in this study. The cross-sectional survey indicates that most patients who have their own mobile phone were willing to receive an SMS text message reminder for medication intake (93%) and clinic appointments (92%). In addition, most patients were also willing to receive phone calls for medication intake (91%) and clinic appointment reminders (93%). These results indicate that patients with personal mobile phones were highly willing to receive SMS text messages and phone calls for medication intake and clinic appointment reminders. Most patients were also willing to receive SMS text message and phone call reminders with their caregiver's or family member's mobile phones, even when they did not own a mobile phone. These findings highlight a significant opportunity to promote TB medication adherence and optimize patient and family engagement in TB treatment. The same finding was shown in the interpretive study; most patients believe that SMS texts and phone calls would help them to remember their medications, and some patients also believe that SMS text messages and phone calls can avoid the unwanted consequences of non-adherence to TB medication. These supports the use of mobile based adherence interventions along with treatment administration options such as DOT, VOT, non-daily DOT for promoting medication adherence and treatment outcomes.⁽⁵⁾

The current research findings are consistent with previous studies conducted in Ethiopia among HIV patients, which found high willingness to receive SMS text messages or phone calls for ART medication reminder in central Ethiopia (91%) and northwest Ethiopia among HIV patients (96%).^(71, 208) However, the current research shows a higher willingness to receive SMS text messages or phone calls for clinic appointment reminder compared to the central Ethiopia (41%) and northwest Ethiopia (60%) studies. A Uganda study also reported that mobile-based SMS text messages and phone calls were found to be an acceptable intervention.⁽²¹¹⁾ A similar finding was reported in Peru and Indian studies which demonstrate that TB patients were willing to receive phone calls or SMS text messages for TB medication reminder and clinic appointments.^(67, 212) Therefore, in most geographical settings a mobile based adherence intervention to promote TB medication has high acceptance by patients and healthcare providers. For example, in an Indian study, 95% of TB patients wished to use mobile phone calls and SMS reminders for TB medication adherence instead of using DOTs.⁽²¹²⁾

Participants in the Ma-MAS intervention had a positive attitude towards the SMS text messages and phone calls though some participants preferred SMS text messages over phone calls, while others preferred phone calls over SMS text messages. This study found that participants had high willingness to adopt a mobile-based adherence intervention using SMS text messages or phone calls not only for TB treatment care but also for other healthcare services. Participants in the Ma-MAS intervention believe that the intervention had a positive effect on their TB medication adherence. The same findings were reported from two Pakistan studies which reported SMS text messages were found to be acceptable and useful.^(132, 204) In a Uganda study patients supported the usefulness of SMS text messages for TB medication intake and clinic appointment reminder.⁽²¹³⁾ In the Saudi Arabia randomised trial, after 8 weeks of daily SMS text message, TB patients were satisfied with the SMS text messages, which were perceived to be useful and they were willing to use them in the future.⁽¹⁹⁹⁾

8.7. Factors affecting adoption intention of Ma-MAS intervention

In the current study, it was found that performance expectancy and social influence can positively predict the acceptance of a mobile-assisted adherence intervention, according with the original theory of acceptance and use of technology model. This indicates that as patients perceive the benefits of the Ma-MAS intervention for adherence, the acceptability of the intervention will increase. And as patient's perceive that their social connections (families, friends, healthcare provider) believe the patient should use the Ma-MAS intervention for their TB medication reminder, the patient's acceptability of the intervention will increase. However, the effort expectancy prediction on behavioural intentions of the Ma-MAS intervention did not have a statistically significant finding. This is not consistent with the original unified theory of acceptance and use of technology.⁽¹⁷⁰⁾ This might be because all participants in the current study had 8 weeks of intervention experience and were digitally literate, which could be the reason the ease of use did not have a significant effect on the intention to use the Ma-MAS intervention.

This study also found that the social influence and attitude can positively predict acceptance of the mobile assisted adherence intervention in the modified unified theory of acceptance and use of technology. This means that when the patient perceived high social influence and a positive attitude towards the Ma-MAS intervention, they were more likely to accept the intervention. This study also found that performance expectancy, social influence and facilitating conditions positively predict the attitude of the participants towards the Ma-MAS intervention. Indeed, performance expectancy and facilitating conditions had a direct effect on attitude and an indirect effect on the acceptance of Ma-MAS intervention. However, the current study contrasts with the modified unified theory of acceptance and use of technology model which indicated performance expectancy and facilitating conditions have a direct effect on the acceptance of technology, and effort expectancy has a direct and indirect effect on attitude and acceptance.⁽¹⁷³⁾ This means the ease of engaging with the SMS text messages or phone calls did not significantly affect the acceptance of the Ma-MAS intervention which might be

due to all the participants of this study digitally literate and experienced. This needs further investigation in a larger sample.

8.8. Facilitators and challenges of Ma-MAS intervention

The high coverage of mobile phones, and the high acceptance of a mobile-assisted medication adherence support intervention by patients and health professionals are an opportunity to sustain the Ma-MAS intervention if it is implemented on a larger scale. This is consistent with the literature review which reported the acceptance and positive attitude of mHealth were driving facilitators to mHealth implementation.⁽²¹⁴⁾ However, the scale up of a mobile-assisted adherence intervention might be challenged by limited infrastructure such as the availability of telephones in the TB clinic, network interruptions, electricity interruptions, healthcare work load, and some patients may not have their own mobile phone or may not be digitally literate. According to the literature review, these challenges are also experienced in many low-resource settings, and include unreliable network coverage, an insufficient power supply, unstable internet services, a lack of policy to protect data privacy, a lack of funding, an inadequate healthcare workforce, and poor digital illiteracy.⁽²¹⁴⁾

To sustain a mobile-assisted medication adherence support intervention the following conditions are highly recommended: improving the infrastructure including telephone access at each TB clinic, increase network performance and mobile coverage, assure reliable electricity supply, address informed patient consent including privacy and confidentiality concerns, utilizing the patient's preferred time for SMS texts and phone calls, using a relative's mobile phone with mutual agreement when the patient does not have a mobile phone or cannot read and understand SMS text messages, checking the functionality of each patient's mobile phone during registration, an immediate substitution of mobile phones when the patient changes or loses their mobile phone, giving a unique health service name for the sender of SMS text messages and phone calls to increase the chance that the recipients will read and answer calls, and drafting the messages in the patient's preferred language.

In summary, the Ma-MAS intervention, based on the content of information, motivation, and behavioural skills combined with behavioural change techniques, can address major factors of TB medication non-adherence and therefore is capable of improving TB medication adherence. The Ma-MAS intervention was found to be useful for TB medication reminder, habit formation, awareness creation, patient motivation, perceived professional care and health consultation. Thus, the current research supports the use of the information, motivation, and behavioural skill model ^(160, 215) combined with a behavioural change technique taxonomy ⁽¹⁵⁸⁾ based on local evidence and cultural context for promoting TB medication adherence.

8.9. Strengths and limitations of the study

This study has several strengths, including the use of a behavioural change model and a behavioural change technique to develop an adherence intervention using SMS text messages and phone calls, based on local evidence about non-adherence to TB medication and cultural context, including utilizing local language. Unlike previous adherence interventions, this trial combined SMS text messages with phone calls, which resulted promising outcomes. Developing the intervention systematically and measuring the outcome correctly are crucial for the validity of the randomised control trial. As a result, the intervention was systematically developed and validated in the local context through a collaborative effort involving senior experts, health professionals, and patients, responding to the needs of the target audience. The primary desired outcome of medication adherence was assessed using a reliable measurement (IsoScreen kit) to detect NIH metabolites in urine, which is a strength of the study. The Ma-MAS intervention was evaluated using interpretive and empirical studies, which is also a strength of the study.

However, the study also has some limitations. The IsoScreen test kit only showed shortterm medication adherence of 72 hours, which is not indicative of long-term adherence. This was the reason that the IsoScreen kit test was conducted at two points; at the fourth and eighth

weeks of intervention. Although self-reporting can imply long-term adherence and is inexpensive, it lacks reliability and can be affected by recall bias and social desirability bias. As result, a discrepancy was observed in the proportion of adherence measured by the IsoScreen test kit and self-reported adherence. This study was conducted only for eight weeks, so the long-term effect of Ma-MAS on adherence and treatment outcomes is unproven and needs further research.

Although the Ma-MAS intervention was expected to improve the attendance at clinic appointments, the assessment of clinic appointment attendance was, in part disrupted by the COVID-19 pandemic, not successful, and needs further research. The Ma-MAS intervention might be affected by repeated use of some SMS text messages during the eight weeks of the intervention. The SMS text messages and phone calls were not scheduled to match with the patient's real-time medication intake or clinic appointment, which could affect the outcome of the Ma-MAS intervention. The Ma-MAS intervention could not address all non-adherence barriers, including economic factors, travel distances, stigma, discrimination, and other moderating factors that require additional interventions. Some participants without personal mobile phones have enrolled in the randomised trial using their family members' or caregivers' mobile phone during the scheduled daily messages and weekly phone calls, and the family members or caregivers may forget to communicate the messages to the patients. This could potentially affect the outcome of the Ma-MAS intervention.

The English version questionnaires, SMS text messages, and phone call checklist guide were translated into Amharic, the Ethiopian national language, by a professional language translator and were checked by the principal investigator. However, the translated questionnaires, SMS text messages, and phone call checklist guide were not retranslated into English to check for consistency.

The sample size for structural equation modelling might not be adequate to examine the construct path relationship in the unified theory of acceptance and use of technology, which needs further investigation. It is important to note that interpretive studies do not necessarily indicate statistical significance, and further hypothesis testing with empirical studies is needed. Overall, while this study has several strengths, it also has some limitations that need to be addressed in future research.

CHAPTER 9: CONCLUSIONS AND RECOMMENDATIONS

9.1. Introduction

This chapter encompasses the conclusions and practical recommendations derived from the studies. The conclusions and recommendations are based on the results and discussions of previous chapters. The recommendations are provided across various sections, including implications for policymakers and program managers, implications for health education and communication, implications for health science education and implications for future research.

9.2. Conclusions

The mobile-assisted medication adherence support intervention using SMS text messages and phone calls is a feasible and acceptable intervention that can improve TB medication adherence, especially if it is designed in the local context and supported by an appropriate behavioural change model and technique. This research has shown that a mobile health intervention, using SMS text messages and phone calls, has the potential to improve TB medication adherence.

The effectiveness of the Ma-MAS intervention on TB medication adherence can be explained by the various ways it has been shown to be useful. First, the Ma-MAS intervention serves as a reminder for patients to take their TB medication. This is important as forgetting to take medication is a common reason for non-adherence to TB medication. Second, the intervention promotes habit formation by helping patients establish a routine for taking their medication at a consistent time. This can be particularly helpful for patients who are struggling to adhere to their medication regimen. Third, the Ma-MAS intervention raises awareness about TB and its treatment. This can help patients understand the importance of taking their TB medication as prescribed and the consequences of non-adherence. Fourth, the intervention motivates patients to adhere to TB medication by providing support and feedback. This can help patients feel more engaged and committed to their treatment plan. Fifth, the intervention improves perceived professional care, as patients feel more supported and cared for, which

can help patients stay on their medication regimen. Finally, the Ma-MAS intervention facilitates access to health consultation, which can help patients to discuss any health issues without going to a health facility. However, the success of this intervention depends on how well it is designed and implemented. It is important to tailor the intervention to the local context and the needs of the patients. Additionally, an appropriate behavioural change model and technique should be incorporated to maximize the effectiveness of the intervention.

The current study showed that patients who received SMS text messages and phone calls had a positive attitude and willingness to adopt the intervention for their TB treatment care, as well as for other healthcare services in the future. This suggests that the use of SMS text messages and phone calls could be an effective way to engage patients in their healthcare and improve their adherence to medication. Additionally, the current study found that some patients preferred to receive phone calls rather than SMS text messages, while others preferred SMS text messages over phone calls to remind them of their TB medication. It is important for healthcare providers to take into account these individual patient preferences when implementing Ma-MAS intervention for promoting TB medication adherence.

Although Ma-MAS was found to be effective in improving medication adherence as measured by IsoScreen urine test of INH metabolites, it did not show a statistically significant improvement in self-reported TB medication adherence. This suggests that further investigation is needed to better understand the reasons for this discrepancy and to identify strategies for improving self-reported adherence.

Privacy and confidentiality are crucial to the safety and well-being of patients, and protecting their privacy is a key component of creating a supportive and safe environment. Although privacy and confidentiality were not major concerns expressed by patents during the implementation of the Ma-MAS intervention, some patients, particularly those with TB and HIV, may still be hesitant to disclose their health status due to stigma. In addition to obtaining

informed consent from patients, protection mechanisms are required to safeguard the privacy and confidentiality of the patient's information.

The high coverage of mobile phones and the acceptance of the mobile-assisted medication adherence support (Ma-MAS) intervention among patients and healthcare professionals underpin the opportunity to sustain the intervention on a larger scale. However, the scale-up of a mobile-assisted adherence support intervention can be challenging due to limited health infrastructure, such as the availability of telephones in the TB clinics, communications network and interruptions to electricity supply, healthcare workload, and the relative digital literacy of patients. Some patients may not have access to their own mobile phones or a network, this is particularly challenging in rural settings with limited mobile phone coverage. Addressing this issue may require additional support to effectively improve TB medication adherence.

Non-adherence to TB medication is a complex issue influenced by various factors, including forgetfulness, inadequate knowledge, psychological distress, drug side effects, co-morbidity, healthcare inaccessibility and transportation cost. The current study identified additional factors such as economic constraints, living away from home, poor healthcare provider-patient relationships, healthcare quality, lack of social support, perceived and experienced stigma and discrimination, beliefs towards the disease and treatment, and substance use that also play a role in non-adherence to TB medication. Healthcare managers, providers, and researchers need to consider the underlying factors when designing adherence interventions and may need to implement additional interventions such as social support, financial assistance, improving communication and patient-provider relationships, ensuring adequate support and counselling at diagnosis and addressing the implementation of DOT standards. The Ma-MAS intervention, although designed based on underlying factors, did not tackle all non-adherence factors such as economic constraints, stigma, and discrimination, that underpin the need for extra interventions.

In conclusion, this study suggests compelling evidence supporting the effectiveness of mobile-assisted intervention approaches to medication adherence. This rigorous combination of randomised control trial and interpretive study provides robust outcomes. The results demonstrate the potential use of mobile-assisted intervention in improving medication adherence, particularly in settings where adherence to medication is vital for positive health outcomes but challenging to achieve using conventional approaches. Based on these findings, the implementation and testing of mobile-assisted intervention in other settings, diseases, and medication adherence contexts are highly recommended.

9.3. Recommendations

9.3.1. Implications for policy makers and program managers

The mobile-assisted medication adherence support intervention has shown a promising result, improving TB medication adherence, which can be scaled up to a larger population and different local languages, and for the entire period of TB treatment. Policy makers and program managers should prioritize the development and validation of a locally tailored adherence intervention that addresses the specific needs of the target audience and utilizing the MRC framework.

Sending daily SMS text messages was useful for promoting TB medication adherence, but the frequency of phone calls should be increased from once a week to at least twice a week. It is recommended to implement the Ma-MAS intervention throughout the entire treatment period to obtain the maximum beneficial effect on TB medication adherence and treatment outcomes. To maximize the effectiveness of the intervention, it is important to schedule SMS text messages and phone calls at the patient's preferred time. In situations where a patient does not have their own mobile phone it may be necessary to utilize a relative's mobile phone with mutual agreement. This should be taken into consideration in any efforts to scale up a Ma-MAS intervention. Additionally, during patient registration, it is essential to check the functionality of each patient's mobile phone and ensure immediate substitution in case of any change or loss. The content of the SMS text messages and phone calls was found useful and acceptable, but the contents of SMS text messages and phone calls may be supplemented, based on the feedback of patients. Drafting messages in the patient's preferred language and giving a unique health service name for the sender of SMS text messages and phone calls may increase the likelihood that recipients will read and answer the phone calls.

Healthcare providers must ensure that privacy and confidentiality of their patients are protected by health policy and procedures, especially for those who may be vulnerable and experience stigma and discrimination. Providing training to healthcare providers on how to support and respect patient privacy will further contribute to the development of a trusting relationship with patients.

The mobile-assisted medication adherence support intervention does not require a smartphone or internet connection, making it highly scalable in low resource settings. However, to sustain the intervention, it is essential to strengthen infrastructure, such as telephone access in TB clinics and other resources. In addition, the SMS text message and phone call intervention can be implemented separately or together with other digital technologies, such as electronic pill boxes or smartphone applications, to further improve TB medication adherence. By combining these technologies and strengthening infrastructure, especially in countries with poorly resourced health systems, it may be possible to sustain and improve the effectiveness of the medication adherence support intervention, ultimately leading to better treatment outcomes.

Moreover, when it comes to sending SMS text messages, choosing the right cloud platform is crucial for ensuring uninterrupted delivery. The selected cloud platform must have the capacity to handle large volumes of SMS traffic and provide reliable service. In addition, a monitoring mechanism is essential to detect any issues that may arise during the delivery process. The monitoring mechanism should be able to identify any failed deliveries and provide insights into the reasons behind those failures. This will enable the sender to take

appropriate action and ensure that all messages are delivered successfully. It is also important to consider factors such as security, scalability, and cost-effectiveness when selecting an SMS text message cloud platform. The platform should have robust security measures in place to protect the privacy and confidentiality of the messages being sent. It should also be able to scale up or down depending on the volume of traffic being handled and be cost-effective for the intended usage.

9.3.2. Implications for health education and communication

The research has demonstrated that healthcare providers can easily communicate with patients and provide them with necessary information and guidance regarding their TB treatment through SMS text messages and phone calls. This can lead to improved patient engagement and adherence to their treatment regimen. Additionally, patients feel they received professional care, leading to increased trust and satisfaction with their care. However, effective communication through phone calls or SMS text messages requires healthcare providers to have good communication skills. Therefore, providing training to healthcare providers in communication skills is essential to ensure that they can effectively communicate with patients and address their concerns and questions. This training may include techniques for active listening, empathy, and clear communication. By improving their communication skills, healthcare providers can build stronger relationships with patients, resulting in improved patient satisfaction, adherence to treatment, and ultimately, better treatment outcomes.

Patient education through SMS text messages and phone calls to promote TB medication adherence is highly recommended. This is because the current study demonstrated that a behavioural intervention using SMS text messages and phone calls is effective in increasing patients' awareness of the consequences of discontinuing TB medication and the precautions they need to take while taking the medication. By using SMS text messages and phone calls, healthcare providers can easily reach out to patients and provide them with the necessary

information and support they need to adhere to their TB medication regimen. This approach is particularly important in the context of TB treatment, as the medication regimen is long and complex, and patients may need ongoing encouragement and education.

In addition to providing information about TB medication adherence using SMS text messages and phone calls, this approach can also help to identify and address any barriers or challenges patients may be facing related to TB treatment. For example, if a patient is experiencing side effects, unable to come to a health facility for medication refill, or travel to other places, healthcare providers can work with them to find solutions that work best for their individual needs.

9.3.3. Implications for health science education

The use of the MRC (Medical Research Council) framework for developing and evaluating complex interventions in a randomised controlled trial (RCT) is considered to be a rigorous and systematic approach to research. Therefore, the current research, which follows MRC framework, can be an exemplar and used in educational material for students, researchers, and practitioners.

The current research can serve as a useful reference for those who want to understand the process of developing and evaluating a complex intervention. The MRC framework used in this research provides a clear and structured pathway for researchers to develop and evaluate an intervention and can lead to better outcomes. By understanding the steps involved in this framework, researchers can learn to develop an intervention and are more likely to succeed.

The research can also be used as a reference for future research. By documenting the intervention development process, the evaluation, and the outcome, the research provides valuable insights for future researchers. The findings can be used to inform the development of new interventions, to identify areas for further research, and to guide the design of future RCTs.

Moreover, the research can be used for educational purposes to teach students about the MRC framework, intervention development, and the importance of RCTs. Students in health-related fields can learn about the systematic approach to intervention development and evaluation, and how to apply this knowledge in their own research.

9.3.4. Implications for future research

The Ma-MAS intervention has shown promising results in a small-scale study. However, the long-term efficacy of this intervention on TB medication adherence and treatment outcomes in large geographic settings is not yet fully understood and requires further investigation.

In order to effectively scale up the Ma-MAS intervention and implement it in a large population, it is important to develop and validate additional content for SMS text messages and phone calls based on the behavioural change technique and model. This will allow for more personalized and targeted SMS text messages that consider individual patient characteristics and needs in different sub-populations or language groups.

This study was conducted among TB patients who took their medication for six months. Therefore, the findings of this study may not be generalizable to patients with multi-drug resistant TB (MDR-TB) or TB patients who take their medication for more than six months, and further research is needed to study the effectiveness of the intervention on TB medication adherence for these populations.

Furthermore, while the current research has focused on the acceptability and effectiveness of the Ma-MAS intervention, it is also important to assess the economic value of this intervention. This will help to determine the cost-effectiveness of the intervention and whether it is a feasible and sustainable solution for improving TB medication adherence in resourcelimited settings.

BIBLIOGRAPHY

1. Bloom BR, Atun R, Cohen T, Dye C, Fraser H, Gomez GB, et al. Tuberculosis. In: Holmes KK, Bertozzi S, Bloom BR, Jha P, editors. Major Infectious Diseases. Washington (DC): The International Bank for Reconstruction and Development / The World Bank; 2017.

2. Agbeko CK, Mallah MA, He B, Liu Q, Song H, Wang J. Mental Health Status and Its Impact on TB Treatment and Its Outcomes: A Scoping Literature Review. Front Public Health. 2022;10:855515.

3. Addo J, Pearce D, Metcalf M, Lundquist C, Thomas G, Barros-Aguirre D, et al. Living with tuberculosis: a qualitative study of patients' experiences with disease and treatment. BMC Public Health. 2022;22(1):1717.

4. WHO. Global tuberculosis report 2022. Geneva: World Health Organization. 2022.

5. WHO. Guidelines for treatment of drug-susceptible tuberculosis and patient care. World Health Organization. 2017.

6. Nezenega ZS, Perimal-Lewis L, Maeder AJ. Factors Influencing Patient Adherence to Tuberculosis Treatment in Ethiopia: A Literature Review. International Journal of Environmental Research Public Health. 2020;17(15):5626.

7. Yew WW. Directly observed therapy, short-course: the best way to prevent multidrug-resistant tuberculosis. Chemotherapy. 1999;45(Suppl. 2):26-33.

8. Pablos-Méndez A, Knirsch CA, R Graham Barr M, Lerner BH, Frieden TR. Nonadherence in tuberculosis treatment: predictors and consequences in New York City. The American journal of medicine. 1997;102(2):164-70.

9. Barclay E. Text messages could hasten tuberculosis drug compliance. The Lancet. 2009;373(9657):15-6.

10. Gandhi NR, Nunn P, Dheda K, Schaaf HS, Zignol M, Van Soolingen D, et al. Multidrugresistant and extensively drug-resistant tuberculosis: a threat to global control of tuberculosis. The Lancet. 2010;375(9728):1830-43.

11. Zegeye A, Dessie G, Wagnew F, Gebrie A, Islam SMS, Tesfaye B, et al. Prevalence and determinants of anti-tuberculosis treatment non-adherence in Ethiopia: A systematic review and meta-analysis. PloS one. 2019;14(1):e0210422.

12. Tola HH, Holakouie-Naieni K, Tesfaye E, Mansournia MA, Yaseri M. Prevalence of tuberculosis treatment non-adherence in Ethiopia: a systematic review and meta-analysis. International journal of tuberculosis and lung disease. 2019;23(6):741-9.

13. Tola HH, Garmaroudi G, Shojaeizadeh D, Tol A, Yekaninejad MS, Ejeta LT, et al. The Effect of Psychosocial Factors and Patients' Perception of Tuberculosis Treatment Non-Adherence in Addis Ababa, Ethiopia. Ethiop J Health Sci. 2017;27(5):447-58.

14. WHO. An expanded DOTS framework for effective tuberculosis control. . Int J Tuberc Lung Dis. 2002;6:378-88.

15. McKay B, Castellanos M, Ebell M, Whalen CC, Handel A. An attempt to reproduce a previous meta-analysis and a new analysis regarding the impact of directly observed therapy on tuberculosis treatment outcomes. PloS one. 2019;14(5):e0217219.

16. Karumbi J, Garner P. Directly observed therapy for treating tuberculosis. Cochrane Database of Systematic Reviews. 2015(5).

17. Fiseha D, Demissie M. Assessment of Directly Observed Therapy (DOT) following tuberculosis regimen change in Addis Ababa, Ethiopia: a qualitative study. BMC Infect Dis. 2015;15:405.

18. Sagbakken M, Frich JC, Bjune G. Barriers and enablers in the management of tuberculosis treatment in Addis Ababa, Ethiopia: a qualitative study. BMC Public Health. 2008;8:11.

19. Adane AA, Alene KA, Koye DN, Zeleke BM. Non-adherence to anti-tuberculosis treatment and determinant factors among patients with tuberculosis in northwest Ethiopia. PLoS One. 2013;8(11):e78791.

20. Mohammed H, Oljira L, Roba KT, Yimer G, Fekadu A, Manyazewal T. Containment of COVID-19 in Ethiopia and implications for tuberculosis care and research. Infectious Diseases of Poverty. 2020;9(1):1-8.

21. Alipanah N, Jarlsberg L, Miller C, Linh NN, Falzon D, Jaramillo E, et al. Adherence interventions and outcomes of tuberculosis treatment: A systematic review and meta-analysis of trials and observational studies. PLoS Med. 2018;15(7):e1002595.

22. Liu Q, Abba K, Alejandria MM, Sinclair D, Balanag VM, Lansang MAD. Reminder systems to improve patient adherence to tuberculosis clinic appointments for diagnosis and treatment. Cochrane Database of Systematic Reviews. 2014(11).

23. WHO. Handbook for the use of digital technologies to support tuberculosis medication adherence. World Health Organization; 2017.

24. WHO. Digital health for the End TB strategy: progress since 2015 and future perspectives: meeting report. World Health Organization; 2017.

25. Belknap R, Holland D, Feng PJ, Millet JP, Cayla JA, Martinson NA, et al. Selfadministered versus directly observed once-weekly isoniazid and rifapentine treatment of latent tuberculosis infection. Annals of Internal Medicine. 2017;167(10):689-97.

26. Hüsler J. Evaluation of the On Cue Compliance Service Pilot: Testing the use of SMS reminders in the treatment of tuberculosis in Cape Town, South Africa. 2005.

27. Lester R, Jay J, Bolten LM, Enjeti A, Johnston JC, Schwartzman K, et al. Mobile Phone Short Message Service for Adherence Support and Care of Patients with Tuberculosis Infection: Evidence and Opportunity. Journal of Clinical Tuberculosis and Other Mycobacterial Diseases. 2019:100108.

28. Ethiopian Public Health Institute (EPHI). Ethiopia Mini Demographic and Health Survey 2019: Final Report. 2021.

29. Central Statistical Agency. Ethiopia Demographic and Health Sruvey 2011. Addis Ababa, Ethiopia; 2012.

30. Aranda-Jan CB, Mohutsiwa-Dibe N, Loukanova S. Systematic review on what works, what does not work and why of implementation of mobile health (mHealth) projects in Africa. BMC Public Health. 2014;14(1):188.

31. Barkman C, Weinehall L. Policymakers and mHealth: roles and expectations, with observations from Ethiopia, Ghana and Sweden. Global Health Action. 2017;10(3):1337356.

32. Davey S, Davey A, Singh JV. Mobile-health approach: a critical look on its capacity to augment health system of developing countries. Indian Journal of Community Medicine 2014;39(3):178.

33. Goel S, Bhatnagar N, Sharma D, Singh A. Bridging the human resource gap in primary health care delivery systems of developing countries with mhealth: narrative literature review. JMIR mHealth uHealth. 2013;1(2):e2688.

34. Tegegne T, Weide T. Is mHealth Viable to Ethiopia: an Empirical Study. [Reserach Report-In Proc. M4D 2012 28–29]. In press 2012.

35. Subbaraman R, de Mondesert L, Musiimenta A, Pai M, Mayer KH, Thomas BE, et al. Digital adherence technologies for the management of tuberculosis therapy: mapping the landscape and research priorities. BMJ Glob Health. 2018;3(5):e001018.

36. Heron KE, Smyth JM. Ecological momentary interventions: incorporating mobile technology into psychosocial and health behaviour treatments. British Journal of Health Psychology. 2010;15(1):1-39.

37. Fjeldsoe BS, Marshall AL, Miller YD. Behavior change interventions delivered by mobile telephone short-message service. American Journal of Preventive Medicine. 2009;36(2):165-73.

38. Scott-Sheldon LAJ, Lantini R, Jennings EG, Thind H, Rosen RK, Salmoirago-Blotcher E, et al. Text Messaging-Based Interventions for Smoking Cessation: A Systematic Review and Meta-Analysis. JMIR mHealth uHealth 2016;4(2).

39. Free C, Hoile E, Robertson S, Knight R. Three controlled trials of interventions to increase recruitment to a randomized controlled trial of mobile phone based smoking cessation support. Clinical Trials. 2010;7(3):265-73.

40. Liao Y, Wu Q, Tang J, Zhang F, Wang X, Qi C, et al. The efficacy of mobile phonebased text message interventions ('Happy Quit') for smoking cessation in China. BMC Public Health. 2016;16(1):833.

41. Riley W, Augustson EM. Mobile phone-based smoking cessation interventions increase long-term quit rates compared with control programmes, but effects of the interventions are heterogeneous. Evidence-based nursing. 2013;16(4):108-9.

42. Whittaker R, McRobbie H, Bullen C, Rodgers A, Gu Y. Mobile phone-based interventions for smoking cessation. Cochrane Database of Systematic Reviews. 2016(4).

43. Ybarra ML, Jiang Y, Free C, Abroms LC, Whittaker R. Participant-level meta-analysis of mobile phone-based interventions for smoking cessation across different countries. Preventive medicine. 2016;89:90-7.

44. Smith C, Gold J, Ngo TD, Sumpter C, Free C. Mobile phone-based interventions for improving contraception use (Review). Cochrane Database of Systematic Reviews 2015(6).

45. Lee SH, Nurmatov UB, Nwaru BI, Mukherjee M, Grant L, Pagliari C. Effectiveness of mHealth interventions for maternal, newborn and child health in low and middle–income countries: Systematic review and meta–analysis. Journal of Global Health. 2016;6(1).

46. Coleman J, Bohlin KC, Thorson A, Black V, Mechael P, Mangxaba J, et al. Effectiveness of an SMS-based maternal mHealth intervention to improve clinical outcomes of HIV-positive pregnant women. AIDS Care. 2017;29(7):890-7.

47. Atnafu A, Otto K, Herbst CH. The role of mHealth intervention on maternal and child health service delivery: findings from a randomized controlled field trial in rural Ethiopia. Mhealth. 2017;3.

48. Kebede AS, Ajayi IO, Arowojolu AO. Effect of enhanced reminders on postnatal clinic attendance in Addis Ababa, Ethiopia: a cluster randomized controlled trial. Glob Health Action. 2019;12(1):1609297.

49. Shiferaw S, Spigt M, Tekie M, Abdullah M, Fantahun M, Dinant G-J. The effects of a locally developed mHealth intervention on delivery and postnatal care utilization; a prospective controlled evaluation among health centres in Ethiopia. PLoS One. 2016;11(7):e0158600.

50. Lund S, Hemed M, Nielsen B, Said A, Said K, Makungu M, et al. Mobile phones as a health communication tool to improve skilled attendance at delivery in Zanzibar: a cluster-randomised controlled trial. BJOG: An International Journal of Obstetrics & Gynaecology. 2012;119:1256–1264.

51. Ngabo F, Nguimfack J, Nwaigwe F, Mugeni C, Muhoza D, Wilson DR, et al. Designing and Implementing an Innovative SMS-based alert system (RapidSMS-MCH) to monitor pregnancy and reduce maternal and child deaths in Rwanda. Pan African Medical Journal. 2012;13(31).

52. Omole O, Ijadunola MY, Olotu E, Omotoso O, Bello B, Awoniran O, et al. The effect of mobile phone short message service on maternal health in south-west Nigeria. Int J Health Plann Manage. 2017.

53. Lund S, Rasch V, Hemed M, Boas IM, Said A, Said K, et al. Mobile Phone Intervention Reduces Perinatal Mortality in Zanzibar: Secondary Outcomes of a Cluster Randomized Controlled Trial. JMIR Mhealth Uhealth. 2014;2(1).

54. De Jongh T, Gurol-Urganci I, Vodopivec-Jamsek V, Car J, Atun R. Mobile phonemessaging for facilitating self-management of long-termillnesses (Review). Cochrane Database of Systematic Reviews 2012(12).

55. Klimis H, Khan ME, Kok C, Chow CK. The Role of Text Messaging in Cardiovascular Risk Factor Optimization. Curent cardiology report. 2017;19:4.

56. Dobson R, Carter K, Cutfield R, Hulme A, Hulme R, McNamara C, et al. Diabetes Text-Message Self-Management Support Program (SMS4BG):A Pilot Study. JMIR mHealth uHealth 2015;3(1).

57. Abbas BB, Fares AA, Jabbari M, Dali AE, Orifi FA. Effect of Mobile Phone Short Text Messages on Glycemic Control in Type 2 Diabetes. Int J Endocrinol Metab. 2015;13(1).

58. Bobrow K, Farmer AJ, Springer D, Shanyinde M, Yu L-M, Brennan T, et al. Mobile Phone Text Messages to Support Treatment Adherence in Adults With High Blood Pressure (StAR): A Single-Blind, Randomized Trial. PMC. 2016;133(6): 592–600. 59. Zurovac D, Sudoi RK, Akhwale WS, Ndiritu M, Hamer DH, Rowe AK, et al. The eff ect of mobile phone text-message reminders on Kenyan health workers' adherence to malaria treatment guidelines: a cluster randomised trial. Lancet 2011;378: 795–803.

60. Sarabi RE, Sadoughi F, Orak RJ, Bahaadinbeigy K. The Effectiveness of Mobile Phone Text Messaging in Improving Medication Adherence for Patients with Chronic Diseases: A Systematic Review. Iran Red Crescent Med J. 2016; 18(5):e25183.

61. Tao D, Xie L, Wang T, Wang T. A meta-analysis of the use of electronic reminders for patient adherence to medication in chronic disease care. Journal of Telemedicine and Telecare. 2015;Vol. 21(1) 3–13.

62. Adler AJ, Martin N, Mariani J, Tajer CD, Owolabi OO, Free C, et al. Mobile phone textmessaging to improve edication adherence in secondary prevention of cardiovascular disease (Review). Cochrane Database of Systematic Reviews 2017(4).

63. Endehabtu B, Weldeab A, Were M, Lester R, Worku A, Tilahun B. Mobile Phone Access and Willingness Among Mothers to Receive a Text-Based mHealth Intervention to Improve Prenatal Care in Northwest Ethiopia: Cross-Sectional Study. JMIR Pediatrics Parenting. 2018;1(2):e9.

64. Hussien FN, Tilahun BC, Gelaye KA, Mamuye AL. Development of automated textmessage reminder system to improve uptake of childhood vaccination in North-West, Ethiopia. Online Journal of Public Health Informatics. 2019;11(2).

65. Little A, Medhanyie A, Yebyo H, Spigt M, Dinant G-J, Blanco R. Meeting community health worker needs for maternal health care service delivery using appropriate mobile technologies in Ethiopia. PLoS One. 2013;8(10):e77563.

66. Ahmed MAA, Gagnon M-P, Hamelin-Brabant L, Mbemba GIC, Alami H. A mixed methods systematic review of success factors of mhealth and telehealth for maternal health in Sub-Saharan Africa. Mhealth. 2017;3.

67. Albino S, Tabb KM, Requena D, Egoavil M, Pineros-Leano MF, Zunt JR, et al. Perceptions and acceptability of short message services technology to improve treatment adherence amongst tuberculosis patients in Peru: a Focus Group Study. PLoS One. 2014;9(5):e95770.

68. Nhavoto JA, Gronlund A, Klein GO. Mobile health treatment support intervention for HIV and tuberculosis in Mozambique: Perspectives of patients and healthcare workers. PLoS One. 2017;12(4):e0176051.

69. Iribarren S, Beck S, Pearce PF, Chirico C, Etchevarria M, Cardinale D, et al. TextTB: A Mixed Method Pilot Study Evaluating Acceptance, Feasibility, and Exploring Initial Efficacy of a Text Messaging Intervention to Support TB Treatment Adherence. Tuberc Res Treat. 2013;2013:349394.

70. Kim J, Zhang W, Nyonyitono M, Lourenco L, Nanfuka M, Okoboi S, et al. Feasibility and acceptability of mobile phone short message service as a support for patients receiving antiretroviral therapy in rural Uganda: a cross-sectional study. Journal of the International AIDS Society 2015;18:20311.

71. Endebu T, Deksisa A, Dugasa W, Mulu E, Bogale T. Acceptability and feasibility of short message service to improve ART medication adherence among people living with HIV/AIDS receiving antiretroviral treatment at Adama hospital medical college, Central Ethiopia. BMC Public Health. 2019;19(1):1315.

72. Dowshen N, Kuhns LM, Johnson A, Holoyda BJ, Garofalo R. Improving adherence to antiretroviral therapy for youth living with HIV/AIDS: a pilot study using personalized, interactive, daily text message reminders. J Med Internet Res. 2012;14(2):e51.

73. Georgette N, Siedner MJ, Zanoni B, Sibaya T, Petty CR, Carpenter S, et al. The Acceptability and Perceived Usefulness of a Weekly Clinical SMS Program to Promote HIV Antiretroviral Medication Adherence in KwaZulu-Natal, South Africa. AIDS Behav. 2016;20(11):2629-38.

74. Musiimenta A, Atukunda EC, Tumuhimbise W, Pisarski EE, Tam M, Wyatt MA, et al. Acceptability and Feasibility of Real-Time Antiretroviral Therapy Adherence Interventions in Rural Uganda: Mixed-Method Pilot Randomized Controlled Trial. JMIR Mhealth Uhealth. 2018;6(5):e122.

75. Spark LC, Fjeldsoe BS, Eakin EG, Reeves MM. Efficacy of a Text Message-Delivered Extended Contact Intervention on Maintenance of Weight Loss, Physical Activity, and Dietary Behavior Change. JMIR mHealth uHealth 2015;3(3).

76. Craig P, Dieppe P, Macintyre S, Michie S, Nazareth I, Petticrew M, et al. Developing and evaluating complex interventions: the new Medical Research Council guidance. BMJ. 2008;337:a1655.

77. Lakshman R, Griffin S, Hardeman W, Schiff A, Kinmonth AL, Ong KK. Using the Medical Research Council framework for the development and evaluation of complex interventions in a theory-based infant feeding intervention to prevent childhood obesity: the baby milk intervention and trial. J Obes. 2014;2014:646504.

78. Horne R, Weinman J, Barber N, Elliott R, Morgan M, Cribb A, et al. Concordance, adherence and compliance in medicine taking. London: NCCSDO. 2005;2005:40-6.

79. Campbell NC, Murray E, Darbyshire J, Emery J, Farmer A, Griffiths F, et al. Designing and evaluating complex interventions to improve health care. BMJ. 2007;334(7591):455-9.

80. Alegria-Flores K, Weiner BJ, Wiesen CA, Lich KLH, Van Rie A, Paul JE, et al. Innovative approach to the design and evaluation of treatment adherence interventions for drug-resistant TB. International journal of tuberculosis and lung disease. 2017;21(11):1160-6.

81. Nglazi MD, Bekker L-G, Wood R, Hussey GD, Wiysonge CS. Mobile phone text messaging for promoting adherence to anti-tuberculosis treatment: a systematic review. BMC infectious diseases. 2013;13(1):566.

82. Chen S, Pan Y, Qu Y, Liu Z, Yao L, Xiang L. Mobile phone based interventions for promoting adherence to tuberculosis treatment: a systematic review and meta-analysis. The Lancet. 2015;386:S29.

83. Tola HH, Shojaeizadeh D, Tol A, Garmaroudi G, Yekaninejad MS, Kebede A, et al. Psychological and educational intervention to improve tuberculosis treatment adherence in Ethiopia based on health belief model: a cluster randomized control trial. PLoS One. 2016;11(5).

84. Gashu KD, Gelaye KA, Lester R, Tilahun B. Combined effect of pill refilling and selfmedication reminder system on patients' adherence to tuberculosis treatment during continuation phase in Northwest Ethiopia: a study protocol for randomised controlled trial. BMJ Health Care Inform. 2019;26(1).

85. Gashu KD, Gelaye KA, Lester R, Tilahun B. Effect of a phone reminder system on patient-centered tuberculosis treatment adherence among adults in Northwest Ethiopia: a randomised controlled trial. BMJ Health Care Informatics. 2021;28(1).

86. Guerra RL, Conde MB, Efron A, Loredo C, Bastos G, Chaisson RE, et al. Point-of-care Arkansas method for measuring adherence to treatment with isoniazid. Respiratory medicine. 2010;104(5):754-7.

87. Nicolau I, Tian L, Menzies D, Ostiguy G, Pai M. Point-of-care urine tests for smoking status and isoniazid treatment monitoring in adult patients. PLoS One. 2012;7(9):e45913.

88. Mkopi A, Range N, Lwilla F, Egwaga S, Schulze A, Geubbels E, et al. Validation of indirect tuberculosis treatment adherence measures in a resource-constrained setting. International journal of tuberculosis and lung disease. 2014;18(7):804-9.

89. Soobratty MR, Whitfield R, Subramaniam K, Grove G, Carver A, O'Donovan GV, et al. Point-of-care urine test for assessing adherence to isoniazid treatment for tuberculosis. Eur Respir J. 2014;43(5):1519-22.

90. (MOH) MoH. Health Sector Transformation Plan 2015/16-2019/20 Addis Ababa, Ethiopia; 2015.

91. Ministry of Health (MOH). Ethiopian National eHealth Strategy. Addis Ababa, Ethiopia; 2014.

92. Ogundele OA, Moodley D, Seebregts CJ, Pillay AW, editors. An ontology for tuberculosis treatment adherence behaviour. Proceedings of the 2015 Annual Research Conference on South African Institute of Computer Scientists and Information Technologists; 2015: ACM.

93. Sahile Z, Yared A, Kaba M. Patients' experiences and perceptions on associates of TB treatment adherence: a qualitative study on DOTS service in public health centers in Addis Ababa, Ethiopia. BMC Public Health. 2018;18(1):462.

94. Mekonnen HS, Azagew AW. Non-adherence to anti-tuberculosis treatment, reasons and associated factors among TB patients attending at Gondar town health centers, Northwest Ethiopia. BMC Res Notes. 2018;11(1):691.

95. Ayele AA, Asrade Atnafie S, Balcha DD, Weredekal AT, Woldegiorgis BA, Wotte MM, et al. Self-reported adherence and associated factors to isoniazid preventive therapy for latent tuberculosis among people living with HIV/AIDS at health centers in Gondar town, North West Ethiopia. Patient preference and adherence. 2017;11:743-9.

96. Tesfahuneygn G, Medhin G, Legesse M. Adherence to Anti-tuberculosis treatment and treatment outcomes among tuberculosis patients in Alamata District, northeast Ethiopia. BMC Res Notes. 2015;8:503.

97. Yasin Mohammed A, Kaso Adem M. Treatment Adherence among Tuberculosis and Human Immuno Deficiency Virus Coinfected Patients in Ginnir Referral Hospital. American Journal of Public Health Research. 2014;2(6):239-43.

98. Mindachew M, Deribew A, Memiah P, Biadgilign S. Perceived barriers to the implementation of Isoniazid preventive therapy for people living with HIV in resource constrained settings: a qualitative study. Pan Afr Med J. 2014;17:26.

99. Eticha T, Kassa E. Non-adherence to anti-TB drugs and its predictors among TB/HIV co-infected patients in Mekelle, Ethiopia. Journal of Bioanalysis & Biomedicine 2014;6:061-4.

100. Kebede A, Wabe NT. Medication adherence and its determinants among patients on concomitant tuberculosis and antiretroviral therapy in South west ethiopia. North American journal of medical sciences. 2012;4(2):67-71.

101. Woimo TT, Yimer WK, Bati T, Gesesew HA. The prevalence and factors associated for anti-tuberculosis treatment non-adherence among pulmonary tuberculosis patients in public health care facilities in South Ethiopia: a cross-sectional study. BMC Public Health. 2017;17(1):269.

102. Kiros YK, Teklu T, Desalegn F, Tesfay M, Klinkenberg E, Mulugeta A. Adherence to anti-tuberculosis treatment in Tigray, Northern Ethiopia. Public Health Action. 2014;4(3):31-6.

103. Tekle B, Mariam DH, Ali A. Defaulting from DOTS and its determinants in three districts of Arsi Zone in Ethiopia. International journal of tuberculosis and lung disease 2002;6(7):573-9.

104. Michael KW, Belachew T, Jira C. Tuberculosis defaulters from the "dots" regimen in Jimma zone, southwest Ethiopia. Ethiopian medical journal. 2004;42(4):247-53.

105. Diriba Daksa M, Melaku Kebede T, H Mariam DA. Patients' adherence to antituberculosis medicines and associated factors for non-adherence at a tertiary teaching hospital, South West Ethiopia. European Journal of Therapeutics. 2016;22(2):55-62.

106. Tesfaye M, Adorjan K, Krahl W, Tesfaye E, Yitayih Y, Strobl R, et al. Khat and Alcohol Use Disorders Predict Poorer Adherence to Anti-Tuberculosis Medications in Southwest Ethiopia: A Prospective Cohort Study, (preprint article). 2019

107. Tola HH, Karimi M, Yekaninejad MS. Effects of sociodemographic characteristics and patients' health beliefs on tuberculosis treatment adherence in Ethiopia: a structural equation modelling approach. Infect Dis Poverty. 2017;6(1):167.

108. Tadesse T, Demissie M, Berhane Y, Kebede Y, Abebe M. Long distance travelling and financial burdens discourage tuberculosis DOTs treatment initiation and compliance in Ethiopia: a qualitative study. BMC Public Health. 2013;13(1):424.

109. Gebremariam MK, Bjune GA, Frich JC. Barriers and facilitators of adherence to TB treatment in patients on concomitant TB and HIV treatment: a qualitative study. BMC Public Health. 2010;10:651.

110. Getahun B, Nkosi ZZ. Is directly observed tuberculosis treatment strategy patientcentered? A mixed method study in Addis Ababa, Ethiopia. PLoS One. 2017;12(8):e0181205.

111. Berhe M, Demissie M, Tesfaye GJAie. Isoniazid preventive therapy adherence and associated factors among HIV positive patients in Addis Ababa, Ethiopia. Advances in Epidemilogy. 2014;2014.

112. Mindachew M, Deribew A, Tessema F, Biadgilign S. Predictors of adherence to isoniazid preventive therapy among HIV positive adults in Addis Ababa, Ethiopia. BMC Public Health. 2011;11:916.

113. Gugssa Boru C, Shimels T, Bilal AI. Factors contributing to non-adherence with treatment among TB patients in Sodo Woreda, Gurage Zone, Southern Ethiopia: A qualitative study. J Infect Public Health. 2017;10(5):527-33.

114. Ayele HT, van Mourik MS, Bonten MJ. Predictors of adherence to isoniazid preventive therapy in people living with HIV in Ethiopia. International journal of tuberculosis and lung disease. 2016;20(10):1342-7.

115. Mesfin MM, Newell JN, Walley JD, Gessessew A, Tesfaye T, Lemma F, et al. Quality of tuberculosis care and its association with patient adherence to treatment in eight Ethiopian districts. Health Policy Plan. 2009;24(6):457-66.

116. Nezenega ZS, Gacho YH, Tafere TE. Patient satisfaction on tuberculosis treatment service and adherence to treatment in public health facilities of Sidama zone, South Ethiopia. BMC Health Serv Res. 2013;13:110.

117. Gube AA, Debalkie M, Seid K, Bisete K, Mengesha A, Zeynu A, et al. Assessment of Anti-TB Drug Nonadherence and Associated Factors among TB Patients Attending TB Clinics in Arba Minch Governmental Health Institutions, Southern Ethiopia. Tuberc Res Treat. 2018;2018:3705812.

118. Shargie EB, Lindtjørn B. Determinants of treatment adherence among smear-positive pulmonary tuberculosis patients in Southern Ethiopia. PLoS medicine. 2007;4(2):e37.

119. Osterberg L, Blaschke TJN. Adherence to medication. England Journal of Medicine. 2005;353(5):487-97.

120. WHO. Digital Health for the end TB strategy: An agenda for Action. World Health Organization. 2015.

121. Mohammed S, Glennerster R, Khan AJ. Impact of a Daily SMS Medication Reminder System on Tuberculosis Treatment Outcomes: A Randomized Controlled Trial. PLoS One. 2016;11(11):e0162944.

122. Bediang G, Stoll B, Elia N, Abena JL, Geissbuhler A. SMS reminders to improve adherence and cure of tuberculosis patients in Cameroon (TB-SMS Cameroon): a randomised controlled trial. BMC Public Health. 2018;18(1):583.

123. Liu X, Lewis JJ, Zhang H, Lu W, Zhang S, Zheng G, et al. Effectiveness of Electronic Reminders to Improve Medication Adherence in Tuberculosis Patients: A Cluster-Randomised Trial. PLoS Med. 2015;12(9):e1001876.

124. Broomhead S, Mars M. Retrospective return on investment analysis of an electronic treatment adherence device piloted in the Northern Cape Province. Telemedicine and e-Health. 2012;18(1):24-31.

125. Fang XH, Guan SY, Tang L, Tao FB, Zou Z, Wang JX, et al. Effect of short message service on management of pulmonary tuberculosis patients in Anhui Province, China: A prospective, randomized, controlled study. Medical Science Monitor. 2017;23:2465-9.

126. Kunawararak P, Pongpanich S, Chantawong S, Pokaew P, Traisathit P, Srithanaviboonchai K, et al. Tuberculosis treatment with mobile-phone medication reminders in northern Thailand. Southeast Asian Journal of Tropical Medicine and Public Health. 2011;42(6):1444.

127. Gashu KD, Gelaye KA, Mekonnen ZA, Lester R, Tilahun BJBID. Does phone messaging improves tuberculosis treatment success? A systematic review and meta-analysis. BMC Infectious Diseases 2020;20(1):42.

128. Wald DS, Butt S, Bestwick JP. One-way versus two-way text messaging on improving medication adherence: meta-analysis of randomized trials. The American journal of medicine. 2015;128(10):1139. e1-. e5.

129. Mills EJ, Lester R, Thorlund K, Lorenzi M, Muldoon K, Kanters S, et al. Interventions to promote adherence to antiretroviral therapy in Africa: a network meta-analysis. The Lancet HIV. 2014;1(3):e104-e11.

130. Kanters S, Park JJ, Chan K, Socias ME, Ford N, Forrest JI, et al. Interventions to improve adherence to antiretroviral therapy: a systematic review and network meta-analysis. The lancet HIV. 2017;4(1):e31-e40.

131. Bediang G, Stoll B, Elia N, Abena J-L, Nolna D, Chastonay P, et al. SMS reminders to improve the tuberculosis cure rate in developing countries (TB-SMS Cameroon): a protocol of a randomised control study. Trials. 2014;15(1):35.

132. Mohammed S, Siddiqi O, Ali O, Habib A, Haqqi F, Kausar M, et al. User engagement with and attitudes towards an interactive SMS reminder system for patients with tuberculosis. Journal of telemedicine and telecare. 2012;18(7):404-8.

133. Hoffman JA, Cunningham JR, Suleh AJ, Sundsmo A, Dekker D, Vago F, et al. Mobile direct observation treatment for tuberculosis patients: a technical feasibility pilot using mobile phones in Nairobi, Kenya. American Journal of Preventive Medicine. 2010;39(1):78-80.

134. Farooqi RJ, Ashraf S, Zaman M. The role of mobile SMS-reminders in improving drugs compliance in patients receiving anti-TB treatment from DOTS program. Journal of Postgraduate Medical Institute. 2017;31(2).

135. de Sumari-de Boer IM, van den Boogaard J, Ngowi KM, Semvua HH, Kiwango KW, Aarnoutse RE, et al. Feasibility of real time medication monitoring among HIV infected and TB patients in a resource-limited setting. AIDS Behav. 2016;20(5):1097-107.

136. Howren MB. Adherence. In: Gellman MD, Turner JR, editors. Encyclopedia of Behavioral Medicine. New York, NY: Springer New York; 2013. p. 33-9.

137. WHO. Adherence to long-term therapies: evidence for action. World Health Organization. 2003.

138. Valencia S, Leon M, Losada I, Sequera VG, Fernandez Quevedo M, Garcia-Basteiro AL. How do we measure adherence to anti-tuberculosis treatment? Expert review of anti-infective therapy. 2017;15(2):157-65.

139. Morisky DE, Ang A, Krousel-Wood M, Ward HJ. Predictive validity of a medication adherence measure in an outpatient setting. Journal of Clinical Hypertension. 2008;10(5):348-54.

140. Kalichman SC, Amaral CM, Swetzes C, Jones M, Macy R, Kalichman MO, et al. A simple single-item rating scale to measure medication adherence: further evidence for convergent validity. Journal of the International Association of Physicians in AIDS Care. 2009;8(6):367-74.

141. Thompson K, Kulkarni J, Sergejew A. Reliability and validity of a new Medication Adherence Rating Scale (MARS) for the psychoses. Schizophrenia research. 2000;42(3):241-7.

142. Svarstad BL, Chewning BA, Sleath BL, Claesson C. The Brief Medication Questionnaire: a tool for screening patient adherence and barriers to adherence. Patient Education Counseling. 1999;37(2):113-24.

143. Chesney MA, Ickovics J, Chambers D, Gifford A, Neidig J, Zwickl B, et al. Self-reported adherence to antiretroviral medications among participants in HIV clinical trials: the AACTG adherence instruments. AIDS Care. 2000;12(3):255-66.

144. Boogaard Jvd, Lyimo RA, Boeree MJ, Kibiki GS, Aarnoutse RE. Electronic monitoring of treatment adherence and validation of alternative adherence measures in tuberculosis patients: a pilot study. Bulletin of the World Health Organization. 2011;89:632-9.

145. Schraufnagel DE, Stoner R, Whiting E, Snukst-Torbeck G, Werhane MJ. Testing for Isoniazid: An Evaluation of the Arkansas Method. Chest. 1990;98(2):314-6.

146. Whitfield R, Cope GF. Point-of-care test to monitor adherence to anti-tuberculous treatment. Annals of clinical biochemistry. 2004;41(5):411-3.

147. Mkopi A, Range N, Lwilla F, Egwaga S, Schulze A, Geubbels E, et al. Adherence to tuberculosis therapy among patients receiving home-based directly observed treatment: evidence from the United Republic of Tanzania. PloS one. 2012;7(12):e51828.

148. Farmer KC. Methods for measuring and monitoring medication regimen adherence in clinical trials and clinical practice. Clinical therapeutics. 1999;21(6):1074-90.

149. Finitsis DJ, Pellowski JA, Huedo-Medina TB, Fox MC, Kalichman SC. Visual analogue scale (VAS) measurement of antiretroviral adherence in people living with HIV (PLWH): a meta-analysis. Journal of Behavioral Medicine. 2016;39(6):1043-55.

150. Hathcoat JD, Meixner C, Nicholas MC. Ontology and epistemology. In Pranee Liamputtong (ed), Handbook of Research Methods in Health Social Sciences, Springer Singapore. 2019:99-116.

151. Rawnsley MMJNsq. Ontology, epistemology, and methodology: A clarification. 1998;11(1):2-4.

152. Creswell JW, Creswell JD. Research design: Qualitative, quantitative, and mixed methods approaches: Sage publications; 2017.

153. Yvonne Feilzer MJJommr. Doing mixed methods research pragmatically: Implications for the rediscovery of pragmatism as a research paradigm. 2010;4(1):6-16.

154. Shannon-Baker PJJommr. Making paradigms meaningful in mixed methods research. 2016;10(4):319-34.

155. Sahile Z, Perimal-Lewis L, Arbon P, Maeder AJ. Protocol of a parallel group Randomized Control Trial (RCT) for Mobile-assisted Medication Adherence Support (Ma-MAS) intervention among Tuberculosis patients. PLoS One. 2021;16(12):e0261758.

156. Federal Democratic Republic of Ethiopia Central Statistical Agency. Population Projection of Ethiopia for All Regions At Wereda Level from 2014 – 2017. Addis Ababa, Ethiopia; 2013.

157. Ministry of Health of Ethiopia (MOH). Tuberculosis, Leprosy and TB/HIV Prevention and Control Programme Manual. 4th ed. Addis Ababa, Ethiopia2008.

158. Michie S, Richardson M, Johnston M, Abraham C, Francis J, Hardeman W, et al. The behavior change technique taxonomy (v1) of 93 hierarchically clustered techniques: building an international consensus for the reporting of behavior change interventions. Annals of behavioral medicine. 2013;46(1):81-95.

159. Campbell M, Fitzpatrick R, Haines A, Kinmonth AL, Sandercock P, Spiegelhalter D, et al. Framework for design and evaluation of complex interventions to improve health. BMJ. 2000;321(7262):694-6.

160. Fisher JD, Fisher WA, Amico KR, Harman J. An information-motivation-behavioral skills model of adherence to antiretroviral therapy. Health psychology. 2006;25(4):462.

161. Bleijenberg N, Janneke M, Trappenburg JC, Ettema RG, Sino CG, Heim N, et al. Increasing value and reducing waste by optimizing the development of complex interventions: Enriching the development phase of the Medical Research Council (MRC) Framework. International Journal of Nursing studies. 2018;79:86-93.

162. Vasileiou K, Barnett J, Thorpe S, Young T. Characterising and justifying sample size sufficiency in interview-based studies: systematic analysis of qualitative health research over a 15-year period. BMC Medical Research Methodology. 2018;18(1):148.

163. van Rijnsoever FJ. (I can't get no) saturation: a simulation and guidelines for sample sizes in qualitative research. PLoS One. 2017;12(7).

164. Marshall B, Cardon P, Poddar A, Fontenot R. Does sample size matter in qualitative research?: A review of qualitative interviews in IS research. Journal of computer information systems. 2013;54(1):11-22.

165. Hsu C-C, Sandford BA. The Delphi technique: making sense of consensus. Practical assessment, research, and evaluation. 2007;12(1):10.

166. Polit DF, Beck CT. The content validity index: are you sure you know what's being reported? Critique and recommendations. Research in Nursing and Health. 2006;29(5):489-97.

167. Bobrow K, Farmer A, Cishe N, Nwagi N, Namane M, Brennan TP, et al. Using the Medical Research Council framework for development and evaluation of complex interventions in a low resource setting to develop a theory-based treatment support intervention delivered via SMS text message to improve blood pressure control. BMC Health Services Research. 2018;18(1):1-15.

168. Ministry of Health (MOH). National Guidelines for TB, DR-TB and Leprosy in Ethiopia. Sixth edition ed2018.
169. Taherdoost H. A review of technology acceptance and adoption models and theories. Procedia Manufacturing. 2018;22:960-7.

170. Venkatesh V, Morris MG, Davis GB, Davis FD. User acceptance of information technology: Toward a unified view. MIS quarterly. 2003:425-78.

171. Venkatesh V, Sykes TA, Zhang X. 'Just what the doctor ordered': a revised UTAUT for EMR system adoption and use by doctors. in 2011 44th Hawaii International Conference on System Sciences. 2011:1-10.

172. Wills MJ, El-Gayar OF, Bennett D. Examining healthcare professionals' acceptance of electronic medical records using UTAUT. Issues in Information Systems. 2008;9(2):396-401.

173. Dwivedi YK, Rana NP, Jeyaraj A, Clement M, Williams MD. Re-examining the unified theory of acceptance and use of technology (UTAUT): Towards a revised theoretical model. Information Systems Frontiers. 2019;21(3):719-34.

174. Davis FD, Bagozzi RP, Warshaw PR. User acceptance of computer technology: a comparison of two theoretical models. Management science. 1989;35(8):982-1003.

175. Ajzen I, Fishbein M. Attitude-behavior relations: A theoretical analysis and review of empirical research. Psychological bulletin. 1977;84(5):888.

176. Taylor S, Todd P. Assessing IT usage: The role of prior experience. MIS quarterly. 1995:561-70.

177. Taylor S, Todd PA. Understanding information technology usage: A test of competing models. Information Systems Research. 1995;6(2):144-76.

178. Fishbein M, Ajzen I. Belief, attitude, intention, andbehavior: An introduction to theory and research. Addison-Wesley 1975.

179. Zhang X, Zaman Bu. Adoption mechanism of telemedicine in underdeveloped country. Health Informatics Journal. 2019:1460458219868353.

180. Davis FD. Perceived usefulness, perceived ease of use, and user acceptance of information technology. MIS quarterly. 1989:319-40.

181. Dwivedi YK, Shareef MA, Simintiras AC, Lal B, Weerakkody V. A generalised adoption model for services: A cross-country comparison of mobile health (m-health). Government Information Quarterly. 2016;33(1):174-87.

182. Braun V, Clarke V. Thematic analysis: American Psychological Association; 2012.

183. Raskind IG, Shelton RC, Comeau DL, Cooper HL, Griffith DM, Kegler MC. A review of qualitative data analysis practices in health education and health behavior research. Health Education & Behavior. 2019;46(1):32-9.

184. Hair EC, Anderson Moore K, Hadley AM, Kaye K, Day RD, Orthner DK. Parent marital quality and the parent–adolescent relationship: Effects on adolescent and young adult health outcomes. Marriage and Family Review. 2009;45(2-3):218-48.

185. Ayaz A, Yanartaş M. An analysis on the unified theory of acceptance and use of technology theory (UTAUT): Acceptance of electronic document management system (EDMS). Computers in Human Behavior Reports. 2020;2:100032.

186. Hair FJ, Black, W. C., Babin, B. J., and Anderson, R. E. Multivariate Data Analysis: A Global Perspective,. 8th edition ed. New York, NY: MacMillan 2019.

187. Pallant J. SPSS survival manual: A step by step guide to data analysis using IBM SPSS: Routledge; 2020.

188. Hair Jr JF, Hult GTM, Ringle CM, Sarstedt M. A primer on partial least squares structural equation modeling (PLS-SEM): Sage publications; 2021.

189. Hu Lt, Bentler PM. Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. Structural equation modeling: a multidisciplinary journal. 1999;6(1):1-55.

190. Philippi P, Baumeister H, Apolinário-Hagen J, Ebert DD, Hennemann S, Kott L, et al. Acceptance towards digital health interventions–Model validation and further development of the Unified Theory of Acceptance and Use of Technology. Internet interventions. 2021;26:100459.

191. Connelly LM. Trustworthiness in qualitative research. Medsurg Nursing. 2016;25(6):435.

192. Polit DF, Beck CT. Essentials of nursing research: appraising evidence for nursing practice 9th ed. Philadelphia: Wolters Kluwer; 2018.

193. Lincoln YS, Guba EG. Naturalistic inquiry: sage; 1985.

194. Guba EG, Lincoln YS. Competing paradigms in qualitative research. Handbook of qualitative research. 1994;2(163-194):105.

195. Lincoln YS, Guba EG. But is it rigorous? Trustworthiness and authenticity in naturalistic evaluation. New directions for program evaluation. 1986;1986(30):73-84.

196. Schulz KF, Altman DG, Moher D, the CG. CONSORT 2010 Statement: updated guidelines for reporting parallel group randomised trials. BMC Medicine. 2010;8(1):18.

197. Munro SA, Lewin SA, Smith HJ, Engel ME, Fretheim A, Volmink J. Patient adherence to tuberculosis treatment: a systematic review of qualitative research. PLoS Med. 2007;4(7):e238.

198. Santra S, Garg S, Basu S, Sharma N, Singh MM, Khanna A. The effect of a mhealth intervention on anti-tuberculosis medication adherence in Delhi, India: A quasi-experimental study. Indian Journal of Public Health. 2021;65(1):34.

199. Alotaibi NN. Effect of Using Mobile Phone Messaging Reminders in Improving Adherence to Treatment of Pulmonary Tuberculosis Patients in Jeddah, During 2016-2017: A Randomized Control Study. Journal of Public Health and Environmental Pollution 2019;3(1).

200. Kumboyono. Short message service as an alternative in the drug consumption evaluation of persons with tuberculosis in Malang, Indonesia. Japan Journal of Nursing Science. 2017;14(2):112-6.

201. Dewi FST, Sudiya S, Supriyati S, Purwanta P, Madyaningrum E, Aulia FU, et al. Preparing short message service reminders to improve treatment adherence among tuberculosis patients in Sleman District, Indonesia. Indian Journal of Community Medicine. 2019;44(2):81.

202. Ali AOA, Prins MH. Mobile health to improve adherence to tuberculosis treatment in Khartoum state, Sudan. Journal of Public Health in Africa. 2019;10(2).

203. Elangovan R, Arulchelvan S. A study on the role of mobile phone communication in tuberculosis DOTS treatment. Indian journal of community medicine. 2013;38(4):229.

204. Farooqi RJ, Ahmed H, Ashraf S, Zaman M, Farooq S, Farooqi JI. Feasibility and acceptability of Mobile SMS reminders as a Strategy to improve drugs adherence in TB Patietns. Pakistan Journal of Chest Medicine. 2017;23(3):93-100.

205. Musiimenta A, Tumuhimbise W, Mugaba AT, Muzoora C, Armstrong-Hough M, Bangsberg D, et al. Digital monitoring technologies could enhance tuberculosis medication adherence in Uganda: Mixed methods study. Journal of clinical tuberculosis other mycobacterial diseases. 2019;17:100119.

206. van der Kop ML, Memetovic J, Smillie K, Coleman J, Hajek J, Natasha Van Borek M, et al. Use of the WelTel mobile health intervention at a tuberculosis clinic in British Columbia: a pilot study. Journal of Mobile Technology in Medicine. 2013;2(3):7-14.

207. Hirsch-Moverman Y, Daftary A, Yuengling KA, Saito S, Ntoane M, Frederix K, et al. Using mHealth for HIV/TB treatment support in Lesotho: enhancing patient–provider communication in the START study. Journal of acquired immune deficiency syndromes. 2017;74(Suppl 1):S37.

208. Kebede M, Zeleke A, Asemahagn M, Fritz F. Willingness to receive text message medication reminders among patients on antiretroviral treatment in North West Ethiopia: A cross-sectional study. BMC medical informatics decision making. 2015;15(1):1-10.

209. Iribarren SJ, Beck SL, Pearce PF, Chirico C, Etchevarria M, Rubinstein F. mHealth intervention development to support patients with active tuberculosis. Journal of mobile technology in medicine. 2014;3(2):16.

210. Ridho A, Alfian SD, van Boven JF, Levita J, Yalcin EA, Le L, et al. Digital Health Technologies to Improve Medication Adherence and Treatment Outcomes in Patients with Tuberculosis: Systematic Review of Randomized Controlled Trials. Journal of medical Internet research. 2022;24(2):e33062.

211. Ggita JM, Ojok C, Meyer AJ, Farr K, Shete PB, Ochom E, et al. Patterns of usage and preferences of users for tuberculosis-related text messages and voice calls in Uganda. International journal of tuberculosis and lung disease. 2018;22(5):530-6.

212. Jose NK, Vaz C, Chai PR, Rodrigues R. The Acceptability of Adherence Support via Mobile Phones for Antituberculosis Treatment in South India: Exploratory Study. JMIR Formative Research. 2022;6(5):e37124.

213. Hermans SM, Elbireer S, Tibakabikoba H, Hoefman BJ, Manabe YC. Text messaging to decrease tuberculosis treatment attrition in TB-HIV coinfection in Uganda. Patient preference and adherence. 2017;11:1479-87.

214. Osei E, Mashamba-Thompson TP. Mobile health applications for disease screening and treatment support in low-and middle-income countries: A narrative review. Heliyon. 2021;7(3):e06639.

215. Fisher WA, Fisher JD, Harman J. The information-motivation-behavioral skills model: A general social psychological approach to understanding and promoting health behavior. Social psychological foundations of health illness. 2003;82:106.

APPENDIX ONE: FACTORS INFLUENCING PATIENT ADHERENCE TO TB MEDICATION IN ETHIOPIA

Table 29 : Summary of reviewed articles showing factors influencing non-adherence or default or loss to follow up from TB treatment

First Author (year)	Region	Study Design	Sample size	Population	Associated factors of non-adherence to TB medication and loss to follow up/default from TB treatment					
Tesfaye et al. (2019) ⁽¹⁰⁶⁾	Oromia	Prospective cohort	268	Active TB Patients	Khat and alcohol use were associated with non-adherence with TB medication. Educational and occupational status were also associated with non-adherence to TB medication.					
Sahile et al.(2018) (93)	Addis Ababa	A qualitative study	10	Active TB Patients	Forgetfulness was the factor of TB medication non-adherence. perceived risk and perceived wellness were also associated with medication non-adherence.					
Mekonnen et al. (2018) ⁽⁹⁴⁾	Amhara	Cross- sectional study	314	Active TB patients	Being in the continuation phase of treatment, the presence of more than one co-morbidity, inadequate knowledge of TB and its treatment, poor-patient provider relationships, and alcohol use were statistically associated with non-adherence to TB treatment. Additionally, forgetting to take medication, being busy with work, and being out of home/town were the reasons mentioned by patients for non-adherence.					
Gube et al. (2018) ⁽¹¹⁷⁾	SNNPR	Cross- sectional study	271	Active TB Patients	Drug side effects, a far distance from the health facility, and prolonged waited time to get healthcare service were statistically associated with non-adherence.					
Woimo et al. (2017) ⁽¹⁰¹⁾	SNNPR	Cross- sectional study	261	Active Pulmonary TB patients	Inadequate knowledge of TB and its treatment, health information at each medication refill visits, a far distance from a health facility (more than 10 km), transportation cost, and cost of medications other than anti-TB.					

Tola et al. (2017) (107)	Addis Ababa	Cross- sectional study	698	Active TB patients	Perceived barrier over and perceived benefits directly associated with non- adherence TB medication. Cue to action and psychological distress were indirectly influence non-adherence mediated through perceived barriers and benefits.
Tola et al. (2017)	Addis Ababa	Cross- sectional study	698	Active TB patients	ART status, economic status, alcohol use, perceived barrier, and psychological distress were statistically associated with non-adherence to TB medication.
Gugsa et al. (2017) ⁽¹¹³⁾	SNNPR	A qualitative study	22	Active TB patients	Lack of adequate food, poor communication between healthcare providers and patients, beliefs in traditional healing, unavailability of the healthcare service nearby, drug side-effect and pill burden, stigma, and discrimination.
Getahun et al. 2017 ⁽¹¹⁰⁾	Addis Ababa	Mixed method design	649	Active TB patients & healthcare provider	Among lost to follow-up patients, inadequate information about TB, healthcare provider-patient relationship (respect and value to patients), lack of support such as transport and nutrition support, less commitment of healthcare provider to towards with the patients were reported as reasons for lost to follow-up.
Ayele et al. (2017) ⁽⁹⁵⁾	Amhara	Cross- sectional study	154	Latent TB-HIV patients	Lack of information about isoniazid preventive therapy (IPT), drug side effect, forgetfulness, and being away from home was found factors for non-adherence to isoniazid preventive therapy (IPT).
Diriba et al. (2016) (105)	Oromia	Cross- sectional study	67	Active TB patients	Lack of family support, a far distance from the health facility, drug side-effect, extremely ill, feeling better, education level, and being HIV positive were the factors that lead to non-adherence to TB medication.
Ayele et al. (2016) ⁽¹¹⁴⁾	SNNPR	Prospective cohort	162	Latent TB-HIV patients	Patients experiencing a high level of HIV stigma and having opportunistic infections statistically associated with non-adherence to isoniazid preventive therapy (IPT).

Tesfahuneygn et al. 2015 ⁽⁹⁶⁾	Amhara	Cross- sectional study	200	Active TB patients	Forgetfulness, being away from home, drug side effects, being unable to go to the health facilities on the date of appointment and being hospitalized. TB-HIV infected, alcohol use, smoking, khat chewing, dissatisfied with healthcare service, long waiting time to get the health service significantly associated.
Yasin et al. (2015) (97)	Oromia	Cross- sectional study	53	Active TB-HIV co-infected	Forgetfulness, a far distance from the health facility, and low income were the factors for non-adherence to TB medication.
Mindachew et al. (2014) ⁽⁹⁸⁾	Addis Ababa	A qualitative study	12	Latent TB-HIV patients	Forgetfulness, lack of patient information, knowledge about the disease and its treatment, mental health status makes them reluctant to attend follow-up and clinic appointment, not disclosing their HIV status to their family members because of fear of stigma and discrimination and lack of support from family lead them to non-adhere to isoniazid preventive therapy (IPT).
Kiros et al. (2014) ⁽¹⁰²⁾	Tigray	Cross- sectional study	278	Active TB patients	The drug-side effect, and knowledge associated non-adherence to TB medication.
Eticha et al. (2014) ⁽⁹⁹⁾	Tigray	Cross- sectional study	120	Active TB-HIV co-infected	Patients who don't have caregivers and people who don't have to remind them to take their medications more likely non-adhere to TB medications. Forgetfulness, feeling sick, and being far away from health facilities were the mention of the main reason by the TB-HIV co-infected patients for missing medication.
Berhe et al. (2014) ⁽¹¹¹⁾	Addis Ababa	Cross- sectional study	381	Latent TB-HIV patients	Patients who took their medication for ≥ 5 or months high likely to be adherent than those who took for 1-2 months. Patient's friend decision to take the medication less likely non-adherent to isoniazid preventive therapy (IPT).

Nezenega et al. (2011) (116)	SNNPR	Cross- sectional study	531	Active TB patients	Employment status, area of residence, perceived time spent with a healthcare provider, perceived accessibility, perceived waiting time, perceived professional care and overall patient satisfaction were associated with non-adherence to TB medication.
Adane et al. (2013) ⁽¹⁹⁾	Amhara	Cross- sectional study	280	Active TB patients	Forgetfulness, being in the continuation phase, HIV co-infection, and persistence symptoms of TB were the factors for non-adherence to TB medication.
Tadesse et al. (2013) ⁽¹⁰⁸⁾	Amhara	A qualitative	26	Active TB	Access to health facilities, financial burdens, quality of health services and social support are the main reasons for failing to fully adhere to TB medication.
Kebede et al. (2012) ⁽¹⁰⁰⁾	SNNPR	Cross- sectional study	24	Active TB-HIV co-infected	The educational status associated with non-adherence to TB medication. Forgetfulness and transportation cost was mentioned as a reason for non- adherence to medication.
Mindachew et al. (2011) ⁽¹¹²⁾	Addis Ababa	Cross- sectional study	319	Latent TB-HIV patients	Patients who did not received information about IPT, patients not comfortable to take IPT in front of other people, who were not attended regular clinic appointments and drug side effects were associated to non-adherence to isoniazid preventive therapy (IPT).
Gebremariam et al. (2010) ⁽¹⁰⁹⁾	Addis Ababa	A qualitative study	38	Active TB-HIV co-infected & Healthcare Providers	Side effects, pill burden, economic constraints, lack of food, stigma with lack of disclosure, and lack of adequate communication with health professionals were barriers for adherence. While beliefs in the curability of TB, beliefs in the severity of TB in the presence of HIV infection and lack of support from families and health professionals were influenced the non-adherence to TB medication.
Mesfin et al. (2009) ⁽¹¹⁵⁾	Tigray	Cross- sectional	237	Active TB patients	Lack of supervision and incapable of dealing with patients' illness more likely to interrupt and default from treatment.

Sagbakken et al. (2008) ⁽¹⁸⁾	Addis Ababa	A qualitative study	50	Active TB patients Healthcare providers & Relatives	Transportation costs, poor health status due to illness or slow progression, have not got social support and those did not manage to restore their health and social status were factors for non-adherence.
Shargie et al. (2007) ⁽¹¹⁸⁾	SNNPR	Prospective cohort study	404	Active TB patients	Continuation phase of treatment, a far distance from the treatment center, and necessity to use public transport were the factors for default from TB treatment.
Michael et al. (2004) ⁽¹⁰⁴⁾	Oromia	Cross- sectional study	114	Defaulted TB patients	Far distance from the health institution, transportation cost and unaware about TB were the major reasons contributing to defaulting.
Tekle et al. (2002) ⁽¹⁰³⁾	Oromia	Case-control	1367	Active TB patients	Being in the continuation phase of treatment, lack of family support, inadequate knowledge about treatment duration and drug side effects.

SNNPR: South Nations, Nationalities, People Region, ART: Antiretroviral therapy

APPENDIX TWO: QUALITATIVE ASSESSMENT FOR Ma-MAS INTERVENTION DEVELOPMENT

Demographic characteristics

A total of 21 health professionals were included in the in-depth interviews conducted for the purpose of intervention development, consisting of 10 males and 11 females. Out of the 21 health professionals, 15 were TB focal persons who worked at health facility, and 6 were TB prevention and control experts who worked at the health office. The participants' ages were uniformly distributed within a range of 26 to 58 years. The majority of participants held a bachelor's degree (n=16) (see Table 30).

		Parti	cipar	ts																		
Characteristics	n	A	В	С	D	E	F	G	н	I	J	к	L	М	Ν	0	Ρ	Q	R	s	Т	U
Gender																						
Male	10	м							М		М		М	М	М	М		М	М			М
Female	11		F	F	F	F	F	F		F		F					F			F	F	
Age group																						
25-29 years	5				29		26						29			28		29				
30-34 years	5									30	32			31	34							32
35-39 years	4		37			35			35												39	
40+ years	7	47		50				42				53					58		45	47		
Education																						
Diploma	2				D													D				
Bachelor's degree	16		В	в			В		В	в	в	В	в	В	В	В	В		В	в	В	в
Master's degree	3	М				М		М														
Job																						
possession																						
TB focal person (TB-FP) at health facility	15			TB- FP	TB- FP		TB- FP			TB- FP	TB- FP										TB- FP	
TB prevention and control expert (TB-PCE) at health office	6		TB- PCE			TB- PCE		TB- PCE				TB- PCE										

Table 30: Demographic characteristics of health professionals, Addis Ababa, Ethiopia

Education: D is Diploma, B is bachelor's degree, M is master's degree

A total of 15 adult TB patients were included in the in-depth interview, consisting of 10 males and 5 females. The majority of the participants were between the ages of 35-39 years (n=5), followed by those between the ages of 25-29 years (n=4). The majority of participants had completed education beyond grade 12 (see Table 31).

		Participants														
Characteristics	n	А	В	С	D	Е	F	G	Н	I	J	K	L	М	Ν	0
Gender																
Male	10	М	М	М			М	М			Μ	М	М	М		М
Female	5				F	F			F	F					F	
Age group																
20-24 years	2											22				24
25-29 years	4				26	26	27		25							
30-34 years	1		33													
35-39 years	5			37				38		35				36	38	
40+ years	3	67									48		67			
Education																
Grade 1-8	3			7				7	8							
Grade 9-12	5										12		12	12	10	10
Grade 12+	7	D		В	D	D	D			D		D				

Table 31: Demographic characteristics of adult TB patients, Addis Ababa, Ethiopia

Education: D is Diploma, B is bachelor's degree

Theme, sub-theme and categories

The thematic analysis of the data identified five themes in relation to the intervention development for TB adherence to medication. The first theme, adherence, explored the factors that influence adherence to TB medication. The second theme, history of health system practice, examined participants' experiences with health system practices related to TB management. The third theme, acceptability of Mobile-assisted Medication Adherence Support (Ma-MAS) intervention, focused on participants' perceptions and attitudes towards mobile-assisted medication adherence support intervention. The fourth theme, convenience, investigated the design of mobile-assisted adherence interventions for optimal convenience. Lastly, the fifth theme, Barriers and Facilitators of Mobile-Assisted Adherence Interventions, explored the factors that hinder or facilitate the success of mobile-assisted adherence interventions. Each of the five themes had several sub-themes, and some of these sub-themes were grouped into categories, which are presented in Table 32.

Theme	Sub-theme	Categories
Adherence: Adherence to TB	1. Adherence to TB medication	
medication and influencing factors	2. Factors affecting TB medication adherence	 B. Patient centered factors C. Social-related factors D. Economic related factors E. Health system factors F. Therapy related factors G. Lifestyle related factors H. Geographic access factors
History of health system practice in TB management	 Adherence intervention and monitoring mechanism DOT standard care during COVID-19 emerged 	
Acceptability of mobile-assisted medication adherence support intervention	 Acceptability and benefits of mobile-assisted adherence intervention Acceptability and preference: SMS versus Phone calls 	
Convenience: mobile-assisted medication adherence support intervention design	 Language of the message Content of the message Comprehension Interactivity Frequency of intervention Time of delivery Duration of the intervention Privacy and confidentiality 	
	1. Facilitators of mobile- assisted adherence intervention	A. Willingness and high acceptanceB. High mobile coverage
Barriers and facilitators of mobile-assisted adherence interventions	2. Barriers of mobile-assisted adherence intervention	 A. Digital illiteracy and ignorance of messages or phone calls B. Have no won mobile or telephone C. Patient's registered wrong phone number or change their phone number D. Health professionals work burden E. Sustainability of the adherence intervention
	3. Suggestions	 A. Availing telephone at TB clinic B. Briefing the mobile-assisted adherence intervention C. Using family member's or caregiver's mobile phone D. Checking and updating the patient's phone number E. Giving the project a unique name

Table 32: Theme, sub-theme and categories

Page Break

Theme 1: Adherence to TB medication and influencing factors

Adherence to TB Medication

According to health professionals, non-adherence to TB medication or lost follow-up from TB treatment was a problem at public health facilities in Addis Ababa, Ethiopia. Moreover, they noted that non-adherence to TB medication was more aggravated after the emergence of the COVID-19 pandemic in which the standard DOT program was not implemented. The 53-year-old female TB prevention and control expert stated that

"Yes, non-adherence to TB medication is a problem in our sub-city. I am saying so based on the evidence that we have in this sub-city. One of the pieces of evidence is that the effort to follow, tracing and returning patients to medication is not always successful..."

Another 37 years female TB prevention and control expert added that

"Yes, non-adherence to TB medication is still a problem in the city. The magnitude of the problem, of course, varies from time to time. But it is still a problem even though we are observing a decline in recent years. The increase of drug resistance TB that we are observing is one of the indicators for the existence of non-adherence to TB medication."

In addition, most health professionals from health service delivery point also reported nonadherence to TB medication is a problem in their health facilities, especially this is devastating after the emergence of COVID-19. The 30-year female TB focal person stated that

"Yes, it is a problem that we experience in our health facility. We observe the problem from time to time. Before the emergence of COVID-19, non-adherence was not a major problem. But since we have stopped the DOT system, we have observed an increase in the magnitude of the problem. The official guideline is that we should give a one-month medication to reduce risks associated with

exposure to COVID-19. But in the second month, we discovered three patients who tested positive. This was something that we do not often observe. As a result of that, we are giving one-week medication instead. We also observed that two patients tested positive at the end of their fifth month of the medication. In addition, there was also one patient who was difficult to track. Based on these experiences we have now reverted to the DOT system."

The 28-year-old male TB focal person reported that

"In our health facility, it was rare to observe patients who stop the medication before COVID-19. There were no lost patients before COVID-19. ... After COVID-19 emerged we had two lost patients."

Another 32-year-old male TB focal person said that

"We have observed patients who fail to forget their medication ranging from a few days up to many weeks."

Another 47 years female TB focal person also added that

"Yes, non-adherence to TB medication is a problem in our health facility. For instance, I have come across three patients who relapsed and tested positive after taking their treatment. But I have not come across patients who interrupt their medication."

However, one health professional respondent reported that non-adherence to TB medication or lost to follow up was a problem before the emergence of COVID-19 because patients must come to the health facility daily to take their medication during the intensive phase of the treatment. The 29-year-old male TB focal person said that "It was a problem in the past, especially before the emergence of COVID-19. The standard procedure for administration of TB medication was the DOT system before COVID-19. Now I am doing follow-up with patients every two weeks. It was a problem before the emergence of COVID-19. The DOT was the standard practice and patients must come to the health facility daily to take the medication. ...Some patients may not have the money to cover transportation costs. In addition, some patients may have a job that they have to attend to feed their family and at times they may not be able to come to the health facility."

Some health professionals hold the view that non-adherence to TB medication or loss to follow-up from TB treatment is not a significant public health concern. The 35-year-old male TB focal person said that

"The standard for loss to follow-up from treatment must be below 5% of the total TB patients. In our sub-city, the proportion of patients that lost follow-up from TB treatment currently is a little over 1%. But I still believe that we should be able to reduce this number even though it is below the maximum percentage. The number shows that we have a positive performance."

Another 29-year-old male TB focal added that

"We do not experience that problem while working in this health facility. We often do take the necessary precautions to make sure that we do not experience this problem. When a patient starts the medication, we offer the proper counselling about the importance of adherence and taking the medication as prescribed. We also try to get patients to fully understand the consequences of non-adherence to the medication."

Another 39-year-old female TB focal person also stated that

"No, it is not a problem. I have been working in this position for the last five years. When I compare adherence between the time I started to work here and now, there is a significant improvement. Patients now are very much aware of the importance of adherence to the medication and the possible negative consequences of non-adherence. Nowadays if we are late arriving at the health facility to give the medication to patients, they would call us and urge us to be there on time so that the time for the medication will not pass."

There was also no indication that patients were not adhering to TB medications in any of the responses of qualitative interview. However, it is important to note that qualitative research may not capture the magnitude of non-adherence to TB medication, as it is not empirical study. The 67-year-old male TB patient said that

"I want to get better as soon as possible. So, I take the medication on time. Every day in the morning I take the medication at 8:00 am. I always make sure that not one-minute passes before taking the medication at the time. I am so eager about getting cured. I also do not have alcohol addiction or any other similar matter. I always take the medication on time."

Another 25-year-old female TB patient also stated that

"No, I have never forgotten to take the medication. Even at one time in the past when I went to visit a relative in another town, I had taken the medication with me to take at my relative's home. At one time I also got extremely sick, and I had to stay with another relative who lives in this town and my husband had to get up early at 5:00 am and bring the medication to the place I was staying at, and I was able to take the medication on time. I have always been able to make sure that I have taken the medication on time."

Factors affecting adherence

Patient-centred factors

According to some health professionals, forgetfulness due to patients' busy work schedules was a contributing factor to non-adherence to TB medication and clinic appointments. The 47year-old female TB prevention and control expert said that

"I have observed that sometimes patients would forget their refill visit because of their hectic work. Sometimes the reason could just be they are attending private activities. Sometimes, they may exhaust their medication on a Saturday, or Sunday and they may forget to collect their medication a day before Saturday..."

Another 32-year-old male TB focal person added that *"They just forget to take the medication, or they may have to attend some personal matter and so they may fail to take the medication."*

According to some health professionals, non-adherence to TB medication is attributed to a lack of awareness about the disease and its consequences of discontinuing TB medication. The 39-year-old female TB focal person said that

"The main factor is the patients do not have awareness about the seriousness of failing to take medication properly. Many patients tell us that they do not know the seriousness of non-adherence. Some patients assume that they may not need to take the medication for the whole medication period. But when we sit with them and inform them about the consequences of non-adherence, we observe how their perception changes. Other patients did not realize how important it is to take the medication for the whole period. It is often patients that are not aware of the consequences of non-adherence that are likely to fail to take their medication." Another 28-year-old male TB focal person added that

"The main factor is lack of awareness of the importance of completing the whole six months period of the medication".

Health professionals reported that beliefs associated with treatment and disease were one of the factors contributing to non-adherence to TB medication, the perceived wellness after taking some TB medications being the most raised reason. The 28-year-old male TB focal person said that

> "For most of them, the reason is that they are feeling better or observing improvements as full recovery and then stop taking the medication. But on the other hand, there are also patients who may not show recovery sooner and who then decide to stop the medication and try other options"

Another 26-year-old female TB focal person added that

"They will still stop their medication when symptoms start to disappear despite repeatedly being told that they should continue to use the medication for the whole six-month period."

Another 37 years female TB prevention and control officer also stated that

".... Some patients have shown substantial progress in their health after taking some medications and feel that they are cured that could lead them to discontinue their medication."

In addition, a 47-year-old female TB focal person reported that denial of the disease is one of the contributing factors for non-adherence to TB medication and not starting the TB medication,

"One of the driving factors for non-adherence to TB medication is that patients may be in denial about their situation. Some of them may be unaware of the issue and may assume that their case is not TB. This is a common issue that we observe with patients that have extra pulmonary TB. The physical pain they experience may be on their knee and ask how their case is TB. These patients have a challenging time accepting the diagnosis and the medication."

Most TB patients reported that they were adhering to the TB medication because they knew the consequences of non-adherence. The 38-year-old female TB patient said that

"I also know the nature of the disease and the importance of taking the medication on time. Just because I saw an improvement in my health does not mean I would take the medication less seriously. I know that the disease can relapse if the patient stops the medication. I have seen a relapsing disease in my neighbor. My neighbor was taking the medication in the past. But she stopped taking the medication at some point when she observed an improvement in her health condition. But at later time, the illness relapsed, and she later died because the medication did not work."

Another 37-year-old male TB patient added that

"There was a one friend that told me he had to take the medication for another six months because he interrupted the medication for the first time. He informed me that the disease relapsed because he did not take the medication properly in the first instance. He shared his story as a warning for me not to make the mistake that he made. My wife is there to check on me and to inform me to take the medication when the time nears. When I am away from home, she also calls me and asks me if I have taken the medication." The 35-year-old female TB patient reported that she does not forget TB medications because she knows the consequences of nonadherence,

"... If I stopped or forget to take the medication, I will be the one who gets the consequence. I know this is a serious matter and I make sure that I always take the medication in the morning."

Another 26 years female TB patient added that

"I have observed an improvement in my health since starting the medication. But that does not mean I will ponder stopping the medication. That is because I have been made aware of the possible consequences of failing to take the medication for the whole six-month period. I know I must take the medication regardless of the health improvements that I observe myself..."

In addition, some other TB patients reported that perceived wellness after taking some TB medication is a contributing factor to non-adherence. The 26-year female TB patient said that "One of the reasons is that patients fail to continue to strictly take the medication when they see an improvement in their health. Patients may stop taking the medication because they think that they have been cured." The 37-year male TB patient also said that "Some patients may not be aware of the seriousness of the illness. As a result, they only take the medication when they when they experience physical pain."

Some health professionals suggest that negligence and fatigue could be contributing factors to non-adherence to TB medication. Patients may become overwhelmed by the demands of their treatment regimen, leading to unintentional non-adherence. Additionally, fatigue or other physical or mental health issues could make it challenging for patients to adhere to their medication regimen. The 50-year-old TB focal person reported that fatigues was commonly observed in young men in the last month of treatment,

"We observe fatigue in some patients at the end month of the treatment, and patients are not being consistent in taking their medications. We observe these problems among young men. Young men who frequently use motorbikes to earn their livelihood are often exposed to the disease and we also observe the problem of non-adherence with them. Adult people are more careful about their treatment."

Another 31 years male TB focal person added that

"One of the reasons for non-adherence to medication is that patients may become negligent and fail to take the medication properly."

Social related factors

According to some health professionals, family support can have a positive impact on TB medication adherence. Patients who receive support from their family members may be more likely to adhere to their medication regimen and attend clinic appointments. The 32-year-old male TB focal person said that

"They also may not have a close family that lives with them and that can support and take care of them as they take the medication."

Another 29-year-old male TB focal person added that

"...there are also patients who may have a job that they have to attend to feed their family and at times they may not be able to come to the health facility. Family or friendship support can influence a patient to consistently take the medication." Additionally, some TB patients indicated that their families could remember and monitor their TB medication usage. The 48-year-old male TB patient said that stated that

"I have a family that depends on me, and I make sure that I take the medication on time for my own and my family's sake. My whole family also checks on me every day to see if I have taken the medication. My wife, my mother, and my sisters always check and ask if I have taken the medication. Therefore, I have never failed to take the medication so far."

Another 25-year-old female TB patient added that

"Initially, my husband used to tell me not to forget taking the medication when he wakes up around 6:00 am to head to his place of work. But later, I became accustomed to taking the medication at that time and I normally wake up at 6:30 am and get water and take the medication."

Economic factors

According to most health professionals, individuals with low economic status face difficulties in adhering to TB medication. Economic constraints, such as the inability to afford sufficient food, can impact the ability to follow up on TB treatment. Health professional respondents also reported that there were some homeless patients who do not have a contact person and a mobile phone to trace and back them to the treatment when they lost. It is impossible for these patients to afford adequate food and, as a result, they are unable to attend TB treatment appointments on a regular basis. The health 30-year-old female TB focal person said that

"The second factor is related to economic status and ability to afford the required diet to take the medication. Taking the medication without the recommended diet supplement is a challenge. This is the case that we observe with HIV AIDS patients who are also taking TB medication."

Another 58 years female TB focal person added that

"One of the reasons for patients to fail to continue to take the medication is related to diet. Patients who may be unable to afford the needed diet may be afraid to take the medication. That is why we observe interruptions with patients."

The 53-year-old male TB prevention and control expert also said that

"Another driving factor for non-adherence is related to economic factors. One of the challenges that patients complain about is taking medication without the appropriate diet. Besides, patients who are at risk of unemployment and loss of income while being at home or coming to a health facility to take the medication may also experience non-adherence. I suspect such a challenge that young patients who are normally engaged in daily and casual labor may be exacerbated due to the impact of COVID-19 on overall economic activity."

According to health professionals, patients who are homeless or live on the streets often face challenges in adhering to medication due to their lack of a fixed address. Bringing such patients back to treatment can be difficult and requires extensive efforts in tracing them. The 47-year-old male TB prevention and control expert added that

"...in the city, there is a large number of TB patients who are homeless. This is the poorest population and as a result, they are often most likely to default from the treatment."

Another 45-year-old male TB focal person stated that

"One of the reasons for non-adherence to the medication is that some of our patients are homeless and people who live on the streets. These patients do not have a fixed address and tracing them and bringing them back to treatment is a difficult matter. Other patients may come from other towns to start the treatment. But after a certain period, they may not be able to continue the medication because they do not have a place to stay here and because they are unable to afford to live here. Therefore, economic instability is one of the drivers of non-adherence to medication with these patients."

Another 26-year-old female added that

"It (non-adherence) is mainly much more with patients who live on the street. These patients do not own a personal mobile phone number and we are not always able to reach out to them whenever they miss a refill day or when they fail to come to the health facility to take the medication in the first two months."

Furthermore, the nature of patients' work has an impact on their adherence to TB treatment, as they may be unable to attend regular treatment appointments. Patients engaged in daily labor or those who work night shifts and sleep during the day may miss their scheduled health facility appointments for medication refills. Barriers to adherence to TB medication also include long-distance drivers, whose job nature makes it challenging to follow up on treatment. As a result, this job category may not be suitable for the Directly Observed Treatment (DOT) program, which requires daily presence at the health facility. The 47-year-old male TB prevention and control expert said that

"The other reason may be related to their work to earn income for their family that may not be fit with their treatment schedule and may not be able to frequently follow their treatment. Sometimes they may work the night shift, and they want to sleep during the daytime and as a result they may not appropriately follow their medications. Some long driver patients may move from one place to the other as a result they may discontinue the treatments."

Another 42-year-old female TB prevention and control expert said that

"I have observed that sometimes patients would forget their refill visit because of hectic work. Sometimes the reason could just be that they are attending a private matter."

A 32-year-old male TB focal person added that

"Some of them may need to go to another town for work and they may at times forget to take the medication with them. So, the movement for work is one of the factors that contribute to non-adherence."

Health system factors

Most health professional respondents reported that changing the Directly Observed Treatment (DOT) standard from daily medication intake to monthly medication refill visits had a negative impact on medication adherence. This change in the DOT standard made it more challenging for healthcare providers to monitor patients' medication adherence and provide support as needed. Additionally, some health professionals reported an increase in loss to follow-up from TB treatment after the DOT program was changed. The 47-year-old male TB prevention and control expert said that

"... Giving one-month medications and not having a follow-up system for checking the patients taking the medication may affect medication adherence. Especially for those in the intensive phase of treatment, the follow-up should have been daily and now it is only done once a month and even for those on a continuation phase of treatment, the follow-up has been pushed to once a month from a weekly basis. This is all done due to the measures taken to slow the spread of COVID-19, but it may affect non-adherence..."

Another 42-year-old female TB prevention and control expert added that

"Changing the DOT system where patients used to come to health facility daily to self-administered medication at home for a month can contribute to nonadherence to medication."

On the other hand, some health professionals reported that the daily Directly Observed Treatment (DOT) standard posed a challenge for some patients who needed to come to the health facility every day. Patients with limited transportation options may find it difficult to attend daily clinic appointments, which can lead to missed doses of TB medication. The 47-year-old female TB focal person stated that "On top of that, having to come to the health facility daily frustrates the patients and leads them to interrupt the medication." Another 29-year-old male TB focal person said that "The DOT was the standard in which patients were expected to come to the health facility daily to take the medication."

Some health professionals reported that a lack of commitment from healthcare providers to monitor patients' treatment can have a negative impact on adherence to TB medication. Patients who do not receive adequate support and monitoring from healthcare providers may be more likely to miss doses or discontinue their medication regimen. The 37-year female TB prevention and control expert said that

"In intensive phase, I observed two factors that may contribute to nonadherence. The first one is that healthcare professionals do not strictly monitor and observe the patients' medication intake. There is also not a custom for health professionals to contact the patients by phone to check whether they are properly taking their medication once the patients start the continuation phase of treatment. Thus, a lack of commitment by a healthcare professional to monitor patients' medication intake may contribute to non-adherence."

Additionally, a lack of communication between healthcare providers and patients can lead to misunderstandings or misinformation about TB treatment, which can further impact TB medication adherence. Unable to provide appropriate counselling to the patient at the diagnosis of TB and poor health professional patient relationship also negatively affect adherence to TB medication. The 29-year male TB focal person stated that

"There are also gaps in health professionals' commitments in providing counselling and essential information regarding the risks and problems associated with non-adherence. The medication is taken over an extended period and there is always a risk of non-adherence, especially if there is a gap on the side of experts in providing proper counselling. Therefore, counselling can affect medication adherence."

Another 47-year-old female TB focal person added that

"The gap here usually is with the physicians that prescribed the TB medication without properly explaining to the patient about their diagnosis. Such patients are likely to stop the medication. Patients are often more likely to trust their physicians than us (health professionals working primary health care) who give them medication. But if the physician did not explain their case fully, we will have a challenging time convincing the patient... The physician may not tell the patient about the length of medication that he/she prescribed to them. When patients start the medication and inform them of the length of the treatment, they interrupt the medication for a certain period to reach their physician to confirm the length of the medication time. They will interrupt the medication for two or three days and come back to continue the medication." Another 39-year-old female TB focal person shared her experience on how poor health professional patient relationship affect adherence to TB medication,

"There was, for instance, one patient who lives in the streets around... and he came to our health facility to take the medication. He started the medication in another health facility. Because he did not like how they treated him in that health facility. He was not informed that he will take the medication for six months. But fortunately, he came to our health facility, and he was given all the information that he needed, and he was finally cured... Physicians do not stress why the patient will take the medication for six months. Providing proper information on the consequences of non-adherence will always contribute to addressing the problem..."

Some patients reported that the health information they received from the health facility about the consequences of non-adherence encouraged them to take their TB medications properly. Patients who are well-informed about the risks associated with non-adherence may be more motivated to adhere to their medication regimen and attend clinic appointments. The 26-year-old female TB patient said that

"They (the health professionals) have told me that I will not be cured if I do not take the medication correctly on time. They told me that if I did not take the medication properly, then I would not be cured and that I might be forced to take the medications for up to two years. To avoid that I am taking the medication properly and at the right time.

Another 36-year-old male TB patient added that

"When I started to take the medication, the health care experts at the health facility informed me about the consequences of failing to take the medications correctly. They informed me that if I missed the medication for a day or two the

illness would relapse. After realizing the possible consequences of failing to take the medication as recommended, I have been taking extra care to ensure that I am taking the medication on time."

Some health professionals reported that TB focal persons' lack of incentive to work over the weekend poses a problem with adhering to TB medication. Therefore, some health facilities did not work on weekends. The 47-year-old male TB prevention and control said that *"One of the gaps from the healthcare system is that professionals assigned to work on weekends to administer the medication do not always adhere to the working schedule. Health professionals are not incentivized to work on the weekends that is one of the gaps from the healthcare system is that professionals are not incentivized to work on the weekends that is one of the gaps from the healthcare structure side."*

Therapy-related factors

Some health professionals reported that drug side effects as one of the factors that negatively influence TB medication adherence. The 32-year-old male TB focal person said that

"... the adverse effects of the medication contribute to non-adherence to the medication. Patients experience adverse reactions such as vomiting and nausea until their body becomes familiar with the medication and there may be patients who skip the medication due to these adverse reactions. Patients complain about the adverse reactions until they enter the continuous phase of the treatment."

Another 58-year-old female TB focal person added that

"In addition to these, some patients may interrupt their medication due to some of the adverse effects of the medication. The adverse effects may be strong on some patients and patients may choose to stop taking the medication for a few days to get relief." According to some health professionals, the pill burden associated with other co-morbidities can have a negative impact on adherence to TB medication. The 53-year-old female TB prevention and control expert stated that "One of the reasons for adherence that we observe with HIV AIDS patients who are also taking TB medication is the Pill Burden." Another 32-year male TB focal person added that "sometimes patients experience another disease in addition to TB and they may not take the TB medication for some time."

Some health professionals have reported that a long treatment regimen can also have a negative impact on adherence to TB medication. The 34-year-old male TB focal person stated that "One of the reasons for non-adherence has to do with the fact that the medication is taken for a longer period. This means that there is a risk for patients to become forgetful." Another 50-year-old female TB focal person added that "The main reason for non-adhering concerning TB patients is the length of the TB treatment period. The medication is expected to be taken for six months, and people who are unstable due to either their working condition or other personal factors are likely to be non-adhere."

Lifestyle factors

According to some health professionals and TB patient respondents, having an addiction to alcohol was a contributing factor to non-adherence to TB medication. Alcohol addiction can lead to forgetfulness, decreased motivation, and an increased risk of adverse drug reactions. The 30-year-old male TB focal person said that

"There are patients who have alcohol and other types of addictions. At the bus station there are a lot of young people working there. Most of them have addiction issues. Those who take TB medication tend to stop taking the medication to revert their addictive behaviour. With some patients, we also observe that they forget to take their time. For instance, one of the patients skips the medication when he takes alcohol. He said he is afraid that taking the medication after having alcohol will damage his liver." Another 37-year male TB patient also said that "...Some other patients may take alcohol and may stop taking the medication."

Geographical access factors

Most health professionals reported travel distance and transportation cost as factors affecting TB medication non-adherence. It was reported that some patients live far from the health facility, which results in high transportation costs. It is especially difficult during the intensive phase of treatment when medication must be taken daily at the health facility. The 47-year-old male TB prevention and control expert said that

"One of the main reasons for non-adherence to medication is that most TB patients are economically low. For some of them, transportation is a challenge because they may not even be able to pay for the taxi, as a result they may default from the treatment."

Another 37-year-old female TB prevention and control expert added that

"...patients may not have sufficient income to pay for transportation to come daily to health facilities to take their medication. Thus, their economic status may have an impact on non-adherence to TB medication."

Another 29-year-old male TB focal person said that

"One of the factors that can lead to non-adherence to TB medication is the distance from patients' residential area to the health facility. The greater the distance between patients' homes and the health facility, the patient is more likely to interrupt medication. To prevent this, when a patient starts the medication, we also check if the distance between the two locations is far..."

Another 32-year-old male TB focal person added that

"The other factor is distance. Some patients may not be able to afford to pay for a transportation cost for a 'Bajaj.' Most of the TB patients in the first stage as they take the medication are very weak and they cannot walk to the health facility daily. Such patients who cannot afford the transportation cost are more likely to skip taking the medication."

Theme 2: History of health system practice

Adherence Intervention and mentoring mechanism

According to health professionals surveyed, there is currently no sustainable governmental or non-governmental organization implementing interventions to improve TB medication adherence. However, there were plans for interventional research aimed at improving TB medication adherence. Most health professionals mentioned the ASCENT project, a large cluster randomised trial that was in the pilot phase before the emergence of the COVID-19 pandemic. At the time of the assessment, the project had not yet started. The ASCENT project was planned to be implemented in Addis Ababa and nearby cities in the Oromia region, using digital equipment to send text messages to remind patients to take their medication and sound an alarm if they missed their medication time. The project was planned for implementation in 36 health facilities in Addis Ababa, Ethiopia. Detailed information of the trial is available at the link below.

https://pactr.samrc.ac.za/TrialDisplay.aspx?TrialID=12241.

The 47-year-old male TB prevention and control expert said that

"There was one project called ASCENT. We did the baseline assessment for the pilot program with them. But the project never progressed to the implementation phase. It uses a digital system that sends a text message to a central server when patients open the medication strip. It is aimed at promoting the effective use of technology to monitor medication adherence."

Another 53-year-old female TB prevention and control expert added

"I know there was a plan by one non-governmental organization that came up with the idea of using a digital system to remind patients about taking their medication. It was initially planned to be implemented in a selected few health facility. They planned to utilize a digital box. It provides a reminder service for the patient. When the patient takes the medication, the TB focal person will get a message indicating the patient has taken the medication."

Most health professionals have reported that they occasionally make phone calls to patients or treatment supporters when patients miss their medication refill appointments, fail to show up for clinic appointments, or are lost from treatment. They may also make phone calls when patients need to undergo sputum tests. Additionally, some health professionals make phone calls to patients who have not taken their medication correctly. The 32-year-old male TB focal person said that

> "We do not call all the patients. We identified three patient groups to make follow-up calls. The first patient group is those who may struggle economically. The second patient group is those who live in a relatively distant place and far away from the health facility. The third patient group is those who have a low educational qualification. These are people who we think are more likely to be non-adherent to taking the medication. On the other hand, we do have patients

who are homeless and who live on the streets. These patients are likely to experience non-adherence because of their living conditions and we call them to check if they are taking the medication properly. We also call them to remind them to come and collect the medication. There are of course patients who may not have a phone. For these patients who may not have their own phone, we get a contact person's phone. The contact person is a person who is close to the patient and that we can call if we cannot reach the patient."

Another 32-year-old male TB focal person stated that

"We call patients when they fail to come to the health facility for a number of days. But we use our phones to call patients. There is no phone available for us to call. The rule is that if a patient does not come to the health facility for a week, we should start the tracing process and one of the ways is to try to call them."

Another 58-year-old female TB focal person added that

"We use phone calls to reach patients in the intensive phase of treatment who forget to come to take the medication. The same goes for patients who forget to come on a refill appointment date in the continuation phase of treatment. We also make phone calls to remind patients to come for a sputum test. But we do not make calls to remind patients to take the medication."

Another 45-year-old male TB focal person added that

"Yes, we call patients when they fail to come on their appointment dates. Back when the DOT system was employed patients used to come to the health facility daily. But now since patients come every other week and some of them once a month, we do not have the opportunity to closely monitor their progress as we used to. A phone call is one way for us to reach patients who fail to come on the appointment date. ...For example, if a patient who takes home a oneweek medication fails to come after seven days, we will call him and remind him to come and take the medication as soon as possible. In this manner, we have been using phone calls to follow up with patients to come and get the medication."

Another 50 years female TB focal person said that "when we see patients default from the treatment, we make phone calls and tell them to continue their TB medications. We often make a phone call to back them to the health facility and give them counselling and guidance services to let them know the importance of taking the TB medications properly and as prescribed by medical experts and if it is discontinued it has a risk." Another 35 years old female TB prevention and control expert added that "When we identify these patients that are defaulted to their TB treatment, we try to trace their contact information. We also keep the contact information of another person that is close to them, and we make calls and try to get them to continue taking the medication."

Some health professionals would make a phone call to remind some patients to refill their medications and schedule sputum tests. The 29-year-old female TB focal person said that

"We make calls to monitor how properly they are using their medications. We make at least one phone call per week for all patients. We do not have many patients and we make the phone call once or sometimes twice a week to monitor if they are taking the medication. I am the one who is responsible for making the phone calls. In the past, we would always make phone calls to patients who forgot to come to the health facility for the medication refill visit. We would also make phone calls to patients who have a sputum sample examination. We always keep the contact information of the patient and a family member or some other person who is close to the patient. So, we make phone calls as needed. But the phone calls in the past were not made as part
of monitoring and follow up mechanism to check if patients are taking the medication."

Another 29 years male TB focal person added that

"We give priority to patients on the DOT stage (the first two months of treatment). In addition, we also prioritize patients who were weaker physical status when they start the medication. We make sure that we make phone calls to these patients more frequently. We make a phone call for patients on the continuation phase of treatment every two or three weeks."

Another 47-year-old female TB focal person reported that she would make a phone call for the medication refill visit and sputum test examination before one or two days of the scheduled date,

"When I give a patient a one-month medication, I will record the date for the refill date after exactly one month. When that date nears, I will call the patient to come to collect additional medication. I usually call patients the day before the refill and remind them to come to collect the medication. Most of the patients will call me themselves right before the refill date and check if I will be available in the health facility. In addition, I also call patients and remind them about the sputum tests that they take at the end of the second, fifth and sixth months of medication. Most of the patients call me themselves. I also tell them to call me two days before the appointment date and check if I am available and take the medication. I also told them that I will not be available over the weekends. If the patient will have a sputum test tomorrow, I will make the call today."

Health professionals reported that the health extension workers have the responsibility to provide health education during home-to-home visits. A 58-year-old female TB focal person from the health facility said that

"Health extension workers are mainly tasked with this responsibility. They conduct a home-to-home visitation and counsel TB patients about the importance of adherence and the consequences of failing to properly take the medication. In addition, we also counsel them about the importance of adherence whenever we get the opportunity to speak to patients as they come to take the medication or on a refill date."

Another 37 years female TB prevention and control expert from the health office added that

"Several activities and efforts have been carried out to address non-adherence to TB medication. The first one that I can mention and still is being carried out is the work done by the health extension workers at the community level. They are conducting an awareness creation campaign to make patients visit health facilities and adhere to their medications. Despite the efforts made, there is still a varying level of non-adherence once after the patients starting their medications."

The health extension workers will also trace patients when they are lost from the treatment through home-to-home visit. The 30-year-old TB focal person said "Another opportunity may be that health extension workers can support the program. Health extension workers working with me are particularly good at tracing and returning patients to treatment. We can give them the patient's phone number and they can even trace patients home to home." Another 42-year-old female TB prevention and control expert also said that "We try to deal with default to TB treatment through the engagement of Family Health Groups and Health Extension workers who would go home to home. We provide the list of TB patients to treatment. This list, which contains the name and phone numbers of each defaulter in the village that given to the health extension worker assigned to cover that village. They will also trace defaulters and bring them back to treatment."

At the beginning of the TB treatment, health professionals usually educate the patient on the consequences of non-adherence to TB medication. The 29-year-old TB focal person said that

"We only make sure that we give patients the proper counselling to ensure that they are taking the medication. We explain in detail to patients in the intensive phase why we are giving a one-week medication. We will tell them that we are giving them a one-week medication because we are trying to reduce their exposure to COVID-19. We also inform them about the importance of strictly adhering to taking the medication. The same counselling is also given to patients' treatment supporters. We inform them that they should support the patient and make sure that they are taking the medication daily. We also explain to them the risk of developing MDR if the patient fails to take the medication properly. We take these counselling and discussions with patients and treatment supporters as potential non-adherence prevention mechanisms. But we do not have any guarantee that the patient will take the medication after taking it home."

Another 58-year-old female TB focal person said that

"We counsel the patients and inform them about the consequences of failing to properly take the medication. We also tell them about MDR TB and the consequences that come with it. Before COVID-19, health extension workers will carry out a home to the home visit of patients who fail to come to take the medication or for a refill visit clinic appointment."

Another 47-year-old TB focal person stated that

"There is no way we can truly confirm if a patient is taking the medication at home. The only means available to encourage patients to consistently take the medication on time and without interruptions is to counsel them about the importance of adherence to medication and the negative consequences of nonadherence and medication interruptions. I inform patients that we are giving them the medication to bring to their homes because we wanted to reduce their exposure to COVID-19. We try to stress as much as possible about the importance of strictly taking the medication and that if they are experiencing any form of adverse reaction to the medication, we tell them to reach out to us instead of just stopping the medication."

Monitoring patients' medication adherence through treatment supporters is a control mechanism that has been reported by health professionals. Treatment supporters can be close family members or caregivers who use a paper card to monitor the patient's correct intake of medication. Patients are expected to bring paper cards when they visit the health facility for medication refills each week. Additionally, if a patient is lost from treatment, the treatment supporters will be contacted. The 47-year-old male TB prevention and control expert said that

"Previously patients in the continuation phase were contacted by treatment supporters every week. The treatment supporter is checking if the patient is taking their medication and recording on the card. The treatment supporters try to see if patients had failed to use the medication at any time of the week. A treatment supporter is a person close to the patient and they are also supported by the health extension workers during the home-to-home visit. The patients will bring that card when they come to the medication refill visit. The health professional checks whether the patient had missed any medication."

Another 53-year-old female TB prevention and control expert added that

"We use the treatment supporters of the patients to check if patients are taking the medication properly. The treatment supporters are usually members of the patients' families. We try to check how well patients are properly taking the medication when they come to the health facility by accompanying patients."

However, according to a report by health professionals, paper-based monitoring cards were not properly implemented in most health facilities. In some health facilities that do implement it, patients may not bring the monitoring card to the health facility during refill visits. For example, the 32-year-old male TB focal person said that *"They are expected to bring the card. Most of them bring the cards with them when they come here. But some of them forget to bring it. But check their overall status at the end of the intensive phase."*

Another method of monitoring patients' adherence is through healthcare professionals who typically check whether patients come and refill their medications on time. The 32-year-old male TB focal person said that

"We put the date the patient comes to the health facility and then we put the date that he will be coming back after a week. We also keep the scheduled date on the card and track the dates. We make sure that the patient comes to collect the medication for sixteen weeks. For instance, if the next appointment is on the 24th and if the patient comes on the 26th, then that is an indication that the patient ^{ha}s skipped the medication for two da^{ys}. It helps us in following up and monitoring patients."

Another 50-year-old female TB focal person added that

"There is no mechanism to monitor the patients to confirm they are taking the medication. Our observation mechanism is whether the patient is coming on the scheduled date of the appointment. There is no way we could effectively monitor non-adherence. We just try to inform patients about the importance of taking the medication as prescribed by experts and about the consequences of failing to adhere to TB medications and their side effects. When they come

back to refill their medications for the next week, we try to have a conversation with them to check how well they are properly taking the medications. Then we will give them two weeks' medication. We try to ask them at what time they are taking the medication."

According to some health professionals, they use various methods to monitor patient adherence, such as counting the number of medication stripes left, checking the refill appointment date, or asking the patients if they are taking their medication. The 28-year-old male TB focal person said that

> "One of the ways we use to check if the patients are taking the medication is to see if they have any medications on their hand that they have not taken yet. If they have a pill on their hand, it means they have not taken the medication properly. If I give the patient seven days of medication to take home and if they do not have any on the eighth day when they come, that is an indication that they have taken the medication. The second way we check is to ask the patient if they have taken the medication. Another indication can also be whether the patient comes on the refill date or not."

Another 39-year-old female TB focal person said that

"We also request patients to bring the empty strips with them when they come for a refill appointment and even though it is not conclusive evidence that they have taken the medication, we consider that as evidence too. An additional thing is that we often inform patients to come for an appointment date on the day they take the last medication that they took from their last refill. When they come, they will also bring the strips and if there is any medication left on the strip, it is an indication that they have not taken the medication properly. If there is no medication on the strip, then it is an indication that they have taken the medication properly. But the main factor is that we inform the patient about the consequences of failing to take the medication properly."

Another 31-year-old male TB focal person said that

"In the period before COVID-19, we use the DOT system and patients come to the health facility daily. We tally patients' attendance. If a patient is absent or if a patient is late, we call them and check why or if they experience anything. In the intensive phase, the medication is taken here (health facility), and patients come daily. We know if a patient did not come. In the continuous phase of the treatment, we check the medication stripes. We give them a two-week medication to take at home."

Another 26-year-old female TB focal person added that

"We try to tell the patients to bring the empty stripes when they come back for a refill. We take that as one of the indications that they have taken the medication. They would tell us they know taking the medication is for their own sake. The sputum test is also taken as an indicator. The symptoms would reappear if they were not taking the medications. We use these indicators to evaluate if patients call them and check things. We also remind them to continue to properly take the medication."

Furthermore, some health professionals have also stated that a patient's sputum test can indicate whether the medication is being taken correctly. The 30-year-old female TB focal person said that "*The first indication is the tests that they take. I have worked with TB for two years now. It was rare to find a patient who tested positive in the second month of the treatment. Now, we have found three patients that have tested positive. This is an indication that they are not taking the medication properly."*

The DOT standard care during COVID-19 emerged

The Directly Observed Treatment (DOT) program for drug-susceptible TB treatment initially required patients to take their TB medication at the health facility for the first two months of treatment, followed by non-daily DOT where patients take their TB medication at home and visit the health facility weekly to refill their medications from two to six months of treatment. However, health professionals have reported that the standard DOT program has been interrupted since the emergence of the COVID-19 pandemic. The 47-year-old male TB prevention and control said that

"The DOT program has been disrupted and is not being applied due to COVID-19 pandemic. Patients at both the intensive and continuous phase of treatment now are taking their medications in their own homes. This has been made to reduce the movement of people and not to contribute to the spread of COVID-19. A one-month medication is given to all patients, and they will come and collect after exhausting the medication."

Another 53-year-old female TB prevention and control expert added that

"Since the COVID-19 pandemic happened, we have changed the medication procedure. Patients in both the intensive and in the continuation, phases are taking a one-month medication home. If the patient is not able to come to a health facility to collect the medication, a decision has been made for a family member to collect the medications on behalf of the patient. This is how currently the medication is being given to patients."

Another 35 years male TB prevention and control expert said that

"In the first two months, patients go to health facility daily and take the medication at home. This is the DOT phase. But since the emergence of COVID-19, there has not been a uniform standard that is used by all health

facilities within the sub-city. The official standard passed down from the ministry of health is that patients should come once a month and take a one-month medication to their home. But in some health facilities, a two-week medication is given to patients."

However, the medication refill visit schedule program was not consistently implemented by health facilities. Some health facilities used weekly or biweekly refill visits during the intensive phase of treatment and then switched to monthly refill visits during the continuation phase, while others used one-month refill visits throughout the treatment. The 32-year male TB focal person stated that

"Since the emergence of COVID-19, the standard DOT has been changed. Now we are giving a week's medication to patients at the intensive phase of the treatment. For those in the continuous phase of medication, we are giving them a one-month medication."

Another 29-year-old male TB focal person stated that

"...But in our health facility, we have been giving patients in the intensive phase a one-week medication while giving patients in the continuation phase a twoweek medication. We adopted this system because it is more convenient for us to follow patients' adherence."

Another 29-year-old male TB focal person said that

"Yes, the directive that we received from the Sub-city health bureau states that we should provide medication for one month for patients to take home since the emergence of cases of COVID-19 in Ethiopia. But in our health facility, we first provide medication for two weeks at the start of TB medication. That is because there are adverse reactions that patients may experience, and we

want to have the opportunity to closely monitor those reactions... But after the first two weeks, patients will be given medication for a month."

Another 34-year-old male TB focal person said that "...But after COVID-19, changes have been made to the standard and since then patients are taking a one-month medication to their home. The change was officially introduced by the Ministry of Health."

According to some health professionals, the schedule period for medication refills depends on the patients' behaviour as well as other factors, such as side effects and distance from the facility. The 58-year-old female TB focal person said that

> "Currently, we are giving the medication depending on the status and behaviour of patients. For patients that we believe will take the medication properly without interruptions, we will give them a two-week medication to take to their home. There are even patients that we give a one-month medication because we do have confidence that they will take the medication properly. For weak patients and for patients whose contact person may not be able to come weekly to get the medication, we will always give them more medication that can last them for more days. For patients that can come weekly, we will give them oneweek medication. But for patients who are newly starting the medication and entering the intensive phase, we would want to have the opportunity to make sure that they are taking the medication properly. We would also want to observe and check their adverse reactions to the medication. We will want to see if they have vomiting and other reactions."

Another 32-year-old male TB focal person said that

"...But we do not necessarily employ the new standard for all patients. We apply it to patients who may likely have challenges with work and who may have to work away from Addis Ababa and who may have to stay there for a longer time.

For instance, one of our patients is a long-distance truck driver who would go to Djibouti and back and for him, we are giving him a one-month medication. But for patients who are just starting the medication, we would not give them a one-month medication. We first give them five days of medication to monitor their progress."

Since the emergence of the COVID-19 pandemic, only one 30-year-old female TB focal person from the health facility reported that the standard DOT program had continued in their health facility. According to the health professional, they are implementing daily DOT for the first two months and weekly refills from two to six months of treatment, even though they were instructed to give one month's worth of medication at each refill visit.,

"The official direction given is that we should give a one-month medication to reduce risks associated with exposure to COVID-19. I have scheduled for all patients to come to the health facility in the morning at around 7 am. It is for patients in the first two months of their medication. For patients in the continuation phase, I am still giving them a one-week medication."

Theme 3: Acceptability of mobile-assisted adherence intervention

Acceptability and benefits of mobile-assisted adherence intervention

According to most health professionals, mobile-assisted medication adherence support intervention can promote TB medication adherence. The healthcare professionals believe that SMS text message and phone calls can avoid forgetfulness of medication intake. The 32-year male TB focal person said that

> "I think it can improve adherence. Phone calls and text messages can be an important reminder for patients because patients may forget to take their medication. Patients may be busy with work or personal matters and sometimes they may be away from town and may forget to be back for a refill

date. As a result, patients may not properly be taking the medication. They may experience interruptions. Phone calls and text messages can help to remind the patients."

Another 53-year-old female TB prevention and control expert also reported that SMS text messages and phone calls could be a solution for non-adherence to TB medication,

"Yes, it will be a solution to the problem of non-adherence especially with those patients who tend to forget to take their medication in time. Some patients may be busily engaged in work and, as a result, may forget to take the medication. We often see patients who forget the medication time. Implementing such an intervention can contribute a lot in reminding patients to take their medication at the right time which in turn can promote adherence to TB medication."

Another 39-year-old female TB focal person reported that mobile-assisted medication adherence intervention especially very important for those who do not have close family support,

> "There are patients that do not have a contact person and a treatment supporter. These patients usually live alone, and their family and relatives may be living in another town. I believe this intervention would be extremely useful for these patients. It can also be helpful for patients who may be living with their families."

In addition to simple reminders, health professional respondents believed that SMS text messages and phone calls could help improve patient awareness of TB medication adherence. The 53 female TB focal person said that "Yes, it can be a solution. We can use phone calls and text messages to promote patients' awareness about the importance of medication adherence." Another 35-year-old male TB prevention and control expert added that "I think it will have a positive role in creating awareness about the importance of

medication adherence... In addition, it can be used to remind patients to take the medication at the right time. Patients always tend to focus on either work or other personal matters as time goes on and they may forget to take the medication. Such a program can be used to address this gap."

Health professionals believed that adherence interventions using phone calls and SMS texts could improve the quality of professional care. By improving communication between health professionals and patients, it can motivate patients to follow health advice and properly take their medication. Additionally, health professionals reported that SMS text messages and phone calls could reduce the spread of TB and workload. The 32-year-old male TB focal person stated that

"Patients will also be satisfied when they observe that we are calling them. One of the main benefits will be that we will be able to reduce interruptions related to medication. In addition, it can also reduce workload by inducing behavioural change in the patients. I believe patients will be motivated to take the medications properly when health care professionals consistently call them and send them reminder text messages to take the medication properly. It can foster a sense of professional care which can also promote behavioural change. In addition, it can also contribute to TB prevention. The behavioural change as well as the fact that patients take the medication properly means that we can also reduce its transmission."

Another 29-year-old male TB focal person added that

"They (patients) could view it as professional care. There are also patients who ask us for our phone numbers and call us every time they have a question to ask or when they experience some adverse effects of the medication. In addition, we also interact with them freely and we try to make them comfortable."

Another 53-year-old female TB prevention and control expert added that

"Yes, it can reduce the workload. In addition, it can also foster interaction and communication between the patient and the health care professional. If we can successfully ensure that our patients are taking the medications properly, then that will contribute to the reduction in the transmission of the disease. The intervention not only benefits patients but also the wider community."

Additionally, some health professionals believed that SMS texts and phone calls have a role in reducing multi-drug resistant TB (MDR-TB) through proper medication adherence. The intervention is particularly important during COVID-19 pandemic in which the standard DOT program was not properly implemented. The 31-year-old male TB focal person said that

"Yes, I believe it will be a useful solution. Firstly, it will play a crucial role in reducing MDR-TB. The prevalence of MDR-TB was high in the past when we were using the DOT system. Now patients are taking a one-month medication at home, and we can expect an increase in MDR-TB because there is no way for us to verify the patient is taking the medication. If a mobile SMS text or a phone call-based intervention is to be implemented, I believe it can promote adherence especially in such a time DOT is not used."

Another 28-year-old male TB focal person added that

"The intervention can promote adherence. It can reduce MDR-TB. It can ensure that we will not lose patients. In addition, it can ensure that we will have information about patients. It can also help the professional in easily following up patients."

Some health professionals also reported that SMS text messages and phone calls could be useful reminders for medication refill visits and clinic appointments. The 47-year-old male TB prevention and control expert said that

"I think it would be helpful if the SMS text and or phone call reminder is done a week before for the monthly clinic attendance schedule to collect their medication for the next month. We ready medication kits for all patients and such reminder text messages or phone calls could be made a week or two to three days before their scheduled date so that patients could schedule early and have time to do the refill medication visit. Patients at times may be in a distant place for work or any other reasons and such reminders could help to avoid non-adherence."

Another male TB focal person 34-year-old added that "...we can also add a short message to remind the patient about their upcoming medication refill date. If we are sending the text daily and if the phone call is also made more frequently, patients can also be reminded about their refill date of medication without having to implement it as a separate intervention."

In addition, most patients believed that SMS texts and phone calls would help them to remember to take their medication. The 67-year-old patient said that

"I believe it can help patients who forget to take the medication on time. I also personally welcome such an intervention. It can be a useful intervention. I normally make sure that I never forget to take the medication on time. My family is also on my side, they are supporting me and making sure that I take the medication on time. But such an intervention can also be helpful for other patients that are especially likely to forget to take the medication on time."

Another patient of the same age and gender as above reported that patients may not give the same attention when they have medication in their hands to take at home and thus SMS text messages and phone calls can be useful reminder,

"...I think people are more careful to take the medication. But I also understand that people may not always pay attention to take their medication at the same

time as they have the medication in their own hands at home and when they do not have to go to the health facility daily to take the medication. I think it can be a useful reminder for patients. We may be busy, and we may at times forget the medication even though it is the most important aspect of our day-to-day life as TB patients. Having such a reminder intervention available to us is an important thing."

Another 22-year-old male patient said that *"I believe the intervention will be useful. I have been taking my medication properly so far. But as time goes on, the attention and focus that I give to the medication may not be as similar as it has been so far. Having such a reminder would help a lot"*

Some patients also reported that SMS text messages and phone calls can avoid the unwanted consequences of non-adherence to TB medication. The 25-year-old female patient said that

"I think it is a particularly good thing that you have come up with this idea. I have been informed about the consequences of non-adherence. I have been told that it can cause serious problems where patients can be forced to take the medication for more than two years. Therefore, trying to address this challenge and supporting patients so that they will take the medication on time is an incredibly positive thing. Patients forget to take the medication, and this is an extremely useful intervention that can address the problem."

Another 33-year-old male patient reported that patients are highly likely to forget the medication when they are taking medication for a long time at home and feel well,

"I believe that it will be useful for patients. Patients at a time may forget to take their medication on time. Patients are likely to forget to take their medication on time because the medication period is extended, and they must take the medication at home. The attention they give to the medication time may be less and less as the time goes on, and as they see improvement in their health status. If the phone calls and the SMS messages are going to reach the patient regularly, then the patient will always be reminded to take his medication on time. In doing so you will prevent the patient from developing MDR."

However, one 25-year-old female patient reported that SMS text messages and phone calls are only good and acceptable only for medication refill visit clinic appointment reminders but not for remembering medication intake, *"I do not think it will be a good idea to call or message patients daily to remind them or to ask them if they have taken their medication. What I think would be a good idea is to remind patients about their refill appointment dates."*

Acceptability and preference: SMS text versus phone calls

Most health professionals preferred phone calls over SMS text messages to promote TB medication adherence. The main reasons for the respondents were that SMS text messages are easily ignored as many texts are currently being sent from multiple organizations, patients may not check their phones for SMS text messages, some patients cannot read and understand SMS text messages, and phone calls are more engaging than SMS text messages for promoting TB medication adherence. The 29-year-old male TB focal person said that

"The phone call would be more impactful and more appropriate to achieve the needed results. This is because text messages are more likely to be ignored by patients. Literate patients may be able to read and understand the texts. But text messages could still be ignored."

Another 45-year-old male TB focal person added that

"In my opinion, the phone call would be the most impactful mechanism to ensure that patients are receiving the message we want to convey. A potential problem with the SMS text messages is that they could be missed because

patients may receive several text messages from different people and organizations."

Another 35-year-old female TB prevention and control expert reported that phone calls communication is more engaging and useful for those who cannot read and understand, "*The phone call is more engaging which is one advantage compared to the text message. For people who may not read, we could not send the texts and we cannot also expect a reply for the text message. So, the text message may not be engaging. But the phone call could be more engaging.*"

There were some health professionals who preferred SMS text messages over phone calls for adherence intervention. The major reasons for choosing SMS texts were that giving a phone call in the morning may be disruptive, it may not be feasible to reach all TB patients through phone calls, repeated phone calls may bore them, patients may not pick their phone when they are busy at work or engaged in other activities, a text message can be seen at any time when patients are free and sending SMS text message is easy. The 26-year-old female TB focal person said that

> "I think sending text messages is better than phone calls. This is because making repeated phone calls to people can be boring for them and at times patients may be at work and busy or handling personal issues. Sending the text message is easier for them and they can see it whenever they have the time."

Another 42-year-old female TB prevention and control expert added that

"Overall, SMS text messages can be a useful reminder. It may not be possible to make phone calls to all TB patients. Some people may miss the phone calls and the text messages can be an important reminder." Some health care professionals suggested both phone calls and SMS text messages are useful to promote TB medication adherence. The 32-year-old male TB focal person said that

"If that is possible, I think it would be more impactful. Having both (SMS and Phone calls) would be the best choice. The text message may be missed. Therefore, having the phone call can be an important guarantee to ensure that we are delivering our text messages to the patients."

Another female TB focal person with the same age as above added that

"I would say both are more appropriate. The phone calls can be used to check if patients are reading SMS messages. You can even decide if it is going to be useful to continue to send the SMS messages especially if patients are not seeing and reading the SMS messages."

According to some other health professionals, the choice between SMS text messages or phone calls will depend on the target audience. For educated patients, SMS text messages are the most appropriate means of communication, and for uneducated patients, phone calls are appropriate to promote TB medication adherence. In this regard, the 47-year-old female TB focal person said that "The SMS would be appropriate to patients that are literate that can read and write. To remind patients, the SMS text message would be more successful in reminding patients. For those unable to read and write, the phone call is more useful." Another 34-year-old male TB focal person said that "The text message will be more impactful with patients who are literate and/or able to read. For illiterate patients, the phone call will have more impact."

Similarly, some patients preferred to get phone calls over SMS text messages while some other patients preferred SMS text messages over phone calls to remind them of the TB medication. Patients who prefer phone calls for medication reminders mainly cited the following reasons: multiple SMS texts sent from various organization that bore the patient,

patients are less likely to read the SMS, SMS text message notifications may not be heard, and patients can ask a question and get answer from health professional through phone call. The 67-year-old male patient said that

> "A potential problem that I see with the SMS message is that, as you might have been aware of it, we receive a lot of SMS messages from different organizations. People are already fed up with these SMS messages and we always do not care to read the message that they sent us. I am afraid patients may consider the reminder messages as one of those other messages that we usually ignore. So, the phone call may be more appropriate and more likely to succeed. I also do not think patients will be disturbed because they will realize that the phone call is made for their own benefit."

Another 24-year-old female patient said that "Patients will start to think about the medication once they see the text message sent from you. But they may not necessarily read the SMS text message when they realize that the text message is sent from you or some known phone number. The phone call may be more acceptable for them." Another 26-year-old female patient believed that both SMS text messages and phone calls are useful, but the phone call is advantageous to ask questions and get an answer and stated that "I think both are applicable and I think most patients own a phone. But the phone call has more advantages because it allows for patients to ask questions and get clarification on issues they do not understand."

There were three major reasons why patients preferred SMS text messages for medication reminders over the phone calls; they do not want to be interrupted by phone calls in the morning, they do not want to discuss their medication in public, and they do not want to pick up unknown phone calls. The 22-year-old male TB patient had preferred an SMS text message, but he also told phone call is important for people may not see the text message,

"If the aim is to remind patients to take their medication by a phone call or a text in the morning, then my preference will be the SMS text message. Some people may not like to be disturbed by a phone call in the early morning. But, for people who may miss looking at a text message, then the phone call would be useful."

Another 35-year-old female TB patient said that

"I think the SMS message is preferable. That is because people may not like to receive a phone call early in the morning. In our culture, when someone calls them in the morning people assume to announce sad news, and phone ringing in the morning may be disturbing them. As a result, it may not be right to call them in the morning. But a text will not be that much of a source of discomfort for them."

Another 33-year-old male patient had concerned the acceptance of phone calls intervention by some patients because of the privacy issues, *"The phone calls may not be acceptable to some patients due to privacy issues. This is because some patients may be with people whom they did not disclose their health status. When you call them, they may not be comfortable to pick up the phone call and discuss their health status. But that still will not be a problem with all patients. There might just be some patients that may not feel comfortable to speak over the phone about their health condition."*

Some patients believed that the choice of SMS text messages or phone calls for medication reminders depends on the patients' literacy level. For example, the 36-year-old male patient suggested *"For people who are literate and can read and understand the content of the text message, I believe the SMS text message will be sufficient. The problem may be for patients who may not be able to read. For these patients, the phone call will be the appropriate choice."*

Theme 4: Convenience: mobile-assisted adherence intervention design

Content of the message

Some health professionals suggested to include information regarding how long the medication must be taken, the necessity of following the prescribed medication, and the consequences of not following the prescribed medication, the disease transmission, prevention, signs and symptoms, the diagnosis category of TB, how the medication should be taken, and nutritional-related information in the SMS text messages. The 28-year-old male TB focal person stated that

"Instead of only sending a reminder text message that tells the patient to take the medication, we can also add a message about the importance of adherence to medication and the consequences of non-adherence."

Another 32-years-old male TB focal person also suggested,

"Concerning the contents of the SMS (Short Messages Service) message, one of the components should be about the methods through which the disease is transmitted. Additionally, preventive methods such as ventilation and the use of a face mask should also be included. Information about the signs and symptoms of TB as well as the existence of types of TB could also be shared with patients through SMS messages. Advice about how the medication should be taken and about the types of nutrition the patient should take could also be included."

Some patients also suggested including information about TB and the consequences of nonadherence to TB medication. The 38-year-old female patient reported that the SMS text messages would be more useful if the SMS text messages include information about the disease and the consequence of non-adherence to TB medication and stated that

"If the intervention is going to be employed to inform patients about the disease, I believe that will be a useful solution too. That is because the attention that you give to taking the medication on time is dependent upon your awareness of its significance and the negative consequences of interrupting the medication or failing to take the medication on time. If people know the seriousness of the issue, they will pay attention to the medication. This is also dependent upon whether the physician has informed the patient about the disease and the medication. If patients are informed about the importance of taking the medication for the whole six months period and the danger of not taking the medication when they see a health improvement, then they will take it more seriously."

Language

Most health professionals suggested using the Amharic language (the national official language) for SMS text messages and phone calls adherence interventions. The health professionals reported that most of the patients can speak and understand the Amharic language in Addis Ababa, Ethiopia. The 31-year-old male TB focal person said that

"The vast majority of TB patients that I have come across in this health facility can speak Amharic. I have not come across a patient that does not speak Amharic. But some may not be able to read."

Another 39-year-old female TB focal person added that

"When it comes to language, it is better to use Amharic to compose the SMS message. That is the language that most people in the city can speak and understand. As far as language is considered, we have always been using Amharic. There is a small portion of patients that are not able to speak and understand Amharic always come with a relative who lives here and who speaks and understands Amharic."

However, most health care professionals also agreed that some patients cannot speak and read the official Amharic language, and thus suggested to include other major languages of the country to make the intervention more inclusive and successful. In this regard, the 47-year-old male TB prevention and control expert said that

"It should consider the language variations and the languages that are comfortable in communicating and best able to comprehend messages. It is possible to use a common language in the country, which is the Amharic language, but all people may not know that. If it is possible, it would be best to find the language preferences of the patients in the city. That is because people from various parts of the country with diverse cultural backgrounds come to the city, and thus they may not be fluent speakers of the Amharic language, even though it is the official working language. Adding Afan Oromo and other languages could be extremely useful."

Another 26-year-old female TB focal person from the health facility added that

"We need to prepare the contents of the SMS text messages or the phone calls in different languages. It may also be important to have callers who speak some of the major languages in the country. That is because even though most people in the city speak Amharic, they come from various parts of the country to get the treatment. These patients may not be fluent speakers of Amharic"

Most patients also agree that the Amharic language can be used to design SMS text messages and phone calls information. The 36-year-old male TB patient said that

"I believe the Amharic language should be used to compose the messages. That is the language that most patients in the city can understand." However, a few other patients suggested to include other major languages in addition to the Amharic language for the SMS text messages and phone calls. "I think that depends on the language that the patient can speak and understand. In the area that I am currently living, ..., there are people who are more comfortable speaking Afan Oromo than Amharic. For people who live in the city, Amharic would of course be the right choice."

Another 33-year-old male patient added that

"I think using Amharic and Afan Oromo would be useful. I do not think English will be relevant. If it is possible, other languages can also be included depending on the number of patients that is comfortable speaking that particular language."

Comprehension

Most health professionals agree that SMS text messages and phone calls designed to encourage TB medication adherence should be short and easy to understand in the way patient's educational level. The 45-year-old male TB focal person said that

> "When designing text messages, we need to make sure that we use a language that is easy to understand for an average person. People receiving the text message must have an easier time understanding the content of the text message. The SMS text messages that we send them must be as short and as easy to understand as possible."

Another 35-year-old female TB prevention and control expert added that

"One of the things with SMS text messages and phone calls content is that we should perhaps refrain from using medical and technical terms. To make sure patients understand everything that we want to convey in the messages, we should use language and words that they understand." Some patients also indicated that the message of the SMS text and the phone calls should be simple and understandable. The 36-year-old male patient said that "In addition, the message must be crafted using simple language and terminologies that the average patient can understand."

Interactivity

Some health professionals believe that a phone calls intervention added to SMS text messages can make the intervention more interactive and engaging than using two-way SMS text messages. The reasons for this, according to health professionals, were that phone calls are more trusted and more inclusive for those who are uneducated, it is a good means of communication to get feedback from patients about their experiences like drug side-effects and ask their health status. The 34-year-old male TB focal person said that

"I think, it (adding phone calls) is important to make it more engaging for participants. It would be more helpful if there was also a way to have patients' feedback and experience as they take the medication. It will be more acceptable for patients to make it more engaging so that they can share their experiences including the adverse reactions to the medication."

Another 26-year-old female TB focal person reported that adding phone calls in the SMS text messages could makes the communication more engaging and interactive, *"Keeping the phone calls in addition to the text message can make it more engaging. The phone calls allow for more discussion to happen. So, phone calls are more engaging than text messages. Designing the text messages to be more of a reminder for the patient to collect his medication would be impactful. Having both the phone call and the SMS text message could be useful and make it more engaging."*

Some health professionals suggested a two-way SMS text messages to make the intervention more successful and engaging. The 45-year-old male TB focal person said that *"If a resource is available, I believe that it (two-way SMS text messages) would make the intervention most*

successful. I think it has an advantage." Another 29-year-old female TB focal person suggested to design the SMS text messages in question-and-answer format that could increase the replay messages from the patients and it engage the patients in the communication, *"I think to do that, it may be important to consider including a question in the SMS text message that patients could respond to so that we would have more information on how they are taking the medication."*

However, some other health professionals had doubted the success of two ways SMS text messages because it requires the patients' ability operate their mobile phone, and the patient may be tired to reply to the SMS text message. The 32-year-old male TB focal person was guessed that 50% of the patients may replay to the text message if two SMS text message is implemented,

"I would say that there is only a 50% chance that it would be successful. That is because there are patients that are not literate and able to write a message. Even for literate people, they may be busy and unable to respond to the message."

Another 31-year-old male TB focal person said that

"It is possible to make it interactive by allowing patients to respond when they receive the reminder text message. But that will depend on whether the patients can type a text message and send it. For digitally literate patients, that is very much possible. For those that are not able to operate their mobile phone, then the reminder text message will be the only choice. Designing the intervention so that patients can reply can be weary for patients and we need to consider that."

Frequency

In order to promote adherence to TB medication, most health professionals recommended sending an SMS text message daily. Most health professionals also believed that phone calls should not be frequent because a daily phone call might disturb patients and might not be possible for many patients in health facility. Indeed, most of the health professionals suggest once a week or twice a week phone calls are adequate. The 32-year-old male TB focal person stated that

"Making a daily phone call may not be good. Patients will get tired of it if the phone calls are going to be made regularly and if it is going to be repetitive. You would need to prepare a schedule and decide the number of calls that you need to make per week. It may depend on the number of the patient. The SMS text messages could be sent as a daily reminder and the phone call could be less frequent in a week."

Another 47-year-old female TB focal person added that

"A daily phone call to the patient may not be appropriate. That is because it can disturb the patients. This is because patients may be on their job, and it can disturb them. Such a patient may not pick their phone at all. But the SMS text messages can be sent once a day."

However, some other health professionals believed that daily SMS text messages and phone calls could make the intervention more successful to promote TB medication adherence. For example, the 32-year-old male TB focal person believed that the daily SMS text messages and phone calls could make a more positive impact on TB medication adherence,

"I believe the higher the frequency of the phone calls and the text messages, then the higher their impact. People would be more alert, and they tend to expect to receive phone calls or text messages when we frequently send it to them. If sending the texts or making the phone calls daily is not possible, then we can either do it every week or every three days."

Some other health professionals also suggested twice a week for both text messages and phone calls to remind the patients to take their medication. The 50-year-old female TB focal person stated that

"I think it would be better not to send a message every day. We do not want to annoy the patients. We can send an SMS text message once every three days. Sending over the weekend could be useful too. People tend to relax at the weekend because they may be off work and sometimes away from home to visit a relative or a friend and they may forget to take the medication."

Another 42-year-old female TB prevention and control expert added that

"I am afraid it may be too frequent to send texts or make the daily phone calls. Twice a week would be good enough. Sending SMS text messages, or phone calls every two or three days will not be too much."

Some other health professionals reported that the frequency of the phone calls could be decided based on the patients' situation that can support them to remember their medication. The 29-year-old male TB focal person suggested,

"The phone calls can be made once a week. Additionally, we can also review the extent of family support for patients and make a varying number of phone calls for patients depending on it. For patients where there is strong family support that ensures patients are taken care of at home, we may need to call only once a week to assess adherence. For patients who do not have a family support system, we can make phone calls twice a week."

Another 39-year-old female TB focal person added that

"Making a phone call daily may not be necessary. Some patients also have their family members who remind them to take the medication. Also, calling patients daily in the morning may be inconvenient for them. In this regard, I do not see a problem with the SMS message."

Some other health professionals believed that the frequency of SMS text messages or phone calls could be determined based on the patient's treatment phases. The 30-year-old female TB focal person suggested, "for patients on the DOT phase, making daily phone calls and text messages are good. For patients in the continuation phase, I think once a week will be enough." Whereas some other health professionals believed that twice a week SMS text messages or phone calls is enough in the intensive phase of treatment, and three times a week is enough in the continuation phases of treatment. The 47-year-old male TB prevention and control expert suggested "If the DOT system is going to continue to be implemented, then phone calls or SMS texts could be sent twice a week for patients in the intensive phase of treatment. Otherwise, if there is a desire to discontinue the DOT and replace it with the current practice it should be done daily for patients in both phases of medication."

Most patients also believed that the daily SMS text messages or phone calls made for TB medication reminder would not be problematic. Most patients believed that the intervention comes from health professionals is for the benefit of the patients. The 48-year-old male patient reported that if patients are convinced about the aim of the intervention frequency would not be a problem,

"I do not see the problem with both (SMS and phone calls) being done daily. That would be a proper thing if the aim is to remind patients to take their medication. As long as patients are convinced about the aim of the intervention, I do not see any inconvenience that it will create." "I do not think people will complain about the frequency. The intervention is designed to support patients and I do not think they will complain about receiving too many texts or phone calls that remind them to take their medication."

Some patients, however, suggested making SMS text messages daily, but less frequently for the phone calls. The 25-year-old female patient suggested "I think the daily SMS message is a more appropriate reminder. The phone call would be more useful to remind patients about their refill medication dates." Another 33-year-old male patient added that "The reminder messages only need to be sent frequently because the medication should be taken daily." Some patients yet believed that weekly or monthly SMS text messages or phone calls frequency is enough. For example, a 22-year-old female patient stated that "I think sending SMS messages or phone calls once a week is an optimum frequency and I do not think patients will have a problem with this frequency." Another 26-year-old patient suggested "I would say once a month would be enough. That is because I think people will remember to take their medication on time once they are made aware of the serious consequence of failing to take the medication on time."

Time of delivery

All health professionals believe that the morning is the appropriate time for SMS text messages and phone calls for TB medication reminders because most patients take their medication in the morning. The health professionals reported that during the late morning, patients may go out for work or may be busy with other activities, which could make it difficult to reach them. The 32-year-old male TB focal person said that

"I prefer to send text messages every morning. TB patients take medication and other treatments early in the morning. Therefore, the right time will be either at 7:30 am or at 8:00 am. Some patients had to go to work after taking the medication at that time. The morning time will therefore be the most appropriate time to make the phone calls and to send the text messages."

Another 34-year-old male TB focal person added that

"I think it will be better if the phone calls are made in the morning. They take the medication in the morning. Therefore, the right time is to send the text or make the phone calls in the morning. Making it in the afternoon after they took the medication or after they forgot to take the medication is not going to be helpful."

Most of the TB patients also believed that the early morning time is appropriate for SMS text message or phone calls at the time of medication intake and before the patients go out for work or other personal activities. For example, the 37-year-old male patient said that *"I think the best time to send the SMS text messages and phone calls is in the morning or in a time that we know the patient will take the medication. If you send text messages at the time the patient should take the medication, it can increase the likelihood of the patient taking that medication. If a patient is taking the medication at 7:00 am, then we should send the medication."*

Duration of the Intervention

Most health professionals believed that the adherence intervention using SMS text messages and phone calls should be implemented for the whole six-month period of TB treatment. The reason for this was that the daily DOT (Directly Observed Treatment) was not implemented in the initial phase of treatment after the emergence of the COVID-19 pandemic, and patients become negligent when they feel well and may forget to take their medication or become lost from their treatment during the continuation phase. The 29-year-old male TB focal person said that

"I think it will be beneficial if we focus on both phases of the treatment because forgetfulness can happen in both phases. Despite health professionals providing all the counselling, negligence can always happen. Patients may become familiar with the routine of taking the medication daily in the beginning stages. But after the first two months, they may start to be negligent because they are taking the medication at home, and they may fail to take the medication in case they move to another town. But in general, I think 70% of the focus must be on the intensive phase and 30% of our effort should focus on the continuation phase. Or we can opt for a 65% focus on the intensive phase and 35% on the continuation phase makes more sense."

Another 31-year-old male TB focal person also suggested that adherence intervention should be implemented for the whole period of treatment because patients were taking medication at home for one month due to the COVID-19 pandemic. The participant also reported that the intervention could focus on the continuation phase if the standard DOT was reintroduced in the initial phase of treatment,

> "I believe it would be most impactful if we implement the intervention throughout the whole period of medication. Now, we are giving a one-month medication for patients in all phases of medication as part of a measure to reduce patients' exposure to COVID-19. Therefore, the intervention must be implemented throughout the whole medication period. If a decision is made to go back to using the DOT in which patients in the intensive phase come and take the medication at the health facility daily, then the focus of the intervention could be the continuous phase..."

Another 37-year-old female TB focal person suggested,

"The medication is taken for six months, and I believe sending reminder SMS text messages is important for the whole period. Especially the continuation phase is where healthcare professionals will not be able to contact patients and it is also a period where patients feel a sense of recovery. Therefore, the continuation phase of treatment is the time where there is likely to be nonadherence and it is important to pay more attention to this phase."

However, some health professionals disagreed about where the focus of the adherence intervention should be the intensive phase or the continuation phase of treatment. Some professionals believed that the intervention is most effective during the intensive phase, while others suggested that it would be more effective during the continuation phase. Health professionals who recommended the intervention in the intensive phase provided three main reasons: the patients' health status is crucial at this stage, patients may experience drug side-effects, and the standard daily DOT has not been implemented after COVID-19. The 50-year-old female TB focal person said that

"I think implementing it in the intensive phase of the treatment would be useful. That is because especially now patients are taking their medication at home due to COVID-19. In the future, it may be important to return towards focusing on the continuation phase."

Another 32-year-old male TB focal person added that

"I believe it would be better to focus on patients who are in the intensive phase of the treatment. More attention should be paid to this phase. But lesser attention should also be focused on the continuation phase. It is important to follow up with patients in this phase and make sure that they are strictly taking the medication. It is a phase of medication where patients experience adverse reactions and there is a need to closely monitor that as well."

The main reasons health professionals believe that the continuation phase of treatment should be emphasized were the patients become more diligent at the beginning of treatment and take their TB medication correctly, but when they feel well, they would start forgetting to take the medication. The 28-year-old male TB focal person said that "The focus of the intervention should be the last four months of the medication period. In the first two months, patients are more likely to be careful about taking the medication. During the intensive phase, patients will come to the health facility, and we observe them as they take the medication. It is in the four months after the intensive phase that patients tend to experience disruptions. Therefore, the focus must be four months after the intensive phase ends."

Privacy and confidentiality

Most health professionals do not believe that privacy issues would be a problem for SMS text messages and phone calls intervention done for TB medication adherence promotion. The 31year-old male TB focal person shared his experience,

> "My experience with TB patients' privacy is not a major problem. We can inform patients that text messages and phone calls are made to remind them to take their medication on time. If there is a patient who is concerned about not allowing their health status to be revealed to other people, they can easily delete the text message once they receive and read it."

Some health professionals reported that there has been some concern about the privacy issues associated with SMS text and phone calls when the goal of the intervention is to ensure ART adherence rather than TB medication. The 34-year-old male TB focal person said that

"Yes, that is true. But the focus of this intervention is TB. As long as we do not raise issues related to the patient's status concerning HIV/AIDS, I do not think it will be an issue. For example, I know about one patient who is taking TB medication and ART. When he wants to take the TB medication, he sends it to his family members. But to collect the ART medication, he will come. We do have similar experiences. So, I do not think we will have issues concerning privacy while the focus is on TB."

Another 32-year-old male TB focal person added that

"I would not say that will be a challenge. It would have been a challenge if the focus was HIV/AIDS. But with TB, other people such as the family members of the patient must be notified of the patient's health status because they must take the necessary preventive measures. So, I do not think privacy is going to be an issue. I have not come across a TB patient who said he is concerned about his privacy. We often inform the patient to inform their family members to take preventive measures. There is a small percentage of patients who come alone. They often come with a friend or a family member."

Another 39 years female TB focal person added that

"I do not think it is going to be a challenge. That is a challenge we experience with ART medication. But we have never had a problem with TB patients. But until patients become familiar with the intervention, they may not pick up their phones. But after they understand the aim of the intervention, I believe they would accept it and be grateful about it."

Some health professionals, however, believed that implementing SMS text messages and phone calls intervention would pose a privacy risk. This would be particularly more problematic for TB/HIV co-infected patients who do not want to receive SMS text messages or phone calls. The 29-year-old male TB focal person said that

"Yes, we may face those challenges. HIV AIDS patients may have a privacy issue with it. In addition, we have also come across patients who do not want their medical status to be revealed to their employer organization."
Another 37 years female TB prevention and control expert added that

"TB patients in our country may not wish to disclose their status to their communities for different reasons. This may be one of the things that should be taken into consideration when designing such interventions. It would be important to make sure that their identities and privacy are always kept confidential."

Another 50 years female TB focal person added that

"One of the challenges can be the issue of confidentiality. Some people do not want to disclose their status or to be at risk of exposure. People, depending on their specific circumstances, may not feel comfortable receiving a phone call or a text. That needs to be explored with the patients by having a conversation with them."

There was also a concern that sending SMS text messages or phone calls through the contact person could compromise the patient's privacy. The 35-year-old male TB prevention and control expert said that "*Privacy will be an issue if the text is sent to the patient through another person's phone. Otherwise, I do not think it will be an issue.*" Another 45-year-old male TB focal person added that "one of these scenarios is for example if the patient does not own a mobile phone and if we opt to reach that patient through a treatment supporter that means we are disclosing the patient's issues with another person."

Most patients did not perceive privacy as a problem when it came to SMS text messages or phone calls because the intervention is aimed at benefiting the patients. The 27-year-old male patient said that

"I do not have issues related to privacy. All my family members and friends are aware of my health status and the medication that I am taking. They have been supportive to me. I do not see any issue related to privacy for getting a message or a phone call that reminds me to take the medication."

Another 25-year female patient added that

"I do not think there will be any challenge that you could face. Currently taking TB medication is not something that people want to hide. So, I do not think there will be problems related to confidentiality."

Another 36-year-old male patient also said that

"I do not think it will be a problem. Patients can decline the calls and can be allowed to decline to receive the SMS messages as well if that is their desire. But I do think that will happen."

However, some patients reported that privacy issues could be a problem to some extent when it comes to sending SMS text messages or making phone calls. The patient respondents reported that some patients may want to keep their health status confidential from friends or family members, and therefore they may not want to receive phone calls or SMS text messages from the health facility. The 33-year-old male patient said that

> "Some patients may be worried about their privacy and about not revealing their health status to other people. Besides that, I do not think there will be other challenges."

Another 35-year female TB patient added that

"I do not see that as a challenge at least as far as I am concerned. But there may be some people who may not want their status to be revealed to anyone."

Theme 5: Barriers and facilitators of mobile-assisted adherence intervention

Facilitators of mobile-assisted adherence intervention

The health professionals mentioned that the willingness of TB focal persons to send SMS text messages and make phone calls, the high acceptance of mobile-assisted adherence intervention by patients, and the high coverage of mobile phones were opportunities to implement and sustain the adherence intervention.

Willingness and high acceptance

Health professional respondents from health office reported that the TB focal persons' willingness to adopt mobile-assisted medication adherence intervention was the opportunity to sustain the intervention. The 47-year-old male TB prevention and control expert said that

"We do have TB focal persons and communicable disease control case teams established at health facilities. Decisions could be made to use the focal persons or members of the case team to make the phone calls or to send and monitor the SMS text messages. Thus, it is possible to use the existing health system structure."

As long as the intervention is integrated into the health system, most TB focal persons working at health facilities had willingness to make phone calls, and to send and monitor SMS text messages. A mobile phone or landline telephone should be available in the TB clinic of the health facility, according to most health professional respondents. The 32-year-old male TB focal person stated that

> "We have been cooperative in making the phone calls. Sometimes they (patients) even leave a text for us to call them back and we would always volunteer and call them back and discuss whatever issue they are

experiencing. In addition, placing a landline phone or purchasing mobile phones to be used by the clinic can make things easier."

Another 31-year-old male TB focal person added that

"The first opportunity may be that health care professionals will be cooperative in the process of implementing this intervention. Since it is a program that can reduce workload, they will be willing to make phone calls."

Some health professionals also mentioned the high acceptance of SMS text messages and phone calls by patients as an opportunity for intervention success and sustainability. The health professional respondents reported that mobile-based SMS text message and phone call intervention can enhance a sense of professional care, which is highly acceptable to patients. The 47-year-old female TB focal person from the health facility stated that *"patients have a positive outlook overall. They are cooperative and respond when we ask them any question. I believe this is a positive thing that can facilitate this intervention (SMS and phone calls)."*

High mobile coverage

Some TB focal person respondents reported that a large number of patients have their own mobile phone which was seen as a good opportunity for mobile-based SMS text messages and phone calls. The 29-year-old male TB focal person reported that we can get at least one mobile phone in the household that we can use for the intervention even when the patient does not have their own mobile phone,

"We can find at least one phone owned by either the patient or a family member. Almost all patients own a phone. But some patients do not have their own mobile or do not operate their mobile phone because of their age or any other factor. But we can find one person within the family."

Challenges of mobile-assisted adherence intervention

Most health professionals reported the challenges for mobile-assisted adherence intervention were; limited infrastructure of the network interruptions may challenge the delivery of SMS text messages or phone calls, unable to charge mobile phones battery due to electricity interruptions, some patients may not own a mobile phone, healthcare professional workload, some patients may not pick up their phone, some patients may not read or understand SMS text messages, and several private companies send SMS text messages so that patients may ignore them.

Digital illiteracy and ignorance of messages or phone calls

Some health professionals reported that digital illiteracy and ignorance of SMS text messages may be the challenges for mobile-assisted adherence intervention. The 32-year male TB focal person said that

"Some patients may not read a text message sent to them in Amharic language. Another challenge that we may face with SMS text messages is the existence of several private organizations that send many SMS text messages... Most people ignore these text messages because they are not very useful. Here the challenge will be to make sure patients who receive the text messages will not be ignored these messages."

Another TB focal person with the same age and gender as above added that

"Concerning the SMS messages, a possible challenge is that everyone receives several messages from different organizations, and we do not always care to check the content of these messages. Patients may be busy with their work sometimes and they may not get the time to check their message at the time that you want them to do. In addition, not all patients can read the messages."

The 53-year-old female TB prevention and control expert reported that some patients may not like to receive too many SMS text messages and phone calls, "One of the challenges may be that patients may be tired of receiving these calls and text messages. Patients may not like having too many phone calls and text messages."

There were some common challenges reported by patients, such as uneducated patients being unable to understand and read text messages or operate their mobile phones. Patients also reported receiving multiple SMS messages sent from a private company that posed a problem, and some patients did not know the SMS messages are from the health facilities. The 33-year-old male patient said that

"One thing is that people are extremely irritated by the various SMS messages that they receive from various sources that typically use four-digit numbers. I am afraid some patients may consider the SMS text that you send them as part of this intervention as one of those other messages."

Another 35-year-old female TB patient added that

"Patients may assume that the message is sent from the telecom company and tend to ignore it. People do not tend to check these messages when they are received from the telecom company or from other firms that are not useful to them. I am afraid there could be a chance that patients may consider the message that you are going to send them as one of those SMS text messages. But it depends on the habit and behaviour of the person that actively checks the messages."

"I think it can be useful. But my concern about that would be how many people would be able to properly operate a mobile phone. I see there are a lot of elderly patients when I go to the health facility to get the medication. These elderly patients do not seem to be able to fully operate a mobile phone and respond to SMS text messages. But for other patients such as myself who can effectively use a mobile phone for communication..."

In some cases, patients with busy schedules or other activities may not read the text messages. The 26-year-old female patient said that "That (reading the SMS text) depends on the patient and their way of life. For instance, I am always busy working in my office. As a result, I am not always able to read the text messages that I receive. I may see the notification and I may forget to read the text until I get home at night. But overall, it is important to send patients a text message to remind them to take the medication." Another 27-year male patient added that "One of the challenges may be that patients may not always pick their phone. Sometimes patients may be at their work or handling a private matter and they may not see their phones ringing. This is the only challenge that you may experience. If patients are made aware of the phone calls and the SMS text messages when they start taking the medication, the extent of the problem will be lesser."

Have no own mobile or telephone

Some health professionals reported that some patients may not have mobile phones and their contact person or family member may be busy with their work to get the patient with them which may challenge the intervention. The 58-year-old female TB focal person said that

"One of the challenges may be there are some patients who do not own a mobile phone. We are only able to reach them through their contact person. The contact persons may be busy because they may be engaged in job. They may also not be happy when you repeatedly call them (contact person)."

Another 32-year-old male TB focal person added that

"One of the challenges may be that there are some patients who do not own a mobile phone. In addition, we may also face network interruptions. This is a rare problem that we experience. It does not happen often. We are always able to contact the patient or the contact person. Such a problem is likely to happen if the patient does not have their own mobile phone. Patients who are homeless do not own mobile phones nor do their contact person have their own mobile phone. Even this is a rare case. For instance, in the past year, I have come across only two patients who said they do not have their own mobile phone."

Another 42-year-old female TB prevention and control expert added that

"I believe it will be useful. But my question is how many patients own phones and how many of them can read a text message. Overall, a text message can be a useful reminder."

Most TB clinics in a health facility did not have their own telephones, nor did health professional have airtime cards to make phone calls or send SMS text messages, which could make it difficult to sustain adherence intervention using SMS text messages and phone calls. Most TB focal persons, however, reported that there was a telephone in another room of the health facility which was not convenient for them because they needed to go to other rooms to call patients. The 32-year-old male TB focal person said that

"I use my own mobile phone to make the phone call. I also spend my own money to make the phone call. No finance is set aside for us to purchase airtime and make the phone calls. But since they are our patients, we have been cooperative in making the phone calls."

Another 50-year-old female TB focal person added that

"We do not even have a landline telephone in the TB clinic room. I must go to another office to make the phone call. Sometimes I am using my own mobile phone to make the calls."

Patient's registered wrong phone number or change their phone number

Some health professionals also reported that some patients provided the wrong phone number, which was recorded in the medical chart abstraction, making it difficult to trace the patients when they were lost from treatment. The 58-year-old female TB focal person said that

> "Yes, that has happened to me before on a limited instance. I have only come across one patient. The patient was a female homeless person...Her phone number never worked when I try to reach her. After multiple trials, the contact person was reached over the phone, and we were able to get the patient to return the patient to treatment. But it is true that patients will give you the wrong phone number."

Another 45-year-old male TB focal person added that

"The first challenge is that you may find out that patients' phone numbers may not work at all. We have experienced this challenge as well when trying to reach patients. Patients may give you a phone number that may not be functional. The option, in this case, may be to record the treatment supporter's phone number and try to reach the patient through the treatment supporter. We do not call them right when we receive their number to see if it works. We only ask the patient if the phone works or not. They will always tell you that their phone number is active and that it works. But when you try to reach them at a later stage during treatment, you find that some patients' numbers do not work."

Some health professionals also reported that patients may change their phone number without informing them which challenge to trace the patients when they lost from the treatment. The 32-year-old male TB focal person said that

"Sometimes patients change their phone numbers, and it could be a challenge to be able to reach them. We also take the phone numbers of patient supporters. But at times the patient supporter may be away, and they may not be immediately available for the patient as well. This could be a challenge for such intervention."

Another 30-year-old female TB focal person reported that TB/HIV co-infected patients were change their phone number more frequently and said that

"The challenge that we experienced is that they frequently change their phone numbers. Even after telling them to notify us if they change their phone number, they still change their phone number without telling us. We checked if it works, but they still changed the number. At times, the number may be disabled later. This is a challenge that we experience concerning patients taking TB medication and ART."

Another 39-year-old female TB focal person also said that

"Some of them do not always give you their correct phone number. They may also change their number at a later stage of medication. Since we know that problem, we always take the time to inform the patient to give us a phone number that they will use constantly and not another phone number."

Health professionals work burden

Some health professionals mentioned that the health professionals may also be burdened with phone calls and SMS text messages if they have the responsibility to operate it. The 53-year-old female TB prevention and control expert said that

"Concerning health care professionals, if the plan is to use them to make phone calls, a challenge is that many of them may be overburdened with other responsibilities.

Another 31-year male TB focal person added that

"The first challenge may be that it could be an additional burden of responsibility for health care professionals working at TB clinics if the plan is to deploy them to make the phone calls."

Another 29-year-old male TB focal person also said that

"That (the phone calls) may be a difficult thing for us. Currently, we have about 55 patients. It may be challenging to make phone calls to all patients daily."

Sustainability of the intervention

Some health professionals had a concern on the sustainability of the adherence intervention. The 34-year-old male TB focal person said that "One of the challenges is the sustainability of such program or intervention. The problem with such programs has to do with their short-term existence which can exacerbate the problem instead of addressing the issue they want to tackle." Another 37-year-old TB prevention and control female expert added that "Sustainability of the intervention is another issue to be considered. If the intervention is carried out by a non-governmental organization that implements the program only for a certain period, or if the organization decides to stop the program, it could have an adverse effect. Therefore, there must be a system where it could be implemented sustainably. To ensure its sustainability we would need to think about having a proper budgeting system. Budgeting and finance should be seen as one of the potential challenges that we may experience in the process of

implementing this program and we can also view it as one of the basic features of its management."

Suggested solutions

Availing telephone at TB clinic

To sustain the adherence intervention, health professionals advised purchasing a telephone or provide airtime cards to health professionals. The 29-year-old male TB focal person said that

> "We need to ensure necessary conditions are met to implement such a program (SMS text message and phone calls). For instance, one of the things has to do with the telephone. As I have mentioned we have to go to another office to make phone calls to patients."

Another 32-year-old male TB focal person added that

"Yes, that is one of the challenges that we are experiencing, and I believe covering the airtime cost would solve the problem. We can use mobile phones if the airtime cost is covered. But so far, we have been forced to go to the ART room to make calls to TB patients."

Briefing the mobile-assisted adherence intervention

Health professionals suggested having clear communication with the patient about the mobileassisted adherence intervention. The patient needs to be clearly informed about the SMS text messages and phone calls aim which is necessary to avoid problems related to patients' privacy and confidentiality. The 47-year-old female TB focal person said that

> "What I would like to add here is the patients should first be aware of the plan for phone calls and sending a text message to them. This is because some patients may want to keep their medication status private, and they may not

want another person or even a family member to be made aware of their situation."

Patient's consent is also crucial to avoid privacy problems and respect their self-determination, according to health professionals. The 28-year-old male TB focal person stated that "...We should start the implementation by first getting patients' consent. If we get their consent, then we will be clear to implement the intervention..." Another 53 years female TB prevention and control expert said that "Firstly, we should get patients' consent and willingness to receive a phone call or a text message that reminds them to take their medication. We should notify them before starting the intervention and we should secure their consent. Some patients may be keeping their status of TB patient as a secret, and we should make necessary preparations to ensure the confidentiality is always respected." Another 37-year-old female TB prevention and control expert added that "Yes, confidentiality is an issue of consideration when designing the intervention. When designing the content of the text message, the expert is responsible for designing the text message, and the patient must discuss and agree on the proper way they preferred to be reminded of their medication."

Health professionals also suggested communicating the patient's preferred time for phone calls and SMS text messages will reduce unanswered calls and increase the readability of the SMS text message. The 29-year male TB focal person said that

"They answer the phone calls because we call them on their private phones. We will need to consider the time of the phone call. We may need to refrain from calling them during working hours. We can speak to them and ask for the proper time they feel comfortable to receive a phone call. We will also need to ask them if the phone will be with them at the time of texting. I have only experienced such issues with two patients (not answering the phone calls) so far."

For the intervention to be effective, some patients also report that clear communication and consent are important. For example, the 33-year male patient said that *"I believe getting their consent is important. What I have not mentioned so far is that when you start the implementation, the target patient needs to be made aware of the intervention and all necessary information about the intervention."*

Using a family member's or caregiver's mobile phone

Some health professionals from the health facility suggested using the phone number of a caregiver or family member when the patient has no mobile phone or cannot read and understand messages. The 45-year-old male TB focal person said that

"Yes, I think it can be one of the solutions that we can employ. We keep patients' contact information including their phone number. For patients who do not own a mobile phone, we also keep the treatment supporter's contact information. Sending a reminder text message for the patient himself or treatment supporter can be an extremely useful solution to patients who may forget to take their medication on time and who forget the health facility appointment dates."

Another 50-year-old female TB focal person added that

"Yes, it is one of the solutions that, I think, can have a positive impact. But a major challenge may be that patients may not always own a mobile phone. To track patients when they missed the refill visit appointment, we often take two or three phone numbers owned by the patient and another person who is close to the patient that we can reach out to if the patient's phone is not working."

Additionally, some patients suggested that family members can be involved when a patient cannot read or understand a text message. The 37-year-old male patient said that

"There are some patients who may not be able to read and understand a text message. For these patients, you can inform them to get another person such as their family member to read the text to them. But if the text message is going to be written in Amharic, I believe the majority of patients can read."

Another 27-year male patient added that

"I do not think there are people today who do not read texts. Even if there are people who are unable to read, there are treatment supporters available that can help the patient on the SMS text messages. There may be elderly patients that may not be able to read a text message and it may be better to contact their treatment supporters. But overall, we have made improvements regarding basic literacy, and I think most of the people are literate."

Making a voice message or phone calls to an illiterate

In addition, some health professionals suggested using phone calls, voice messages, and treatment supporters for those who cannot read and understand SMS text messages. For example, the 53-year female TB prevention and control expert stated that

"A possible solution for illiterate patients is to send a voice message ... In addition, we can also use treatment supporters who are literate and able to read a text message. You can provide orientation to treatment supporters and use them to deliver the contents of the SMS text messages to patients."

Additionally, some patients also reported that phone calls are appropriate for those patients who are unable to read and understand SMS text messages. The 25-year-old patient suggested using a phone call when unable to read text messages, *"There are patients who may not be able to read a text message. So instead of a text message, a phone call may be more appropriate. Yes, not all patients may be able to read a message. In addition, some people will also ignore a text message sent to them from a phone number that they do not*

know even if they see a notification. For these reasons, I think the phone call may be a better option."

Confirming and updating the patient's phone number

Some health professional respondents suggested immediately checking whether the patient's mobile phone number is working when the patient provides it to the health professionals. Patients should be made aware of the reason why their phone number is required. The 47-year-old female TB focal person shared her experience,

"When I take patient's phone number, I will call back them immediately while they are in front of me, and I confirm if the number they gave me is working. When taking their phone number, we provide them with all the information as to why we need their phone number and why we give them our phone number too. We inform them that they can call us at any time if they experience any problems. Some of them ask for my phone number on their own."

Another 29 female TB focal person said that

"When we take their contact information, we first check the mobile phone is working. In addition, we also take additional phone number for the contact person who is most time a family member or a friend of the patient."

The 53-year-old female TB focal person also said that

"Sometimes, I call back the phone numbers right after registering to check the phone number is working. So, making sure that the phone number is functional is an important thing."

The 30-year female TB focal person said that

"We check if it (the phone number) works. But they still change the number. At times, the phone number may be disabled later. This is a challenge that we experience with patients taking TB medication and ART. You must continuously update their contact information as they change their phone number."

Giving the project a unique name

Some health professionals suggested using a unique code or project name for the short number that sends SMS text messages and makes phone calls, in order to distinguish the messages sent from the health facility from other messages. This could help patients to easily identify and read the text messages. The 35-year-old male TB prevention and control expert said that

> "The text message that we want to send to the patients must be distinct from the rest of text message services. There are many text messages services that the public receives without requesting to receive them. As a result, many people tend to ignore these text messages. We must make sure that we should make the text message for this program is distinct so that patients will not ignore it."

Some patients also suggested a way to differentiate the SMS text messages and phone calls made from the health facility from those made by other organizations. In this regard, the 33-year-old female patients said that

"I do not know how but there needs to be a way for you to differentiate the SMS messages that you plan to send to patients from the other messages sent from different organizations...People do not even bother to check these messages. When we see a mobile SMS message sent from a four-digit number, we usually do not even check the content of the message. So, it will be better to ensure that patients will see a unique identifier instead of four-digit numbers."

APPENDIX THREE: PROTOTYPE SMS TEXT MESSAGES FOR MA-MAS INTERVENTION USING BEHAVIOUR CHANGE TECHNIQUES TAXONOMY AND IMB MODEL

Table 33: Prototype SMS text for Ma-MAS Intervention using Behaviour Change Techniques (BCT) taxonomy (Michie et al. 2013) ⁽¹⁵⁸⁾ and IMB model ⁽¹⁶⁰⁾

Code	BCT	Definition	SMS Text content	IMB
1	Goals and Planning			
1.1	Goal setting (Behaviour)	"Set or agree a goal defined in terms of the behaviour to be achieved."	A person infected with TB must take their medications every day at a regular time for at least six months to be cured. Please plan to take all your medications with no interruptions	АМ
1.2		"Analyse, or prompt the person to analyse, factors influencing the behaviour and generate or select strategies that include overcoming barriers and/or increasing facilitators"	Why do some patients forget to refill their medication? You can come to refill any time before you run out of medications. Keep your calendar card in a place where you can see it	ABS
1.2.1	Problem-solving		Does feeling well from TB symptoms cause forgetfulness about medications and lead people to discontinue medications TB medications must be taken for at least six months to be cured	ABS
1.2.3			Some patients miss their regular medical check-up. If the appointment scheduled is not comfortable to you, please let know to the healthcare provider to reschedule	ABS
1.2.4			Please ask your family member to pick up your medications for you if you are not able to come to the clinic to refill your medications	ABS

1.3	Goal setting (outcomes)	"Set or agree a goal defined in terms of a positive outcome of wanted behaviour"	If you want to be fully cured from TB, you must take all your medications for at least six months. Please do not forget to come to refill your medication supply	AM
1.4	Action planning	"Prompt detailed planning of performance of the behaviour (must include at least one of context, frequency, duration and intensity). Context may be environmental (physical or social) or internal (physical, emotion or cognitive) (includes 'implementation intentions')"	Do you experience difficulties to remember to take your medications every day at the same time as directed? Ask a family member or friend to help you remember	ABS
1.6	Discrepancy between current behaviour and goal	"Draw attention to discrepancies between a person's current behaviour (in terms of the form, frequency, duration, or intensity of that behaviour) and the person' previously set outcome goals, behavioural goals or action plans (goes beyond self-monitoring of behaviour)."	Consider some form of self-tracking such as a medication calendar to mark the number of consecutive days that you are adhering. How many days in a row have you taken your medication as prescribed?	ABS
1.7	Review of outcome goal(s)	"Review outcome goal(s) jointly with the person and consider modifying goal(s) in light of achievement. This may lead to resetting the same goal, a small change in that goal or setting a new goal instead of, or in addition to the first."	Consider discussing how your health is improved while you are appropriately adhering your medication. Tell them if you missed any dose of medications	AM
1.8	Behavioural contract	"Create a written specification of the behaviour to be performed, agreed by the person, and witnessed by another"	Every day when you wake up in the morning, remember to take your medication before your breakfast. Consider asking a family member to remind you	AM

1.9	Commitment	"Ask the person to affirm or reaffirm statements indicating commitment to change the behaviour."	For how many days have you correctly taken your medication as prescribed? To be cured from TB, you must take all your medications for at least six months	AI
2	Feedback and monitoring			
2.2	Feedback and monitoring	"Monitor and provide informative or evaluative feedback on performance of the behaviour".	Thank you for picking up your medication as per your scheduled date. Keep it up, and take your daily medication as directed	AM
2.3	Self-monitoring of behaviour	"Establish a method for the person to monitor and record their behaviour(s) as part of a behaviour change strategy."	Plan ahead to visit the health facility to refill your medication supply. Use a calendar card to check off the days when you have taken your medicine	ABS
2.4	Self-monitoring outcome of behaviour	"Establish a method for the person to monitor and record the outcome(s) of their behaviour a part of a behaviour change strategy."	You can monitor your body weight and health improvement with appropriate medication adherence. Keep taking your medication as prescribed	ABS
2.5	Monitoring of outcome(s) of behaviour without feedback	"Observe or record outcomes of behaviour with the person's knowledge as part of a behaviour change strategy."	Plan with healthcare provider to get sputum test and to regular health check-up during treatment while you are adhering your medication	ABS
2.6	Biofeedback	"Provide feedback about the body (e.g physiological or biochemical state) using an external monitoring device as part of a behaviour"	Your healthcare provider monitors your health progress to check the medication is working. Keep up taking your daily medications as directed unless you are told to stop by a healthcare provider	AI
2.7	Feedback on outcome(s) of behaviour	"Monitor and provide feedback on the outcome of performance of the behaviour"	Do you know that those who adhere will have a better chance to be cured? If you missed any dose of TB medications, please tell the healthcare provider	AM

3	Social support			
3.1	General	"Advise on, arrange or provide social support (e.g. from friends, relatives, colleagues, or staffs) or non-contingent praise or reward for encouragement and counselling, but only when it is directed at the behaviour."	We are here to better serve you. Please tell your health care provider whether you correctly adhere or not	AM
3.2	Practical	"Advise on, arrange, or provide practical help (e.g from friends, relatives, colleagues, or staffs) for performance of behaviour"	Please tell the healthcare provider if you are traveling somewhere away from home. They can give you extra medications or link you with another healthcare worker, so you will not run out of medications	ABS
3.2.1	_		Please tell us if you are feeling ill while you are on your medications. Ask the healthcare provider about the common side effects	ABS
3.3	Emotional	"Advise on, arrange, or provide practical help (e.g from friends, relatives, colleagues, or staffs) for performance of behaviour"	Your health status is our priority. Please Keep in touch with your healthcare provider and family support team for help when needed	AM
4	Shaping knowledge			
4.1	Instruction on how to perform a behaviour	"Advise or agree on how to perform the behaviour (include 'skill training')"	It is common to experience some side effects while you are taking TB medication. Please tell to your healthcare provider if you experience mild side effects	ABS
4.2	Information about Antecedents	"Provide information about antecedents (e.g. social and environmental situations and events, emotions, cognitions) that	People tend to forget to take their medications when they are away from home. Please remember to have an adequate amount of medications with you when you are away from home	AI

4.21`		reliably predict performance behaviour"	of the	TB can be cured in almost all cases by taking the medications as prescribed without interruption	AI
4.2.2				Some patients will stop taking TB medications when they experience side effects. Please tell the health care provider if you have any side effects	AI
4.2.3				TB is not caused by exposure to cold air, insect bite, by opening the door/window at home or public transport. It is caused by a type of bacteria, Mycobacterium Tuberculosis	AI
4.2.4				TB is caused by Mycobacterium Tuberculosis that is not seen by eye. It is only cured by taking at least six months of medications	AI
4.2.5				TB is transmitted from one person to the other through inhaling air when infected person cough or sneezing or talking	AI
4.2.6				A cough that lasts 2 weeks or more, chest pain, coughing up blood or sputum, weakness or fatigue, and weight loss are the most typical signs of Tuberculosis	AI
4.2.7				Please bring your household members to a health facility if you see any TB symptoms on them. Early diagnosed TB will be effectively cured	AI
4.3	Reattribution	"Elicit perceived causes of behavi suggest alternative explanation external or internal and stable or un	is (e.g	Please remember that TB bacteria die very slowly. Simply because you are feeling well does not mean you have been	AI

			cured. You must maintain regular treatment for at least 6 months to be cured	
5	Natural consequences			
5.1	Information about Health consequences	"Provide information (e.g written, verbal, visual) about health consequence of performing the behaviour"	Be aware that if you are unable to adhere to TB medication regularly as directed, this can lead to failed treatment, relapse, and multidrug resistance	AM
5.2	Salience of consequences	"Use methods specifically designed to emphasise the consequence of performing the behaviour with the aim of performing them more memorable (goes beyond informing about consequences)"	A person with TB who takes all the medications as directed will be cured and will not be a risk of TB transmission to his/her family and other communities	AM
5.3	Information about social and environmental consequences	"Provide information (e.g written, verbal, visual) about social and environmental consequences of performing the behaviour"	Untreated TB can be transmitted to close family members, friends and other people. Please take your medications until the end of the treatment period as per prescription	AM
5.4	Monitoring of Emotional consequences	"Prompt assessment of feeling after attempts at performing the behaviour."	Did you take your medications at the same time every day? If you did, congratulations! If you didn't adhere, try again, but tell your healthcare provider the doses that you missed	AM
5.5	Anticipated regret	"Induce or raise awareness of expectations of future regret about performance of the unwanted behaviour"	Do you know that if TB treatment fails, you will become sick again, and you may transmit TB to loved family members and friends? Treatment fails if a patient is not adhering to medications	AM
5.6	Self-assessment of affective consequences	"Provide information (e.g written, verbal, visual) about emotional consequence of performing the behaviour"	Imagine what you would feel if you had a multi-drug resistant TB because of non-adherence to medications. It would need long-term treatment and may not ever be cured	AM

6	Comparison of behaviour			
6.1	Demonstration of the behaviour	"Provide an observable sample of the performance of the behaviour, directly in person or indirectly."	Consider using a strategy like setting an alarm in your mobile phone, or tell a family member to remember your daily TB medication and refill visit schedule	AM
6.2	Social comparison	"Draw attention to others' performance to allow comparison with the person's own performance."	I know some patients with TB who adhere to their medications. Adhered Patients have a higher chance of cure than non-adhered patients	AM
6.3	Information about other approval	"Provide information about what other people think about the behaviour. The information clarifies whether others will like, approve or disapprove of what the person is doing or will do."	Your healthcare professional, families and friends support you to take your everyday TB medications as per directed. Together you will succeed!	AM
7	Associations			
7.1	Prompts/Cues	"Introduce or define environmental or social stimulus with the purpose of prompting or cueing the behaviour. The prompt or cue would normally occur at the time or place of performance"	You must decide the first task in the morning is always to take your daily medication. Keep taking your daily medications at the same time every day as directed	AI
7.2	Cue signalling reward	"Identify an environmental stimulus that reliably predicts that reward will follow the behaviour"	Good morning! Before you eat your breakfast, please take your daily medication	AI
7.4	Remove access to the reward	"Advise or arrange for the person to be separated from situations in which unwanted behaviour can be rewarded in order to reduce the behaviour"	Cigarette smoking during TB treatment can lower your treatment success. Avoid people or places that make you want to smoke as much as possible	AI

7.6	Satiation	"Advise or arrange repeated exposure to a stimulus that reduces or extinguishes a drive for unwanted behaviour"	When patients drink alcohol while taking TB medications they would increase the side effect of the drug so that most people want to avoid alcohol during treatment	АМ
7.8	Associative learning	"Present a neutral stimulus jointly with a stimulus that already elicits the behaviour reputedly until the natural stimulus elicits that behaviour"	Some patients take their medication in the morning immediately after they wash their faces. Keep taking your daily medication	AI
8	Repetition and substitution			
8.1	Behavioural practice/rehearsal	"Prompt practice or rehearsal of the performance of the behaviour one or more times in a context or at a time when the performance may not be necessary, in order to increase habit and skill"	Please remember to take your medication regularly every day at the same time. Please keep your medications in a safe place where you can see it	ABS
8.3	Habit formation	"Prompt rehearsal and repetition of the behaviour in the same context repeatedly so that the context elicits the behaviour"	Please take your medications every day at the same time in the morning. This helps you to remember to take your medications regularly	ABS
8.6	Generalization of target behaviour	"Advise to perform the wanted behaviour, which is already performed in a particular situation, in another situation."	Please keep taking your daily medications regularly until the end of your treatment. TB is a curable disease if you are fully adhering to treatment	AM
9	Comparison of outcomes			
9.2	Pros and cons	"Advise the person to identify and compare reasons for wanting (pros) and not wanting to (cons) change the behaviour."	Adhering to TB medications for a long period is not easy. Sometimes it has side effects. But not bad compared to getting a resistance TB and become a risk of transmission to families	AI

11	Regulation			
10.11	Future punishment	"Inform that future punishment or removal of reward will be a consequence of performance of an unwanted behaviour (may include fear arousal), includes 'threat'."	Do you know that non-adherence to medications may cause multi-drug resistant TB that may need complex treatment for a long time, has more side effects, and may not work or allow you to be cured	AM
10.10	Reward (outcome)	"Arrange for the delivery of a reward if and only if there has been effort and/or progress in achieving the behavioural outcome (includes 'positive reinforcement')"	You are a hero. We appreciate your courage and commitment to adhere and pick-up your medications regularly schedule dates from clinics	AM
10.9	Self-reward	"Prompt self-praise or self-reward if and only if there has been effort and/or progress in performing the behaviour."	Plan celebrate your medication adherence commitment at every 2 months of TB treatment progress until the end of treatment	AM
10.7	Self-incentive	"Plan to reward self in future if and only if there has been effort and/or progress in performing the behaviour"	Plan to celebrate your completion of TB treatment with your families and friends together. Keep taking your daily medication	AM
10.4	Social reward	"Arrange verbal or non-verbal reward if and only if there has been effort and/or progress in performing the behaviour (includes 'positive reinforcement')"	Thank you for picking up your medications. Please remember to take your daily medications at the same time every day	AM
10	Reward and threat			
9.3	Comparative imagining of future outcomes	"Prompt or advise the imagining and comparing of future outcomes of changed versus unchanged behaviour."	Do you know that patients who fully adhere to their TB medications are more likely to get cured than those who don't adhere to their TB medications?	AI/AM

11.1	Pharmacological support	"Provide, or encourage the use of or adherence to, drug to facilitate behaviour change"	Please ask us about your medications and common side effect. Your health is always our first concern.	AM
11.2	Reduce negative emotions	"Advise on ways of reducing negative emotions to facilitate performance of the behaviour (include 'stress management')"	If you have experienced any difficulties during treatment, please do not afraid (fear) to ask for help.	AM
12	Antecedents			
12.3	Avoidance/reducing exposure to cues for	"Advise on how to avoid exposure to specific social and contextual/physical cues for the behaviour, including changing daily	Please remember to keep your medications with you anytime when you are away from home	AI
12.3.1	the behaviour	or weekly routines."	Please remember to avoid alcohol and smoking that will affect your TB treatment success	AI
12.6	Body changes	"Alter body structure, functioning or support directly to facilitate behaviour change"	You will gain body weight and become healthy if you appropriately adhere to your TB medications and lead a healthy lifestyle	AM
13	Identity			
13.1	Identification of self as role model	"Inform that one's own behaviour may be an example to other"	Be a role model and save others from TB transmission by fully adhering to your TB medications regularly to the end of treatment	AM
13.3	Incompatible beliefs	"Draw attention to discrepancies between current or past behaviour and self-image, in order to create discomfort (includes 'cognitive dissonance')."	Do you want to be free from TB disease and be healthy? If so, are you taking your everyday medication as per directed?	AM

13.5	Identity associated with changed behaviour	"Advise the person to construct a new self- identity as some who 'used to engage with the unwanted behaviour"	Do you remember those days when you had a hard time sticking with treatment—that wasn't really you? Now, remember all the times you stayed on track—that's who you are!	AM
14	Scheduled consequences			
14.4	Reward approximation	"Arrange for reward following any approximation to the target behaviour, gradually rewarding only performance closer to the wanted behaviour (includes 'shaping')"	You are almost near to finish your TB treatment. Please contact us for more information about your treatment status.	AM
14.5	Rewarding completion	"Build up behaviour by arranging reward following final component of the behaviour; gradually add the components of the behaviour that occur earlier in the behavioural sequence (includes 'Backward chaining')"	Congratulations! You are completed your TB medications. Please contact us for further health examinations.	AM
15	Self-belief			
15.1	Verbal persuasion about capability	"Tell the person that they can successfully perform the wanted behaviour, arguing against self-doubts and asserting that they can and will succeed."	You have the capacity to fully adhere to all your TB medications every day to be cured. Yes, you can do it!	АМ
15.2	Mental rehearsal of successful performance	"Advise to practice imagining performing the behaviour successfully in relevant contexts."	Imagine that you can remember to take your daily TB medications to the end of treatment. Your imagination will become true!	AM
16	Covert learning			

16.1	Imaginary punishment	"Advise to imagine performing the unwanted behaviour in a real-life situation followed by imagining an unpleasant consequence (includes 'covert sensitisation')."	My worst enemy will laugh at me if I fail to adhere my medication and sick again. My dearest family will cry if I fail to adhere my medication	АМ
16.2	Imaginary reward	"Advise to imagine performing the wanted behaviour in a real-life situation followed by imagining a pleasant consequence (includes 'covert conditioning)"	Imagine that you can remember to take your daily TB medications to end of treatment and get cured. Your imagination has the power to become true!	АМ
16.3	Vicarious consequences	"Prompt observation of the consequences (including rewards and punishments) for others when they perform the behaviour"	I know that there is someone just like you who has stuck with their TB treatment successfully.	АМ

Am: Adherence Information , AM: Adherence Motivation , ABS: Adherence Behavioural skills

APPENDIX FOUR: VALIDATED SMS TEXT MESSAGES USING BEHAVIOUR CHANGE TECHNIQUES AND IMB MODEL

Table 34: Validated SMS text for Ma-MAS Intervention using Behaviour Change Techniques (BCT)taxonomy (Michie et al. 2013)⁽¹⁵⁸⁾ and IMB model (Fisher et al. 2006) ⁽¹⁶⁰⁾

S.NO	вст	IMB	SMS text messages English Version
1	Commitment	Information	For how many days have you correctly taken your medication as prescribed? To be cured from TB, you must take all your medications for at least six months
2	Information about Antecedents	Information	People tend to forget to take their medications when they are away from home. Please remember to have an adequate amount of medications with you when you are away from home
3	Information about Antecedents	Information	TB can be cured in almost all cases by taking the medications as prescribed without interruption
4	Information about Antecedents	Information	Some patients will stop taking TB medications when they experience side effects. Please tell the health care provider if you have any side effects
5	Reattribution	Information	Please remember that TB bacteria die very slowly. Simply because you are feeling well does not mean you have been cured. You must maintain regular treatment for at least 6 months to be cured
6	Prompts/Cues	Information	You must decide the first task in the morning is always to take your daily medication. Keep taking your daily medications at the same time every day as directed
7	Cue signalling reward	Information	Good morning! Before you eat your breakfast, please take your daily medication
8	Associative learning	Information	Some patients take their medication in the morning immediately after they wash their faces. Keep taking your daily medication
9	Avoidance/reducing exposure to cues for the behaviour	Information	Please remember to keep your medications with you anytime when you are away from home

	Avoidance/reducing		
10	exposure to cues for the	Information	Please remember to avoid alcohol and smoking
	behaviour		that will affect your TB treatment success
11	Demous economics to the		Cigarette smoking during TB treatment can lower
	Remove access to the	Information	your treatment success. Avoid people or places that
	reward		make you want to smoke as much as possible
	Goal setting (Behaviour)	Motivation	A person infected with TB must take their
10			medications every day at a regular time for at least
12			six months to be cured. Please plan to take all your
			medications with no interruptions
			If you want to be fully cured from TB, you must take
40			all your medications for at least six months. Please
13	Goal setting (outcomes)	Motivation	do not forget to come to refill your medication
			supply
ļ			Every day when you wake up in the morning,
		Motivation	remember to take your medication before your
14	Behavioural contract		breakfast. Consider asking a family member to
			remind you
	Feedback and monitoring		Thank you for picking up your medication as per
15		Motivation	your scheduled date. Keep it up, and take your
			daily medication as directed
	Feedback on outcome(s) of behaviour	Motivation	Do you know that those who adhere will have a
10			better chance to be cured? If you missed any dose
16			of TB medications, please tell the healthcare
			provider
	Emotional	Motivation	Your health status is our priority. Please Keep in
17			touch with your healthcare provider and family
			support team for help when needed
	Information about Health consequences	Motivation	Be aware that if you are unable to adhere to TB
18			medication regularly as directed, this can lead to
			failed treatment, relapse, and multidrug resistance
	Salience of consequences	Motivation	A person with TB who takes all the medications as
40			directed will be cured and will not be a risk of TB
19			transmission to his/her family and other
			communities
	Information about social and environmental consequences	Motivation	Untreated TB can be transmitted to close family
00			members, friends and other people. Please take
20			your medications until the end of the treatment
			period as per prescription
	Monitoring of Emotional		Did you take your medications at the same time
21	consequences	Motivation	every day? If you did, congratulations! If you didn't

			adhere, try again, but tell your healthcare provider the doses that you missed
22	Anticipated regret	Motivation	Do you know that if TB treatment fails, you will become sick again, and you may transmit TB to loved family members and friends? Treatment fails if a patient is not adhering to medications
23	Self-assessment of affective consequences	Motivation	Imagine what you would feel if you had a multi-drug resistant TB because of non-adherence to medications. It would need long-term treatment and may not ever be cured
24	Demonstration of the behaviour	Motivation	Consider using a strategy like setting an alarm in your mobile phone, or tell a family member to remember your daily TB medication and refill visit schedule
25	Generalization of target behaviour	Motivation	Please keep taking your daily medications regularly until the end of your treatment. TB is a curable disease if you are fully adhering to treatment
26	Social reward	Motivation	Thank you for picking up your medications. Please remember to take your daily medications at the same time every day
27	Future punishment	Motivation	Do you know that non-adherence to medications may cause multi-drug resistant TB that may need complex treatment for a long time, has more side effects, and may not work or allow you to be cured
28	Body changes	Motivation	You will gain body weight and become healthy if you appropriately adhere to your TB medications and lead a healthy lifestyle
29	Identification of self as role model	Motivation	Be a role model and save others from TB transmission by fully adhering to your TB medications regularly to the end of treatment
30	Verbal persuasion about capability	Motivation	You have the capacity to fully adhere to all your TB medications every day to be cured. Yes, you can do it!
31	Mental rehearsal of successful performance	Motivation	Imagine that you can remember to take your daily TB medications to the end of treatment. Your imagination will become true!
32	Problem-solving	Behavioural Skills	Why do some patients forget to refill their medication? You can come to refill any time before you run out of medications. Keep your calendar card in a place where you can see it

			Please ask your family member to pick up your		
33	Problem-solving	Behavioural	medications for you if you are not able to come to		
		Skills	the clinic to refill your medications		
		Skills	Consider some form of self-tracking such as a		
			, i i i i i i i i i i i i i i i i i i i		
	Discrepancy between current behaviour and goal	Behavioural Skills	medication calendar to mark the number of		
34			consecutive days that you are adhering. How many		
			days in a row have you taken your medication as		
			prescribed?		
	Behavioural	Behavioural skill	Please remember to take your medication regularly		
35	practice/rehearsal		every day at the same time. Please keep your		
			medications in a safe place where you can see it		
	Habit formation	Behavioural skill	Please take your medications every day at the		
36			same time in the morning. This helps you to		
			remember to take your medications regularly.		
	Self-monitoring of behaviour	Behavioural Skills	Plan ahead to visit the health facility to refill your		
37			medication supply. Use a calendar card to check off		
			the days when you have taken your medicine		
	Practical	Behavioural Skills	Please tell the healthcare provider if you are		
			traveling somewhere away from home. They can		
38			give you extra medications or link you with another		
			healthcare worker, so you will not run out of		
			medications		
	Practical	Behavioural Skills	Please tell us if you are feeling ill while you are on		
39			your medications. Ask the healthcare provider		
			about the common side effects		
			It is common to experience some side effects while		
	Instruction on how toBehaviouralperform a behaviourSkills		you are taking TB medication. Please tell to your		
40			healthcare provider if you experience mild side		
			effects		
			a Gara - Dahar da washa killa		

BCT: Behaviour Change technique, IMB: Information, Motivation, Behavioural skills

APPENDIX FIVE: PHONE CALLS CHECKLIST GUIDE

Checklist for phone calls intervention

(To be completed by a health professional at sub-city clinic when participant get a phone call)

Identification				
Sub-city Name:				
Health facility Name:				
Participant registration code:				
Participant mobile phone number:				

Weekly phone calls intervention checklist form

Questions to be asked of the participant by a health professional:

Please answer the following questions by saying 'Yes' or 'No" based on the intervention you are currently receiving.

		Ask the patient the following questions when you call			
Date (DD/MM/YY)	Are the patient answered for the phone calls (minimum of 3 repetitive calls)	Good morning, do you take your daily medication?	Are you reading and following the SMS text message instructions?	Do you have any health problems or side-effects you want to discuss?	Do you have a question regarding your treatment?
		Date (DD/MM/YY) answered for the phone calls (minimum of 3	Date (DD/MM/YY) Are the patient answered for the phone calls (minimum of 3 repetitive calls) Good morning, do you take your daily	Date (DD/MM/YY)Are the patient answered for the phone calls (minimum of 3 repetitive calls)Good morning, do you take your dailyAre you reading and following the SMS text message	Date (DD/MM/YY)Are the patient answered for the phone calls (minimum of 3 repetitive calls)Good morning, do you take your dailyAre you reading and following the SMS text messageDo you have any health problems or side-effects you want to

APPENDIX SIX: DEMPGRAPHIC CHARACTEROSTICS OF ADULT TB PATIENTS WITHOUT PERSONAL MOBILE PHONES

Demographic characteristics

Among the 48 participants without personal mobile phones, 29 (60.4%) were female. More than half of the participants (54.2%) were married, while 17 (35.4%) had never been married. Regarding educational background, 23 (47.9%) participants had no formal education, and 13 (27.1%) had completed primary education grade 1-8. Occupationally, 15 (31.3%) participants identified as housewife, while 11 (22.9%) were unemployed. Family size distribution revealed that 22 (45.8%) participants had a family size of 4-5 people in the household, followed by 15 (31.3%) with a family size of 2-3 people. Concerning income, 31 (64.6%) participants had a monthly income less than 2373 ETB (57 USD). Additionally, 32 (66.7%) participants relied on walking as their primary mode of transportation to reach health facilities (see Table 35).

Table 35: Socio-demographic characteristics of adult TB patients without personal mobile phones, Addis Ababa, Ethiopia

Variables	Frequency (percent)			
Gender	n=48			
Male	19 (39.6%)			
Female	29 (60.4%)			
Marital status	n=48			
Never married	17 (35.4%)			
Married	26 (54.2%)			
Divorced	0			
Widowed	5 (10.4%)			
Educational level	n=48			
No formal education	23 (47.9%)			
Primary education (1-8 grade)	13 (27.1%)			
Secondary education (9-10 grade)	6 (12.5%)			
Technical/Vocational education (11-12 grade)	4 (8.3%)			
Higher education (above 12 grade)	2 (4.2%)			
Occupational status	n=48			
Daily laborer	7 (14.6%)			
Governmental employee	0			
Unemployment	11 (22.9%)			
Merchant	3 (6.3%)			
Housewife	15 (31.3%)			
Student	2 (4.2%)			
Driver	1 (2.1%)			
Private employee	2 (4.2%)			
Pensioner	3 (6.3%)			
Guard	1 (2.1%)			
Other*	3 (6.3%)			
Family size	n=48			
Living alone	3 (6.3%)			
2-3 people	15 (31.3%)			
4-5 people	22 (45.8%)			
6 and above people	8 (16.7%)			
Monthly income in ETB (USD)	n=48			
Less than 2373 (57\$)	31 (64.6%)			
2373 (57\$) and above	17 (35.4%)			
Transportation used to reach the health facility	n=48			
Walk	32 (66.7%)			
Taxi	16 (33.3%)			
Bajaj	0			
Bus/Public transport	0			

*Farmer, House Servant, ETB: Ethiopia Birr, USD: United States Dollar, Ethiopian Birr (ETB) to USD exchange was made based on the data collection time conversion rate
APPENDIX SEVEN: CHARACTERSTICS OF LOSS TO FOLLOW-UP PARTICIPANTS AND SENSITIVITY ANALYSIS

Demographic characteristics of loss to follow-up participants

Four of five lost to follow-up participants in the Ma-MAS group were males, while one of three lost to follow-up participants in the control group was male. In the Ma-MAS and control groups, the median age of lost to follow-up participants was 38 and 28 years, respectively. Three lost to follow-up participants in both Ma-MAS and control groups had never been married. Three lost to follow-up participants in the Ma-MAS group and one lost to follow-up participant in the control group had monthly income less than 2738 ETB (57 USD), putting them in poverty. Two lost to follow-up participants in the Ma-MAS group and one lost to follow-up participant in the control group did not have adequate food, while these lost to follow-up participants in both groups received food support. Two out of five lost to follow-up participants in the Control group participants in the Ma-MAS group participants in the Ma-MAS group had TB/HIV co-infection, whereas all three lost to follow-up participants in the control group had only infected with TB (see table 36).

Table 36: Demographic characteristics of lost to follow-up adult TB patients enrolled in Ma-MAS trial, Addis Ababa, Ethiopia

Variables	Ma-MAS group n=5	Control group n=3
Sex	11=5	11=5
Male	4	1
Female	1	2
Age in years		
Mean (Sd)	40.2 (16.39)	31 (12.76)
Median (IQR)	38 (25)	28 (19)
Marital status		
Never Married	3	3
Married	2	0
Educational status		-
1-8 grade	1	2
9-10 grade	1	0
11-12 grade	1	0
Above 12 th grade	1	1
Occupation		
Governmental	1	0
Daily laborer	3	0
Housewife	1	1
Student	0	1
Unemployment	0	1
Monthly family Income (ETB)		
Less than 2738	3	1
2738 and above	2	2
Access to adequate food		
Yes	3	2
No	2	1
Food support (any source)		
Yes	2	1
No	3	2
TB-HIV co-infection		
Yes	2	0
No	3	3

Sd: standard deviation, IQR: Interquartile range

Sensitivity analysis

Effectiveness of Ma-Mas intervention on primary outcome

The effect of the mobile-assisted medication adherence support intervention on TB medication adherence measured by the IsoScreen urine test was unchanged after the inclusion of the loss to follow-up patients in the sensitivity analysis after 4 and 8 weeks of follow-up (see table 37 and 38).

Assumptions

Assumption one: The worst scenario; the lost to follow up patients considered as non-adherent to TB medication.

After 8 weeks follow-up, the Ma-MAS intervention had improved adherence by 12.90% (95%CI:

2.70, 23.16), and after 4 weeks of follow-up, Ma-MAS had improved adherence by 13.97% (95%CI:

4.91, 23.03) compared to the control group who only received the standared care (see table 37).

Table 37: Sensetivity analysis of Ma-MAS effect on TB medication adherence meausred IsoScreen urine test in the worst scenario

Primary outcomes	Adherence measured by IsoScreen test	95% CI	P value, 1- tailed
After 8 weeks of follow-up			
Ma-MAS group	81.72%	[73.86, 89.59]	
Control group	68.81%	[59.40, 78.23]	
Absolute Difference (AD)	12.90%	[2.61, 23.19]	0.020
After 4 weeks of follow-up			
Ma-MAS group	89.24%	[82.95, 95.54]	
Control group	75.26%	[66.50, 84.03]	
Absolute Difference (AD)	13.97%	[4.91, 23.03]	0.0063

Absolute Difference (AD) calculated Ma-MAS group – control group with 95% confidence interval (CI) of 1-tailed, n1=n2=93

Assumption two: The best scenario; the loss to follow-up and missing outcomes considered as adherent to TB medication.

The Ma-MAS intervention had a higher adherence by 15.05% after 4th and 8th weeks of follow-up

time compared to the standard care in the control group (see table 38).

Table 38: Sensetivity analysis of Ma-MAS intervention effect on TB medication adherence meausred IsoScreen urine test in best scenario

Primary outcomes	Adherence measured by IsoScreen test	95% CI	P value, 1- tailed
After 8 weeks of follow-up			
Ma-MAS group	87.09%	[80.28, 93.91]	
Control group	72.04%	[62.92, 81.16]	
Absolute Difference (AD)	15.05%	[5.49, 24.60]	0.0054
After 4 weeks of follow-up			
Ma-MAS group	92.47%	[87.11, 97.83]	
Control group	77.41%		
Absolute Difference (AD)	15.05	[6.62, 23.48]	0.0020

Absolute Difference (AD) is calculated Ma-MAS group – Control group with 95% confidence interval (CI) of 1-tailed, n1=n2=93

Effectiveness of Ma-Mas intervention on secondary outcomes

The Ma-MAS intervention effect on self-reported TB medication adherence did not statistically differnent after including missing outcomes in the sensetivity analysis. Ma-MAS had no effect after the inclusion of the lost to follow-up (see Table 39 and 40).

Assumptions

Assumption one: The worst scenario, the lost to follow up and missing outcomes were considered

as non-adherent to TB medication.

After accounting for lost to follow-up as non-adherence, the effect of the Ma-MAS intervention had no statistically significant differnce on self-reported TB medication adherence (see Table 39).

Table 39: Sensetivity analysis of Ma-MAS intervention effect on self report TB medication adherence in worst senario

Secondary outcomes	Adherence measured by ACTG [95%CI]	P value	Adherence measured by VAS [95%CI]	P value	
Before intervention					
Mo MAS group	70.97%		91.40%		
Ma-MAS group	[61.74, 80.19]		[85.69, 97.09]		
Control group	79.57%		92.47%		
Control group	[71.37, 87.76]		[87.11, 97.83]		
Absolute Difference (AD)	-8.6%	0.174	-1.0%	0.787	
Absolute Difference (AD)	[- 20.9, 3.7]	0.174	[-8.8, 6.7]	0.707	
After intervention		P value, 1-		P value, 1-	
Alter Intervention		tailed		tailed	
Ma-MAS group	77.41%		89.24%		
Ma-MAS group	[68.92, 85.91]		[82.95, 95.54]		
	78.49%		87.09%		
Control group	[70.14, 86.84]		[80.28, 93.91]		
Absolute Difference (AD)	-1.0%	0.430	2.15%	0.325	
Absolute Difference (AD)	[-10.9, 8.9]	0.430	[-5.6, 9.8]	0.320	
DIFF	-7.5% [-21.9, 6.9]	0.193	-3.2% [-13.3, 6.9]	0.302	

Absolute Difference (AD) is calculated Ma-MAS group – control group with 95% confidence interval (CI) of 1-tailed, DIFF-Difference in difference, n1=n2=93, ACTG: AIDS Clinical Trial Group adherence questionnaire, VAS: Visual Analogue Scale

Assumption two: The best scenario; the missing outcomes were considered as adherent to TB

medication.

Ma-MAS had no statistically significant positive effect on self-reported TB medication adherence

after accounting for lost to follow-up patients as adhered to TB medication (seeTabel 40).

Table 40: Sensetivity analysis of Ma-MAS intervention effect on self report TB medication adherence in best scenario

Secondary outcomes	Adherence measured by ACTG [95%CI]	P value 2- tailed	Adherence measured by VAS [95%CI]	P value 2- tailed
Before				
intervention				
Ma-MAS group	70.97% [61.74, 80.2]		91.40% [85.7, 97.09]	
Control group	79.57% 92.47% [71.37, 87.7] [87.11, 97.83]			
Absolute Difference (AD)	-8.6% [- 20.9, 3.7]	0.174	-1.0% [-8.8, 6.70]	0.787
After intervention		P value, 1- tailed		P value, 1- tailed
Ma-MAS group	AS group 84.94% 96.77% [77.6, 92.2] [93.18, 100]		96.77% [93.18, 100]	
Control group	83.87% [76.3, 91.34]		92.47% [87.1, 97.8]	
Absolute Difference (AD)	1.1% [-7.6, 9.8]	0.4199	4.3% [-1.1, 9.7]	0.0967
DIFF	-9.7% [-23.3, 3.9]	0.122	-5.4% [-13.9, 3.1]	0.151

Absolute Difference (AD) is calculated Ma-MAS group - Control group with 95% confidence interval (CI) of 1-tailed, DIFF-Difference in difference, n1=n2=93, AIDS Clinical Trial Group adherence questionnaire, VAS: Visual Analogue Scale

APPENDIX EIGHT: PUBLICATIONS FROM THE THESIS

Article 1: Factors influencing patient adherence to Tuberculosis treatment in Ethiopia: a literature review



International Journal of Environmental Research and Public Health



Review

Factors Influencing Patient Adherence to Tuberculosis Treatment in Ethiopia: A Literature Review

Zekariyas Sahile Nezenega ^{1,2,*}, Lua Perimal-Lewis ³ and Anthony John Maeder ²

- ¹ Department of Public Health, Ambo University, Ambo, Ethiopia
- ² Flinders Digital Health Research Centre, College of Nursing & Health Sciences, Flinders University,
- Adelaide SA 5001, Australia; anthony.maeder@flinders.edu.au
- ³ Flinders Digital Health Research Centre, College of Science & Engineering, Flinders University, Adelaide SA 5001, Australia; lua.perimal-lewis@flinders.edu.au
- Correspondence: zekariyas.nezenega@flinders.edu.au

Received: 12 June 2020; Accepted: 28 July 2020; Published: 4 August 2020



Abstract: Background: Tuberculosis (TB) is a major global public health problem and one of the leading causes of death among infectious diseases. Although TB can be cured with first-line antibiotics treatment of 6 months regimen, non-adherence to the treatment remains the main challenge for TB prevention and control. Interventions to promote adherence need to address multiple underlying factors linked to non-adherence, which requires a synthesis of studies to understand these factors in the local context. Our review accordingly examines these factors for TB treatment in Ethiopia. Methods: Articles were searched from PubMed and ScienceDirect databases, as well as manual searches through Google and Google Scholar search engines. Both quantitative and qualitative studies that showed factors associated with or reasons for non-adherence, default or loss to follow up from TB treatment were included. A total of 276 articles were screened, and 29 articles were ultimately included in the review. Findings: The extracted factors were synthesized thematically into seven dimensions of patient-centred, social, economic, health system, therapy, lifestyle, and geographic access factors. More than 20 distinct factors were identified under these headings. Some of these factors may also apply quite widely in other settings, with greater or lesser influence, but some are particularly applicable to the Ethiopian setting. Conclusion: Helping patients to achieve full adherence to TB medication is a complex problem as it is influenced by interplay between many factors. Healthcare managers, providers, and researchers need to consider and address multiple underlying factors when designing adherence interventions. This work provides a reference set of such factors for Ethiopian interventions.

Keywords: tuberculosis treatment; medication adherence; Ethiopia

1. Introduction

Tuberculosis (TB) is an infectious disease that is one of the major causes of death, being in the top ten causes of death worldwide and the leading cause of death from infectious disease, ranking above HIV/AIDS in 2018. Globally, one-fourth of the population is either infected with TB or at risk of developing the disease, with an estimated 10 million people infected with TB worldwide in 2018 [1]. Ethiopia has achieved a 50% reduction of TB through the Millennium Development Goals (MDGs) [2]. However, Ethiopia remains one of 14 high-burden TB countries for TB, TB-HIV, and Multi-drug-resistant tuberculosis (MDR-TB). An estimated 165,000 cases of TB incidence were reported for Ethiopia in 2018 [1].

Although TB can be cured with first-line antibiotics treatment of 6 months regimen, non-adherence is the main challenge for TB control and prevention programs. The patient needs to take >90% of TB

Int. J. Environ. Res. Public Health 2020, 17, 5626; doi:10.3390/ijerph17155626

www.mdpi.com/journal/ijerph

medication to facilitate TB cure, and a patient who takes at least 95% is said to be 'high adherence'. Treatment default is defined by the World Health Organization as a patient who interrupts treatment for 2 or more months [3]. The default rate is thus a crude method to monitor adherence [4]. Non-adherence to TB treatment increases the risk of morbidity, mortality, and drug resistance at both the individual and community level [5].

The World Health Organization has recommended Direct Observation of Treatment (DOT) by a trained supervisor, in which a healthcare worker watches the patient take the medication every day, to ensure adherence to treatment [6]. However, implementing DOT in Ethiopia is challenging for both the patient and healthcare provider. For example, one study conducted in Addis Ababa found patients reported that a daily visit to a health facility for the first two months was very difficult for a range of reasons, including severe illness at the initiation of treatment, distance too far for walking, and high transportation cost. Because of these challenges, DOT has not been implemented on a daily basis in Ethiopian standard care after the first two months of treatment [7].

A systematic review found that the pooled prevalence of non-adherence to TB treatment in Ethiopia was 21.3%. Forgetfulness, fear of drug side-effect, waiting time for 1 h or more during the service, and feeling a long distance to health facility were identified as factors associated with this non-adherence [8]. Another systematic review in the same setting found that the pooled prevalence of non-adherence to TB treatment and loss to follow-up were 20% and 5%, respectively. Being TB-HIV-co-infected, transport costs, lack of knowledge, drug side-effect, educational status, forgetfulness, being in continuation phase, perceived physical and psychological barriers, and psychological distress were identified as associated factors [9].

Interventions to promote adherence require addressing multiple components to overcome the barriers to adherence [10,11]. This requires a synthesis of studies to understand the causal factors for non-adherence to TB treatment in the local context. The above-mentioned systematic reviews and meta-analysis conducted in Ethiopia by Zegeye et al. (2019) [8] and Tolla et al. (2019) [9] had their main objective to estimate the pooled prevalence of non-adherence to TB treatment. The associated factors for non-adherence to TB treatment were not comprehensively identified by them, and their reviews included only quantitative studies.

This literature review synthesises both qualitative and quantitative studies to thematically present multiple factors that have been identified as influencing non-adherence to TB treatment in Ethiopia. This includes all types of TB such as active TB, latent TB, and TB-HIV co-infected patients. Therefore, this review could help healthcare managers, providers, and researchers to design and implement adherence interventions based on established contextual factors rather than ad hoc generalisations.

2. Methods

Research articles were searched for from PubMed and ScienceDirect databases, as well as manual search through Google and Google Scholar search engines. Search expressions were developed for TB medication adherence or loss to follow up or default from TB treatment that were published in the English language with no publication date restriction (see Table 1). The same expression of search strategy was used for all databases and search engines. Article searching was undertaken from 15 April to 5 May 2020.

Table 1. Searching strategy.

AND	(TB) OR (Tuberculosis)
AND	("medication adherence") OR (adherence) OR ("treatment adherence") OR ("medication compliance") OR (compliance) OR (default) OR ("loss to follow up")
AND	Ethiopia

Both qualitative and quantitative research articles were included, and articles that did not report original research were excluded. Articles that did not assess factors or reasons associated with TB medication non-adherence or default or loss to follow up from TB treatment were excluded, following the protocol of Figure 1. Data were extracted using an Excel template comprising the author, year, region, sample size, study design, population, and major findings. The article selection and data extraction were performed by the first author of this paper, and consistency was checked by the two other authors. Factors associated with TB treatment non-adherence were extracted from the quantitative type of studies, and reasons for non-adherence or default or loss to follow up were extracted from the qualitative type of studies. The extracted data were synthesized into groups based on the seven thematic dimensions of TB medication adherence factors proposed by Ogundele et al. (2015) [12]. These seven thematic dimensions are patient-centred, social, economic, health system, therapy, lifestyle, and geographic access factors.



Figure 1. Flow chart for selection of reviewed articles.

3. Findings

A total of 276 research articles were screened, and 29 studies were ultimately included in this literature review. Of these, 6 articles were qualitative studies, 22 articles were quantitative studies (18 cross-sectional, 3 prospective cohort, and 1 case-control), and 1 article used a mixed-method design.

These studies were conducted in Addis Ababa (9); Southern Nations, Nationalities, and Peoples' Region (SNNPR) (7); Amhara (5), Oromia (5); and Tigray (3) regions of Ethiopia. Five studies were conducted among latent TB-HIV co-infected patients, and the remaining 24 studies were conducted among active TB infected patients. Approximately 7382 participants in total were involved in the studies reported in these 29 articles.

The synthesised findings from both quantitative and qualitative studies are presented below, grouped in seven dimensions of adherence influencing factors. The individual studies that showed factors or reasons linked with TB medication non-adherence are presented in Appendix A Table A1.

3.1. Patient-Centred Factors

Forgetfulness [13–21] and inadequate knowledge about tuberculosis and its treatment regimen [14,18,22–25] were the two major patient-centred factors. Three studies conducted in Oromia [26,27] and SNNPR [21] regions showed that the patient's educational status was associated with non-adherence to TB medications: the more the patient was educated, the less likely was non-adherence to TB medication. Psychological distress was another factor: two studies conducted in Addis Ababa reported that this indirectly positively influences non-adherence to TB medication [28,29]. Another qualitative study conducted in Addis Ababa also stated that poor mental health status of a patient would make them reluctant to regularly attend follow up and clinic appointments [18].

3.2. Social Factors

Several studies reported that patients not getting social support from families and neighbours in remembering to take their medication, food, and financial assistance were the major social factors that influenced non-adherence to TB medication [18,19,24,26,30–33]. Additionally, one study in Addis Ababa conducted among latent TB-HIV co-infected patients reported that the patients' friends' decision to take the medication would make them less likely to be non-adherent to isoniazid preventive therapy (IPT) [34]. Another study among the same subjects and setting found that patients who were comfortable to take IPT in front of other people were less likely to be non-adherent [35]. Being busy with work [14] and away from home for work or other social-related activities were also found to influence non-adherence to TB medication [14–16]. Perceived and experienced stigma and discrimination also led the patient to non-adherence [18,36–38]. These particular factors were highly noted in studies conducted among TB-HIV co-infected patients [18,31,37,38]. As one study indicated, because of fear of stigma and discrimination, the patients were not disclosing their HIV status to their family, which in turn influenced their non-adherence to TB medication [18].

Beliefs about the disease and treatment, such as perceived wellness or cure, perceived risk, and perceived barriers over the benefits, were influencing factors for non-adherence to TB medication [13,26,28,29]. One study conducted in Addis Ababa reported that a patient's belief in curability and severity of TB in the presence of HIV infection would make them less likely to be non-adherent [31]. Another study in Addis Ababa found that the perceived risk of discontinuing TB medication was the reason for adherence, while perceived wellness was the reason for patients have intention to discontinue TB treatment [13]. One study conducted in SNNPR also reported that belief in traditional healing influenced non-adherence to TB medication [36].

3.3. Economic Factors

The patient's economic constraints (which impact the financial burden) was the main economic factor that influences non-adherence to TB medication [17,29–31]. Economic constraints limit the patient's ability to have adequate food which influences non-adherence [31,36]. The cost of medication other than TB medications is also a factor for non-adherence in one study conducted in SNNPR. Another study conducted in SNNPR reported that the patient being not employed was associated with non-adherence [39].

3.4. Health System Factors

Poor healthcare provider–patient relationship with communication gaps was a major factor that influenced non-adherence to TB medication [14,31,32,36,39]. For example, one study conducted in Addis Ababa among loss-to-follow-up patients reported that the healthcare providers were seen as disrespectful of their patients and less committed to their profession [32]. The quality of healthcare service and a patient's satisfaction with healthcare service affect non-adherence to TB medication [16,30,39]. When patients perceived that they received less professional care and less time spent with the healthcare providers, and waited a long time to get healthcare service, they were more likely to be non-adherent [16,32,39,40]. Health information/education is also crucial for adherence: a few studies showed that the patients who did not receive health information/education from health facilities were more likely to be non-adherent [15,18,22]. Additionally, one study done in Addis Ababa found that cues to action were reported as a factor for non-adherence [28]. Lack of supervision and healthcare providers incapable of managing the patient's illness were also reported as influencing factors for interruption and default from TB treatment [38].

3.5. Therapy Factors

Many studies reported that drug side-effects were the major therapy-related reason for non-adherence to TB medication [15,16,23,24,26]. Pill burden was also reported as a factor for non-adherence to TB medication among active TB and TB-HIV co-infected patients [31,36]. The presence of more than one co-morbidity including TB-HIV co-infection was also reported as a factor for non-adherence to TB medication [14,16,20,26,37]. One study conducted in Addis Ababa also found that being on Antiretroviral Therapy (ART) was a factor for non-adherence to TB medication [29]. Symptom presence after initiation of anti-TB treatment and slow progression of the health status were also found as non-adherence factors [19,20,26,33]. Being in the continuation phase of the treatment (after the initial 2-month clinic-based treatment period) was a factor for non-adherence and default [14,20,24,34,41]. This might be due to the patient's perceived wellness or cure because a daily DOT was not implemented after the first two months of treatment in Ethiopia.

3.6. Lifestyle Factors

Alcohol consumption was reported as a factor that influenced non-adherence to TB medication in several studies [14,16,27,29]. Cigarette smoking and khat (herbal stimulant) chewing were also found as factors associated with non-adherence [16,27].

3.7. Geographical Access Factors

Healthcare inaccessibility from residence location was a major geographical access factor for non-adherence and default from TB treatment [17,19,22,26,30,36,40]. Due to inaccessibility, patients were unable to keep regular clinic appointments and follow up treatment in two studies, conducted in Addis Ababa and Amhara regions [16,35]. Distance of the health facility was related to transportation cost, which was also a factor for non-adherence [21,22,25,33].

4. Discussion

Adherence to TB medication is a complex and dynamic matter as it is affected by multiple factors. This review has identified the range of multiple factors that have been found to affect non-adherence to TB medications in Ethiopia. The influence of these factors individually and in combination might vary from one social or geographic setting to the other. Healthcare managers need to consider the underlying factors for non-adherence to TB medication in the local setting (ideally using locally available evidence) when they design and implement an intervention.

Our literature review has found some similar influencing factors of non-adherence to TB treatment as those described in the previous two systematic reviews conducted in Ethiopia [8,9]. These factors were forgetfulness, inadequate knowledge about TB and its treatment regimen, psychological distress (poor mental health condition), perceived barriers, long waiting time, drug side-effects, TB-HIV co-infection, being on the continuation phase of treatment, healthcare inaccessibility, and travelling costs. Adherence interventions such as providing health information about TB and its treatment regimen, dealing with side-effects, providing reminders, and other health system interventions could be used to resolve these factors.

However, in this review, we have found several additional factors of non-adherence to TB treatment that were not identified by the previous two systematic reviews. These factors were lack of social support, being busy with work, being away from home, perceived and experienced stigma and discrimination, beliefs such as perceived wellness/cured, perceived risk, financial constraints to buy food and medication cost other than anti-TB, poor healthcare provider–patient relationship such as communication gaps, disrespecting patients, quality of healthcare service, patient satisfaction, lack of health information/education, pill burden, the persistence of symptoms after initiation of treatment, and use of substances such as alcohol, smoking, and khat chewing. These factors and the possible interventions to overcome the factors are discussed below.

We have found that lack of social support can influence non-adherence to TB medication. Thus, providing social support such as financial assistance for transport costs, food assistance, and reminders for medication intake from families and neighbours may help the patient to adhere to TB medication. In the same way, not getting social acceptance to take medication from families, friends, and neighbours can lead to non-adherence. A systematic review in developing countries reported the same finding [42]. Similarly, the perceived stigma and discrimination were reported as factors for non-adherence to TB medication. These were higher especially among TB-HIV co-infected patients. The other new social factors were being busy with work and away from home for work and social-related activities. These factors influence non-adherence might be due to the patient forgetting to take their medication when they are busy and away from home. A reminder from family or through mobile SMS text could be considered as a solution for these factors. Beliefs related to TB and its treatment can also influence non-adherence. This might be due to the patient's perceived wellness or cure after taking some medications and thus interrupting their treatment. On the other hand, when the patient perceives that the disease is severe and the risk of discontinuing TB medication leads to poor health outcomes, their non-adherence to TB medication would be less likely. A systematic review conducted in developing countries also found that feeling well after the initiation of treatment was a factor for non-adherence to TB medication [42]. Providing patients with health information to change such wrong beliefs about the disease and its treatment could be considered as an adherence intervention to address these factors.

The financial constraints to get adequate food, transport cost, and medication cost other than anti-TB medication were found to influence non-adherence or loss to follow up from TB treatment. A systematic review conducted in developing countries also reported that the financial burden can cause TB medication non-adherence [42]. Financial assistance or support of food and transport and making all medications freely available to the patient may overcome this factor.

We also found that healthcare-system-related factors were other major factors for non-adherence to TB medication. Poor healthcare-provider relationship with communication gaps, disrespect of the patient, and lack of professional commitment influenced non-adherence to TB medication. A systematic review conducted in developing countries also found that poor patient-healthcare worker communication was a factor for non-adherence to TB medication [42]. The quality of health service as perceived by the patient and the patient satisfaction also influenced non-adherence to TB medication. It was also found that patients who did not receive health information/education were more likely to be non-adherent to TB medication. Therefore, health system interventions such as training for healthcare provider–patient communication and their relationships and health system strengthening to shorten waiting times and to raise the quality of health services could address these factors.

Pill burden was also a factor for non-adherence to TB medication both in TB and TB-HIV co-infected patients. TB-HIV co-infection has additional burden from the pills and drug side effects. Persistence of symptomatic conditions after initiation of TB treatment was another influence for non-adherence.

This might be due to the patient's belief in the curability of the disease becoming less when symptoms persist despite the treatment initiation. Health information about TB, its treatment, and side effects could reduce the influence of these factors. New drug investigations to reduce the drug side effect and to make a shorter treatment regimen could also be considered if it is possible.

Substance use such as alcohol use, cigarette smoking, and khat chewing was a factor for non-adherence to TB medication as reported by a few studies. The use of these substances might make the patient reluctant to follow the regular clinic appointment and follow up treatment. Other systematic reviews conducted in developing countries also showed that alcohol and tobacco were factors for non-adherence to TB medication [42]. Health information on how the use of substances affected treatment adherence and treatment outcome may help to resolve this factor.

5. Limitations

This review included studies conducted in Ethiopia; therefore the identified factors may not be generalizable to other settings. Adherence to TB medication is a complex problem as it is influenced by multiple factors so a single factor may not be shown as a cause–effect relationship.

6. Conclusions

This review describes more than 20 factors that influence adherence to TB treatment in Ethiopia, demonstrating that it is a complex problem that is affected by the interplay of multiple factors. We have found major additional factors for TB medication non-adherence or default or loss to follow up. These were social support from families and neighbours such as food support, reminders, and encouragement; being busy with work; being away from home; perceived and experienced stigma and discrimination; beliefs such as perceived wellness/cure; perceived risk; economic constraints for having adequate food and medication cost other than anti-TB medication; poor healthcare provider-patient relationships such as communication gaps, disrespecting patients, quality healthcare service, and patient satisfaction; health information/education, pill burden, the persistence of symptoms after treatment initiation; and use of substances. Healthcare managers, providers, and researchers need to address these underlying factors when they design and implement adherence interventions.

Definition of Term: Adherence to a medication regimen is defined by Osterberg and Blaschke (2005) as "the extent to which patients take medications as prescribed by their health care providers"(10).

Author Contributions: Z.S.N., L.P.-L. and A.J.M. contributed to the conceptual design, data extraction, synthesis and analysis, and preparation of the manuscript. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Acknowledgments: The authors received advice from Flinders University library in developing the protocol.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Table A1. Summary of reviewed articles showing factors or reasons linked with TB medication adherence or default or loss to follow up from TB treatment.

First Author (Year)	Region	Study Design	Sample Size	Population	Associated Factors of Non-Adherence to TB Medication and Loss to Follow up/Default from TB Treatment
Daba et al. (2019) [27]	Oromia	Prospective cohort	268	Active TB patients	Khat (herbal stimulant) and alcohol use was associated with non-adherence with TB medication. Educational and occupational status were also associated with non-adherence to TB medication.
Sahile et al. (2018) [13]	Addis Ababa	A qualitative study	10	Active TB patients	Forgetfulness is the factor of TB medication non-adherence. The perceived risk and perceived wellness were also associated with TB medication non-adherence.
Mekonnen et al. (2018) [14]	Amhara	Cross-sectional study	314	Active 'TB patients	Being in the continuation phase of treatment, the presence of more than one co-motivity inadequate knowledge of TB and its treatment, poor patient-provider relationships, and alcohol use were statistically associated non-adherence to TB treatment. Additionally, forgetting to take medication, being busy with work, and out of homs/town were the reasons mentioned by patients for non-adherence.
Gube et al. (2018) [40]	SNNPR	Cross-sectional study	271	Active TB patients	Drug side effects, a far distance from the health facility, and prolonged wait time to get healthcare service were statistically associated with non-adherence.
Woimo et al. (2017) [22]	SNNPR	Cross-sectional study	261	Active pulmonary TB patients	Inadequate knowledge of TB and its treatment, health information at each medication refill visits, a far distance from a health facility (more than 10 km), transportation cost, and cost of medications other than anti-TB.

Int. J. Environ. Res. Public Health 2020, 17, 5626

9 of 15

First Author (Year)	Region	Study Design	Sample Size	Population	Associated Factors of Non-Adherence to TB Medication and Loss to Follow up/Default from TB Treatment
Tola et al. (2017) [28]	Addis Ababa	Cross-sectional study	698	Active TB patients	Perceived barrier over and perceived benefits directly associated with non-adherence TB medication. Cue to action and psychological distress were indirectly influence non-adherence mediated through perceived barriers and benefits.
Tola et al. (2017) [29]	Addis Ababa	Cross-sectional study	698	Active TB patients	ART status, economic status, alcohol use, perceived barrier, and psychological distress were statistically associated with non-adherence to TB medication.
Gugsa et al. (2017) [36]	SNNPR	A qualitative study	22	Active TB patients	Lack of adequate food, poor communication between healthcare providers and patients, beliefs in traditional healing, unavailability of the healthcare service in nearby, drug side-effect and pill burden, stigma, and discrimination.
Getahun et al. 2017 [32]	Addis Ababa	Mixed method design	649	Active TB patients and healthcare provider	Among lost to follow-up patients inadequate information about TE, bachkoare provider-patient relationship (respect and value to patients), lack of support such as transport and nutrition support, and less commitment of healthcare provider towards the patients were reported as reasons for lost to follow-up.
Ayele et al. (2017) [15]	Amhara	Cross-sectional study	154	Latent TB-HIV patients	Lack of information about isoniazid preventive therapy (IPT), drug side effect, forgetfulness, and being away from home were found as factors for non-adherence to isoniazid preventive therapy (IPT).
Diriba et al. (2016) [26]	Oromia	Cross-sectional study	67	Active TB patients	Lack of family support, a far distance from the health facility, drug side-effects, extremely ill, feeling better, education level, and being HIV positive were the factors that led to non-adherence to TB medication.

Int. J. Environ. Res. Public Health 2020, 17, 5626

Table A1. Cont.

First Author (Year)	Region	Study Design	Sample Size	Population	Associated Factors of Non-Adherence to TB Medication and Loss to Follow up/Default from TB Treatment
Ayele et al. (2016) [37]	SNNPR	Prospective cohort	162	Latent TB-HIV patients	Patients experiencing a high level of HIV stigma and having opportunistic infections statistically associated with non-adherence to isoniazid preventive therapy (IPT).
Tesfahuneygn et al. 2015 [16]	Amhara	Cross-sectional study	200	Active TB patients	Forgerfulness, being away from home, drug side effects, being unable to go to the health facilities on the date of appointment, being hospitalized, TB-HIV infected, alcohol use, smoking khat (herbal situulant) chewing, unsatisfied with healthcare service, and long waiting time to get he health service were significantly associated.
Yasin et al. (2015) [17]	Oromia	Cross-sectional study	53	Active TB-HIV co-infected	Forgetfulness, a far distance from the health facility, and low income were the factors for non-adherence to TB medication.
Mindachew et al. (2014) [18]	Addis Ababa	A qualitative study	12	Latent TB-HIV patients	Forgetfulness, lack of patient information, knowledge about the disease and its treatment, mental health status make them reluctant to attend follow-up and climic appointment, not disclosing their HIV status to their family members because of fear of stigma and discrimination and lack of support from family led them to non-adhere to isoniazid preventive therapy (UT).
Kiros et al. (2014) [23]	Tigray	Cross-sectional study	278	Active TB patients	The drug side effect and knowledge about TB prevention were associated with non-adherence to TB medication.
Eticha et al. (2014) [19]	Tigray	Cross-sectional study	120	Active TB-HIV co-infected	Patients who do not have caregivers and people who do not have to remind them to take their medications more likely non-adhere to TB medications. Forgetfulness, feeling sick, and being far away from health fadilities were the mention of the main reasons of the TB-HIV o-indiceded patients for missing medication.

Int. J. Environ. Res. Public Health 2020, 17, 5626

11 of 15

First Author (Year)	Region	Study Design	Sample Size	Population	Associated Factors of Non-Adherence to TB Medication and Loss to Follow up/Default from TB Treatment
Berhe et al. (2014) [34]	Addis Ababa	Cross-sectional study	381	Latent TB-HIV patients	Patients who took their medication for ≥ 5 or months were highly likely to be adherent compared to those who took it for 1-2 months. Patients' friends' decision to take the medication made them less likely to be non-adherent to isoniazid preventive therapy (IPT).
Nezenega et al. (2011) [39]	SNNPR	Cross-sectional study	531	Active TB patients	Employment status, area of residence, perceived time spent with a healthcare provider, perceived accessibility, perceived waiting time, perceived professional care, and overall patient satisfaction were associated with non-adherence to TB medication.
Adane et al. (2013) [20]	Amhara	Cross-sectional study	280	Active TB patients	Forgetfulness, being in the continuation phase, HIV co-infection, and persistent symptoms of tuberculosis were the factors for non-adherence to TB medication.
Tadesse et al. (2013) [30]	Amhara	A qualitative	26	Active TB	Access to health facilities, financial burdens, quality of health services, and social support were the main reason for failing to fully adhere to TB medication.
Kebede et al. (2012) [21]	SNNPR	Cross-sectional study	24	Active TB-HIV co-infected	The educational status was associated with non-adherence to TB medication. Forgetfulness and transportation cost were mentioned as reasons for non-adherence to medication.
Mindachew et al. (2011) [35]	Addis Ababa	Cross-sectional study	319	Latent TB-HIV patients	Patients who did not receive information about IPT, patients not comfortable taking IPT in from of other people, patients who did not attend regular dimic appointments, and drug side effects were associated to non-adherence to isoniazid preventive therapy (IPT).

10 of 15

Int. J. Environ. Res. Public Health 2020, 17, 5626

Table A1. Cont.

First Author (Year)	Region	Study Design	Sample Size	Population	Associated Factors of Non-Adherence to TB Medication and Loss to Follow up/Default from TB Treatment
Gebremariam et al. (2010) [31]	Addis Ababa	A qualitative study	38	Active TB-HIV co-infected and healthcare providers	Side effects, pill burden, economic constraints, lack of food, stigma with lack of disclosure, and lack of adequate communication with health professionals were barriers for adherence. While beliefs in the curability of TB, beliefs in the severity of TB in the presence of HIV infection and lack of support from families and health professionals influenced non-adherence to TB medication.
Mesfin et al. (2009) [38]	Tigray	Cross-sectional	237	Active TB patients	Lack of supervision and incapability of dealing with patients' illness more likely to interrupt and default from treatment.
Sagbakken et al. (2008) [33]	Addis Ababa	A qualitative study	50	Active TB patients, healthcare providers and relatives	Transportation costs, poor health status due to illness or slow progression, not having social support, and not managing to restore their health and social status were factors for non-adherence.
Shargie et al. (2007) [41]	SNNPR	Prospective cohort study	404	Active TB patients	Continuation phase of treatment, a far distance from the treatment centre, and necessity to use public transport were the factors for default from TB treatment.
Michael et al. (2004) [25]	Oromia	Cross-sectional study	114	Defaulted TB patients	Far distance from the health institution, transportation cost, and being unaware about TB were the major reasons contributing to defaulting.
Tekle et al. (2002) [24]	Oromia	Case-control	1367	Active TB patients	Being in the continuation phase of treatment, lack of family support, inadequate knowledge about treatment duration and drug side effects.

SNNPR-Southern Nations, Nationalities, and Peoples' Region, TB-Tuberculosis.

12 of 15

References

- 1. World Health Organization. Global Tuberculosis Report 2019; WHO: Geneva, Switzerland, 2019.
- Deribew, A.; Deribe, K.; Dejene, T.; Tessema, G.A.; Melaku, Y.A.; Lakew, Y.; Amare, A.T.; Bekele, T.; Abera, S.F.; Dessalegn, M.; et al. Tuberculosis Burden in Ethiopia from 1990 to 2016: Evidence from the Global Burden of Diseases 2016 Study. *Ethiop. J. Health Sci.* 2018, 28, 519–528.
- World Health Organization. Treatment of Tuberculosis Guidelines for National Programmes; WHO: Geneva, Switzerland, 2003.
- Awofeso, N. Anti-tuberculosis medication side-effects constitute major factor for poor adherence to tuberculosis treatment. *Bull. World Health Organ.* 2008, *86*, B–D. [CrossRef]
- World Health Organization. Adherence to Long-Term Therapies Evidence for Action; WHO: Geneva, Switzerland, 2003.
- World Health Organization. An expanded DOTS framework for effective tuberculosis control. Int. J. Tuberc. Lung Dis. 2002, 6, 378–388.
- Fiseha, D.; Demissie, M. Assessment of Directly Observed Therapy (DOT) following tuberculosis regimen change in Addis Ababa, Ethiopia: A qualitative study. *BMC Infect. Dis.* 2015, 15, 405. [CrossRef] [PubMed]
- Zegeye, A.; Dessie, G.; Wagnew, F.; Gebrie, A.; Islam, S.M.S.; Tesfaye, B.; Kiross, D. Prevalence and determinants of anti-tuberculosis treatment non-adherence in Ethiopia: A systematic review and meta-analysis. *PLoS ONE* 2019, 14, e0210422. [CrossRef] [PubMed]
- Tola, H.H.; Holakouie-Naieni, K.; Tesfaye, E.; Mansournia, M.A.; Yaseri, M. Prevalence of tuberculosis treatment non-adherence in Ethiopia: A systematic review and meta-analysis. *Int. J. Tuberc. Lung Dis.* 2019, 23, 741–749. [CrossRef]
- 10. Osterberg, L.; Blaschke, T. Adherence to medication. N. Engl. J. Med. 2005, 353, 487–497. [CrossRef] [PubMed]
- Subbaraman, R.; de Mondesert, L.; Musiimenta, A.; Pai, M.; Mayer, K.H.; Thomas, B.E.; Haberer, J. Digital adherence technologies for the management of tuberculosis therapy: Mapping the landscape and research priorities. *BMJ Glob. Health* 2018, *3*, e001018. [CrossRef] [PubMed]
- Ogundele, O.A.; Moodley, D.; Seebregts, C.J.; Pillay, A.W. (Eds.) An ontology for tuberculosis treatment adherence behaviour. In SAICSIT 2015, Proceedings of the 2015 Annual Research Conference on South African Institute of Computer Scientists and Information Technologists, Stellenbosch, South Africa, 28–30 September 2015; Association for Computing Machinery: New York, NY, USA, 2015.
- Sahile, Z.; Yared, A.; Kaba, M. Patients' experiences and perceptions on associates of TB treatment adherence: A qualitative study on DOTS service in public health centers in Addis Ababa, Ethiopia. *BMC Public Health* 2018, 18, 462. [CrossRef]
- Mekonnen, H.S.; Azagew, A.W. Non-adherence to anti-tuberculosis treatment, reasons and associated factors among TB patients attending at Gondar town health centers, Northwest Ethiopia. *BMC Res. Notes* 2018, 11, 691. [CrossRef]
- Ayele, A.A.; Asrade Atnafie, S.; Balcha, D.D.; Weredekal, A.T.; Woldegiorgis, B.A.; Wotte, M.M.; Gebresillasie, B.M. Self-reported adherence and associated factors to isoniazid preventive therapy for latent tuberculosis among people living with HIV/AIDS at health centers in Gondar town, North West Ethiopia. *Patient Prefer. Adher.* 2017, 11, 743–749. [CrossRef] [PubMed]
- Tesfahuneygn, G.; Medhin, G.; Legesse, M. Adherence to Anti-tuberculosis treatment and treatment outcomes among tuberculosis patients in Alamata District, northeast Ethiopia. *BMC Res. Notes* 2015, *8*, 503. [CrossRef] [PubMed]
- Yasin Mohammed, A.; Kaso Adem, M. Treatment Adherence among Tuberculosis and Human Immuno Deficiency Virus Coinfected Patients in Ginnir Referral Hospital. Am. J. Public Health Res. 2014, 2, 239–243. [CrossRef]
- Mindachew, M.; Deribew, A.; Memiah, P.; Biadgilign, S. Perceived barriers to the implementation of Isoniazid preventive therapy for people living with HIV in resource constrained settings: A qualitative study. *Pan Afr. Med. J.* 2014, 17, 26. [CrossRef]
- Eticha, T.; Kassa, E. Non-adherence to anti-TB drugs and its predictors among TB/HIV co-infected patients in Mekelle, Ethiopia. J. Bioanal. Biomed. 2014, 6, 061–064. [CrossRef]
- 20. Adane, A.A.; Alene, K.A.; Koye, D.N.; Zeleke, B.M. Non-adherence to anti-tuberculosis treatment and determinant factors among patients with tuberculosis in northwest Ethiopia. *PLoS ONE* **2013**, *8*, e78791. [CrossRef]

- 21. Kebede, A.; Wabe, N.T. Medication adherence and its determinants among patients on concomitant tuberculosis and antiretroviral therapy in South West Ethiopia. *N. Am. J. Med. Sci.* **2012**, *4*, 67–71.
- Woimo, T.T.; Yimer, W.K.; Bati, T.; Gesesew, H.A. The prevalence and factors associated for anti-tuberculosis treatment non-adherence among pulmonary tuberculosis patients in public health care facilities in South Ethiopia: A cross-sectional study. *BMC Public Health* 2017, *17*, 269. [CrossRef]
- 23. Kiros, Y.K.; Teklu, T.; Desalegn, F.; Tesfay, M.; Klinkenberg, E.; Mulugeta, A. Adherence to anti-tuberculosis treatment in Tigray, Northern Ethiopia. *Public Health Action* **2014**, *4* (Suppl. 3), S31–S36. [CrossRef]
- 24. Tekle, B.; Mariam, D.H.; Ali, A. Defaulting from DOTS and its determinants in three districts of Arsi Zone in Ethiopia. *Int. J. Tuberc. Lung Dis.* 2002, *6*, 573–579.
- Michael, K.W.; Belachew, T.; Jira, C. Tuberculosis defaulters from the "dots" regimen in Jimma zone, southwest Ethiopia. *Ethiop. Med. J.* 2004, 42, 247–253. [PubMed]
- Diriba Daksa, M.; Melaku Kebede, T.; Dahjejot, M. Patients' adherence to anti-tuberculosis medicines and associated factors for non-adherence at a tertiary teaching hospital, South West Ethiopia. *Eur. J. Ther.* 2016, 22, 55–62.
- Daba, M.; Tesfaye, M.; Adorjan, K.; Krahl, W.; Tesfaye, E.; Yitayih, Y.; Strobl, R.; Grill, E. Khat and Alcohol Use Disorders Predict Poorer Adherence to Anti-Tuberculosis Medications in Southwest Ethiopia: A Prospective Cohort Study. *Prepr. Lancet* 2019. [CrossRef]
- Tola, H.H.; Karimi, M.; Yekaninejad, M.S. Effects of sociodemographic characteristics and patients' health beliefs on tuberculosis treatment adherence in Ethiopia: A structural equation modelling approach. *Infect. Dis. Poverty* 2017, 6, 167. [CrossRef] [PubMed]
- Tola, H.H.; Garmaroudi, G.; Shojaeizadeh, D.; Tol, A.; Yekaninejad, M.S.; Ejeta, L.T.; Kebede, A.; Kassa, D. The Effect of Psychosocial Factors and Patients' Perception of Tuberculosis Treatment Non-Adherence in Addis Ababa, Ethiopia. Ethiop. J. Health Sci. 2017, 27, 447–458.
- Tadesse, T.; Demissie, M.; Berhane, Y.; Kebede, Y.; Abebe, M. Long distance travelling and financial burdens discourage tuberculosis DOTs treatment initiation and compliance in Ethiopia: A qualitative study. BMC Public Health 2013, 13, 424. [CrossRef]
- Gebremariam, M.K.; Bjune, G.A.; Frich, J.C. Barriers and facilitators of adherence to TB treatment in patients on concomitant TB and HIV treatment: A qualitative study. *BMC Public Health* 2010, 10, 651. [CrossRef]
- Getahun, B.; Nkosi, Z.Z. Is directly observed tuberculosis treatment strategy patient-centered? A mixed method study in Addis Ababa, Ethiopia. *PLoS ONE* 2017, 12, e0181205. [CrossRef]
- 33. Sagbakken, M.; Frich, J.C.; Bjune, G. Barriers and enablers in the management of tuberculosis treatment in Addis Ababa, Ethiopia: A qualitative study. *BMC Public Health* **2008**, *8*, 11. [CrossRef]
- 34. Berhe, M.; Demissie, M.; Tesfaye, G. Isoniazid preventive therapy adherence and associated factors among HIV positive patients in Addis Ababa, Ethiopia. *Adv. Epidemiol.* **2014**, 2014, 230587. [CrossRef]
- Mindachew, M.; Deribew, A.; Tessema, F.; Biadgilign, S. Predictors of adherence to isoniazid preventive therapy among HIV positive adults in Addis Ababa, Ethiopia. *BMC Public Health* 2011, *11*, 916. [CrossRef] [PubMed]
- Gugssa Boru, C.; Shimels, T.; Bilal, A.I. Factors contributing to non-adherence with treatment among TB patients in Sodo Woreda, Gurage Zone, Southern Ethiopia: A qualitative study. J. Infect. Public Health 2017, 10, 527–533. [CrossRef] [PubMed]
- Ayele, H.T.; van Mourik, M.S.; Bonten, M.J. Predictors of adherence to isoniazid preventive therapy in people living with HIV in Ethiopia. Int. J. Tuberc. Lung Dis. 2016, 20, 1342–1347. [CrossRef] [PubMed]
- Mesfin, M.M.; Newell, J.N.; Walley, J.D.; Gessessew, A.; Tesfaye, T.; Lemma, F.; Madeley, R.J. Quality of tuberculosis care and its association with patient adherence to treatment in eight Ethiopian districts. *Health Policy Plan.* 2009, 24, 457–466. [CrossRef] [PubMed]
- 39. Nezenega, Z.S.; Gacho, Y.H.; Tafere, T.E. Patient satisfaction on tuberculosis treatment service and adherence to treatment in public health facilities of Sidama zone, South Ethiopia. *BMC Health Serv. Res.* 2013, 13, 110. [CrossRef]
- Gube, A.A.; Debalkie, M.; Seid, K.; Bisete, K.; Mengesha, A.; Zeynu, A.; Shimelis, F.; Gebremeskel, F. Assessment of Anti-TB Drug Nonadherence and Associated Factors among TB Patients Attending TB Clinics in Arba Minch Governmental Health Institutions, Southern Ethiopia. *Tuberc. Res. Treat.* 2018, 2018, 3705812. [CrossRef]

Int. J. Environ. Res. Public Health 2020, 17, 5626

- 41. Shargie, E.B.; Lindtjørn, B. Determinants of treatment adherence among smear-positive pulmonary tuberculosis patients in Southern Ethiopia. *PLoS Med.* 2007, *4*, e37. [CrossRef]
- Tola, H.H.; Tol, A.; Shojaeizadeh, D.; Garmaroudi, G. Tuberculosis treatment non-adherence and lost to follow up among TB patients with or without HIV in developing countries: A systematic review. *Iran. J. Public Health* 2015, 44, 1.



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).

Article 2: Protocol of a parallel group Randomised Control Trial (RCT) for Mobile-assisted Medication Adherence Support (Ma-MAS) intervention among Tuberculosis patients

PLOS ONE

STUDY PROTOCOL

Protocol of a parallel group Randomized Control Trial (RCT) for Mobile-assisted Medication Adherence Support (Ma-MAS) intervention among Tuberculosis patients

Zekariyas Sahile^{1,2}*, Lua Perimal-Lewis³, Paul Arbon⁴, Anthony John Maeder²

1 Department of Public Health, Ambo University, Ambo, Ethiopia, 2 Flinders Digital Health Research Centre College of Nursing & Health Sciences, Flinders University, Adelaide, SA, Australia, 3 College of Science & Engineering, Flinders Digital Health Research Centre, Flinders University, Adelaide, SA, Australia, 4 College of Nursing & Health Sciences, Flinders University, Adelaide SA, Australia

* zekariyas.nezenega@flinders.edu.au, zekitiru@gmail.com

Abstract

Background

Non-adherence to Tuberculosis (TB) medication is a serious threat to TB prevention and control programs, especially in resource-limited settings. The growth of the popularity of mobile phones provides opportunities to address non-adherence, by facilitating direct communication more frequently between healthcare providers and patients through SMS texts and voice phone calls. However, the existing evidence is inconsistent about the effect of SMS interventions on TB treatment adherence. Such interventions are also seldom developed based on appropriate theoretical foundations. Therefore, there is a reason to approach this problem more rigorously, by developing the intervention systematically with evidence-based theory and conducting the trial with strong measurement methods.

Methods

This study is a single-blind parallel-group design individual randomized control trial. A total of 186 participants (93 per group) will be individually randomized into one of the two groups with a 1:1 allocation ratio by a computer-generated algorithm. Group one (intervention) participants will receive daily SMS texts and weekly phone calls concerning their daily medication intake and medication refill clinic visit reminder and group two (control) participants will receive the same routine standard treatment care as the intervention group, but no SMS text and phone calls. All participants will be followed for two months of home-based self-administered medication during the continuation phases of the standard treatment period. Urine test for the presence of isoniazid (INH) drug metabolites in urine will be undertaken at the random point at the fourth and eighth weeks of intervention to measure medication adherence will also be assessed by self-report measurements using the AIDS Clinical Trial Group adherence (ACTG) and Visual Analogue Scales (VAS) questionnaires, and clinic appointment attendance registration. Multivariable regression model analysis will be employed to assess the effect of the Ma-MAS intervention at a significance level of P-value < 0.05 with a 95% confidence interval.





GOPEN ACCESS

Citation: Sahile Z, Perimal-Lewis L, Arbon P, Maeder AJ (2021) Protocol of a parallel group Randomized Control Trial (RCT) for Mobileassisted Medication Adherence Support (Ma-MAS) intervention among Tuberculosis patients. PLoS ONE 16(12): e0261758. <u>https://doi.org/10.1371/</u> journal.oone.0261758

Editor: Kathleen Finlayson, Queensland University of Technology, AUSTRALIA

Received: April 4, 2021

Accepted: November 20, 2021

Published: December 31, 2021

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pone.0261758

Copyright: © 2021 Sahile et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This trial protocol is partially funded by Flinders University. The funder and sponsor had no role in trial design, and will not have a role in the trial implementation, analysis, and interpretation of the findings. All trial information is the responsibility of the authors.

Competing interests: The authors declared that they have not any competing interests.

Discussion

For this trial, a mobile-assisted medication adherence intervention will first be developed systematically based on the Medical Research Council framework using appropriate behavioural theory and evidence. The trial will then evaluate the effect of SMS texts and phone calls on TB medication adherence. Evidence generated from this trial will be highly valuable for policymakers, program managers, and healthcare providers working in Ethiopia and beyond.

Trial registration

The trial is registered in the Pan-Africa Clinical Trials Registry with trial number PACTR202002831201865.

Background

Non-adherence to Tuberculosis (TB) treatment is a risk factor for further transmission, treatment failure, relapse, acquired multi-drug resistance, or extensively drug-resistant TB and death [1-3]. In Ethiopia, non-adherence to TB treatment, specifically a medication regimen involving a combination of antibiotics, is a serious threat to TB prevention and control programs. A recent systematic review and meta-analysis conducted in Ethiopia found that the pooled prevalence of non-adherence to TB medication was 21.3% [4].

Non-adherence to TB treatment is influenced by many interconnected factors. These include patient-centred-related factors, economic-related factors, social-related factors, health system-related factors, therapy-related factors, geographical access-related factors and life-style-related factors [5, 6]. The major factors affecting TB treatment adherence in Ethiopia are patient-centred-related factors such as knowledge about TB, its treatment duration and consequence of non-adherence [7–9], educational status [7, 10], psychological state of the patients like forgetfulness [10–13] and psychological distress [7, 14] and social-related factors such as social support, stigma and discrimination, perceived risk, perceived wellness/cure, and perceived barriers over the benefits [7, 14].

The World Health Organization (WHO) recommended Direct Observation of Treatment (DOT) by a trained supervisor; a health worker or a treatment supporter watches the patient take their antibiotics every day to ensure adherence [15]. However, evidence shows that DOT does not necessarily provide a reliable solution to poor TB treatment adherence [16, 17]. In Ethiopia, implementing DOT is also very challenging for patients as it requires patients' daily visits to the health facility, which has undesired implications in their work, social life and high transportation cost [18, 19]. The current WHO recommendations are to consider DOT implementation with different administration options including conventional DOT, video-observed treatment (VOT), and non-daily DOT [20]. The non-daily DOT has been implemented in Ethiopia, in which during the first two months of the intensive phase of treatment patients take their daily medications with the supervision of the healthcare provider, and from two to six months in the continuation phase of treatment patients take all their medications at home and visit a clinic weekly to refill their medications [11, 19]. However, the non-daily DOT schedule has been interrupted by the COVID-19 pandemic [21].

Digital health interventions through mobile health can create new opportunities for healthcare improvements, especially in situations with limited infrastructure, expertise, and human

resources in the healthcare system [22–25]. Digital health interventions that enhance TB medication adherence are seen as a promising option to bring health benefits to the community [26]. The growth in popularity of mobile phones provides opportunities to address adherence challenges related to DOT, as mobile health can facilitate direct communication between healthcare providers and TB patients through SMS and voice calls along with DOT or nondaily DOT [27, 28]. Along with SMS text reminders, phone calls are also suggested to improve a patient's clinic attendance and treatment outcomes, and can also be used to make the SMS text intervention interactive [29, 30]. In Ethiopia, access to mobile phones is expanding widely: according to the Ethiopia Demography Health Survey (EDHS) 2016, 88% of urban households and 47% of rural households have mobile phone access [31]. This is substantial growth over 5 years when compared to the EDHS 2011, which indicated 65% of urban households and 13% of rural households had mobile phone access [32].

There is some evidence that interventions that rely on communications, like educating and counselling the patient, or medication monitoring and reminders, can improve treatment adherence and success [28, 33-35]. In Ethiopia, for example, a randomized trial in 2016 found that psychological counselling and educational intervention can significantly improve a patient's treatment adherence [36]. Another trial in Ethiopia in 2019 has investigated the effect of a daily SMS text and weekly medication refill reminders on patient's medication adherence [37]. However, the combination of SMS text and phone calls (as in the intervention proposed here) based on the application of formal behavioural theories and behavioural techniques has not been previously designed and tested. This trial will also apply a combination of direct and indirect methods of adherence measurements for outcomes assessment. The mobile-assisted medication adherence support intervention will be developed systematically based on the Medical Research Council model, incorporating appropriate theory and evidence. The intervention design will be informed through qualitative in-depth interviews with different stakeholder groups such as patients, healthcare providers and healthcare managers, and the SMS content will be validated with experts in behavioural science, healthcare system, pulmonary disease and digital health, using a Delphi technique and will be revalidate by health professionals working at ground level.

Methods

Hypothesis

We plan to test whether our Mobile-assisted Medication Adherence Support (Ma-MAS) intervention using mobile SMS text reminders and phone calls will improve Tuberculosis medication adherence in addition to routine standard non-daily DOT care.

Trial design

A parallel-group individual randomized controlled trial (RCT) with two groups will be employed to investigate the effect of Ma-MAS intervention on TB medication adherence. Participants will be individually randomized into one of the two groups (one intervention and one control group) in a 1:1 allocation ratio (Fig 1).

Study setting

The study will be conducted at the primary public health facilities of Addis Ababa, Ethiopia. Addis Ababa is the capital city of Ethiopia, and the city has 11 administrative sub-city regions. Based on population projection the estimated total population of Addis Ababa in 2017 was 3,433,999 with 1,624,999 Males and 1,809,000 Females [38]. In Ethiopia, a DOT program has

	Study	Period			
	Enrolment	Allocation	Post-allocatio	on	close-out
Timeline	- t1	to	t1	t2	t3
Timeline	Week 1	Week 12	Week 16	Week 20	week 21
Enrolment					
Eligibility screening	**	*			
Informed consent	**	*			
Allocation	*	*			
Intervention					
Daily SMS text					
Weekly Phone calls					
Assessment					
Baseline					
Socio-demographic	**	*			
Clinical characterstics	**	**			
Substance Use-related information	**	**			
Health Service-related information	*	**			
Adherence measured by AIDS Clinical Trial Group adherence questionnaire	**	**			
Adherence measured by Visual Analogue Scales (VAS)	**	**			
Outcome					
Adherence measured by IsoScreen test				**	*
Adherence measured by AIDS Clinical					\$
Trial Group adherence questionnaire					
Adherence measured by Visual					*
Analogue Scales (VAS)					
Adherence measured by clinic					
appointment attendance			V	V	

https://doi.org/10.1371/journal.pone.0261758.g001

been implemented in two phases, "intensive" and "continuation". In the intensive phase (daily DOT), a patient takes their medication in front of the healthcare worker every day at a health facility. In the continuation phase (non-daily DOT), a patient is given the responsibility to take

the medication at home and come to a health facility weekly for health checking and medication refilling [39].

Participants

Adult Tuberculosis patients who are receiving anti-TB treatment at primary public health facilities of Addis Ababa, Ethiopia will be prospective participants for the trial.

Eligibility

Inclusion criteria. To be included in the trial:

- participants must be enrolled in a primary public health facility for anti-TB treatment and have attained their first two months of the intensive phase of treatment
- participants must be aged 18 years and above
- participants must be able to read and understand SMS text that is written in the national official language (Amharic) of Ethiopia
- · participants must have their own mobile phone, or
- participants who do not have a mobile phone can be included if they have a shared mobile phone in the household with a collaborative agreement; (i.e. a voluntary agreement between the patient and family member living in the same household).

Exclusion criteria. Participants will be excluded from the trial under the following conditions:

- participants whose anti-tuberculosis treatment has been underway for more than six months.
- participants who are enrolled or have agreed to enroll in any other interventional study at the same time as this study is conducted.

Intervention group

The intervention will be given in arm-1 (intervention) group participants will receive a daily SMS text and weekly phone calls for medication intake and medication refill visit reminders. All participants will receive the intervention at a similar time for two months during the continuation phase of anti-TB treatment (Fig 2).

Control group

The control group (Arm-2) participants will receive the same routine standard treatment care as the intervention group, but no SMS text and phone calls intervention. Patients in the control group will be followed for the same two-month period as the intervention group during the continuation phase of their treatment.

Intervention development procedure

The mobile SMS text messages will be developed systematically with the best available evidence in the context area and with the application of appropriate behavioural theory and technique. The intervention will be developed and evaluated based on the Medical Research Council



Fig 2. Mobile-assisted Medication Adherence Support (Ma-MAS) intervention diagram.

https://doi.org/10.1371/journal.pone.0261758.g002

(MRC) framework for complex interventions in healthcare (Table 1) [40]. The non-sequential development stage of the framework as adapted by Bleijenberg et al. (2018) will be used as a more specific guide for the intervention design [41]. This has seven different phases:

Phase 1: problem definition (identification and analysis)

Phase 2: systematically identifying the evidence

Phase 3: identifying and developing a theory

Phase 4: determining the needs (patients and/or providers)

Phase 5: examining the practice

Phase 6: modeling process and outcome and

Phase 7: intervention design.

$Table 1. \ Mobile-assisted \ medication \ adherence \ support \ intervention \ development \ process \ using \ an \ adapted \ development \ phase \ of \ MRC \ framework \ and \ the \ associated \ methods.$

Development phases	Methods
Phase 1: Problem identification and definition	Review of literature Qualitative research
Phase 2: Identifying the evidence base	 Review of literature Qualitative research Quantitative Survey research
Phase 3: Identifying and/or developing theory	Review of literature
Phase 4: Determine the need	Review of literature Qualitative research
Phase 5: Examine current practice and context	Qualitative research
Phase 6: Modeling process and outcome	 Modeling a prototype of intervention components Identity interrelation with the outcome Experts opinion through Delphi method and survey
Phase 7: Intervention design	Experts opinion Refine the intervention

https://doi.org/10.1371/journal.pone.0261758.t001

Phase 1: Problem identification and definition. The problem of non-adherence to the anti-TB medication has been identified and defined through a literature review. The literature review is published in the title of "Factors Influencing Patient Adherence to Tuberculosis Treatment in Ethiopia? A Literature Review" [6]. This review will be further supported by qualitative research using in-depth interviews with TB patients and healthcare providers.

Phase 2: Identifying the evidence base. Evidence has been collected through a review of literature, to identify the effectiveness of mobile health SMS interventions including acceptability, feasibility, and characteristics of the intervention. Acceptability and feasibility survey will be conducted before the trial is implemented. This phase aims to identify previous interventions and their effectiveness and to assess the acceptability and feasibility of the target groups of the interventions, as well as the success and failure factors (S1 and S2 Tables).

Phase 3: Identifying and/or developing theory. The theories or models that can best explain the proposed intervention mechanism and action have been selected through a review of the literature. This phase of development aims to identify the theory or model that can best explain the intervention mechanism of action. We have selected an information motivation behavioural skill model [42], and have shown the relationship between the model construct of information, motivation and behavioural skills with adherence behaviour based on the factors affecting TB medication adherence identified in phase one (Fig.3).

Phase 4: Determine the needs. The needs, preferences, perceptions, and capacities of the patient and healthcare providers with the identified problems and proposed solutions will be determined through a review of the literature and qualitative research by using in-depth interviews. This will generate information about which characteristics of the intervention are likeable and adaptable, and how the SMS texts can be tailored regarding language, content, frequency, length, and time of the day.

Phase 5: Examine current practice and context. Identification of barriers and facilitators of mobile health interventions through examining the current practice and context from the patient's and healthcare provider's perspectives will be made. This would help the functionality of the proposed mobile health intervention best fits with the existing practice and context. This activity will be done through qualitative research by using in-depth interviews.

Phase 6: Modelling process and outcome. Modeling the active component of the mobile health intervention and its relationship with outcomes will be refined with multidisciplinary team involvement. The initial SMS text content will be refined using the taxonomy of evidence-based behaviour change [43]. The Delphi technique will be employed to validate the content of the SMS text further. Eight experts from different professional backgrounds of behavioural science, healthcare system, pulmonary disease, and digital health experts will be involved in this process. A survey will be conducted among health professionals who working as TB focal persons and communicable disease control officers from health facilities and health offices in Addis Ababa, Ethiopia.

Phase 7: Intervention design. A full prototype mobile SMS text intervention will be developed and will be given to experts for their feedback to make final refinements. The experts will be asked to comment on the timing and frequency of information, clarity, and arrangement of the SMS text messages. Based on the expert's comments the final SMS text message set will be prepared.

Outcome

The outcome of the trial is the medication adherence of the patient, measured by both direct and indirect methods of measurement. The primary outcome will be measured using an Iso-Screen test (GFC Diagnostics Ltd, Bicester, England) for the presence of isoniazid (INH) drug



Fig 3. Information, motivation and behavioural skills model to Tuberculosis medication adherence adapted from Fisher et al. 2006 [42].

https://doi.org/10.1371/journal.pone.0261758.g003

metabolites in urine in accordance with the vendor-supplied user manual. The IsoScreen test result is negative when there is no colour change (i.e. the strip remains yellow) after 5 minutes, and the test result is positive when the colour changes to dark purple, or blue/ green in that time, depending on the concentration of the medication (i.e. if the medication was taken in the last 24 hours or 48 hours respectively) [44, 45]. The IsoScreen test will be done at the fourth and eighth weeks of the intervention. The proportion of adherence measured by the IsoScreen test at the eighth week of the intervention (primary end point) will be used as the primary interest of analysis. The proportion of adherence measured by the IsoScreen test at the fourth week of the intervention will be used as the secondary interest of analysis.

The secondary outcome will also be measured by self-report of medication adherence using the adapted AIDS Clinical Trial Group adherence questionnaire (ACTG) [46], Visual Analogue Scales (VAS) [47], and by clinic appointment attendance registration [48]. The adapted version of ACTG has three sections: section B (social support), section C (possible reasons for non-adherence) and section D (adherence behaviour). Participant scores less than "somewhat satisfied" to any questions in section B or other than "never" to any questions in sections C

and D are considered as non-adherence [48]. VAS is a single adherence scale that requires a patient to estimate their adherence level from 0% to 100%: 0% means the patient has taken no medication, 50% means the patient has taken half of their prescribed medications, and 100% means the patient has taken all of their prescribed medications, for the last 30 days. Participant scores less than 90% for VAS are considered as non-adherence [36, 49]. Clinic appointment attendance registration for weekly medication refills visits will also be used for adherence measurement. Patients who delay for at least one medication refill visit are considered as non-adherence [48]. The ACTG and VAS questions will be asked at bassline assessment and end of two months of intervention (Fig 1). The proportion of adherence measured by self-report and clinic appointment attendance registration at the eighth week of the intervention will be used as the secondary interest of analysis.

Sample size

A STATA code power twoproportions was used to calculate the sample size considering the trial outcome of anti-TB medication non-adherence. The rate of anti-TB medication nonadherence in Ethiopia varies considerably in the literatures, ranging from 10% to 26% depending on the patient's treatment phase and different settings [8, 11, 14]. It was found specifically that the rate of non-adherence was 25.6% (po = 0.256) during the continuation phase of treatment from a recent study conducted at Addis Ababa, Ethiopia [14] in the control group. We used a 15% absolute reduction of non-adherence due to a mobile SMS text intervention found in a similar study [37] to estimate a non-adherence rate of 10.6% (pa = 0.106) in the intervention group. Considering that we are using a simple randomized controlled trial parallel-group design, choosing a 95% confidence level (alpha = 0.05), 80% power (beta = 0.80) and a onesided p-value of 0.025 for the initial sample size would yield n = 81 participants per arm. Assuming an attrition rate of 15%, the final sample size would be a total of n = 186 with each of the 2 arms having n = 93. From 10 sub-cities in Addis Ababa, a total of 36 primary public health facilities will be included in the trial study by random selection, based on the assumption that on average 6 patients per public health facility will be available to be included in the study.

Recruitment

Patients enrolled in the intensive phase of anti-TB treatment will be recruited by the research support team of healthcare professionals from each health facility, based on the eligibility and exclusion criteria. The recruitment will be continued until the required sample size is obtained. All the recruitment process details will be recorded on paper forms and kept securely by the facilities until the project is completed.

Randomization and allocation

Participants will be centrally randomized into one of the two groups using simple randomization in computer-generated algorithm, with an equal allocation ratio (1:1), by members of the research support team, who will not be involved in measuring the outcome ($\underline{\operatorname{Fig}} 4$).

Follow up

Both the intervention and control groups' patients will be followed for an equal period of 2 months. Intervention group participants will receive a total of 60 daily SMS text reminders and 8 phone calls for medication intake and refill visits reminders.



https://doi.org/10.1371/journal.pone.0261758.g004

Blinding

Single-blinding will be applied to avoid bias in outcome measurement, by differentiating the person who measures the outcome from the person who randomizes and provides the intervention. Thus, the person who measures the outcome will be blinded as to which group

participants are assigned. Due to the nature of the trial intervention participants will not be blinded to which group they are assigned.

Implementation

All patients assigned to any group will get the same standard treatment based on the Ethiopia National Tuberculosis Treatment guideline which is rifampicin, isoniazid, pyrazinamide and ethambutol for two months followed by rifampicin and isoniazid for four months [50]. Patients screening, enrolment, consenting and randomization will be undertaken by the research support team, who have health professional backgrounds. An open SMS automation software platform will be used to store, monitor, and send SMS texts. Global system for mobile (GSM) sim card and short code SMS credit will be used to send the SMS texts. SMS text traffic will be managed and monitored in one centralised computer database server by trained IT personnel. The maximum size of an SMS text will not be more than 160 characteristics. A weekly phone calls will be made to the intervention group participants from the health facility by delegated members of the research team. A checklist of questions will be recorded at each time the weekly phone calls are made to the participants (S2 Checklist). Bothe the SMS text and Phone calls intervention will be made using the national official language which is Amharic. For patients who do not answer phone calls, three repetitive attempts will be made at five-minute intervals. All SMS texts and phone calls will be delivered at a similar time in the morning before the actual medication time. The message delivery reports will be continuously monitored to ensure the participants have received the SMS texts as planned. Participants will also be asked whether they are receiving the SMS text every week through phone calls and when they come for their medication refill visit. Participants will be briefed on the duration, frequency, and timing of SMS text and phone calls intervention and advised to turn on their mobile in the morning. Participants will be informed about the intervention and instructed not to tell and/or share the SMS texts or phone calls with other people. If participants lose or change their phone numbers, the new phone number will be replaced immediately after they have informed a support team member. The trial will be audited with participant enrolment, consent, allocation, and during the intervention period.

Data collection and management

A trained healthcare professional team will collect and supervise the data collection. Data collection will be made after the patient has received the routine non-daily DOT services in a different room from the TB clinic room, to ensure the confidentiality and privacy of the patient's information. The data will be collected electronically using a smart mobile app through a questionnaire coded using Open Data Kit (ODK). Data will be collected from the primary source through interviews and sample urine collection, and secondary source by a review of the health facility attendance registration book to confirm the patient's medication refill visit attendance. All interviews will be conducted after the patients finish the non-daily DOT service. Interviews will be conducted in a different room from the TB clinic to be convenient for discussion and privacy.

To ensure the security of the patients' information, all data will keep in Open Data Kit (ODK) database, principal investigator laptop computer, and backup file with password protectors. All identifiable information will be removed and replaced by code. All hard-copy research documents will be locked in a cabinet.

Statistical analysis

The electronically collected data will be exported to STATA and cleaned using STATA coding programing language. Descriptive statistics of mean, standard deviation, median, and range will be used for continuous variables and frequency and percentage for categorical variables.

Intention to treat (ITT) analysis will be performed for all randomized participants according to their group. A chi-square test for binary outcomes and Student's t-test for continuous outcomes will be used as appropriate. For primary and secondary interest of analysis, multivariable regression model analysis will be employed to adjust confounding factors in the statistical effect. A P-value of less than 0.05 will be used as a cut-off point to show a statistically significant association between Ma-MAS intervention and medication adherence with a 95% confidence interval.

Ethical considerations

The research proposal has been approved by the Southern Adelaide Clinical Health Research Ethics committee (SAC HREC) on April 8, 2020, with Office for Research (OFR) number: 23.20. The proposal has also be approved by the Addis Ababa Health Bureau ethics committee on August 21, 2020, with reference number A/A/H/930/227 and an official permission letter is provided from Addis Ababa Health Bureau to respective sub-city and health facilities. Each participant will complete a written informed consent form after being informed about the research objectives, procedures, anticipated risks, and benefits of the research. The research team will be confirming or renegotiating consent with participants for additional data. The principal investigator will be ensuring participants are confirming their consent to participation. Participants have the right to withdraw from the study at any time when they feel uncomfortable or unable to continue in the study, and they will be informed of its limitations or consequences on the research during consent requests. Any change in the study protocol and participant informed consent will be submitted to SAC HREC before implementing the change. This trial has no forms of inducement, coercion and the study does not bring any risks that incur compensation in cash or kind.

Publication and dissemination of the result

The trial findings will be reported based on the Consolidated Standard of Reporting Trials (CONSORT) guidelines. The research outcomes will be published in an international peerreviewed open access journal. The research outcomes will also be presented at selected national and international conferences. The principal investigator is responsible for disseminating the finding to Health facilities, Addis Ababa Health Bureau, Federal Ministry of Health (FMOH), Flinders University, national and international conferences. The participants will be asked whether they wish to get a summary of the research work, and if so in what form, and will be informed about how they can access the research results. In the case of a formal request for further research purposes, non-identifiable secondary data may be granted to others. A participant's identifiable information will be removed and replaced by code to protect the anonymity of the participants. Identifiable data and/or information will not be accessed or granted to others.

The trial is registered in the Pan-Africa Clinical Trials Registry with trial number <u>PACTR202002831201865</u>.

Discussion

An adherence intervention needs to address the underlying factors that cause medication nonadherence [26]. Medical Research Council (MRC) guidelines recommend using appropriate theory and evidence when developing an intervention [40]. However, most adherence interventions are developed without a strong theoretical base, which might be one of the reasons they have not been effective [51]. In this randomized controlled trial, the Mobile-assisted Medication Adherence Support (Ma-MAS) intervention will be developed systematically using the MRC conceptual framework with an appropriate selection of behavioural theory.

Some evidence has shown that interactive SMS texts may have a better impact than using simple SMS text reminders in improving medication adherence [52, 53] although there is also inconsistent evidence found in some two-way [30, 54] and one-way SMS text studies among TB patients [55, 56]. This trial was designed to test an intervention with a combination of daily SMS text and weekly phone calls for medication intake and medication refill visit reminders during the continuation phase of treatment for a two months period, which was not studied previously.

When testing an intervention with randomized trials, correctly measuring outcomes in the same way for both the intervention and control groups is crucial. The accuracy of adherence measurement is very important to assess the true effect of an adherence intervention. Here the adherence outcome will be measured through the combination of a direct method (IsoScreen test) and several indirect methods (VAS, ACTG, and clinic attendance registration) of measurement. This will help to avoid bias in outcome measurement. Therefore, this research would provide evidence to help healthcare managers and policymakers to develop strategies that improve Tuberculosis medication adherence and treatment outcomes in general.

Supporting information

S1 Checklist. SPIRIT 2013 checklist: Recommended items to address in a clinical trial protocol and related documents*.

(DOC)

S2 Checklist. English version checklist for phone calls intervention. (PDF)

S1 Table. SMS text messaging interventions for TB treatment support. (DOCX)

S2 Table. SMS interventions for TB: Acceptability and feasibility study design. (DOCX)

S1 File. Participant information sheet/consent form intervention and post-intervention participants.

(PDF)

S2 File. Mobile-assisted Medication Adherence Support (Ma-MAS) intervention for Tuberculosis patients: Adoption of intention, effectiveness, and experiences in Addis Ababa, Ethiopia. (PDF)

(121)

S3 File. Pan African Clinical Trials Registry. (PDF)

Acknowledgments

We would like to thank the Addis Ababa Health Bureau and Federal Ministry of Health of Ethiopia for facilitating the necessary conditions to conduct the trial. We would also like to thank Shahid Ullah from Flinders University, College of Medicine and Public Health, for advising us on sample size calculation.

Author Contributions

Conceptualization: Zekariyas Sahile, Lua Perimal-Lewis, Paul Arbon, Anthony John Maeder. Funding acquisition: Zekariyas Sahile, Anthony John Maeder.

Methodology: Zekariyas Sahile, Lua Perimal-Lewis, Paul Arbon, Anthony John Maeder.

Project administration: Zekariyas Sahile.

Supervision: Lua Perimal-Lewis, Paul Arbon, Anthony John Maeder.

- Writing original draft: Zekariyas Sahile, Lua Perimal-Lewis, Paul Arbon, Anthony John Maeder.
- Writing review & editing: Zekariyas Sahile, Lua Perimal-Lewis, Paul Arbon, Anthony John Maeder.

References

- Yew WW. Directly observed therapy, short-course: the best way to prevent multidrug-resistant tuberculosis. Chemotherapy. 1999; 45(Suppl. 2):26–33. <u>https://doi.org/10.1159/000048479</u> PMID: <u>10449895</u>
- Pablos-Méndez A, Knirsch CA, R Graham Barr M, Lerner BH, Frieden TR. Nonadherence in tuberculosis treatment: predictors and consequences in New York City. The American journal of medicine. 1997; 102(2):164–70. https://doi.org/10.1016/s0002-9343(96)00402-0 PMID: 9217566
- Barclay E. Text messages could hasten tuberculosis drug compliance. The Lancet. 2009; 373 (9657):15–6. https://doi.org/10.1016/s0140-6736(08)61938-8 PMID: 19125443
- Zegeye A, Dessie G, Wagnew F, Gebrie A, Islam SMS, Tesfaye B, et al. Prevalence and determinants of anti-tuberculosis treatment non-adherence in Ethiopia: A systematic review and meta-analysis. PloS one. 2019; 14(1):e0210422. <u>https://doi.org/10.1371/journal.pone.0210422</u> PMID: <u>30629684</u>
- Ogundele OA, Moodley D, Seebregts CJ, Pillay AW, editors. An ontology for tuberculosis treatment adherence behaviour. Proceedings of the 2015 Annual Research Conference on South African Institute of Computer Scientists and Information Technologists; 2015: ACM.
- Nezenega ZS, Perimal-Lewis L, Maeder AJJIJoER, Health P. Factors Influencing Patient Adherence to Tuberculosis Treatment in Ethiopia: A Literature Review. 2020; 17(15):5626. <u>https://doi.org/10.3390/ ijerph17155626</u> PMID: <u>32759876</u>
- Tola H, Holakouie-Naieni K, Tesfaye E, Mansournia M, Yaseri MJTIJoT, Disease L. Prevalence of tuberculosis treatment non-adherence in Ethiopia: a systematic review and meta-analysis. 2019; 23 (6):741–9.
- Mekonnen HS, Azagew AW. Non-adherence to anti-tuberculosis treatment, reasons and associated factors among TB patients attending at Gondar town health centers, Northwest Ethiopia. BMC Res Notes. 2018; 11(1):691. https://doi.org/10.1186/s13104-018-3789-4 PMID: 30285907
- Habteyes Hailu T, Azar T, Davoud SHOJAEIZADEH GGJljoph. Tuberculosis treatment non-adherence and lost to follow up among TB patients with or without HIV in developing countries: a systematic review. 2015; 44(1):1. PMID: 26060770
- Kebede A, Wabe NT. Medication adherence and its determinants among patients on concomitant tuberculosis and antiretroviral therapy in South west ethiopia. N Am J Med Sci. 2012; 4(2):67–71. https://doi.org/10.4103/1947-2714.93376 PMID: 22408750
- Adane AA, Alene KA, Koye DN, Zeleke BM. Non-adherence to anti-tuberculosis treatment and determinant factors among patients with tuberculosis in northwest Ethiopia. PLoS One. 2013; 8(11):e78791. https://doi.org/10.1371/journal.pone.0078791 PMID: 24244364
- Tesfahuneygn G, Medhin G, Legesse M. Adherence to Anti-tuberculosis treatment and treatment outcomes among tuberculosis patients in Alamata District, northeast Ethiopia. BMC Res Notes. 2015; 8:503. https://doi.org/10.1186/s13104-015-1452-x PMID: 26420164
- Eticha T, Kassa EJJoB, Biomedicine. Non-adherence to anti-TB drugs and its predictors among TB/HIV co-infected patients in Mekelle, Ethiopia. 2014; 6(6):61.
- Tola HH, Garmaroudi G, Shojaeizadeh D, Tol A, Yekaninejad MS, Ejeta LT, et al. The effect of psychosocial factors and patients' perception of tuberculosis treatment non-adherence in Addis Ababa, Ethiopia. 2017; 27(5):447–8.

- 15. World Health Organization. An expanded DOTS framework for effective tuberculosis control. Int J Tuberc Lung Dis. 2002; 6:378–88. PMID: <u>12019913</u>
- Karumbi J, Gamer P. Directly observed therapy for treating tuberculosis. Cochrane Database of Systematic Reviews. 2015(5). <u>https://doi.org/10.1002/14651858.CD003343.pub4</u> PMID: 26022367
- McKay B, Castellanos M, Ebell M, Whalen CC, Handel A. An attempt to reproduce a previous metaanalysis and a new analysis regarding the impact of directly observed therapy on tuberculosis treatment outcomes. PIoS one. 2019; 14(5):e0217219. <u>https://doi.org/10.1371/journal.pone.0217219</u> PMID: <u>31120965</u>
- Sagbakken M, Frich JC, Bjune G. Barriers and enablers in the management of tuberculosis treatment in Addis Ababa, Ethiopia: a qualitative study. BMC public health. 2008; 8(1):11.
- Fiseha D, Demissie M. Assessment of Directly Observed Therapy (DOT) following tuberculosis regimen change in Addis Ababa, Ethiopia: a qualitative study. BMC infectious diseases. 2015; 15(1):405.
- 20. World Health Organization. Guidelines for treatment of drug-susceptible tuberculosis and patient care. 2017.
- Mohammed H, Oljira L, Roba KT, Yimer G, Fekadu A, Manyazewal TJIDoP. Containment of COVID-19 in Ethiopia and implications for tuberculosis care and research. 2020; 9(1):1–8.
- Barkman C, Weinehall L. Policymakers and mHealth: roles and expectations, with observations from Ethiopia, Ghana and Sweden. Global health action. 2017; 10(sup3):1337356. <u>https://doi.org/10.1080/</u> 16549716.2017.1337356 PMID: <u>28838303</u>
- 23. Davey S, Davey A, Singh JVJljocmopolAoP, Medicine S. Mobile-health approach: a critical look on its capacity to augment health system of developing countries. 2014; 39(3):178.
- Goel S, Bhatnagar N, Sharma D, Singh AJJm, uHealth. Bridging the human resource gap in primary health care delivery systems of developing countries with mhealth: narrative literature review. 2013; 1 (2):e25.
- 25. Tegegne T, Weide T. Is mHealth Viable to Ethiopia: an Empirical Study. 2011.
- Subbaraman R, de Mondesert L, Musiimenta A, Pai M, Mayer KH, Thomas BE, et al. Digital adherence technologies for the management of tuberculosis therapy: mapping the landscape and research priorities. BMJ Glob Health. 2018; 3(5):e001018. <u>https://doi.org/10.1136/bmjgh-2018-001018</u> PMID: 30364330
- LESTER R, Jay J, BOLTEN LM, ENJETTI A, JOHNSTON JC, SCHWARTZMAN K, et al. Mobile Phone Short Message Service for Adherence Support and Care of Patients with Tuberculosis Infection: Evidence and Opportunity. Journal of Clinical Tuberculosis and Other Mycobacterial Diseases. 2019:100108. <u>https://doi.org/10.1016/j.jctube.2019.100108</u> PMID: <u>31720432</u>
- Liu Q, Abba K, Alejandria MM, Sinclair D, Balanag VM, Lansang MAD. Reminder systems to improve patient adherence to tuberculosis clinic appointments for diagnosis and treatment. Cochrane Database of Systematic Reviews. 2014(11). https://doi.org/10.1002/14651858.CD006594.pub3 PMID: 25403701
- Kunawararak P, Pongpanich S, Chantawong S, Pokaew P, Traisathit P, Srithanaviboonchai K, et al. Tuberculosis treatment with mobile-phone medication reminders in northern Thailand. 2011; 42 (6):1444. PMID: <u>22299414</u>
- Mohammed S, Glennerster R, Khan AJ. Impact of a Daily SMS Medication Reminder System on Tuberculosis Treatment Outcomes: A Randomized Controlled Trial. PLoS One. 2016; 11(11):e0162944. <u>https://doi.org/10.1371/journal.pone.0162944</u> PMID: <u>27802283</u>
- 31. Central Statistical Agency CSAE, Icf. Ethiopia Demographic and Health Sruvey 2016. Addis Ababa, Ethiopia: CSA and ICF; 2017.
- 32. Central Statistical Agency. Ethiopia Demographic and Health Sruvey 2011. Addis Ababa, Ethiopia; 2012.
- Nglazi MD, Bekker L-G, Wood R, Hussey GD, Wiysonge CS. Mobile phone text messaging for promoting adherence to anti-tuberculosis treatment: a systematic review. BMC infectious diseases. 2013; 13 (1):566.
- Chen S, Pan Y, Qu Y, Liu Z, Yao L, Xiang L. Mobile phone based interventions for promoting adherence to tuberculosis treatment: a systematic review and meta-analysis. The Lancet. 2015; 386:S29.
- Alipanah N, Jarlsberg L, Miller C, Linh NN, Falzon D, Jaramillo E, et al. Adherence interventions and outcomes of tuberculosis treatment: A systematic review and meta-analysis of trials and observational studies. PLoS Med. 2018; 15(7):e1002595. <u>https://doi.org/10.1371/journal.pmed.1002595</u> PMID: 29969463
- 36. Tola HH, Shojaeizadeh D, Tol A, Garmaroudi G, Yekaninejad MS, Kebede A, et al. Psychological and educational intervention to improve tuberculosis treatment adherence in Ethiopia based on health belief model: a cluster randomized control trial. 2016; 11(5).

PLOS ONE | https://doi.org/10.1371/journal.pone.0261758 December 31, 2021

15/16

- 37. Gashu KD, Gelaye KA, Lester R, Tilahun B. Combined effect of pill refilling and self-medication reminder system on patients' adherence to tuberculosis treatment during continuation phase in Northwest Ethiopia: a study protocol for randomised controlled trial. BMJ Health Care Inform. 2019; 26(1). https://doi.org/10.1136/bmjhci-2019-100050 PMID: 31484661
- Strathdee SA, Magis-Rodriguez C, Mays VM, Jimenez R, Patterson TL. The emerging HIV epidemic on the Mexico-U.S. border: an international case study characterizing the role of epidemiology in surveillance and response. Ann Epidemiol. 2012; 22(6):426–38. https://doi.org/10.1016/j.annepidem.2012.04. 002 PMID: 22626001
- Kamath-Rayne BD, Thukral A, Visick MK, Schoen E, Amick E, Deorari A, et al. Helping Babies Breathe, Second Edition: A Model for Strengthening Educational Programs to Increase Global Newborn Survival. Global health, science and practice. 2018; 6(3):538–51. <u>https://doi.org/10.9745/GHSP-D-18-00147</u> PMID: <u>30287531</u>
- 40. Craig P, Dieppe P, Macintyre S, Michie S, Nazareth I, Petticrew M, et al. Developing and evaluating complex interventions: the new Medical Research Council guidance. BMJ. 2008; 337:a1655. <u>https://doi.org/10.1136/bmj.a1655</u> PMID: <u>18824488</u>
- Bleijenberg N, Janneke M, Trappenburg JC, Etterna RG, Sino CG, Heim N, et al. Increasing value and reducing waste by optimizing the development of complex interventions: Enriching the development phase of the Medical Research Council (MRC) Framework. 2018; 79:86–93.
- Fisher JD, Fisher WA, Amico KR, Harman JJJHP. An information-motivation-behavioral skills model of adherence to antiretroviral therapy. 2006; 25(4):462.
- 43. Michie S, Richardson M, Johnston M, Abraham C, Francis J, Hardeman W, et al. The behavior change technique taxonomy (v1) of 93 hierarchically clustered techniques: building an international consensus for the reporting of behavior change interventions. 2013; 46(1):81–95.
- Mkopi A, Range N, Lwilla F, Egwaga S, Schulze A, Geubbels E, et al. Validation of indirect tuberculosis treatment adherence measures in a resource-constrained setting. Int J Tuberc Lung Dis. 2014; 18 (7):804–9. https://doi.org/10.5588/ijttd.13.0675 PMID: 24902556
- 45. Soobratty MR, Whitfield R, Subramaniam K, Grove G, Carver A, O'Donovan GV, et al. Point-of-care urine test for assessing adherence to isoniazid treatment for tuberculosis. Eur Respir J. 2014; 43 (5):1519–22. <u>https://doi.org/10.1183/09031936.00132613</u> PMID: 24435008
- Chesney MA, Ickovics J, Chambers D, Gifford A, Neidig J, Zwickl B, et al. Self-reported adherence to antiretroviral medications among participants in HIV clinical trials: the AACTG adherence instruments. 2000; 12(3):255–66.
- Kalichman SC, Amaral CM, Swetzes C, Jones M, Macy R, Kalichman MO, et al. A simple single-item rating scale to measure medication adherence: further evidence for convergent validity. 2009; 8 (6):367–74.
- Boogaard Jvd, Lyimo RA, Boeree MJ, Kibiki GS, Aarnoutse RE. Electronic monitoring of treatment adherence and validation of alternative adherence measures in tuberculosis patients: a pilot study. Bulletin of the World Health Organization. 2011; 89:632–9. https://doi.org/10.2471/BLT.11.086462 PMID: 21897483
- Valencia S, León M, Losada I, Sequera VG, Fernández Quevedo M, García-Basteiro AL. How do we measure adherence to anti-tuberculosis treatment? Expert Review of Anti-infective Therapy. 2017; 15 (2):157–65. https://doi.org/10.1080/14787210.2017.1264270 PMID: 27910715
- 50. The Federal Ministry of Health. National Guidelines for TB, DR-TB and Leprosy in Ethiopia. 2018.
- Horne R, Weinman J, Barber N, Elliott R, Morgan M, Cribb A, et al. Concordance, adherence and compliance in medicine taking. 2005; 2005:40–6.
- Kanters S, Park JJ, Chan K, Socias ME, Ford N, Forrest JI, et al. Interventions to improve adherence to antiretroviral therapy: a systematic review and network meta-analysis. The lancet HIV. 2017; 4(1):e31– e40. https://doi.org/10.1016/S2352-3018(16)30206-5 PMID: 27863996
- Bediang G, Stoll B, Elia N, Abena JL, Geissbuhler A. SMS reminders to improve adherence and cure of tuberculosis patients in Cameroon (TB-SMS Cameroon): a randomised controlled trial. BMC Public Health. 2018; 18(1):583. https://doi.org/10.1186/s12889-018-5502-x PMID: 29720146
- Liu X, Lewis JJ, Zhang H, Lu W, Zhang S, Zheng G, et al. Effectiveness of Electronic Reminders to Improve Medication Adherence in Tuberculosis Patients: A Cluster-Randomised Trial. PLoS Med. 2015; 12(9):e1001876. <u>https://doi.org/10.1371/journal.pmed.1001876</u> PMID: 26372470
- Fang XH, Guan SY, Tang L, Tao FB, Zou Z, Wang JX, et al. Effect of short message service on management of pulmonary tuberculosis patients in Anhul Province, China: A prospective, randomized, controlled study. Medical Science Monitor. 2017; 23:2465–9. <u>https://doi.org/10.12659/msm.904957</u> PMID: 28534476
- Broomhead S, Mars MJT, e-Health. Retrospective return on investment analysis of an electronic treatment adherence device piloted in the Northern Cape Province. 2012; 18(1):24–31.

PLOS ONE | https://doi.org/10.1371/journal.pone.0261758 December 31, 2021

16/16

Article 3: Mobile-assisted medication adherence support intervention among Tuberculosis patients: a parallel group randomised control trial (Conference Abstract)



< >

Mobile-assisted medication adherence support intervention among tuberculosis patients: a parallel group randomized control trial

Zekariyas Sahile¹, Anthony John Maeder², Lua Perimal-Lewis², Paul Arbon²

More details

Popul. Med. 2023;5(Supplement):A585

> DOI: https://doi.org/10.18332/popmed/163638

🕒 Article (PDF)

ABSTRACT

Background:

There is conflicting evidence regarding the efficacy of SMS texts to promote TB medication adherence. Thus, we developed a mobile-assisted medication adherence support (Ma-MAS) intervention using the medical research council (MRC) framework based in a local context targeting audience needs and the effectiveness of this intervention evaluated in a sample of TB patients in Addis Ababa, Ethiopia.

Methods:

A parallel group randomized control trial design was used to evaluate the effectiveness of Ma-MAS intervention. In total, 186 adult TB patients (93 per group) were randomly assigned 1:1 to one of the two groups. Participants in the Ma-MAS group received daily SMS texts and weekly phone calls regarding their daily medication intake and reminders to attend clinic visits for 8 weeks. Participants in the control group did not receive SMS texts or phone calls but received the same routine standard care as the Ma-MAS group. The primary outcome was the proportion of adherence measured by urine tests for isoniazid (INH) metabolites at the end of the 4th and 8th week of follow-up period. Analysis of a multivariable binomial generalized linear model was employed to evaluate the effect of Ma-MAS at P value< 0.05.

Results:

Ma-MAS significantly improved adherence to TB medications by 15.25% (95%CI: 5.38, 25.12; P-value=0.0065) after 8 weeks of intervention compared to the standard care alone in the control group. At the end of the 8 weeks follow-up period, the predicted probability of adherence to TB medication in the Ma-MAS group was 86% (95%CI: 81, 93), and in the control group was 70% (95%CI: 61, 79). Ma-MAS also improved adherence to TB medication by 15.30% (95%CI: 6.68, 23.90; P-value=0.0022) after 4 weeks of intervention compared with the control group.

Conclusion:

Ma-MAS intervention based on information-motivation-behavioural skills model and behavioural change techniques is useful to improve adherence to TB medication.

ISSN: 2654-1459

 \wedge
APPENDIX NINE: QUESTIONNAIRE

In-depth Interview Guide for Ma-MAS Intervention Development

Introduction

Hello my name is______. I am going to ask you a few questions regarding your perceptions and experiences on mobile SMS text and phone calls intervention development for TB adherence support during your treatment follow-up.

Section 1: Participant characteristics

S. No	Characteristics	Response
1	Sex	
2	Age	
3	Educational status	
4	Occupational status	

Section 2: In-depth interview guide questions

- 1. Is non-adherence to TB medication is a problem in Addis Ababa, Ethiopia? (Probe: why that?)
- 2. What interventions are available to address the problem of non-adherence? (Probe: what current and past interventions have been implemented?)
- 3. Are mobile SMS text and/or phone call-based interventions a possible solution for nonadherence to TB medications? (Probe: how?)
- 4. How can mobile SMS text and/or phone call-based interventions be designed to make them desirable and culturally appropriate? (Probe: content, frequency, length of intervention, time of day?)
- How can mobile SMS text and/or phone call be made more interactive and engaging? (Probe: two-way SMS? phone call conversation?)
- Would a daily programme of mobile SMS text and/or phone call-based reminders to improve TB medication adherence be desirable? (Probe: how? what do you suggest in the programme?)
- 7. Would a weekly programme of mobile SMS text and/or phone call-based reminders to encourage medication refill visits improve a patient's clinic attendance? (Probe: how? what do you suggest?)
- 8. Is a mobile SMS text and/or phone call-based intervention compatible with the current nondaily DOT service? (Probe: what opportunities and challenges are there?)
- 9. What would be the challenges if mobile SMS and/or phone call interventions are being implemented? (Probe: to system level, healthcare provider, patient, and any risk associated)
- Would mobile SMS text and/or phone call intervention help healthcare professionals in delivering better care services? (Probe: how?)

Feasibility and Acceptability Questionnaire

Identification					
Sub-city Name:	_				
Health facility Name:					
Participant registration code:	Participant registration code:				
Date of interview	Time started	Time finished			
Data collector's name	Signature	Date			
Supervisor's name	Signature	Date			

Section 1: Socio-Demographic Characteristics Information

	QEUSTION	Response	Code
S.		-	
No 101	Your sex	1= Male 2=Female	
101			
102	Your age in years	1 Marriad	
103	Your marital status	1=Married 2=Divorced/separated 3=Widowed 4=Never Married	
104	Your educational status	1=No formal education 2=Primary 1-8 3=Secondary 9-10 4=Technical/Vocational 11-12 5= Higher Education 12+	
105	Your occupational status	1=Governmental employee 2=Daily labourer 3=Merchant 4=Housewife 5=Student 6=Driver 7=Guard 8=Pensioner 9= Unemployment Other (specify)	
106	What is the family size of your household (include all related persons residing there)		
107	What is the average monthly household income (sum of incomes contributed by all family members)		
108	How much time does it take you on average to travel to this health facility?		
109	Which kind of transportation do you use to travel to this health facility?	1=Taxi 2=Public Transport/Bus 3=Bajaj 4=Bicycle/ Motor bicycle 5=Walk Other (specify)	
110	If you paid for your transportation here, how much Ethiopian birr did you pay per visit?		

Intervention description

Mobile-assisted medication adherence support (Ma-MAS) intervention is a systematically developed behavioural intervention that is delivered through mobile SMS text and phone calls to improve a TB patient's medication adherence and clinic appointment attendance. Mobile SMS text and phone calls will be sent to remind patients to take their medication at home and to collect refills of their medication during the TB treatment. I will ask you a few questions based on this intervention.

S.NO	Question	Response	Code	
200	De veu have vour own Mehile phone?	1=Yes	1 1	
200	Do you have your own Mobile phone?	2=NO	II	
		1=Smart phone (install		
201	Type of mobile phone	and use applications)		
201		2. Basic (only used for		
		voice and text messages)		
	Do you have a trusted family or caregiver in the	1=Yes		
202	house that can share their mobile phone with	2=NO		
	you?			
203	Are you confident using your mobile phone to	1=Yes	1 1	
203	receive and open SMS text?	2=NO	II	
204	Are you confident to read an SMS text message	1=Yes	1 1	
204	on your mobile phone?	2=NO	II	
205	Are you confident using your mobile phone to	1=Yes	1 1	
205	send an SMS text?	2=NO		
206	Are you confident using your mobile phone to	1=Yes	1 1	
200	make a phone call?	2=NO	II	
207	Are you confident using your mobile phone to set	1=Yes	1 1	
207	a reminder?	2=No	II	
208	Are you confident using your mobile phone to	1=Yes	1 1	
200	send a "Please Call Me"?	2=NO	II	
209	Are you confident using your mobile phone to add	1=Yes		
203	a contact?	2=NO	II	
210	Are you willing to receive SMS reminders to take	1=Yes		
210	your medications?	2=NO	II	
211	Are you willing to receive SMS reminders to your	1=Yes		
~ ~ ~ ~	clinic appointments?	2=NO	II	

Section 2: Feasibility and Acceptability of Mobile-assisted Medication Adherence Support (Ma-MAS) Intervention

r			
212	Are you willing to receive a phone call reminder to	1=Yes	1 1
	take your medications?	2=NO	II
213	Are you willing to receive a phone call reminder to	1=Yes	1 1
215	your clinic appointment?	2=NO	II
	Are you willing to receive a SMS text reminder to	1=Yes	
214	take your medications on a trusted family	2=NO	
	member's shared mobile phone?		
	Are you willing to receive SMS text reminder to	1=Yes	
215	your clinic appointment on a trusted family	2=NO	
	member's shared mobile phone?		
	Are you willing to receive a phone call reminder to	1=Yes	
216	take your medications on a trusted family	2=NO	
	member's shared mobile phone?		
	Are you willing to receive a phone call reminder to	1=Yes	
217	your clinic appointment on a trusted family	2=NO	
	member's shared mobile phone?		
218	Total cost for SMS text message (from record		
210	review)		II
219	Total cost for a phone call (from record review)		
220	Total cost for Ma-MAS intervention (from record		1 1
220	review)		II
	1	I	1

Section 3: Disease-related Information (from record review)

S.NO	Question	Response	Code
300	TB diagnosis category?	1=Smear positive pulmonary TB 2=Smear Negative pulmonary TB	
		3=Extra Pulmonary TB	
301	TB HIV co-infection?	1=Yes	
		2=No	
302	Any concurrent non-communicable disease?	1=Diabetes	
		2=Cancer	
		3=Hypertension	
		4=Other	
		(specify)	
303	History of Tuberculosis?	1=New case	
	-	2=Failure	
		3=Relapsed case	
		4=Default	

Ma-MAS Intervention Trial Questionnaire

	Identification		
Sub-city Name:			
Health facility Name:			
Participant registration code:			
Participant Phone no 1:	Phone no 2:		_
Date of interview	Time started	Time finished	
Data collector's name	Signature	Date	
Supervisor's name	Signature	Date	

Section 1: Socio-Demographic Characteristics information

S.NO	Question	Response	Cod e
101	Your sex	1= Male 2=Female	
102	Your age in years		
103	Your marital status	1=Married 2=Divorced/separated 3=Widowed 4=Never Married	
104	Your educational status	1=No formal Education 2=Primary 1-8 3=Secondary 9-10 4=Technical/Vocational 11- 12 5=Higher Education 12+	
105	Your occupational status	1=Governmental employee 2=Daily laborer 3= Merchant 4=Housewife 5=Student 6=Driver 8=Guard 7=Pensioner Other (specify)	
106	What is the family size of your household (include all related persons residing there)		
107	What is the average monthly household income (sum of incomes contributed by all family members)		
108	How much time does it take you on average to travel to this health facility?		
109	Which kind of transportation do you use to travel to this health facility?	1=Taxi 2=Public transport/Bus 3= Bajaj 3=Walk Other (specify)	
110	If you paid for your transportation here, how much Ethiopian birr did you pay per visit?		
111	Do you have access to adequate food?	1=Yes 2=No	
112	Do you get food support from any other source?	1=Yes 2=No	
113	Do you get financial/social support from any other source?	1=Yes 2=No	
114	Does your family member or caregiver remind you to take your medications?	1=Yes 2=No	

115	Does your family member or caregiver remind you to attend your clinic appointment?	1=Yes	2=No	
116	Have you experienced any stigma and/or discrimination because of your disease?	1=Yes	2=No	

Section 2: Disease-related information

S.NO	Question	Response	Code
200	TB diagnosis category	1=Smear positive pulmonary TB 2=Smear Negative pulmonary TB 3=Extra Pulmonary TB	
201	TB HIV co-infection	1=Yes 2=No	
202	Any other co-morbidity?	1=Diabetes 2=Cancer 3=Hypertension 4=Other (specify	
203	History of Tuberculosis	1=New case 2=Failure 3=Relapsed case 4=Default	

Section 3: Substance Use-related information

S.NO	Question	Responses	Code
300	Have you ever drunk alcohol?	1=Yes 2=NO (Skip Q 301)	
301	If yes for Q. 300, do you drink alcohol currently?	1=Daily 2=Weekly 3=Occasionally	
302	Have you ever smoked cigarettes?	1=Yes 2=NO (Skip Q 303)	
303	If Yes for Q. 302, do you currently smoke cigarettes?	1=Daily 2=Weekly 3=Occasionally	
304	Do you ever chew khat?	1=Yes 2=NO (Skip Q 305)	
305	If yes for Q. 304, do you chew khat currently?	1=Daily 2=Weekly 3=Occasionally	
306	Do you ever use any other substance like shisha, ganja?	1=Yes (specify) 2=NO (Skip Q 307)	
307	If yes for Q. 306, do you use it currently?	1=Daily 2=Weekly 3=Occasionally	

Section 4: Health Service-related information

S.NO	Question	Response	Code
400	Is the waiting time to get the health service acceptable?	1=Yes 2=No	
401	How do you judge your relationship with health care providers?	1=Very good 2=Good 3=Poor 4=Very poor	
402	How do you judge the quality of health service you obtained from this facility?	1=Very good 2=Good 3=Poor 4=Very poor	

Section 5: Treatment adherence questionnaire

Instructions: Please answer the following questions by placing a " $\sqrt{"}$ in the appropriate box.

A. The following questions ask about your medication. (Please check one box for each question.)

	Not at all Sure	Some what Sure	Very Sure	Extremely Sure
1. How sure are you that you will be able to take all or most of the medication as directed?				
2. How sure are you that the medication will have a positive effect on your health?				
3. How sure are you that if you don't take this medication exactly as instructed, the TB in your body will become resistant to TB medications?				

B. The following questions ask about your social support. (Please check one box for each question.)

	Very Dissatisfie d	Somewhat Dissatisfie d	Somewhat Satisfied	Very Satisfie d
1. In general, how satisfied are you with the overall support you get from your friends and family?				
2. To what extent do your friends or family members help you remember to take your medication?				

C. If you have NOT taken any medication within the past month, please check this box and skip to section D:

People may miss taking their medication for various reasons. Here is a list of possible reasons why you may have missed taking medication. (*Please check one box for each question*.) In the <u>past month</u>, how often have you missed taking your medication because you:

		Never	Rarely	Sometimes	Often
1.	Were away from home?				
	Were busy with other things?				
3.	Simply forgot?				
4.	Had too many pills to take?				
5.	Wanted to avoid side effects?				
6.	Did not want others to notice that you were taking medication?				
7.	Had a change in daily routine?				
	0.40				

8. Felt like the drug was toxic/harmful?		
Fell asleep/slept through dose time		
10. Felt sick/ill		
11. Felt depressed/overwhelmed?		
12. Had problem taking pills at specified times (with meals, on empty stomach, etc.)?		
13. Ran out of pills		
14. Felt good		

D. When was the last time you missed taking any of your medication? (*Please check only one box*)

1. Within the past week	
2. 1-2 weeks ago	
3. 2-4 weeks ago	
4. Never skip medication or not applicable	

Visual Analogues scale

Put a cross on the line below at the point showing your best guess about how much of your TB medication you have taken (baseline) in the past one month. For example, 0% means you have taken no medication, 50% means you have taken half of your prescribed medications, and 100% means you have taken all of your prescribed medications.

										i I	
0 100%	%	10%	20%	30%	40%	50%	60%	70%	80%	90%	

Section 6: Isoscreen test (Urine test)

S.No	Urine Test	Test result	Code
		1=Purple	
600	At the fourth week of the study	2=Blue	
600	assessment	3=Green	
		4=Yellow	
		1=Purple	
601	At the eighth week of the study	2=Blue	
601		3=Green	
		4=Yellow	

Adoption Intention of Ma-MAS Intervention Questionnaire

	Identification	
Sub-city Name:	_	
Health facility Name:		
Participant registration code:		
Date of interview	Time started	Time finished
Data collector's name	Signature	Date
Supervisor's name	Signature	Date

Section 1: Socio-Demographic Characteristics Information

	QEUSTION	Response	Code
S.No		-	
101	Your sex	1= Male 2=Female	
102	Your age in years		
103	Your marital status	1=Married 2=Divorced/separated 3=Widowed 4=Never Married	
104	Your educational status	1=No formal education 2=Primary 1-8 3=Secondary 9-10 4=Technical/Vocational 11-12 5= Higher Education 12+	
105	Your occupational status	1=Governmental employee 2=Daily labourer 3=Merchant 4=Housewife 5=Student 6=Driver 7=Guard 8=Pensioner 9= Unemployment Other (specify)	
106	What is the family size of your household (include all related persons residing there)		
107	What is the average monthly household income (sum of incomes contributed by all family members)		
108	How much time does it take you on average to travel to this health facility?		
109	Which kind of transportation do you use to travel to this health facility?	1=Taxi 2=Public Transport/Bus 3=Bajaj 4=Bicycle/ Motor bicycle 5=Walk Other (specify)	
110	If you paid for your transportation here, how much Ethiopian birr did you pay per visit?		

S.NO	Question	Response	Code
200	TB diagnosis category?	1=Smear positive pulmonary TB	
		2=Smear Negative pulmonary	
		ТВ	
		3=Extra Pulmonary TB	
201	TB HIV co-infection?	1=Yes	
		2=No	
202	Any concurrent non-communicable disease?	1=Diabetes	
		2=Cancer	
		3=Hypertension	
		4=Other	
		(specify)	
203	History of Tuberculosis?	1=New case	
		2=Failure	
		3=Relapsed case	
		4=Default	

Section 2: Disease-related Information (from record review)

Intervention description

Mobile-assisted medication adherence support (Ma-MAS) intervention is a systematically developed behavioural intervention that is delivered through mobile SMS text and phone calls to improve a TB patient's medication adherence and clinic appointment attendance. Mobile SMS text and phone calls that sent to you to remind you to take their medication at home and to collect refills of their medication during TB treatment. I will ask you a few questions based on your experience to this intervention.

Section 3: Intention to adopt Mobile-assisted Medication Adherence Support (Ma-MAS) intervention

S.NO	Items	Strongly	disagree	Disagree	Neutral (3)	Agree (4)	Strongly Agree (5)
PE	Performance expectancy						
PE1	Using mobile SMS and calls would help me to remember my daily medication intake.						
PE2	Using mobile SMS and calls would help me to remember my refill medication visits at the scheduled date.						
PE3	Mobile SMS and calls would be very useful to my follow-up treatment.						
PE4	Mobile SMS and calls would provide very useful services and information to me.						
EE	Effort of expectancy						
EE1	Learning to operate mobile SMS and calls would be easy for me.						
EE2	I would find the mobile SMS and calls easy to use.						
EE3	My interaction with health professionals through mobile SMS and calls would be clear and understandable.						

EE4	It will be easy for me to become skillful at sending, receiving,			
	reading and interpreting mobile SMS text.			
SI	Social Influence			
SI1	People who are important to me think that I should use mobile			
511	SMS and calls.			
SI2	People who influence my behaviour think that I should use			
512	mobile SMS and calls.			
SI3	The health professionals of this health facility have been			
010	helpful in explaining the use of mobile SMS and calls.			
SI4	The health facility has indicated that it would support the use			
014	of mobile SMS and calls for my treatment follow-up			
FC	Facilitating Conditions			
FC1	I have the resources necessary to use mobile SMS and calls.			
FC2	I have the knowledge necessary to use mobile SMS and calls.			
FC3	I believe I can get reliable help from health professionals when			
105	I have difficulties using mobile SMS and calls.			
FC4	The mobile SMS and calls are not compatible with other			
104	services I use.			
ATT	Attitude			
-	Attitude Using mobile SMS and calls is a good idea for medication			
ATT ATT1	Using mobile SMS and calls is a good idea for medication reminders			
ATT1	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve			
ATT1 ATT2	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits			
ATT1	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits Mobile SMS and calls are easy to work			
ATT1 ATT2	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits Mobile SMS and calls are easy to work I would enjoy using mobile SMS and calls			
ATT1 ATT2 ATT3	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits Mobile SMS and calls are easy to work			
ATT1 ATT2 ATT3 ATT4 BI	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits Mobile SMS and calls are easy to work I would enjoy using mobile SMS and calls Behavioural Intention I predict I will use mobile SMS and calls in the next months for			
ATT1 ATT2 ATT3 ATT4	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits Mobile SMS and calls are easy to work I would enjoy using mobile SMS and calls Behavioural Intention I predict I will use mobile SMS and calls in the next months for my treatment follow-up			
ATT1 ATT2 ATT3 ATT4 BI BI1	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits Mobile SMS and calls are easy to work I would enjoy using mobile SMS and calls Behavioural Intention I predict I will use mobile SMS and calls in the next months for			
ATT1 ATT2 ATT3 ATT4 BI	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits Mobile SMS and calls are easy to work I would enjoy using mobile SMS and calls Behavioural Intention I predict I will use mobile SMS and calls in the next months for my treatment follow-up			
ATT1 ATT2 ATT3 ATT4 BI BI1 BI2	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits Mobile SMS and calls are easy to work I would enjoy using mobile SMS and calls Behavioural Intention I predict I will use mobile SMS and calls in the next months for my treatment follow-up I will always try to use mobile SMS and calls in my daily life for continued health services I prefer to use mobile SMS and calls to remember medications			
ATT1 ATT2 ATT3 ATT4 BI BI1	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits Mobile SMS and calls are easy to work I would enjoy using mobile SMS and calls Behavioural Intention I predict I will use mobile SMS and calls in the next months for my treatment follow-up I will always try to use mobile SMS and calls in my daily life for continued health services I prefer to use mobile SMS and calls to remember medications intake and refill visits			
ATT1 ATT2 ATT3 ATT4 BI BI1 BI2 BI3	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits Mobile SMS and calls are easy to work I would enjoy using mobile SMS and calls Behavioural Intention I predict I will use mobile SMS and calls in the next months for my treatment follow-up I will always try to use mobile SMS and calls in my daily life for continued health services I prefer to use mobile SMS and calls to remember medications intake and refill visits I intend to use the mobile SMS and calls for my treatment			
ATT1 ATT2 ATT3 ATT4 BI BI1 BI2	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits Mobile SMS and calls are easy to work I would enjoy using mobile SMS and calls Behavioural Intention I predict I will use mobile SMS and calls in the next months for my treatment follow-up I will always try to use mobile SMS and calls in my daily life for continued health services I prefer to use mobile SMS and calls to remember medications intake and refill visits I intend to use the mobile SMS and calls for my treatment follow-up when it is available			
ATT1 ATT2 ATT3 ATT4 BI BI1 BI2 BI3	Using mobile SMS and calls is a good idea for medication reminders I believe that mobile SMS and calls can be utilized to improve my medication intake adherence and refill visits Mobile SMS and calls are easy to work I would enjoy using mobile SMS and calls Behavioural Intention I predict I will use mobile SMS and calls in the next months for my treatment follow-up I will always try to use mobile SMS and calls in my daily life for continued health services I prefer to use mobile SMS and calls to remember medications intake and refill visits I intend to use the mobile SMS and calls for my treatment			

In-depth Interview Guide for Post-Ma-MAS Intervention Experiences

Introduction

Hello my name is______. I am going to ask you a few questions regarding your perceptions and experiences on mobile SMS text and phone calls that you received in the past two months during your treatment follow-up.

S.No	Characteristics	Response
1	Sex	
2	Age	
3	Educational status	
4	Occupational status	

Section 1: Participant characteristics

Section 2: In-depth interview questions

- 1. Were the daily mobile SMS texts and/or weekly phone calls beneficial? (Probe: to remind you to take your medications and attend your refill visits, how? What? Why?)
- 2. Were the mobile SMS texts and/or phone calls easy to understand? (Probe: What? Why?)
- 3. Were there any difficulties or challenges for you in receiving the mobile SMS texts and/or phone calls? (Probe: What? Why?)
- 4. Did the mobile SMS text and/or phone calls give you any risk or make you feel uncomfortable? (Probe: What? Why?)
- 5. Was the frequency of the mobile SMS texts and/or phone calls enough? (Probe: What? Why?)
- Was the timing of the mobile SMS texts and/or phone calls appropriate? (Probe: What? Why?)
- Was the duration of the mobile SMS texts and/or phone calls intervention enough? (Probe: What? Why?)
- 8. Was the content of messages good for you? (Probe: What? Why?)
- What would you recommend in the use of mobile SMS texts and/or phone calls in the future? (Probe: What? Why?)

APPENDIX TEN: INFORMED CONSENT FORM





Participant Information Sheet/Consent Form In-depth Interview Participants

Health/Social Science Research - Adult providing own consent

Title	Mobile-assisted Medication Adherence Support (Ma-MAS) intervention for Tuberculosis Patients: Effectiveness, Experiences and Adoption Intention in Addis Ababa, Ethiopia.	
Sub-title	Mobile-assisted Medication Adherence Support intervention development using the Medical Research Council (MRC) framework	
Short Title Ma-MAS intervention development using the MRC framework		
Protocol Number	[NA]	
Project Sponsor	Flinders University, South Australia	
Coordinating Principal Investigator/ Principal Investigator	Zekariyas Sahile	
Associate Investigator(s)	Prof. Paul Arbon; Dr Lua Perima-Lewis, Prof. Anthony Maeder	
Location	Addis Ababa, Ethiopia	

Part 1 What does my participation involve?

1 Introduction

You are invited to take part in this research project for mobile-assisted medication adherence support Intervention development. You have been invited because you are in one of the groups of interest to the project: Tuberculosis-infected persons, Health care providers at Tuberculosis (TB) clinic, health care system service providers experts and managers. Your contact details were obtained by/from the Addis Ababa Health Bureau.

This Participant Information Sheet/Consent Form tells you about the research project. It explains the processes involved in taking part. Knowing what is involved will help you decide if you want to take part in the research.

Please read this information carefully. Ask questions about anything that you don't understand or want to know more about. Before deciding whether or not to take part, you might want to talk about it with a relative, friend or local health worker.

Participation in this research is voluntary. If you don't wish to take part, you don't have to. If you decide you want to take part in the research project, you will be asked to sign the consent section. By signing it you are telling us that you:

• Understand what you have read

- Consent to take part in the research project
- Consent to be involved in the research described
- Consent to the use of your personal and health information as described.

You will be given a copy of this Participant Information and Consent Form to keep.

2 What is the purpose of this research?

The research aims to systematically develop a mobile-assisted medication adherence support intervention for Tuberculosis patients (objective 1). In Ethiopia, past interventions of this type using SMS text and phone calls have not been not well developed using locally available evidence and appropriate behavioural theory. This research will develop an intervention systematically for those aspects, to provide new evidence that could help health care managers and policymakers to improve Tuberculosis medication adherence. The mobile-assisted medication adherence intervention will be tested for feasibility, effectiveness and adoption intention in the next stages of this research project (see Figure 1). The research findings will also be used as a base for future research and serve as educational material.



Figure 1: The flow of the study objectives

This research is conducted by collaboration between Flinders University and Addis Ababa Health Bureau. The research results will be used in the study programme for the Doctor of Philosophy degree being undertaken by the candidate Mr Zekariyas Sahile at Flinders University, Australia.

3 What does participation in this research involve?

If you consent to participate in the research project, you will first complete an initial screening questionnaire taking approximately 5 minutes which will determine if you are eligible to take part. You will then take part in an In-depth Interview by one or more members of the research team which will take approximately 60 minutes. You will be asked only for a one-time interview. This interview will be made in a quiet place that is convenient for privacy and confidentiality. The research team

will record the interview discussion by using a digital audio recorder. Your information provided will be kept confidential and used only for research purposes, and your personal and identifiable information will not be presented, reported, or published.

If the screening questionnaire shows that you meet the requirements, then you will be able to start the research project. If the screening questionnaire shows that you cannot be in the research project, the research coordinator will discuss other options with you.

This research project has been designed to make sure the researchers interpret the results fairly and appropriately and to avoid the study team or participants jumping to conclusions. There are no costs associated with participating in this research project, nor will you be paid.

4 Other relevant information about the research project

This study is conducted at Addis Ababa, Ethiopia. Participant groups such as TB-infected persons, Health care providers at TB clinic, health care system service providers experts and managers at Addis Ababa health Bureau, and sub-city will participate in the study. This study does not affect regular care and/or service provided. You will be interviewed at a convenient place for privacy and confidentiality. Once you voluntarily agreed to be involved in the study, you will be interviewed about the intervention development process. After developing this intervention, it will be tested for acceptability and effectiveness in the different participants.

5 Do I have to take part in this research project?

Participation in any research project is voluntary. If you do not wish to take part in this research, you do not have to. If you decide to take part and later change your mind, you are free to withdraw from the project at any stage. If you do decide to take part, you will be given this Participant Information and Consent Form to sign and you will be given a copy to keep.

Your decision whether to take part or not to take part in this research, or to take part and then withdraw, will not affect your routine care, your relationship with healthcare professional staff or your relationship with this health facility.

6 What are the possible benefits of taking part?

We cannot guarantee or promise that you will receive any benefits from this research, however, possible benefits may include the evidence produced from this study which may be helpful to systematically develop mobile-assisted medication adherence support interventions based on the needs of the local population and the culture, that will be tested for effectiveness in near future. The study mobile-assisted medication adherence support intervention may be scaled up if it found to be effective. Further, the study will serve as educational material and could be a base for future research.

7 What are the possible risks and disadvantages of taking part?

You may feel that some of the questions we ask you are stressful or upsetting. If you do not wish to answer a question, you may skip it and go to the next question, or you may stop immediately. If you become upset or distressed as a result of your participation in the research project, the research team will be able to arrange for counselling or other appropriate support. Any counselling or support will be provided by qualified staff who are not members of the research team. This counselling will be provided to you free of charge.

8 What if I withdraw from this research project?

If you do consent to participate, you may withdraw at any time. If you decide to withdraw from the research project, please notify a member of the research team before you withdraw. A member of

the research team will inform you if there are any special requirements linked to withdrawing. If you do withdraw, you will be asked to complete and sign a 'Withdrawal of Consent' form; this will be provided to you by the research team.

If you decide to leave the research project, the researchers will not collect additional personal information from you, although personal information already collected will be retained to ensure that the results of the research project can be measured properly and to comply with law. You should be aware that data collected up to the time you withdraw will form part of the research project results. If you do not want your data to be included, you must tell the researchers when you withdraw from the research project.

9 Could this research project be stopped unexpectedly?

This research project may be stopped unexpectedly for a variety of reasons. These may include reasons such as decisions made by local regulatory/health authorities or by research sponsors.

10 What happens when the research project ends?

Participants will be asked whether they wish to get a summary of the research work and if so in what form and will be informed about how they can access the research results. The research outcomes will be disseminated to the Federal Ministry of Health, Addis Ababa Health Bureau, and respective health facilities for decision making for the purpose of program improvement and informing future research directions.

Part 2 How is the research project being conducted?

11 What will happen to information about me?

By signing the consent form, you consent to the research team collecting and using personal information about you for the research project. Any information obtained in connection with this research project that can identify you will remain confidential. Our discussion will be recorded the audio when you interview with the research team is conducting. All participant's information or data will keep in the University server and investigators computers and backup file with protected password and will be locked in cabinet storage for hard-copy document file. All identifiable information will be removed and replaced by code. Participant's identifiable information or data will not be reported or transferred to third party. Your information will only be used for the purpose of this research project, and it will only be disclosed with your permission, except as required by law. The personal information that the research team collects, and uses is information from interview questionnaires. The interview discussion will be used only for this research purpose. The non-identifiable transcription information will be reported.

It is anticipated that the results of this research project will be published and/or presented in a variety of forums. In any publication and/or presentation, the information will be provided in such a way that you cannot be identified, except with your express permission. Your identifiable information or data will not be presented, reported or published.

You have the right to request access to the information about you that is collected and stored by the research team. You also have the right to request that any information with which you disagree be corrected. Please inform the research team member named at the end of this document if you would like to access your information.

Any information obtained for the purpose of this research project and for any future research described that can identify you will be treated as confidential and securely stored. It will be disclosed only with your permission, or as required by law.

12 Complaints and compensation

If you suffer any distress or psychological injury as a result of this research project, you should contact the research team as soon as possible. You will be assisted with arranging appropriate treatment and support.

13 Who is organising and funding the research?

This research project is being organised by Zekariyas Sahile and the research is being conducted through Flinders University. It is being financially supported by Flinders University, Australia and NIH/Fogarty International Centre Global Infectious Disease.

No member of the research team will receive a personal financial benefit from your involvement in this research project (other than their ordinary wages). The research investigators declare that have no conflict of interest.

14 Who has reviewed the research project?

All research in Australia involving humans is reviewed by an independent group of people called a Human Research Ethics Committee (HREC). The ethical aspects of this research project have been approved by the Southern Adelaide Clinical Human Research Ethics Committee and Addis Ababa Health Bureau Human Research Ethics Committee, Ethiopia.

This project will be carried out according to the National Statement on Ethical Conduct in Human Research (2007). This statement has been developed to protect the interests of people who agree to participate in human research studies.

15 Further information and who to contact

The person you may need to contact will depend on the nature of your query. If you want any further information concerning this project or if you have any problems which may be related to your involvement in the project, you can contact the researcher on +251911072963 or any of the following people:

Research contact person

Name	Zekariyas Sahile
Position Principal Investigator	
Telephone +2519 11072963	
Email	zekariyas.nezenega@flinders.edu.au

For matters relating to research at the site at which you are participating, the details of the local site complaints person are:

Complaints contact person

Name	
Position	
Telephone	
Email	

If you have any complaints about any aspect of the project, the way it is being conducted or any questions about being a research participant in general, then you may contact:

Reviewing HREC approving this research and HREC Executive Officer details

Reviewing HREC name	Addis Ababa Health Bureau Human Research Ethics Committee
HREC Executive Officer	Dr Yohannes Channea
Telephone	
Email	yoha2wok@yahoo.com

Local HREC Office contact

Name	Dr Yohannes Channea
Position	HREC co-ordinator Addis Ababa Health Bureau, Ethiopia
Telephone	
Email	yoha2wok@yahoo.com

Consent Form - Adult providing own consent

Title	Mobile-assisted medication adherence support intervention design for Tuberculosis patients using the Medical Research Council (MRC) framework
Short Title	Ma-MAS intervention design using the MRC framework
Protocol Number	[NA]
Project Sponsor	Flinders University, South Australia
Coordinating Principal	
Investigator/ Principal	Zekariyas Sahile
Investigator	
Associate Investigator(s)	Professor Paul Arbon, Dr Lua Perinmal-Lewis, Professor Anthony
Location	Maeder Addis Ababa, Ethiopia

Declaration by Participant

I have read the Participant Information Sheet or someone has read it to me in a language that I understand.

I understand the purposes, procedures, and risks of the research described in the project.

I have had an opportunity to ask questions and I am satisfied with the answers I have received.

I freely agree to participate in this research project as described and understand that I am free to withdraw at any time during the project without affecting my future care.

I understand that I will be given a signed copy of this document to keep.

Name of Participant (please print)		
Signature	Date	

Declaration by Researcher[†]

I have given a verbal explanation of the research project, its procedures, and risks and I believe that the participant has understood that explanation.

Name of Researcher [†] (please print)		
Signature	Date	

[†] An appropriately qualified member of the research team must provide the explanation of, and information concerning, the research project.

Note: All parties signing the consent section must date their own signature.

Form for Withdrawal of Participation - Adult providing own consent

It is recommended that this form NOT be included as part of the PICF itself, but that it be developed at the same time and made available to researchers for later use, if necessary.

Title	Mobile-assisted medication adherence support intervention design for Tuberculosis patients using MRC framework
Short Title Ma-MAS intervention design using MRC framework	
Protocol Number	[NA]
Project Sponsor	Flinders University, South Australia
Coordinating Principal Investigator/ Principal Investigator	Zekariyas Sahile
Associate Investigator(s)	Prof. Paul Arbon, Dr Lua Perinmal-Lewis, Prof. Anthony Maeder
Location	Addis Ababa, Ethiopia

Declaration by Participant

I wish to withdraw from participation in the above research project and understand that such withdrawal will not affect my routine care, or my relationships with the researchers.

Name of Participant (please print)		
Signature	Date	

In the event that the participant's decision to withdraw is communicated verbally, the Senior Researcher must provide a description of the circumstances below.

Declaration by Researcher[†]

I have given a verbal explanation of the implications of withdrawal from the research project and I believe that the participant has understood that explanation.

Name of Researcher (please print)	
Signature	Date

[†] An appropriately qualified member of the research team must provide information concerning withdrawal from the research project.

Note: All parties signing the consent section must date their own signature.





Participant Information Sheet/Consent Form Feasibility and Acceptability Participants

Health/Social Science Research - Adult providing own consent

Title	Mobile-assisted Medication Adherence Support (Ma-MAS) intervention for Tuberculosis Patients: Effectiveness, Experiences and Adoption Intention in Addis Ababa, Ethiopia.
Sub-title	Feasibility and acceptability and of Mobile-assisted Medication Adherence Support intervention
Short Title	Feasibility and acceptability of Ma-MAS intervention
Protocol Number	[NA]
Project Sponsor	Flinders University, South Australia
Coordinating Principal Investigator/ Principal Investigator	Zekariyas Sahile
Associate Investigator(s)	Prof. Paul Arbon; Dr Lua Perima-Lewis, Prof. Anthony Maeder
Location	Addis Ababa, Ethiopia

Part 1 What does my participation involve?

1 Introduction

You are invited to take part in this research project for assessing feasibility and acceptability of a Mobile-assisted Medication Adherence Support intervention. You have been invited because you are following anti-Tuberculosis treatment at this health facility. Your contact details were obtained by/from the health facility.

This Participant Information Sheet/Consent Form tells you about the research project. It explains the processes involved in taking part in this research project. Knowing what is involved will help you decide if you want to take part in the research.

Please read this information carefully. Ask questions about anything that you don't understand or want to know more about. Before deciding whether or not to take part in this research, you might want to talk about it with a relative, friend or local healthcare professional.

Participation in this research is voluntary. If you don't wish to take part, you don't have to. If you decide you want to take part in the research project, you will be asked to sign the consent section. By signing the consent form you are telling us that you:

- Understand what you have read
- Consent to take part in the research project
- · Consent to be involved in the research described
- Consent to the use of your personal and health information as described.
- You will be given a copy of this Participant Information and Consent Form to keep.

2 What is the purpose of this research?

The research aims to assess the acceptability and feasibility of mobile-assisted medication adherence support intervention (objective 2). These aspects have not previously been well investigated. The mobile-assisted medication adherence intervention has been systematically developed and will be evaluated for effectiveness in the next stage of this research project (see Figure 1). This research, therefore, must be tested for the acceptability and feasibility of intervention to provide evidence for health care managers and policymakers to develop strategies to improve Tuberculosis medication adherence. The research findings will also be used as a base for future research studies and used for educational material.



Figure 1: The flow of the study objectives

This research is conducted by collaboration between Flinders University and Addis Ababa Health Bureau. The research results will be used in the study programme for the Doctor of Philosophy degree being undertaken by the candidate Mr Zekariyas Sahile at Flinders University, Australia.

3 What does participation in this research involve?

If you decide to participate in the research project, you will first complete an initial screening questionnaire taking approximately 5 minutes which will determine if you are eligible to take part in this research project. In this research, only TB-infected persons who are undergoing anti-TB treatment will participate. You will then take part in a survey by interview conducted by one or more members of the research team which will take approximately 20 minutes. You will be asked only for a one-time interview. The interview will be made in a quiet place that is convenient for privacy and confidentiality. The research team will record your answers to the survey questions given during the interview on a form. Your information provided will be kept confidential and used only for research purposes, and your personal and identifiable information will not be presented, reported, or published.

If the screening questionnaire shows that you meet the requirements, then you will be able to start the research project. If the screening questionnaire shows that you cannot be in the research project, the research coordinator will discuss other options with you.

This research project has been designed to make sure the researchers interpret the results fairly and appropriately and to avoid study doctors or participants jumping to conclusions. There are no costs associated with participating in this research project, nor will you be paid.

4 Other relevant information about the research project

This study is conducted at Addis Ababa, Ethiopia, Public Health Facilities. A total of 36 public health centers that are providing the Tuberculosis (TB) treatment service will be enrolled for this study. All adult TB-infected participants who are following anti-TB treatment for a six-month period will be eligible for this study. This study does not affect your regular anti-TB treatment care provided or your relationship with a health facility. Once you voluntarily agree to be involved in the study, you will be asked questions about acceptability and feasibility of the intervention.

5 Do I have to take part in this research project?

Participation in any research project is voluntary. If you do not wish to take part in this research, you do not have to. If you decide to take part and later change your mind, you are free to withdraw from the project at any stage. If you do decide to take part, you will be given this Participant Information and Consent Form to sign and you will be given a copy to keep with you.

Your decision whether to take part or not to take part in this research, or to take part and then withdraw, will not affect your routine care, your relationship with professional staff or your relationship with this health facility.

6 What are the possible benefits of taking part?

We cannot guarantee or promise that you will receive any benefits from this research; however, possible benefits may include the evidence produced from this study which may be helpful to improve a mobile-assisted medication adherence support intervention, that will be tested for effectiveness in near future. The study will also generate evidence for health care managers and policymakers on ways of improving TB medication adherence interventions and strategy. Further, the study will serve as educational material and could be a base for future research.

7 What are the possible risks and disadvantages of taking part?

You may feel that some of the questions we ask are stressful or upsetting. If you do not wish to answer a question, you may skip it and go to the next question, or you may stop immediately. If you become upset or distressed as a result of your participation in the research project, the research team will be able to arrange for counselling or other appropriate support. Any counselling or support will be provided by qualified staff who are not members of the research team. This counselling will be provided free of charge.

8 What if I withdraw from this research project?

If you do consent to participate, you may withdraw at any time. If you decide to withdraw from the project, please notify a member of the research team before you withdraw. A member of the research team will inform you if there are any special requirements linked to withdrawing. If you do withdraw, you will be asked to complete and sign a 'Withdrawal of Consent' form; this will be provided to you by the research team.

If you decide to leave the research project, the researchers will not collect additional personal information from you, although personal information already collected will be retained to ensure that

the results of the research project can be measured properly and to comply with the law. You should be aware that data collected up to the time you withdraw will form part of the research project results. If you do not want your data to be included, you must tell the researchers when you withdraw from the research project.

9 Could this research project be stopped unexpectedly?

This research project may be stopped unexpectedly for a variety of reasons. These may include a reason such as decisions made by local regulatory/health authorities or by research sponsors.

10 What happens when the research project ends?

Participants will be asked whether they wish to get a summary of the research work and if so in what form and will be informed about how they can access the research results. The research outcomes will be disseminated to the Federal Ministry of Health, Addis Ababa Health Bureau, and respective health facilities for decision making for the purpose of program improvement and informing future research directions.

Part 2 How is the research project being conducted?

11 What will happen to information about me?

By signing the consent form, you consent to the research team collecting and using personal information about you for the research project. Any information obtained in connection with this research project that can identify you will remain confidential. All participant's information or data will keep in the University server and principal investigator laptop computer and backup file with a protected password and will be locked in the cabinet storage for hard-copy document files. All identifiable information will be removed and replaced by code. All participant information will be encrypted. Participant's identifiable information or data will not be reported or transferred to others. Your information will only be used for this research project, and it will only be disclosed with your permission, except as required by law. The personal information that the research team collects, and uses is information from interview questionnaires. The interview discussion will be used only for this research purpose. The non-identifiable transcription information will be reported.

It is anticipated that the results of this research project will be published and/or presented in a variety of forums. In any publication and/or presentation, the information will be provided in such a way that you cannot be identified, except with your express permission. Your identifiable information or data will not be presented, reported or published.

You have the right to request access to the information about you that is collected and stored by the research team. You also have the right to request that any information with which you disagree be corrected. Please inform the research team member named at the end of this document if you would like to access your information.

Any information obtained for the purpose of this research project and for any future research described that can identify you will be treated as confidential and securely stored. It will be disclosed only with your permission, or as required by law.

12 Complaints and compensation

If you suffer any distress or psychological injury as a result of this research project, you should contact the research team as soon as possible. You will be assisted with arranging appropriate treatment and support.

13 Who is organising and funding the research?

This research project is being organised by Zekariyas Sahile and the research is being conducted through Flinders University. It is being financially supported by Flinders University, Australia and NIH/Fogarty International Centre Global Infectious Disease.

No member of the research team will receive a personal financial benefit from your involvement in this research project (other than their ordinary wages). The research investigators declare that have no conflict of interest.

14 Who has reviewed the research project?

All research in Australia involving humans is reviewed by an independent group of people called a Human Research Ethics Committee (HREC). The ethical aspects of this research project have been approved by the South Adelaide Clinical Human Research Ethics committee and Addis Ababa Health Bureau, Ethiopia.

This project will be carried out according to the National Statement on Ethical Conduct in Human Research (2007). This statement has been developed to protect the interests of people who agree to participate in human research studies.

15 Further information and who to contact

The person you may need to contact will depend on the nature of your query. If you want any further information concerning this project or if you have any problems which may be related to your involvement in the project, you can contact the researcher on +251911072963 or any of the following people:

Research contact person

Name	Zekariyas Sahile
Position	Principal Investigator
Telephone	+251911072963
Email	Zekariyas.nezenega@flinders.edu.au

For matters relating to research at the site at which you are participating, the details of the local site complaints person are:

Complaints contact person

Name	
Position	
Telephone	
Email	

If you have any complaints about any aspect of the project, the way it is being conducted or any questions about being a research participant in general, then you may contact:

This person should be someone independent of the research, such as the Executive Officer of the reviewing HREC that approved the project. Contact your local HREC administrator for the requirements at your institution.

Reviewing HREC approving this research and HREC Executive Officer details

Local HREC Office contact

Reviewing HREC name	Addis Ababa Health Bureau Human Research Ethics Committee
HREC Executive Officer	Dr Yohannes Channea
Telephone	
Email	yoha2wok@yahoo.com

Name	Dr Yohannes Channea
Position	HREC co-ordinator Addis Ababa Health Bureau, Ethiopia
Telephone	
Email	yoha2wok@yahoo.com

Consent Form - Adult providing own consent

Title	Patient's acceptability and feasibility of mobile-assisted medication adherence support (Ma-MAS) intervention
Short Title	Feasibility and acceptability of Ma-MAS intervention
Protocol Number	[NA]
Project Sponsor	Flinders University, South Australia
Coordinating Principal Investigator/ Principal	Zekariyas Sahile
Associate Investigator(s)	Prof. Paul Arbon; Dr Lua Perinmal-Lewis, Prof. Anthony Maeder
Location	Addis Ababa, Ethiopia

Declaration by Participant

I have read the Participant Information Sheet or someone has read it to me in a language that I understand.

I understand the purposes, procedures and risks of the research described in the project.

I have had an opportunity to ask questions and I am satisfied with the answers I have received.

I freely agree to participate in this research project as described and understand that I am free to withdraw at any time during the project without affecting my future care.

I understand that I will be given a signed copy of this document to keep.

Name of Participant (please print)	
Signature	Date

Declaration by Researcher[†]

I have given a verbal explanation of the research project, its procedures and risks and I believe that the participant has understood that explanation.

Name of Researcher [†] (please print)	
Signature	Date

[†] An appropriately qualified member of the research team must provide the explanation of, and information concerning, the research project.

Note: All parties signing the consent section must date their own signature.

Form for Withdrawal of Participation - Adult providing own consent

It is recommended that this form NOT be included as part of the PICF itself, but that it be developed at the same time and made available to researchers for later use, if necessary.

Title	Feasibility and acceptability of mobile-assisted medication adherence support intervention	
Short Title	Feasibility and acceptability of Ma-MAS intervention	
Protocol Number	[NA]	
Project Sponsor	Flinders University, South Australia	
Coordinating Principal		
Investigator/ Principal Investigator	Zekariyas Sahile	
Associate Investigator(s)	Prof. Paul Arbon; Dr Lua Perinmal-Lewis, Professor Anthony Maeder	
Location	Addis Ababa, Ethiopia	

Declaration by Participant

I wish to withdraw from participation in the above research project and understand that such withdrawal will not affect my routine care, or my relationships with the researchers.

Name of Participant (please print)	
Signature	Date

In the event that the participant's decision to withdraw is communicated verbally, the Senior Researcher must provide a description of the circumstances below.

Declaration by Researcher[†]

I have given a verbal explanation of the implications of withdrawal from the research project and I believe that the participant has understood that explanation.

Name of Researcher (please print)	
Signature	Date

[†] An appropriately qualified member of the research team must provide information concerning withdrawal from the research project.

Note: All parties signing the consent section must date their own signature.





Participant Information Sheet/Consent Form Intervention and post intervention Participants

Interventional Study - Adult providing own consent

Title	Mobile-assisted Medication Adherence Support (Ma-MAS) intervention for Tuberculosis Patients: Effectiveness, Experiences and Adoption Intention in Addis Ababa, Ethiopia.
Sub-title	Mobile-assisted Medication Adherence Support Intervention Effectiveness: Randomised control Trial (RCT)
Short Title	Ma-MAS intervention Effectiveness: RCT
Protocol Number	PACTR202002831201865
Project Sponsor	Flinders University, South Australia
Coordinating Principal Investigator/ Principal	Zekariyas Sahile
Associate Investigator(s)	Prof. Paul Arbon; Dr Lua Perima-Lewis, Prof. Anthony Maeder
Location	Addis Ababa, Ethiopia

Part 1 What does my participation involve?

1 Introduction

You are invited to take part in this research project because you have been following your anti-Tuberculosis treatment at this health facility. The research project is testing a new behavioural intervention for Tuberculosis medications adherence. The new behavioural intervention is called Mobile-assisted Medication Adherence Support (Ma-MAS).

This Participant Information Sheet/Consent Form tells you about the research project. It explains the tests and behavioural intervention involved. Knowing what is involved will help you decide if you want to take part in the research.

Please read this information carefully. Ask questions about anything that you don't understand or want to know more about. Before deciding whether or not to take part, you might want to talk about it with a relative, friend or your doctor or Health care provider.

Participation in this research is voluntary. If you don't wish to take part, you don't have to. You will receive the best possible healthcare whether or not you take part in this study.

If you decide you want to take part in the research project, you will be asked to sign the consent section. By signing the consent form you are telling us that you:

- Understand what you have read
- Consent to take part in the research project
- Consent to have the tests and behavioural intervention that are described
- Consent to the use of your personal and health information as described

You will be given a copy of this Participant Information and Consent Form to keep.

2 What is the purpose of this research?

The research aims to measure the effectiveness of a mobile-assisted medication adherence support intervention on patients' treatment adherence. This is an experimental behavioural intervention that is given through mobile SMS texts and phone calls in Ethiopia. The mobile-assisted medication adherence intervention has been developed systematically and tested for feasibility, and acceptability in previous stages of this research project (see Figure 1).

At present, there is a lack of well-investigated knowledge on how behavioural interventions using mobile SMS and phone calls could be effective for Tuberculosis medication adherence. This research, therefore, would provide evidence that would help healthcare managers and policymakers to develop strategies to improve Tuberculosis medication adherence. The research findings will also be used as a basis for future research and used for educational purposes.



Figure 1: The flow of the study objectives

This research is conducted by collaboration between Flinders University and Addis Ababa Health Bureau. The research results will be used in the study programme for the Doctor of Philosophy degree being undertaken by the candidate Mr Zekariyas Sahile at Flinders University, Australia.

3 What does participation in this research involve?

To be involved in this research, you must agree voluntarily before any assessment or interview is made, by signing the consent form. You cannot be involved in this study unless you fulfil all the following conditions:

- participants are adult (age 18 years and above)
- participants will be following their anti-Tuberculosis treatment for six months
- participants are at the intensive phase of treatment during the screening
- participants can read and understand the national official language (Amharic)

• Participants have their own mobile phone, however, participants who don't have their own mobile phone may be included if they have a trusted family member's mobile phone in the household with a collaborative agreement

However, you will not be involved if you are deemed not to have the capacity to decide reasonably to participate, or if you are enrolled in other behavioural interventions.

You will be participating in a randomised controlled research project. Sometimes we do not know which behavioural intervention is best for treating a condition. To find out, we need to compare different behavioural interventions. We put people into groups and give each group a different behavioural intervention which may use mobile SMS text messages and phone calls (see Figure 2). The results are compared to see if one intervention is better. To try to make sure the groups are unbiased, each participant is put into a group by chance (random). If you agree to participate in the study, you will be assigned into one of two groups.

- 1. Group one participants will receive daily SMS text messages and weekly phone calls for daily medication intake and medication refill visit reminders
- 2. Group two participants will receive the standard treatment as group one but not receive an additional intervention on a mobile phone.

You will know which of the group you have to be assigned, but you cannot choose yourself.



Figure 2: Mobile-assisted Medication Adherence Support (Ma-MAS) intervention

You are asked to participate in this research because the information found from this study can help the improvement of Tuberculosis treatment services. Moreover, your participation is confirmed by the probability sampling technique which provides an equal chance of selection. You have a 50% chance to participate in the intervention group and a 50% chance of being in the control group. If you are assigned to the intervention group, you will receive a daily SMS text and weekly phone calls reminder of medication intake and refill visit reminders for 8 weeks while you are on anti-Tuberculosis treatment during the continuation phase of treatment. If you have a local doctor or health care provider, we strongly recommend that you inform them of your participation in this research project.

This research project has been designed to make sure the researchers interpret the results fairly and appropriately and avoid study researchers or participants jumping to conclusions. If you agree to participate in this research, the following procedures will be undertaken:

- One or more members of the research team will interview you about a few sociodemographic, health service, and disease-related questions at the beginning of the study. This interview takes about 20 minutes and will be conducted in your usual clinic session.

- At subsequent clinic visits for your medication refills, the data collector will assess your medications adherence level by asking you questions about your medication usage and by taking a sample of urine for testing at a random point after 4 and 8 weeks follow up time. The maximum amount of time you will spend in contact with data collectors for this purpose is 10 minutes for each time of assessment. All adherence assessments will be made immediately after you have finished your regular treatment services.
- After 8 weeks of intervention and follow up period, if you have participated in the intervention group listed above, you will also be asked to participate in another study and you may be a part of an in-depth interview and survey interview to assess the adoption, perceptions and experiences of mobile-assisted medication adherence support. The average time needed for this one-time In-depth interview and survey interview is 60 and 15 minutes, respectively.

The information obtained from this study will be used for research proposes only to inform possible solutions to improve future anti-Tuberculosis medication adherence. Your identifiable information will not be disclosed in the presentation, report, or publication and will not be stored after completion of the study. The information that you give us in this study will be kept confidential within the study research team. For ensuring your privacy, all discussions that we will have with you will be held in a safe place where no one can hear what we are discussing. Your daily SMS text messages and weekly phone calls, to be delivered for 8 weeks, will not be accessed by anybody other than the study research team and our record of your phone number will be deleted after the study is completed. In your medication refill visits to the clinic, a research team member will ask you whether you are receiving the intervention regularly if you are assigned into the intervention group.

There are no additional costs to you associated with participating in this research project, nor will you be paid. All SMS and phone calls within the intervention and tests required as part of the research project will be provided to you free of charge.

4 What do I have to do?

All participants in the study will follow their regular anti-Tuberculosis treatment for six months period as per standard care, in whichever study group they assigned. Once you have voluntarily agreed to be involved in the study and been assigned into the intervention group, you will accept to receive the daily SMS text and weekly phone calls medications intake and refill visits reminders for 8 weeks. If you are assigned to the no-intervention (control) group, you will still be monitored for those 8 weeks.

You will need to disclose your own mobile phone number to the research team to receive an SMS text or phone calls. If you don't have your own mobile phone, you will need to be able to use your trusted family member's mobile phone number in the household with a collaborative agreement, to receive the intervention. Participants are expected to read and answer the daily and weekly medications intake and refill visits reminders of SMS text message and phone call, and regularly fellow the information as directed by mobile SMS text or phone calls. Participants need to report to the research team if they change their mobile phone or lose their mobile phone, to immediately replace the old mobile phone number by the new one. You must not forward, tell or show your mobile SMS text and phone call information to other participants as this will affect the study.

5 Other relevant information about the research project

A total of 36 government health centres from Addis Ababa, Ethiopia will be included in the study and a total of 186 participants will be expected to be involved in the study. Of 186 participants, 93 participants will be assigned in the intervention group (SMS text plus phone call reminders). This intervention has been systematically developed through the involvement of different participants such as Tuberculosis infected person, health care providers, health care managers, and Tuberculosis prevention and control experts. It has also been evaluated for the acceptability and feasibility of the intervention.

6 Do I have to take part in this research project?

Participation in this research project is voluntary. If you do not wish to take part in this study, you do not have to. If you decide to take part in this study and later change your mind, you are free to withdraw from the project at any stage. If you do decide to take part in this study, you will be asked to sign this Participant Information and Consent Form and you will be given a copy to keep with you. Your decision whether to take part in this study or not to take part in this study, to take part and then withdraw, will not affect your routine standard treatment care, your relationship with healthcare professionals treating you or your relationship with this health facility.

7 What are the alternatives to participation?

This is an additional behavioural intervention delivered through mobile SMS text and phone call that evaluates the effectiveness of medication adherence improvement. It is not an alternative treatment to standard care. Your study healthcare provider will discuss this aspect with you before you decide whether or not to take part in this research project.

8 What are the possible benefits of taking part?

We cannot guarantee or promise that you will receive any benefits from this research; however, possible benefits may include participants receiving additional information through daily SMS text messages and weekly phone calls, to remind you to take your daily medications and to attend your medications refill visits. The study may provide relevant evidence for program managers and policymakers to improve Tuberculosis treatment adherence in the future.

9 What are the possible risks and disadvantages of taking part?

This research has no anticipated risk for you, or your family, or your Tuberculosis treatment services you get in future from this health facility. The research has no intention to cause any physical harm, social discrimination, psychological trauma and economic loss. You may feel that some of the questions we ask are stressful or upsetting. If you do not wish to answer a question, you may skip it and go to the next question, or you may stop immediately. If you become upset or distressed as a result of your participation in the research, the study healthcare professional will be able to arrange for counselling or other appropriate support. Any counselling or support will be provided by qualified staff who are not members of the research project team. This counselling service will be provided to you free of charge.

10 What will happen to my test samples?

The proposed sample urine test is part of the medication adherence assessment and will be taken at a random point after 4 and 8 weeks of follow up. You will receive information how sample of urine collected before a urine test is made. You will also receive medical advice from the research team after a urine test is completed. Your urine sample will be individually re-identifiable when the test procedure is undertaken, but it will not be stored or transferred for further tests. This study will also collect data from your health record about your treatment adherence. Only non-identifiable information or data about you will be stored for the study. Thus, your urine test and treatment adherence results will be confidential, known only by you and the person who has tested your urine. Signing the consent form means that you agree to have this testing; it will not be done without your consent.

11 What if new information arises during this research project?

Sometimes during the course of a research project, new information becomes available about the intervention that is being studied. If this happens, your study healthcare provider will tell you about it and discuss with you whether you want to continue in the research project. If you decide to continue in the research project, you will be asked to sign an updated consent form. On receiving new information, your study health care provider might consider it to be in your best interests to withdraw you from the research project. If this happens, he/she will explain the reasons and arrange for your regular standard of healthcare to continue.

12 Can I have other treatments during this research project?

Whilst you are participating in this research project, you may not be able to a part of any other behavioural interventional research. It is important to tell your study healthcare provider and the study staff about if you are already enrolled or may want to enrol in another research project.

13 What if I withdraw from this research project?

If you decide to withdraw from this research project, please notify a member of the research team before you withdraw. This notice will allow that the research supervisor to discuss with you about any health risks or special requirements linked to withdrawing If you do withdraw your consent during the research project, the study healthcare provider and relevant study staff will not collect additional personal information from you, although personal information already collected will be retained to ensure that the results of the research project can be measured properly and to comply with the law. You should be aware that data collected by the research team up to the time you withdraw will form part of the research project results. If you do not want them to do this, you must tell the research team when you withdraw from the research project.

14 Could this research project be stopped unexpectedly?

This research project may be stopped unexpectedly for a variety of reasons. These may include reasons such as decisions made by local regulatory/health authorities or by the project sponsor.

15 What happens when the research project ends?

You will finish the study after 8 weeks of intervention period. You will be asked whether you wish to get a summary of the research work and be informed about how you can access the research results. The research outcomes will be also disseminated to the Federal Ministry of Health, Addis Ababa Health Bureau, and respective government health facilities for decision making for programme improvement and informing future research directions.

Part 2 How is the research project being conducted?

16 What will happen to information about me?

By signing the consent form, you consent to the study research staff collecting and using personal information about you for the research project. Any information obtained in connection with this research project that can identify you will remain confidential. All participant's information or data related to the research project will keep in the university server and investigators computer and backup electronic file with a protected password and lock cabinet storage for the hard-copy document file. All identifiable information will be removed and replaced by code. All participant information or data will be encrypted. All participant's identifiable information or data will not be reported or transferred to others. Non-identifiable information will be stored for 5 years after the study in Flinders University repository server and will be destroyed after that period. Your

information will only be used for this research project, and it will only be disclosed with your permission, except as required by law.

Your health records and any information obtained during the research project are subject to inspection (to verify the procedures and the data) by the relevant authorities and authorised representatives of the project sponsor [Flinders University], by the institution relevant to this Participant Information Sheet, [South Adelaide clinical HREC], or as required by law. By signing the Consent Form, you authorise the release of, or access to, this confidential information to the relevant study personnel and regulatory authorities as noted above.

It is anticipated that the results of this research project will be published and/or presented in a variety of forums. In any publication and/or presentation, the information will be provided in such a way that you cannot be identified, except with your permission. Your identifiable information or data will not be reported or published. Any information obtained for this research project and for the future research described in Section 16 that can identify you will be treated as confidential and securely stored. It will be disclosed only with your permission, or as required by law.

You have the right to request access to your information collected and stored by the research team. You also have the right to request that any information with which you disagree be corrected. Please contact the study team member named at the end of this document if you would like to access your information.

17 Complaints and compensation

If you suffer any distress or psychological injury as a result of this research project, you should contact the research team as soon as possible. You will be assisted with arranging appropriate counselling support.

18 Who is organising and funding the research?

This research project is being conducted by Zekariyas Sahile, a Doctor of Philosophy candidate of Flinders University. It is being financially supported by Flinders University, Australia and NIH/Fogarty International Centre Global Infectious Disease.

No member of the research team will receive a personal financial benefit from your involvement in this research project (other than their ordinary wages). The research investigators declare that have no conflict of interest.

19 Who has reviewed the research project?

All research in Australia involving humans is reviewed by an independent group of people called a Human Research Ethics Committee (HREC). The ethical aspects of this research project have been approved by the South Adelaide Clinical Human Research Ethics Committee and Addis Ababa Health Bureau HREC, Ethiopia.

This project will be carried out according to the Australian National Statement on Ethical Conduct in Human Research (2007). This statement has been developed to protect the interests of people who agree to participate in human research studies.

20 Further information and who to contact

The person you may need to contact will depend on the nature of your query.

If you want any further information concerning this project or if you have any medical problems which may be related to your involvement in the project (for example, any side effects), you can contact the principal investigator on +251911072963 or any of the following people:

Clinical contact person

Name	[Name]
Position	[Position]
Telephone	[Phone number]
Email	[Email address]

For matters relating to research at the site at which you are participating, the details of the local site complaints person are:

Complaints contact person

Name	Zekariyas Sahile
Position	Associate Investigator
Telephone	+251911072963
Email	Neze0002@flinders.edu.au

If you have any complaints about any aspect of the project, the way it is being conducted or any questions about being a research participant in general, then you may contact:

Reviewing HREC approving this research and HREC Executive Officer details

Reviewing HREC name	Addis Ababa Health Burea Human Research Ethics Committee	
HREC Executive Officer	Dr Yohannes Channea	
Telephone		
Email	yoha2wok@yahoo.com	

Local HREC Office contact (Single Site Research Governance Officer)

Name	Dr Yohannes Channea
Position	HREC-coordinator Addis Ababa Health Bureau, Ethiopia
Telephone	
Email	yoha2wok@yahoo.com

Consent Form - Adult providing own consent

Title	Mobile-assisted Medication Adherence Support (Ma-MAS) intervention for Tuberculosis Patients: Effectiveness, Experiences and Adoption Intention in Addis Ababa, Ethiopia.		
Sub-title	Mobile-assisted medication adherence support (Ma-MAS) Intervention Effectiveness among Tuberculosis patients		
Short Title	Ma-MAS intervention Effectiveness among Tuberculosis patients		
Protocol Number	PACTR202002831201865		
Project Sponsor	Flinders University, Australia		
Coordinating Principal Investigator	Zekariyas Sahile		
Associate Location	Prof. Paul Arbon; Dr Lua Perinmal-Lewis Prof. Anthony Maeder Addis Ababa, Ethiopia		

Declaration by Participant

I have read the Participant Information Sheet or someone has read it to me in a language that I understand. I understand the purposes, procedures and risks of the research described in the project.

I give permission for my doctors, other health professionals, health facility to release information to Flinders University concerning my disease and treatment for the purposes of this research project. I understand that such information will remain confidential.

I have had an opportunity to ask questions and I am satisfied with the answers I have received.

I freely agree to participate in this research project as described and understand that I am free to withdraw at any time during the study without affecting my current or future health care.

I understand that I will be given a signed copy of this document to keep.

Name of Participant (please print)	
Signature	Date

Name of Witness* to Participant's Signature (please print)	
Signature	Date

* Witness is <u>not</u> to be the investigator, a member of the study team or their delegate. In the event that an interpreter is used, the interpreter may <u>not</u> act as a witness to the consent process. Witness must be 18 years or older.

Declaration by Study Doctor/Senior Researcher[†]

I have given a verbal explanation of the research project, its procedures and risks and I believe that the participant has understood that explanation.

Name of Study Doctor/ Senior Researcher [†] (please print)	
Signature	_Date

⁺ A senior member of the research team must provide the explanation of, and information concerning, the research project. Note: All parties signing the consent section must date their own signature.

Form for Withdrawal of Participation - Adult providing own consent

It is recommended that this form NOT be included as part of the PICF itself, but that it be developed at the same time and made available to researchers for later use, if necessary. Note that a participant's decision to withdraw their separate consent to the use and storage of tissue will need to be documented separately and linked to the PICF used for that purpose.

Title	Mobile-assisted Medication Adherence Support (Ma-MAS) intervention for Tuberculosis Patients: Effectiveness, Experiences and Adoption Intention in Addis Ababa. Ethiopia		
Sub-title	Mobile-assisted medication adherence support (Ma-MAS) Intervention Effectiveness among Tuberculosis patients		
Short Title	Ma-MAS intervention Effectiveness among Tuberculosis patients		
Protocol Number	PACTR202002831201865		
Project Sponsor Coordinating Principal	Flinders University. Australia		
Investigator/	Zekariyas Sahile		
Principal Investigator			
Associate Investigator(s)	Prof. Paul Arbon; Dr Lua Perimal-Lewis, Prof. Anthony Maeder		
Location	Addis Ababa, Ethiopia		
Declaration by Participant			

I wish to withdraw from participation in the above research project and understand that such withdrawal will not affect my routine standard care treatment, my relationship with those treating me or my relationship with this health facility.

Name of Participant (please print)	
Signature	_ Date

In the event that the participant's decision to withdraw is communicated verbally, the Study Doctor/Senior Researcher will need to provide a description of the circumstances below.

Declaration by Study Doctor/Senior Researcher[†]

I have given a verbal explanation of the implications of withdrawal from the research project, and I believe that the participant has understood that explanation.

Name of Study Doctor/ Senior Researcher [†] (please print)		
Signature	Date	

[†] A senior member of the research team must provide the explanation of and information concerning withdrawal from the research project.

Note: All parties signing the consent section must date their own signature.

APPENDIX ELEVEN: ETHICS APPROVAL

Office for Research

Flinders Medical Centre Ward 6C, Room 6A219 Flinders Drive, Bedford Park SA 5042 Tel: (08) 8204 6453 E: Health.SALHNOfficeforResearch@sa.gov.au





Final Approval for Ethics Application

8 April 2020

Professor Anthony Maeder College of Nursing and Health Sciences Flinders University

Dear Professor Maeder,

OFR Number: 23.20

Project title: Nobile-assisted Medication Adherence Support (Ma-MAS) intervention for Tuberculosis Patients: Adoption of Intention, Effectiveness, and Experiences in Addis Ababa, Ethiopia. Chief Investigator: Professor Anthony Maeder

Ethics Approval Period: 8 April 2020 – 8 April 2023

The Southern Adelaide Clinical Human Research Ethics Committee (SAC HREC) (EC00188) has approved this application, which was first reviewed at the 2 March 2020 full committee meeting, through the greater than low risk pathway, and provided approval which meets the requirements of the *National Statement on Ethical Conduct in Human Research (2007,* updated 2018).

You are reminded that this letter constitutes Ethics approval only. Ethics approval is one aspect of the research governance process.

You must not commence this research project at any SA Health sites listed in the application until a Site Specific Assessment (SSA), or Access Request for data or tissue form, has been approved by the Chief Executive or delegate of each site.

The below documents have been reviewed and approved:

Document	Version	Date
Human research ethics application form	AU/1/DF3B316	24 March 2020
Project description	3	23 March 2020
Objective 1 in depth interview participant information sheet and consent form	-	23 March 2020
Objective 2 feasibility and adoption participant information sheet and consent form	-	23 March 2020
Objective 3 & 4 intervention and focus group participant information sheet and consent form	-	23 March 2020
Leaflet	-	23 March 2020
Eligibility screening tool for objective 1	-	-

Page 1 of 3

Eligibility screening tool for objective 2	-	-
Eligibility screening tool for objective 3	-	-
Objective 1 in depth interview guide	-	22 March 2020
Objective 2 participant feasibility questionnaire	-	22 March 2020
Objective 3 participant monitoring questionnaire	-	22 March 2020
Objective 3 compliance follow up	-	22 March 2020
Objective 4 participant focus group discussion guide	-	22 March 2020
Sample prototype SMS text	-	22 March 2020
Letter of support from Ethiopian Ministry of Health	-	23 March 2020

TERMS AND CONDITIONS OF ETHICS AND GOVERNANCE APPROVAL

The Principal Investigator must ensure this research complies with the National Statement on Ethical Conduct in Human Research (2018) & the Australian Code for the Responsible Conduct of Research (2007 updated 2018) by immediately reporting to the Office for Research (OFR) anything that may change the ethics or scientific integrity of the project. Final approval is granted subject to the researcher agreeing to meet the following terms and conditions:

- 1. This approval is subject to local Addis Ababa Health Bureau IRB approval.
- 2. Confidentiality of research participants MUST be maintained at all times.
- If the research involves the recruitment of participants, a signed copy of the 'Consent Form' must be given to the participant. Any changes to the Participant Information Sheet/Consent Form must be approved by the lead HREC prior to being used.
- 4. No promotion of a study can commence until final ethics and SALHN executive approval has been obtained. All advertisements/flyers need to be approved by the committee and media contact should be coordinated through the FMC media unit.
- Non-SA Health researchers viewing confidential SALHN data are required to complete and sign a SALHN Confidentiality Disclosure Deed
- All approved requests for access to medical records at any SALHN site must be accompanied by this approval letter.
- If your study involves a tertiary institution, contact the University to ensure compliance with University requirements prior to commencement of this study. This includes any insurance and indemnification.
- The PI must adhere to Monitoring and Reporting requirements for both ethics and governance which are available on the SALHN Research Website.
- The PI must immediately report to SAC HREC anything that may change the ethics or scientific integrity of the project
- An annual report must be submitted to the SAC HREC and SALHN governance on each anniversary of the date of final approval. Please visit the Office for Research website for the current template.
- 11. Non-SA Health researchers coming onsite at SALHN must provide evidence of a recent (<3 years) screening check. It is the responsibility of the Principal Investigator to ensure any non-SA Health personnel who conducts or monitors research meets SA Health screening requirements as per the SA Health Criminal & Relevant History Screening Policy Directive before they access any SA Health site. The cost of any such screening is the responsibility of the individual accessing the site or their employer.</p>
- Any reports or publications derived from the research should be submitted to the Committee at the completion of the project.

Page 2 of 3

- 13. Once the research project has concluded, any new product/procedure/intervention cannot be conducted in the SALHN as standard practice without the approval of the SALHN New Medical Products and Standardisation Committee or the SALHN New Health Technology and Clinical Practice Innovation Committee (as applicable). Please refer to the relevant committee link on the SALHN intranet for further information.
- 14. SALHN site-monitoring of authorised studies this approval/authorisation is subject to participation in this monitoring process. You will be notified in advance if your site has been selected for an inspection.

Please visit the SALHN Research website regularly and comply with all submission requirements as they may change from time to time.

For any queries about this matter, please contact The Office for Research on (08) 8204 6453 or via email to <u>Health.SALHNOfficeforResearch@sa.gov.au</u>

Yours sincerely,

Professor Bill Heddle Chair Southern Adelaide Clinical Human Research Ethics Committee

Page 3 of 3



አዲስ አበባ ከተማ አስተዳደር ጤና ቢሮ City Government of Addis Ababa Health Bureau

TO:

- Addis Ketema Sub- city Health Office
- Arada Sub- city Health Office
- Bole Sub- city Health Office
- Akaki-kality Sub- city Health Office

Kirkos Sub- city Health Office

Addis Ababa

.....

Cc

Ref.No. Date

- Gulelle Sub- city Health Office
- Nefas Silk Lafto Sub- city Health Office
- Kolfe Keranio Yeka Sub- city
 Health Office
- Lideta Sub- city Health Office
- Yeka Sub- City Health Office

Subject: Request to access Facilities to conduct approved research

The letter is to support Zekariyas Sahile to conduct research, which is entitled as "Mobile-assisted Medication Adherence Support (Ma-MAS) Intervention for TuberculosisPatients: Adoption of Intention, Effectiveness and Experiences Based in Addis Ababa, Ethiopia." The study proposal was duly reviewed and approved by Addis Ababa Health Bureau IRB, and the principal investigator is in formed with a copy of this letter to report any changes in the study procedures and submit an activity progress report to the Ethical Committee as required. Therefore we request the facility and staffs to provide support to the principal investigator.

Zekariyas Sahile **To Ethical Clearance Committee**

With Regards nce Committee @/h.43 በረተሰብ ጤና ምርምር በድን መሪ

377

5/12/23, 2:29 PM

SAC HREC amendment approval - 23.20

Health:SALHN Office for Research <Health.SALHNOfficeforResearch@sa.gov.au>

Wed 7/04/2021 11:09 AM

To: Zekariyas Nezenega <zekariyas.nezenega@flinders.edu.au>

Cc: Paul Arbon cpaul.arbon@flinders.edu.au>;Lua Perimal-Lewis <lua.perimal-lewis@flinders.edu.au>;Anthony Maeder <anthony.maeder@flinders.edu.au>

Dear Zekariyas,

OFR reference: 23.20

Title: Mobile-assisted Medication Adherence Support (Ma-MAS) intervention for Tuberculosis Patients: Adoption of Intention, Effectiveness, and Experiences in Addis Ababa, Ethiopia. Principal Investigator: Professor Paul Arbon

The amendment to the above study has been reviewed and approved by the SAC HREC.

Approval period: 8 April 2020 - 8 April 2023

The following documents have been reviewed and approved:

Document	Version	Date
Project amendment form	-	5 March 2021
Project description	4	5 March 2021
PICF	4	5 March 2021

The terms and conditions of ethics and governance approval remain unchanged from the original approval. Please note a formal approval letter will not be provided. Please retain a copy of this email as evidence of approval.

Kind Regards,

Dominic How

On behalf of

Professor Bill Heddle Chair SAC HREC

about:blank

APPENDIX TWELVE: CONFERENCE PRESENTATIONS CERTIFICATES



January 18, 2023

Zekariyas Sahile Nezenega Flinders University PhD candidate 2-4 Gladstone Fullarton, 5063 Australia

Dear Zekariyas Sahile Nezenega,

Thank you for attending the 2023 American Statistical Association International Conference on Health Policy Statistics (ICHPS) which was held January 9-11, 2023. This letter is to verify that you were present at the conference and that you presented a poster presentation entitled "Mobile-assisted Medication Adherence Support intervention among Tuberculosis Patients: A Parallel Group Randomized Control Trial" on Tuesday, January 10, 2023, from 7:30 p.m. - 8:30 p.m.

We hope that you enjoyed the conference and will be able to join us at ICHPS 2025.

Sincerely,

Clarissi Dano

Clarissa Davis Customer Services Manager

